reference into Part III.

SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, D.C. 20549

FORM 10-K ANNUAL REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934 For the fiscal year ended December 31, 2009 TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934 П Commission File No. 0-29092 PRIMUS TELECOMMUNICATIONS GROUP, INCORPORATED (Exact name of registrant as specified in its charter) **Delaware** 54-1708481 (State or other jurisdiction of (LR.S. Employer incorporation or organization) Identification No.) 7901 Jones Branch Drive, Suite 900, McLean, VA 22102 (Address of principal executive offices) (Zip Code) (703) 902-2800 (Registrant's telephone number, including area code) Securities registered pursuant to Section 12(b) of the Act: Title of each class Name of each exchange on which registered None Securities registered pursuant to Section 12(g) of the Act: Common Stock, par value \$0.001 per share **Contingent Value Rights** Indicate by check mark if the registrant is a well-known seasoned issuer, as defined in Rule 405 of the Securities Act. Yes 🗆 No 🗵 Indicate by check mark if the registrant is not required to file reports pursuant to Section 13 or Section 15(d) of the Act. Yes \square No \boxtimes Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes \boxtimes No \square Indicate by check mark whether the registrant has submitted electronically and posted on its corporate Website, if any, every Interactive Data File required to be submitted and posted pursuant to Rule 405 of Regulation S-T during the preceding 12 months (or for such shorter period that the registrant was required to submit and post such files). Yes \square No \square Indicate by check mark if disclosure of delinquent filers pursuant to Item 405 of Regulation S-K is not contained herein, and will not be contained, to the best of the registrants' knowledge, in definitive proxy or information statements incorporated by reference in Part III of this Form 10-K or any amendment to this Form 10-K. ⊠ Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, or a smaller reporting company. See the definitions of "large accelerated filer," "accelerated filer," and "smaller reporting company" in rule 12b-2 of the Exchange Act. (Check one): Large accelerated filer \square Accelerated filer □ Non-accelerated filer \square (Do not check if a smaller reporting company) Smaller reporting company ⊠ Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Act). Yes \Box No \boxtimes Non-affiliates of Primus Telecommunications Group, Incorporated held 119,500,670 shares of Common Stock as of June 30, 2009. The fair market value of the stock held by non-affiliates is \$1,792,510 based on the sale price of the shares on June 30, 2009. Indicate by check mark whether the registrant has filed all documents and reports required to be filed by Section 12, 13 or 15(d) of the Securities Exchange Act of 1934 subsequent to the distribution of securities under a plan confirmed by a court. Yes \boxtimes No \square As of March 31, 2010, 9,743,157 shares of Common Stock, par value \$.001, were outstanding.

Documents Incorporated by Reference:Portions of the definitive Proxy Statement to be delivered to Stockholders in connection with the Annual Meeting of Stockholders are incorporated by

Item

TABLE OF CONTENTS

Description

Page

	Part I			
1.	Business	3		
1A.	Risk Factors	26		
1B.	Unresolved Staff Comments	40		
2.	Properties	40		
3.	Legal Proceedings	41		
4.	(Removed and Reserved)	41		
	Part II			
5.	Market for Registrant's Common Equity, Related Stockholder Matters and Issuer Purchases of Equity Securities	42		
6.	Selected Financial Data	44		
7.	Management's Discussion and Analysis of Financial Condition and Results of Operations	47		
7A.	Quantitative and Qualitative Disclosures about Market Risk	74		
8.	Financial Statements and Supplementary Data	75		
9.	Changes in and Disagreements with Accountants on Accounting and Financial Disclosure	76		
9A.	Controls and Procedures	76		
9B.	Other Information	77		
Part III				
10.	Directors, Executive Officers and Corporate Governance	78		
11.	Executive Compensation	78		
12.	Security Ownership of Certain Beneficial Owners and Management and Related Stockholder Matters	78		
13.	Certain Relationships and Related Transactions, and Director Independence	78		
14.	Principal Accountant Fees and Services	79		
Part IV				
15.	Exhibits and Financial Statement Schedule	80		

PART I

ITEM 1. BUSINESS

General

We are an integrated facilities-based communications services provider offering a portfolio of international and domestic voice, wireless, Internet, VoIP, data and data center services to customers located primarily in the United States Australia, Canada, Brazil, the United Kingdom and western Europe. Our primary markets are Australia and Canada where we have deployed significant network infrastructure. We classify our services into three categories: Growth Services, Traditional Services and Wholesale Services. Our focus is on expanding our Growth Services, which includes our broadband, IP-based voice, local, wireless, data and data center services, to fulfill the demand for high quality, competitively priced communications services. This demand is being driven, in part, by the globalization of the world's economies, the global trend toward telecommunications deregulation and the migration of communication traffic to the Internet. We manage our Traditional Services, which includes our domestic and international long-distance voice, prepaid cards, dial-up Internet services and Australian offnetwork local services for cash flow generation that we reinvest to develop and market our Growth Services, particularly in our primary markets of Australia and Canada. We provide our wholesale voice termination services to other telecommunications carriers and resellers requiring IP or time-division multiplexing access.

Through our increased investments in sales and marketing spending focused on our Growth Services we believe that over time our growth in revenue from Growth Services will exceed the decline in our revenue from Traditional Services and Growth Services will increase as a percentage of our total retail revenue.

We target customers with significant telecommunications needs, including SMEs, multinational corporations, residential customers, and other telecommunication carriers and resellers. We provide services over our global, facilities-based network, which consists of:

- 10 data centers in 8 cities in Canada, Australia, and Brazil
- 18 carrier-grade international gateway and domestic switching systems (the hardware/software devices that direct voice traffic across the network), Internet routers and media gateways in the U.S., Canada, western Europe and the Asia-Pacific region;
- approximately 500 interconnection points to our network, or POPs, which includes DSLAMs, which is equipment that allows digital traffic to flow over copper wiring, within our service regions and other markets;
- · undersea and land-based fiber optic transmission line systems that we own or lease and that carry voice and data traffic across the network;
- a global network that uses a high-bandwidth network standard ATM+IP; and
- a global VoIP network based on routers and gateways with an open network architecture which connects our partners in over 150 countries.

Our Services

Within our three main service categories, Growth Services, Traditional Services and Wholesale Services, we offer our customers a wide range of products, including:

Growth Services:

- data center services (colocation and managed services).
- Internet, including business and residential broadband services;
- residential and business VoIP services through SIP and hosted IP-PBX;

- local services:
- wireless Mobile Virtual Network Operator (MVNO) services; and
- data, including frame relay.

Traditional Services:

- domestic and international long-distance voice services;
- prepaid phone card services;
- · local landline services in Australia that do not utilize our owned network (or "off-net" or "off-network" services); and
- dial-up Internet services.

Wholesale Services:

Internet-based protocol and time-division multiplexing wholesale access and transport.

We continue to selectively target growth opportunities in areas of growing demand for telecommunications, including broadband, VoIP, local, wireless, data and data hosting center services. In order to accomplish this objective, we have sought, and continue to seek to:

- manage our Traditional Services business for cash flow generation and prudently reinvest that cash flow in additional sales and marketing spending
 to gain market share, develop innovative and differentiated services, expand coverage and capacity of our existing network infrastructure to service
 new customers and to migrate existing customers onto our network and meeting our debt service obligations;
- prudently expand Growth Services through our enhanced sales and marketing programs in order to steadily improve net revenue and contribution from these Growth Services; and
- make targeted and prudent investments in support systems intended to reduce customer churn and customer service costs.

Strategy

We have refocused our strategy and efforts to enhance the revenue growth from our Growth Services while managing the decline in revenue from our Traditional Services. Key elements of our strategy to achieve these objectives are the following:

- Capitalize on our competitive position in our primary markets to expand Growth Services revenue. We have refocused efforts to transition away from our traditional businesses of domestic and international long-distance voice, prepaid cards and dial-up Internet services to an integrated provider of broadband, VoIP, local, wireless, data and data center services. We believe there is strong demand for high quality, competitively priced communications services that is being driven by the globalization of the world's economies, the global trend toward telecommunications deregulation and the migration of communication traffic to the Internet. We believe that we can capitalize on this demand given our competitive position by offering bundled communications services at attractive prices, particularly where we have a competitive cost structure as a result of our facilities-based network. We are leveraging our brand recognition, network infrastructure and existing customer base in Australia and Canada where we have competitive operations by virtue of our scale to bundle Growth Services and capture new customers.
- Focus on our primary markets of Australia and Canada where we have established brand-recognition. We are focused on our primary markets of Australia and Canada where we believe we have the

scalability and service offerings to effectively compete. For the year ended December 31, 2009, net revenues from our Australia and Canada operations, collectively, generated approximately 57% of our total net revenues and 78% of our total retail net revenue. We are focused on making prudent and effective investments in Australia and Canada where we believe that these markets present us with the highest potential for future growth. Supporting our strategy, during the year ended December 31, 2009 we spent in excess of 90% of our total capital expenditures in our primary markets of Australia and Canada. Our target customer base consists of SMEs, multinational corporations and residential customers.

- Bundle traditional voice services with growth products to increase margins and create customer loyalty. By bundling our traditional long-distance voice services with one or more of our local, broadband, wireless, data and data center services, combined with a prudent increase in sales and marketing investment, we seek to increase net revenue per customer and improve our competitive ability to attract and retain additional new business and residential customers.
- Leverage our global facilities-based network infrastructure. We have a significant amount of capital already invested in our global, voice and data network and our product capabilities and we are making additional investments in support of our Growth Services. By increasing the volume of voice and data traffic that we carry over our network, we are able to reduce transmission costs and other operating costs as a percentage of net revenue and enhance our ability to introduce new products and services with more efficient capital expenditures through utilization of available network capacity. In addition, by leveraging multiple customer segments in different geographical regions, including retail and carrier customers, we achieve greater utilization of our network assets, because our network experiences multiple periods of peak usage throughout each day. We are committed to migrating existing customers to our owned facilities (or "on-net") that in most instances generate significantly higher margins than off-network customers.
- Retail portfolio optimization. Given our retail focus in our primary markets of Australia and Canada where we have extensive operations, we believe there should be opportunities to optimize our remaining retail portfolio to accomplish our strategic objective of positioning us in Growth Services. We will opportunistically assess our alternatives to optimize the economic return from such on-going portfolio restructuring efforts.
- Manage Traditional Services for cash flow which we will reinvest into our Growth Services and use to reduce our debt. The overall market for domestic and international long-distance voice, prepaid cards and dial-up Internet services continues to decline in favor of Internet-based, wireless and broadband communications. To mitigate this revenue decline and preserve cash flow, we focus on customer retention by bundling services, enhancing the customer service experience, and by prudently investing in effective marketing campaigns. For the year ended December 31, 2009, Traditional Services comprised 43% of total net revenue. To further manage cash flow from Traditional Services, we will continue to explore and implement outsourcing or off-shoring options for our non-sales and marketing operations in order to lower costs, improve coordination among our business units to deliver synergy savings and run an aggressive cost management program. We seek to redeploy associated savings and resulting available cash flow into sales and marketing activities for our Growth Services.

2009 Voluntary Reorganization under Chapter 11

On March 16, 2009, Primus Telecommunications Group, Incorporated ("Group" or "PTGI") and three of its non-operating holding company subsidiaries, Primus Telecommunications Holding, Inc. ("Holding" or "PTHI"), Primus Telecommunications International, Inc. ("PTII") and Primus Telecommunications IHC, Inc., ("IHC" and together with Group, Holding and PTII, collectively, the "Holding Companies") each filed a voluntary petition in the United States Bankruptcy Court for the District of Delaware (the "Bankruptcy Court") for reorganization relief ("Reorganization") under chapter 11 of title 11 of the United States Code, 11 U.S.C. §§ 101 et seq., as amended (the "Bankruptcy Code").

None of Group's operating companies in the United States, Australia, Canada, India, Europe and Brazil were included in the Bankruptcy Court filing. These operating units continued to manage and to operate their businesses normally during the financial reorganization of the Holding Companies.

The consensual plan of reorganization (the "Plan" or "Plan of Reorganization") was confirmed by the Bankruptcy Court on June 12, 2009 (the "Confirmation Date"). On July 1, 2009 (the "Effective Date"), the Holding Companies consummated their reorganization under the Bankruptcy Code and the Plan became effective. Upon our emergence under the Plan of Reorganization on July 1, the Holding Companies' principal debt was reduced by \$316 million, or over 50%, interest payments were reduced by over 50% and certain debt maturities were extended. For additional detail concerning the effects of the Plan of Reorganization, see Item 7. Management's Discussion and Analysis of Financial Condition and Results of Operations—Emergence from Voluntary Reorganization under Chapter 11 Proceedings.

Operating Segments

We operate our business segments through a corporate shared services model and through operating segment managers responsible for specific geographic regions in which we operate. We assess performance and allocate resources based on the following operating segments:

Australia. Our Australian operations represented 30% of our net revenue for the year ended December 31, 2009. We believe we are the fourth largest full-service provider in Australia, based upon net revenue. Our Australian operations offers a comprehensive range of international and domestic long-distance voice, broadband and dial-up Internet, wireless, local, VoIP, data and data hosting center products, servicing both residential and business sectors. Our network employs Nortel DMS-100 telephone exchanges in Sydney, Melbourne, Brisbane, Perth and Adelaide operating through 66 POPs to provide national coverage for voice services. We operate a prepaid calling platform as well as a platform for delivery of enhanced toll-free service. The voice network supports direct access integrated services digital network ("ISDN") and telephone line services across Sydney, Melbourne, Brisbane, Perth and Adelaide and some select regional areas of the country. VoIP services are offered to business and consumer customers using softswitch platforms operating in Sydney and Melbourne. Primus Australia owns a national IP network for delivery of business and consumer Internet service. Dial-up Internet is available nationally under a single access code. DSL service is provided on-net through 252 Primus DSLAMs and via wholesale DSLAM access providing reach to more than 400 exchanges nationally. The DSLAMs are capable of delivering a full suite of telecommunication products including asymmetric and symmetric IP services, telephone line, ISDN, frame relay and ATM. Primus Australia offers ATM, frame relay, IP virtual private networking ("VPN") with quality of service ("QoS") and managed routers. Metropolitan fiber networks exist in Sydney, Melbourne, Brisbane, Perth and Adelaide to provide high capacity backhaul for domestic carrier interconnects, DSLAM backhaul and fiber connectivity to select customer premises.

A data center in Melbourne offers managed hosting, colocation, and e-commerce applications. Our data center in Sydney is primarily used for colocation services. The Company plans to add growth capacity to these locations in 2010.

We market our services primarily through a combination of direct sales to independent agents, corporate and SME customers, internet based marketing aimed at residential customers, and telemarketing and media advertising aimed at residential customers.

We operate a call center in Melbourne that services all of Australia. We employ staff in Sydney who run our Australian network management center that operates seven days per week, 24 hours per day, 365 days per year.

Canada. Our Canadian operations represented 27% of net revenue for year ended December 31, 2009. We believe we are the largest alternative consumer service providers in Canada based on net revenue. We provide international and domestic long-distance voice, local, broadband and dial-up Internet, data and data center services, VoIP and wireless services to SMEs, residential customers and government agencies. We have sales and customer service offices in key cities throughout Canada, including Vancouver, Toronto, Ottawa, Calgary

and Edmonton. We operate international gateway switches in Toronto and Vancouver and a nationwide SS7 network with signal transfer points ("STPs") in Vancouver and Toronto. We maintain POPs in all major cities in Canada, and have use of a nationwide integrated network backbone for our voice, data, Internet and private line services. Each of the 24 nodes on the backbone is equipped with synchronous optical networking ("SONET") add/drop, ATM, and IP equipment to provide a complete spectrum of voice and data communications products to our customers. In September 2009, we signed a new contract with Rogers Wireless Partnership which will significantly expand our coverage area and provide increased wireless data functionality to customers for future smart phone offerings. Additionally, we operate one Nortel DMS 500 switch in Toronto and next generation IP voice switches in Toronto, Vancouver, Montreal, Ottawa, London and Edmonton which provide on-net equal access coverage to an estimated 90% of the population of Canada. Together with and through a competitive local exchange carrier ("CLEC") partner, we are co-located at 70 incumbent local exchange carriers ("ILECs") central offices to provide DSL services, T1 access, network interconnection and local dial-tone. We operate a voice dial access network which consists of 70 POPs across the country.

We also own and operate a total of seven data centers in five major cities in Canada: two data centers in Ottawa (totaling 30,000 square feet), two data centers in Toronto (totaling 16,000 square feet), a 18,000 square foot data center in London, Ontario, a 6,000 square foot data center in Edmonton, Alberta and a 3,000 square foot data center in Vancouver through which we offer colocation, managed hosting, storage and other value added services. We have an extensive Internet network and we believe that we provide dial-up and ISDN Internet coverage to over 700 communities across Canada through a network of 51 POPs.

We market our services primarily through a combination of direct sales to independent agents, corporate and SME customers, internet based marketing aimed at residential customers, and telemarketing and media advertising aimed at residential customers.

Wholesale. Our Wholesale operations represented 26% of our net revenue for the year ended December 31, 2009. Our Wholesale business segment provides domestic and international minute transport and termination services and targets a diverse customer base including Tier 1 international carriers, multinational carriers, wireless providers, VoIP providers, cable companies and ISPs.

United States. Our U.S. operations represented 8% of our net revenue for the year ended December 31, 2009. In the U.S., we provide international and domestic voice, data, broadband, hosted IP-PBX, SIP trunking and VoIP services to SMEs and residential customers. We operate international gateway telephone switches in the New York City area, Los Angeles, and Miami which are connected with Canada and countries in Europe, Latin America and the Asia-Pacific region through owned and leased international fiber cable systems. We have deployed intelligent softswitch architecture to our gateways in New York, Los Angeles and Miami. We lease and own domestic fiber in the U.S. to interconnect our switches, data centers, and POPs. We use a direct sales organization to sell to business customers. Our direct sales personnel offer business customers voice and hosted IP-PBX services. See "—Sales and Marketing; Direct Sales Force." To reach residential customers, we advertise in national and regional newspapers, other publications, and television channels whose market targets are expatriates, to offer competitive rates for international and domestic telephone calls, data, Internet and VoIP services. We have telemarketing centers in Florida and Iowa. We also sell retail VoIP services through Web-based on-line interactive marketing. We utilize independent agents to reach and enhance sales to both business and residential customers. We maintain customer service centers in Florida, Virginia and Iowa and also outsource selected customer service functions. We operate a 24-hour global network management control center in India which monitors our global voice, Internet and data traffic. Additionally, we offer local and long-distance voice, and DSL services to SMEs in Puerto Rico.

Europe. We operate as a licensed service provider in the United Kingdom, France, Spain, Italy, Belgium and the Netherlands. The European market represents 7% of our net revenue for the year ended December 31, 2009. Our network consists of voice/data switches in London, Paris, Milan and Madrid and an Internet router in Frankfurt. Our European network interconnects with our global network. We have data centers for hosting and related services in London and Paris.

Our European operations are headquartered in London. We provide voice, data, wireless and hosted PBX services to residential customers, SMEs and callshops. We market our services across Europe using a combination of a direct sales force, independent sales agents and direct media advertising.

Brazil. Our Brazilian operations, headquartered in São Paulo, represented 2% of our net revenue for the year ended December 31, 2009. Our Brazilian business segment primarily consists of data hosting center services and retail and wholesale VoIP services.

Network

General. We operate a global telecommunications network consisting of traditional and next-generation international gateway and domestic switches and related peripheral equipment, carrier-grade routers and switches for Internet and data services, data hosting and co-location centers, and undersea and transcontinental fiber optic cable systems. To ensure high-quality communications services, our network employs digital switching and fiber optic technologies, incorporates the use of SS7/C7 signaling, and is supported by comprehensive network monitoring and technical support services.

Switching Systems. Our network makes use of 18 carrier-grade domestic and international gateway switch systems, Internet routers and media gateways throughout the North America, Europe, and Asia-Pacific regions.

Time of Cartal

The locations and types of our switching systems are as follows:

Location	Type of Switch
New York City area (two locations)	International Gateway
Los Angeles	International Gateway
Toronto	International Gateway
Vancouver	International Gateway
London	International Gateway
Paris	International Gateway
Sydney	International Gateway
Milan	International Gateway
Madrid	International Gateway
Hong Kong	International Gateway
Miami	Internet Peering POP
Frankfurt	Internet Peering POP
Makati City	Internet Peering POP
Adelaide	Domestic
Brisbane	Domestic
Melbourne	Domestic
Perth	Domestic

We also operate a global VoIP network with an open network architecture which connects with our partners in over 150 countries through the use of open settlement protocol.

Fiber Optic Cable Systems. We have purchased and leased undersea and land-based fiber optic cable transmission capacity to connect our various switching systems. We either lease lines on a term basis for a fixed cost or purchase economic interests in transmission capacity through minimum assignable ownership units or indefeasible rights of use to international traffic destinations.

Throughout the previous years we have purchased or acquired through acquisitions various oceanic fiber capacity and land based capacity. This capacity along with leased fiber capacity allows our switching platforms in our operating units to be connected and pass voice and data traffic.

Foreign Carrier Agreements. In selected countries where competition with the traditional Post Telegraph and Telecommunications companies ("PTTs") is limited, we have entered into foreign carrier agreements with PTTs or other service providers which permit us to provide traffic into and receive return traffic from these countries.

Network Management and Control. We own and operate network management control centers in Toronto, Canada; London, England; New Delhi, India; and Sydney, Australia, which are used to monitor and control our switching systems, global data network, and other digital transmission equipment used in our network. These network management control centers operate seven days per week, 24 hours per day, 365 days per year.

Network for Data and Internet Services. We have built an Internet backbone network that enables our global network to carry Internet and data traffic for our business, residential, carrier and ISP customers. This network uses packet switched technology, including IP and ATM. This network allows us to offer to customers data and voice communications services, including, dial-up, broadband and dedicated Internet access, hosting, e-commerce, managed VPN services, VoIP, ATM and frame relay data services.

Data Centers. We offer world-class data center facilities with advanced 24 hours per day, 365 days per year customer access, onsite engineering support and help desk services; dedicated HVAC and environmental control systems; multi-stage fire suppression systems; uninterruptible power supply and backup generator; redundant data connections and services; routing and switching; shared and secure rack space; physical access technologies and practices; closed-circuit television and video security systems; and 24 x 7 building system and network monitoring. Our Australian data center facilities in Melbourne and Sydney occupy approximately 15,000 square feet. Canada offers national data center coverage with locations in Toronto, Vancouver, Edmonton, London, and Ottawa, with a total combined square footage of 73,000 square feet. Additionally, we own a data center facility in Brazil.

Customers

Our residential customers are attracted to us because of competitive pricing as compared to traditional carriers and responsive customer service and support. We offer VoIP, broadband and dial-up Internet access, local access and wireless products to our residential customers in select markets. We are expanding our local and broadband offerings to additional markets and bundling them with traditional voice services.

Our business sales and marketing efforts primarily target SMEs with data, voice, internet and managed hosting needs. We also target large multinational businesses.

We compete for the business of other telecommunications carriers and resellers primarily on the basis of price, service quality and presence. Sales to other carriers and resellers help us to negotiate better termination costs for our retail business units and increase the utilization of our network, thereby reducing our fixed costs per minute of use, as well as permitting our network to be interconnected with other major carriers to provide global coverage.

No single customer accounted for greater than 10% of net revenue for the year ended December 31, 2009 or for the year ended December 31, 2008.

Sales and Marketing

We market our services through a variety of sales channels, as summarized below:

• *Direct Sales Force.* Our direct sales force is focused on business customers, including multinational businesses and international governmental organizations. They are engaged in generating new accounts and in the retention of current customers and cross selling products to current customers. A variety of products are utilized to maximize the customer's lifecycle, based on the customers current product mix.

Direct sales personnel are generally compensated with a base salary plus commissions. We have sales offices in Edmonton, Calgary, Toronto, Vancouver, Ottawa, London, Glasgow, Paris, Adelaide, Brisbane, Melbourne, Perth, Sydney, Sao Paulo, Chicago and Tampa. In addition, we maintain a direct sales team which exclusively sells services to wholesale customers including other telecommunications carriers and resellers.

- *Independent Sales Agents*. We also sell our services through independent sales agents and representatives, who typically focus on SMEs and residential consumers. An agent receives commissions based on revenue generated by customers obtained for us by the agent. The agents are nonexclusive to us and we require our agents and representatives to maintain minimum revenues.
- *Telemarketing*. We employ full-time and part-time inbound and outbound telemarketing sales personnel, and we selectively outsource certain telemarketing functions to supplement sales efforts targeted at residential and small business customers.
- *Direct Marketing*. We use a variety of print, television and radio advertising to increase name recognition and generate new customers. We reach potential customers by advertising in newspapers and on select radio and television programs.
- Web-based Marketing. We use a variety of web-based tools, including search marketing, banner ads and pop-up windows to target Internet users who prefer to purchase and research their purchase decisions on-line.
- Third Party Distribution Agreements and Affinity Channels. Through use of the Primus brand, we have been able to establish relationships to market our services through external retailers, manufacturers, affinity and preferred partnerships and programs. These relationships allow us to increase awareness of our services among customers and reduce the cost of customer acquisition.

Management Information and Billing Systems

We operate various management information, network and customer billing systems in our different operating subsidiaries to support the functions of network and traffic management, customer service and customer billing. For financial reporting, we consolidate information from each of our markets into a single database. For our billing requirements we use several systems developed in-house as well as a few third party systems.

We believe that our financial reporting and billing systems are generally adequate to meet our business needs. We continually consider and evaluate whether our financial reporting and billing systems are generally adequate to meet our business needs, and in the future, we may determine that we need to invest additional capital to purchase hardware and software, license more specialized software and increase our capacity.

Competition

Voice

The telecommunications industry is highly competitive and significantly affected by regulatory changes, marketing and pricing decisions of the larger industry participants and the introduction of new services made possible by technological advances. We believe that long-distance service providers compete on the basis of price, customer service, product quality and breadth and bundling of services offered. In each country of operation, we have numerous competitors including wireline, wireless, VoIP and cable competitors. We believe that as the international telecommunications markets continue to deregulate, competition in these markets will increase. Prices for long-distance voice calls in the markets in which we compete have declined historically and are likely to continue to decrease. In addition, many of our competitors are significantly larger, have substantially greater financial, technical and marketing resources, larger networks and more products for bundling.

Privatization and deregulation have had, and are expected to continue to have, significant effects on competition in the industry. For example, as a result of legislation enacted in the U.S., regional bell operating companies ("RBOCs") entered the long-distance market; long-distance carriers have entered the local telephone services market; and cable television companies and utilities are allowed to enter both the local and long-distance telecommunications markets. Consolidation among these large companies is also occurring, which could change the dynamics of pricing and marketing. In addition, competition has begun to increase in the European Union ("EU") communications markets in connection with the deregulation of the telecommunications industry. In addition, alternatives to wireline services, such as wireless and VoIP services, are significant competitive threats. This increase in competition adversely affects net revenue per minute and usage of traditional wireline services, and these trends are expected to continue.

The following is a brief summary of the competitive environment in our principal service regions:

United States. In the U.S., which is among the most competitive and deregulated long-distance markets in the world, competition is based on pricing, customer service, network quality and the ability to provide value-added services and the bundling of services. AT&T, Inc. and Verizon Communications, Inc. are the largest suppliers of long-distance services. Wireless carriers have gained significant ground particularly in the domestic long-distance markets, and VoIP cable-based service providers present a growing source of competition.

Australia. Australia is one of the most deregulated and competitive communications markets in the Asia-Pacific region. Our principal competitors in Australia are Telstra, the dominant carrier, SingTel Optus Pty Limited, iiNet Limited, SP Telemedia Limited (known as TPG) and AAPT, an affiliate of Telecom New Zealand. Repeated pricing threats and actions by Telstra present serious competitive challenges (see "Regulation—Australia"). In September, 2009, the Australian government has proposed the separation of Telstra's wholesale and retail arms in order to promote more competition and better prices.

Canada. The Canadian communications market is highly competitive and is dominated by a few established carriers whose marketing and pricing decisions have a significant impact on the other industry participants, including us. In residential markets, we compete with each of the incumbent telecommunication companies (of which the largest are those owned by Bell Canada in eastern Canada, and TELUS and MTS Allstream, Inc. in western Canada) in their respective territories and the large cable companies, including Rogers and Shaw Communications Inc., who have launched their telecom service portfolio. We also compete against smaller resellers. In the highly competitive business market, we compete with Bell Canada and TELUS, who are both expanding beyond their traditional territories and competing with each other across the country, and with the national division of MTS Allstream, Inc., Rogers, Shaw Communications, Inc. and other smaller carriers, including Bragg Communications Inc., COGECO Inc. and Quebecor Inc. Major wireless carriers are also a significant source of competition. In 2008 the CRTC conducted a review of the regulatory framework for wholesale services and the definition of essential services. See "Regulation—Canada."

United Kingdom. Our principal competitors in the United Kingdom are BT Group plc, the dominant provider of telecommunications services in the United Kingdom, ntl:Telewest Business, Carphone Warehouse Group plc, Cable & Wireless plc, Colt Telecom Group S.A. and MCI/Verizon Communications Inc. Major wireless carriers are also a significant source of competition.

Internet and Data

The market for Internet services and data services is extremely competitive. We anticipate that competition will continue to intensify. Our current and prospective competitors offering these services include national, international, regional and local ISPs, web hosting companies, other long-distance and international long-distance telecommunications companies, local exchange carriers ("LECs"), cable television, direct broadcast satellite, wireless communications providers and on-line service providers. Many of these competitors have significantly greater resources, product portfolios, market presence and brand recognition than we do.

Government Regulation

We are subject to varying degrees of regulation in each of the jurisdictions in which we operate. Local laws and regulations, and the interpretation of such laws and regulations, differ among those jurisdictions. There can be no assurance that (1) future regulatory, judicial and legislative changes will not have a material adverse effect on us; (2) domestic or international regulators or third parties will not raise material issues with regard to our compliance or noncompliance with applicable regulations; or (3) regulatory activities will not have a material adverse effect on us.

Regulation of the telecommunications industry continues to change rapidly both domestically and globally. Privatization, deregulation, changes in regulation, consolidation, and technological change have had, and will continue to have, significant effects on competition in the industry. Over time we have become subject to increasing competition in all of the markets in which we operate, and we expect such increases in competitive pressure to continue. Although we believe that continuing deregulation with respect to portions of the telecommunications industry will create opportunities for firms such as us, there can be no assurance that deregulation and changes in regulation will be implemented in a manner that would benefit us.

The regulatory frameworks in certain jurisdictions in which we provide services are described below:

United States

In the U.S., our services are subject to the provisions of the Communications Act, the FCC's regulations, and the applicable laws and regulations of the various states.

Certain of our services (our "telecommunications services") constitute common carrier offerings subject to Title II of the Communications Act and associated FCC regulations and similar state laws. Among other things, these requirements impose an obligation to offer service on a nondiscriminatory basis at just and reasonable rates, and to obtain regulatory approval prior to withdrawing from the provision of regulated services, to any assignment of authorizations, or to any transfer of legal or actual control of the company.

Our interstate telecommunications services are subject to various specific common carrier telecommunications requirements set forth in the Communications Act and the FCC's rules, including operating, reporting and fee requirements. Both federal and state regulatory agencies have broad authority to impose monetary and other penalties on us for violations of regulatory requirements.

International Service Regulation. The FCC has jurisdiction over common carrier services linking points in the U.S. to points in other countries. We provide such services. Providers of such international common carrier services must obtain authority from the FCC under Section 214 of the Communications Act. We have obtained the authorizations required to use, on a facilities and resale basis, various transmission media for the provision of international switched services and international private line services on a non-dominant carrier basis. The FCC is considering a number of possible changes to its rules governing international common carriers. We cannot predict how the FCC will resolve those issues or how its decisions will affect our international business. FCC rules permit non-dominant carriers such as us to offer some services on a detariffed basis, where competition can provide consumers with lower rates and choices among carriers and services. With some exceptions, current FCC rules require facilities-based U.S. carriers, like us, with operating agreements with dominant foreign carriers, to abide by the FCC's International Settlements Policy by following uniform accounting rates, an even split in settlement rates, and proportionate return of traffic, for agreements with carriers on certain routes. U.S. carrier arrangements with non-dominant foreign carriers or on a substantial number of international routes where competition exists are not subject to these requirements. We may take advantage of these more flexible arrangements with non-dominant foreign carriers, and the greater pricing flexibility that may result, but we may also face greater price competition from other international service carriers.

Domestic Service Regulation. With respect to our domestic U.S. telecommunications services, we are considered a non-dominant interstate carrier subject to regulation by the FCC. We are not required to obtain FCC authority to initiate or expand our domestic interstate operations, but we may be required to obtain FCC approval to assume control of another telecommunications carrier or its assets, to transfer control of our operations to another entity, or to discontinue service. We are also required to file various reports and pay various fees and assessments to the FCC and various state commissions. Among other things, interstate common carriers must offer service on a nondiscriminatory basis at just and reasonable rates. The FCC has jurisdiction to hear complaints regarding our compliance or non-compliance with these and other requirements of the Communications Act and the FCC's rules, including, for example, alleged unauthorized changes in a customer's preferred carrier. We are also subject to CALEA and associated FCC regulations which require telecommunications carriers and VoIP providers to configure their networks to facilitate law enforcement authorities to perform electronic surveillance; to the Do-Not-Call Registry and related restrictions set out the specific parameters for telemarketing solicitation and prohibit outbound telemarketing in some circumstances; and to the Controlling the Assault of Non-Solicited Pornography and Marketing Act of 2003, which places certain restrictions on commercial electronic mail messages.

Normally our long-distance customers, and the recipients of calls from our customers, do not connect directly to our network but, instead, are connected to our network by means of the local facilities of LECs, which impose regulated charges on us, called switched access charges, to originate and terminate calls over their local facilities. FCC rules cap these charges, and, in the case of incumbent LECs, impose additional restrictions on their ability to change their charges. The FCC may modify its rules regarding LEC access charges, including both the caps and the restrictions on incumbent LECs, which could result in an increase in our termination and origination costs over time.

Interstate telecommunications carriers and VoIP providers are required to contribute to the federal USF. The FCC is considering revising its USF mechanisms and the services considered when calculating the USF contribution. We cannot predict the outcome of these proceedings or their potential effect on our USF contributions. Some changes to the USF under consideration by the FCC may affect different entities more than others, and we may be disadvantaged as compared to our competitors as a result of FCC decisions regarding USF. In addition, the FCC may extend the obligation to contribute to the USF to certain services that we offer but that are not currently assessed USF contributions.

Voice-over-Internet Protocol (VoIP). VoIP services that permit subscribers to send calls to and receive calls from the traditional telephone network, known as interconnected VoIP services, are generally subject to the regulatory authority of the FCC. We offer such services. For many years the FCC has been considering whether to classify such services as "telecommunications" services or "information" services under the Communications Act. Information services are generally subject to less regulation than telecommunications services. However, the FCC has extended a number of regulatory obligations normally applicable to telecommunications services to VoIP services, irrespective of the lack of clarity regarding the classification of such services. These include (1) the obligation to contribute to the USF; (2) the obligation to comply with CALEA; (3) the obligation to comply with the FCC's rules regarding protecting CPNI; (4) the obligation to provide E911 services, and to disclose limitations on such services to end users; (5) the obligation to permit customers to keep their telephone number when changing from one carrier to another; (6) the obligation to deploy equipment and provide services that make reasonable accommodations for the needs of disabled persons; and (7) the obligation to pay a variety of regulatory fees. We cannot predict what other regulatory obligations, if any, will be imposed on our VoIP operations, nor can we predict whether the FCC will eventually classify VoIP as a telecommunications service and, if it does, what additional regulatory obligations, if any, the FCC would choose to impose on such services. The uncertainty regarding the regulatory obligations applicable to interconnected VoIP services presents a number of situations which could have a material negative impact on our operating costs and profitability. These potential regulatory developments include:

• The FCC has ruled that a certain type of interconnected VoIP service is interstate in nature, and not subject to most state-level public utility regulations. We believe that our own VoIP service is subject to

- this ruling, but individual states may nonetheless attempt to regulate that service. We cannot predict whether any individual states would make such an effort, nor can we predict the outcome of any such effort.
- While we believe that our VoIP network is in compliance with the requirements of CALEA, we cannot predict whether law enforcement or the FCC will agree, nor can we predict whether we may be subject to fines or penalties if we are found to be not in compliance with CALEA.
- With respect to our VoIP-related 911 obligations, we have contracted with a third-party provider that is a market leader in emergency 911 service solutions to provide these services. Our ability to expand our VoIP services in the future may depend on the ability of that provider to provide E911 access or on any additional 911-related obligations the FCC may impose. At present, and similar to many other VoIP providers, for some of our customers we cannot offer VoIP E911 services that route emergency calls in a manner fully consistent with the FCC rules. We are addressing this issue with our VoIP E911 solutions provider. The FCC may determine that our VoIP E911 solution for some of our customers does not satisfy the FCC's requirements, in which case the FCC could require us to disconnect a significant number of our VoIP subscribers, which could have a material adverse effect on the Company's financial position, results of operations and cash flows. In addition, the FCC is considering whether it should impose additional VoIP E911 obligations on interconnected VoIP providers including consideration of a requirement that interconnected VoIP providers automatically determine the physical location of their customer rather than allowing customers to manually register their location. We cannot predict the outcome of this proceeding, nor its impact on us, at this time.
- The FCC's CPNI rules, to which our VoIP operations are subject, are intended to protect customer information from unauthorized use or disclosure. Those rules currently require us to use enhanced passwords for customer on-line and call-in access to CPNI and account information, as well as customer notification of account or password changes. Complying with these rules may impose additional costs on us and reduce our profitability or cause us to increase the retail price for our services. We may be subject to enforcement actions including, but not limited to, fines, cease and desist orders, or other penalties if we are not able to comply with these new CPNI obligations.
- Section 255 of the Communications Act and associated FCC rulings require traditional phone providers and manufacturers of associated equipment
 to ensure that their equipment and services meet certain obligations regarding access by persons with disabilities. The FCC has extended those
 requirement to providers of interconnected VoIP services and to manufacturers of specially designed equipment used to provide those services. While
 we believe that we are in compliance with these requirements, we can have no assurance that the FCC might not find us not to be in compliance,
 which could subject us to sanctions including fines.
- The FCC has also extended to VoIP providers the obligations of Section 225 of the Communications Act, which (among other things) requires contributing to the Telecommunications Relay Services fund, and the offering of 711 abbreviated dialing for access to relay services. At this time, we cannot predict the impact of these rules on our business or our ability to comply with these obligations, and we may be subject to enforcement actions including, but not limited to, fines, cease and desist orders, or other penalties if we are not able to comply with them. For example, the FCC mandates that interconnected VoIP providers like us must transmit 711 calls to a relay center that facilitates the ability of speech- and hearing-impaired individuals to communicate. The FCC issued a temporary waiver of that requirement insofar as it requires such providers to transmit the 711 call to an "appropriate relay center," meaning the relay center(s) serving the state in which the caller is geographically located or the relay center(s) corresponding to the caller's last registered address. We are currently not able to route such calls in this manner, but we are working on implementing a call routing solution which will route 711 calls to the appropriate relay center as defined in the FCC's order. We may be subject to enforcement actions including, but not limited to, fines, cease and desist orders, or other penalties if we are not able to comply with these new disability provisions.

• The FCC now requires interconnected VoIP service providers like us to pay regulatory fees to the FCC. The assessment of these fees has increased our costs and reduced our profitability and caused us to increase the price of our retail service offerings.

State Regulation. Traditional long-distance calls for which the caller and recipient are both located in the same state are subject to regulation by that state. Our intrastate long-distance operations are therefore subject to various state laws and regulations, including, in most jurisdictions, certification and tariff filing requirements. Primus Telecommunications, Inc. ("PTI"), our principal operating subsidiary in the U.S., maintains the necessary certificate and tariff approvals, to provide intrastate long-distance service in 49 states and Puerto Rico. PTI also maintains the necessary certificate to provide local services in Puerto Rico. Certain of our other subsidiaries also maintain certificates and tariffs in some states. Some states also require the filing of periodic reports, the payment of various fees and surcharges and compliance with service standards and consumer protection rules. States often require prior approval or notification for certain stock or asset transfers or, in several states, for the issuance of securities, debt or for name changes. As a certificated carrier, consumers may file complaints against us at the public service commissions. Certificates of authority can generally be conditioned, modified, cancelled, terminated, or revoked by state regulatory authorities for failure to comply with state law and/or rules, regulations and policies of the state regulatory authorities. Fines and other penalties also may be imposed for such violations. Public service commissions also regulate access charges and other pricing for telecommunications services within each state. The RBOCs and other LECs have been seeking reduction of state regulatory requirements, including greater pricing flexibility which, if granted, could subject us to increased price competition. We may also be required to contribute to universal service funds in some states.

State Taxes and Fees Applicable to VoIP Services. In general, we do not collect or remit state or municipal taxes (such as sales and use, excise, utility user, and ad valorem taxes), fees or surcharges on the charges to our customers for our VoIP services. With limited exceptions, we do not believe that we have sufficient nexus in most states/local jurisdictions to be subject to state or municipal taxes, surcharges or other fees. Should this change or found to be otherwise, we may be subject to retroactive liability for VoIP-specific taxes, fees and surcharges in a number of states and potentially, penalties and interest. Retroactive liability for such taxes, fees or surcharges, as well as penalties and interest, may adversely impact our financial position and we would not be able to recoup any of these liabilities from our customers. If we must collect such state taxes, fees and surcharges on a going-forward basis, we will likely pass such charges through to our customers. The impact of this price increase on our customers or our inability to recoup our costs or liabilities could have a material adverse effect on the financial position, results of operations and cash flows of our VoIP business.

Several states have attempted to impose state-level USF charges on a portion of the revenues from interconnected VoIP service. Thus far, such efforts have been unsuccessful. For example, the U.S. Circuit Court of Appeals for the Eighth Circuit recently affirmed a lower court ruling enjoining the state of Nebraska from attempting to collect such fees. On the basis of this litigation, we do not believe that existing law allows states to subject us to state universal service fund contribution obligations with respect to our VoIP services. However, we cannot predict the final outcome of this litigation, nor any similar litigation that other states might initiate, nor its impact on us.

On July 16, 2009, Kansas and Nebraska filed a petition with the FCC requesting a declaratory ruling that states are not preempted from requiring nomadic interconnected VoIP providers, such as us, to contribute to state universal service funds. The petition also seeks a retroactive ruling finding that states have been able to collect such contributions for a time period that is not clearly defined in the petition. At this time, we cannot predict the outcome of this proceeding nor its impact on our business. If we were required to retroactively pay into state-level universal service funds, that could have a material negative impact on our earnings. However, if the FCC were to adopt retroactivity but limit it to Kansas and Nebraska, we do not believe our potential liability would be material.

If we become subject to state USF fees in additional states or other telecommunications-related surcharges, we will likely pass such charges through to our customers. The impact of this price increase on our customers or

our inability to recoup its costs or liabilities in remitting USF contributions or other factors could have a material adverse effect on our financial position, results of operations and cash flows.

Australia

The provision of our telecommunications services is subject to federal regulation in Australia. The two primary instruments of regulation are the Telecommunications Act 1997 (the "Telecommunications Act 1997") and federal regulation of network access pricing and anti-competitive practices pursuant to the Trade Practices Act 1974 (the "Trade Practices Act"). The current regulatory framework came into effect in July 1997.

We are licensed as a carrier under the Telecommunications Act 1997, which permits us to own or operate transmission infrastructure that is used to supply carriage services to the public. With respect to providing carriage services to the public, we must comply with legislated "service provider" rules contained in the Telecommunications Act 1997 covering matters such as operator services, regulation of land access, directory assistance, provision of information to allow maintenance of an integrated public number database and itemized billing.

Two federal regulatory authorities principally exercise control over the broad range of issues affecting the operation of the Australian telecommunications industry. The Australian Communications & Media Authority (the "ACMA") is the authority regulating matters including the licensing of carriers and technical matters, and the ACCC has the role of promotion of competition and consumer protection and in particular dealing with carrier to carrier interconnection and network access. Telstra, the dominant carrier and former Government owned monopoly, has typically challenged many of the key principles applied by the ACCC to access pricing and endeavors to have some key decisions removed from the charter of the ACCC. For example, Telstra has previously applied to the High Court of Australia to overturn its obligation to provide access to unbundled local loop lines (and also access to those lines pursuant to spectrum sharing). While that petition to the highest court in Australia was unsuccessful, Telstra continues to lodge challenges against regulatory decisions, with recent unsuccessful Federal Court challenges relating to the unconditional local loop service ("ULLS") and leased line sharing ("LLS") pricing. Telstra has also lodged a number of exemption applications with the ACCC submitting that it should be exempted from an obligation to provide wholesale telecommunications services on various routes and locations in Australia. Some of these applications have been successful, particularly in respect to wholesale line rental, the local carriage service and public switched telephone network originating access services supplied from certain CBD and metro exchanges, and also some transmission services where the ACCC considered sufficient alternative supply was available. There is a risk that Telstra will significantly increase its prices in respect of these exempt services and/or deny us access to essential services we need to compete. Separately, Telstra has also petitioned the ACCC for the right to impose significantly higher unconditional local loop ("ULL") pricing. The ACCC has made a decision to reject this, however Telstra has challenged this decision in the Federal Court. Telstra and the ACCC are also presently contesting the appropriate pricing for all the fixed line regulated services supplied by Telstra. There is currently a significant divergence of view on some of this pricing and in August 2009 the ACCC published its draft pricing proposing increased ULL pricing. The ACCC subsequently indicated it would withdraw from that proposed pricing, and is expected to publish its final indicative pricing decision in the second quarter of 2010. The ACCC will also be establishing SSS and ULL pricing in the context of access disputes during the second quarter of 2010.

In September 2009 the Australian Government announced a regulatory reform package that if enacted, we expect will significantly reform the industry for the better. The key reforms are rules designed to separate Telstra's retail arm from its wholesale business (proposing scope for either functional or structural separation), and the reform of the access pricing regime to enable the ACCC to conduct more efficient and forward looking pricing inquiries, give the ACCC greater price setting powers and provide greater certainty by removing merits review of ACCC decisions. These reforms generally represent improvement in the conditions and prospects for competitors such as us. However, the removal of merits review does expose us to some risk, as negative outcomes will not be open to challenge. The reforms were subject to debate and consideration by the Federal

Senate during November 2009, and again debated in March 2010, however at this time its not clear whether the Government will obtain the necessary support to pass the legislation. Combined with other measures (including amendment of undertakings and exemption processes), the legislative reforms if and when enacted should ultimately lead to better price certainty and less regulatory risk. Given the reluctance of Telstra to take the lead in constructing a high speed broadband network in Australia, and the failure of a competitive tender to construct such a network (given global conditions at the time), the Government has moved to establish a Government owned company (NBN Co) to construct and operate a National Broadband Network. It is expected that draft legislation will be introduced within the next 6 months setting out the governance and regulatory framework for NBN Co. At this time NBN Co is in discussion with industry in connection with its construction plans. This development represents the prospect of long-term improvement in the conditions for industry competitors such as Primus Telecom, given the Government's intention that NBN Co operate as a stand alone wholesale company, therefore removing some of the friction that currently exists between competitors and the monopoly wholesale service provider, Telstra, which because if its vertically integrated nature has incentives to favour its own retail operations over the interests of its wholesale customers.

The Government has recently announced a policy to require internet service providers to impose mandatory filtering of certain objectionable and illegal material through blocking access to specified URLs. At this stage it's not clear what cost and operational implications this policy will have on Primus. More detail on these implications should emerge over coming months.

In addition to the Trade Practices Act and the Telecommunications Act 1997, the industry is also subject to other federal legislation, state legislation, various regulations pursuant to delegated authority and legislation, ministerial declarations, codes, directions, licenses, and statements of Australian government policy.

Licensed carriers are subject to charges that are intended to cover the costs of regulating the telecommunications industry and are obliged to comply with license conditions (including obligations to comply with the Telecommunications Act 1997 and with the telecommunications access regime and related facilities access obligations). Carriers also must meet the cost of the Universal Service Obligations ("USO"), which assists in providing all Australians, particularly in remote areas, with reasonable access to standard telephone services. Telstra is currently the sole universal service provider. Since 2000, the responsible Minister of the Australian government may make a determination of the amount of USO subsidies, with advice from the ACMA. No methodology is provided in legislation and the Minister could make a determination of a Universal Service Levy ("USL") that would be material to us. However, the USL has been set previously at levels that we do not consider to have a material impact. The ACMA is presently proposing a determination for 2010 that is consistent with earlier determinations

Fair Trading Practices. The ACCC enforces legislation (the Trade Practices Act) for the promotion of competition and consumer protection, and is responsible for regulating rights of access to services (including pricing for access) and interconnection. The ACCC can issue a competition notice to a carrier which has engaged in anti-competitive conduct. Where a competition notice has been issued, the ACCC can seek pecuniary penalties, and other carriers can seek damages. Under the Trade Practices Act other carriers can also initiate their own proceedings in the event of violation of competition laws.

As indicated, the Trade Practices Act provides for a telecommunications access regime that sets out a framework for regulating access rights to specific carriage services and related services through the declaration of access services by the ACCC. The regime establishes mechanisms by which the terms and conditions of access to a declared service can be determined by the ACCC in an access arbitration. The telecommunications specific provisions in the Trade Practices Act are intended to ensure fair and competitive access to essential facilities and monopoly services, and address anti-competitive conduct of carriers, particularly those with substantial market power such as Telstra. The right of access provisions also apply to any carrier that owns or controls essential infrastructure or services that have been declared by the ACCC on grounds that access should be provided for the purposes of competition.

Consumer Protection. The ACCC's consumer protection role as set out in the Trade Practices Act is also shared with other state based regulators. Each state has its own Fair Trading Act administered by consumer affairs authorities and ACMA undertakes some activities in consumer protection predominantly in connection with industry codes of conduct. As a service provider we must also be a member of the Telecommunications Industry Ombudsman ("TIO") Scheme. The TIO is responsible for handling complaints from consumers about carriers and Internet service providers. The TIO may impose financial penalties upon carriers that do not satisfactorily deal with consumer complaints.

Canada

We are a reseller of telecommunications services in Canada. Because we do not own or operate transmission facilities in Canada, we are not subject to direct regulation by the CRTC pursuant to the Telecommunications Act (as amended, the "Canadian Telecommunications Act"). We may resell long-distance service, local telephone service, wireless service and Internet access without the regulation of our rates, prices or the requirement to file tariffs. In addition, as described below, as a reseller we are not subject to restrictions on foreign ownership or control.

Regulation. The CRTC has issued various decisions and constructed certain frameworks that directly affect our obligations and operations as a reseller.

In 2000, the CRTC implemented a revenue-based contribution regime to replace the per minute contribution charge formerly in place to support universal access. The revenue-based contribution mechanism collects from a wider base of telecommunications service providers, including resellers, and has lowered our contribution expenses since 2001.

We are required to hold a license to provide Basic International Telecommunications Services (a "BITS" license) pursuant to which we are subject to the requirements not to engage in anti-competitive conduct in relation to the provision of international telecommunications services and the requirement to observe the contribution regime described immediately above.

We operate as a local services reseller, both in reselling CLEC local services and in the provision of local VoIP services interconnected with the public switched telephone network. As such, we are subject to a number of obligations that are passed on through our contractual arrangements with LECs from which we purchase services. These include:

- *Registration.* We are required to be registered with the CRTC to provide these services.
- The provision of 911 services. For fixed services such as our circuit-switched local services, we are required to provide the same standard of 911 service as the underlying LEC. For our VoIP services, however, which are considered to be "nomadic" by the CRTC, we are required to implement an interim solution under which the CRTC requires us to employ the zero-dialed emergency routing services provided by the ILEC in its serving territory in order to route 911 calls to an operator, who determines the caller's location in the first instance. A CRTC Interconnection Steering Committee submitted a report to the CRTC in 2007 establishing some of the parameters for a permanent solution for the provision of emergency services by nomadic VoIP providers. A CRTC proceeding is currently underway to identify remaining parameters for such a solution.
- *Customer notifications*. Local VoIP providers are subject to stringent and detailed customer notification requirements regarding the limitations associated with their 911 services. These include warnings to be included in all of the providers' print, online and broadcasting marketing materials.
- Number portability. Although local voice resellers and VoIP providers must obtain their numbers from a LEC, they are subject to the requirement to
 permit porting out of local numbers.

- Privacy safeguards. Local resellers are required to provide certain privacy safeguards, including the provision of the privacy indicator, automated
 universal per-call blocking of calling line identification and others. To the extent that VoIP providers are unable to provide these privacy safeguards,
 they are required to inform their customers of this fact.
- *Message Relay Service*. All local resellers, including VoIP providers, are required to provide access to message relay service ("MRS"). This is to include both Teletypewriter Relay Service ("TTY Relay") and, as of July 21, 2010, Internet Protocol Relay Service ("IP Relay").

In a Price Cap decision issued in May 2002, the CRTC lowered the prices incumbent providers can charge competitors for a range of competitor services, i.e. certain facilities and services required by competitors to provide telecommunications services to their end-customers and mandated by the CRTC (see discussion of the new framework for wholesale services, below). The Price Cap formula set out in this decision required the ILECs to revise the rates of selected services (primarily local telecommunications services) yearly by the rate of inflation minus a productivity offset of 3.5%. The rates of other service groupings were "capped" and others were "uncapped" with upward pricing constraints. The CRTC typically relied on a four-year Price Cap period, but in 2005 it decided to extend the period by one year in order to complete a public proceeding to establish the parameters of the next Price Cap period. In a decision dated April 2007, the CRTC issued the parameters of the new Price Cap framework. The new Price Cap framework is similar to the previous framework in that rates for service groupings are "capped," "uncapped" and subject to upward or downward pricing constraints. The productivity offset still applies to selected services. CRTC decided not to impose an expiry date for the current Price Cap framework. Accordingly, the current Price Cap framework continues to allow for certain savings on competitor services for resellers.

In 2005 the federal government appointed a Telecom Policy Review Panel to review Canada's telecommunications policy framework. The Panel's report was released in March of 2006. Following the release of the report, the federal government issued a Policy Direction to the CRTC on December 18, 2006 that required, among other things, that in exercising its powers and duties, it rely on market forces to the maximum extent feasible. The Policy Direction has had an impact on the CRTC's recent decision involving essential facilities and wholesale services, described immediately below. The policy directs the CRTC to take into account the principles of technological and competitive neutrality, the potential for incumbents to exercise market power in the wholesale and retail markets for the service in the absence of mandated access to wholesale services, and the impediments faced by new and existing carriers seeking to develop competing network facilities.

On March 3, 2008, the CRTC issued a decision revising the definition of an essential service and the classifications and pricing principles under which facilities-based carriers are to offer wholesale services to competitors at regulated rates. Under this revised regulatory framework for wholesale services, the CRTC has classified existing wholesale services into six categories: essential, conditional essential, conditional mandated non-essential, public good, interconnection, and non- essential subject to phase-out. More than a third of wholesale services will be deregulated by 2012, including intra-exchange transport services which we use to interconnect our sites where we have located DSLAM equipment. The area likely to have the next-largest impact involves high-speed access to business services, which is to be deregulated in 2013.

In a regulatory policy update related to the new wholesale framework, dated January 2009, the CRTC determined that it would be appropriate to allow for negotiated agreements for certain services that were mandated pursuant to the wholesale framework (i.e., those services classified as "conditional essential" and "conditional mandated non-essential"). Accordingly, we now have the ability to purchase these services on a tariff basis or through a negotiated agreement. As we utilize a significant amount of these services, it is expected that the ILECs will attempt to convince us to enter into negotiated agreements for these services going forward. However, it is not yet certain whether such agreements will offer or result in preferable terms for us.

In addition to the above, our competitors have filed applications to review or appeal other aspects of the CRTC's new wholesale framework, as well as other wholesale data and Internet service obligations. In a decision

dated January 2009, the CRTC withdrew a directive contained in the wholesale framework decision that would have required the ILECs to provide unbundled access to the facilities used to provide Internet service to all end-users that they are able to serve. In parallel, the CRTC initiated a proceeding to consider the appropriateness of mandating certain ILEC and cable company wholesale high-speed access services, including unbundled access to next-generation facilities. In March 2009, several of our competitors applied to the Governor in Council for the variance or rescission of CRTC decisions related to wholesale data or Internet services, in which the CRTC directed the ILECs to provide wholesale ADSL services at speeds equivalent to their retail offerings. In an Order in Council dated December 2009, the Governor in Council referred these decisions back to the CRTC for reconsideration. The proceeding in which the CRTC is considering the appropriateness of mandating certain ILEC and cable company wholesale high-speed access services has been expanded to include this reconsideration.

In a decision dated August 2009, the CRTC granted interim approval to an application by an ILEC, Bell Canada, to implement usage-based billing on its wholesale Internet access service. If this application is granted final approval, it will require us to pass down the cost of usage-based billing incurred by us through the Internet usage of our end-users that are served via Bell Canada's wholesale Internet access service to those customers. It is possible that other ILECs will also implement usage-based billing on their wholesale Internet services. This would also permit us to implement usage-based billing on our retail Internet service that is provided via ILEC unbundled facilities obtained through CLEC arrangements.

In a decision dated October 2009, the CRTC set out a framework which will guide the implementation of Internet traffic management practices by ISPs and facilitate reviews of such practices against that framework. This framework is focused on four rules: Internet traffic management practices must be designed to address the stated need and achieve the stated purpose and effect, and nothing else; they must result in as little discrimination or preference as reasonably possible; they must result in as little harm to end-users and others as reasonably possible; and economic approaches, such as network investment, would not have reasonably addressed the need or effectively achieved the same purpose. As a result of this decision, we may continue to utilize the traffic management practices that we have implemented on our retail Internet service that is provided via ILEC unbundled facilities obtained through arrangements with CLECs. However, we may be obliged to notify our customers of such practices, with a 30-day wait period after notification before we can implement them. In addition, our arrangements with ILECs or with CLECs may require that the practices we implement respect the above-mentioned framework. Finally, the CRTC also determined that it would not require prior approval of the continued application of traffic management practices by the providers of wholesale Internet access services, provided that these are not more restrictive than those imposed on its retail customers and do not affect its wholesale customers disproportionately. As a result, our customers that are served via wholesale Internet services are, or may be, subject to the traffic management practices of the wholesale provider.

In an application dated June 2009, an ILEC, Bell Canada, requested approval to increase the tariffed rates for its unbundled local loops. This application has the potential to increase the cost of the unbundled facilities that we obtain through CLEC arrangements and use to provide retail telephone and Internet services. A review of Bell Canada's application is ongoing.

Competition. Long-distance competition has been in place in Canada since 1990 for long-distance resellers and since 1992 for facilities-based carriers. In June 1992, the CRTC issued Telecom Decision CRTC 92-12 requiring the ILECs to interconnect their networks with their facilities-based, as well as reseller, competitors. Since 1994, the ILECs have been required to provide "equal access," which eliminated the need for customers of competitive long-distance providers to dial additional digits when placing long-distance calls. The ILECs disbanded the Stentor alliance in 1999, and former Stentor companies, Bell Canada and TELUS Corporation, the two largest ILECs in Canada, compete against each other. MTS Allstream, Inc., the ILEC serving the Canadian province of Manitoba, has acquired Allstream, Inc. (formerly AT&T Canada Corp.) in 2004 and is now competing nationally as well. The other nationwide competitor, Call-Net Enterprises Inc., which operated as Sprint Canada, was acquired by Rogers in 2005. Cable TV companies, such as Rogers, Shaw Communications

Inc. and Vidéotron Limited, launched their local telephone services in July 2005 and have had a great deal of success thus far. Their local service is provided either via their cable network or acquired CLEC (i.e., Call-Net) or on a resold basis from an underlying LEC.

As markets become competitive, the Canadian Telecommunications Act permits the CRTC to forbear from regulating the rates and terms of service of Canadian carriers. The long-distance rates of Canadian ILECs have largely been forborne from regulation since 1997. The ILECs' rates for retail Internet services were forborne from regulation in 1999. On April 6, 2006, the CRTC established a framework for forbearance in respect of local services by defining the geographic markets in which the CRTC will forbear from regulating as the criteria for deregulation are met, and by establishing these criteria. On April 4, 2007 the Governor-in-Council varied this framework by reducing to the size of the geographic market in which the criteria are to be met and by relaxing the criteria to be met. As of June 2009 approximately 77% of all residential local lines and 68% of commercial local lines in Canada have been deregulated.

The Competition Bureau issued an Information Bulletin on the Abuse of Dominance Provisions as applied to the Telecommunications Industry on June 6, 2008, in which the Bureau describes its approach in reviewing abuse of dominance complaints in telecommunications markets where the CRTC has forborne from regulating conduct.

On March 12, 2009, the Competition Act was amended to provide the Competition Tribunal with the power to order the payment of an administrative monetary penalty of up to 15 million CAD in cases of abuse of dominant position. Subject to any relevant defenses, these amendments apply to telecommunications service providers.

Foreign Ownership Restrictions. Under the Canadian Telecommunications Act and the Radiocommunication Act, and certain regulations promulgated thereunder (i.e. the Radiocommunication Regulations and the Canadian Telecommunications Common Carrier Ownership and Control Regulations), foreign ownership restrictions apply to telecommunications common carriers and radiocommunication carriers ("Canadian carriers"), a concept which includes, for example, CLECs and microwave license holders, but not to companies, such as resellers, that do not own or operate transmission facilities. Accordingly, resellers may be wholly foreign-owned and controlled.

The restrictions applicable to Canadian carriers limit the amount of foreign investment in Canadian carriers to no more than 20% of the voting equity of a Canadian carrier operating company and no more than 33 ½% of the voting equity of a Canadian carrier holding company. The restrictions also limit the number of seats which may be occupied by non-Canadians on the board of directors of a Canadian carrier operating company to 20%. In addition, under Canadian law, a majority of Canadians must occupy the seats on the board of directors of a Canadian carrier holding company. Although there is no restriction on foreign investors holding non-voting equity in a Canadian carrier, the law requires that the Canadian carrier not be "controlled in fact" by non-Canadians. The concept of "control in fact" is very broad, and takes into account all commercial arrangements between the Canadian carrier and third parties.

The Canadian issuer, along with several other telecommunications service providers, has sought to have the Canadian government review foreign ownership restrictions with a view to lowering these restrictions or eliminating them. In April 2003, the Industry Committee of the House of Commons recommended removing these restrictions in their entirety, for both telecommunications common carriers and for broadcasting distribution undertakings ("BDUs") such as cable companies. In June 2003, however, another committee of the House of Commons (the Heritage Committee) expressed concerns that changes in ownership restrictions for either telecommunications common carriers or BDUs could have an adverse impact on the broadcasting system. In its September 2003 response to the Industry Committee's recommendation, the government acknowledged the appropriateness of the committee's conclusion that removing foreign investment restrictions would benefit the telecommunications industry. However, the government also noted the concerns expressed by the Heritage

Committee. The government recognized that it has a responsibility to determine how best to reconcile the conflicting recommendations of the two committees and undertook to analyze this question and be in a position to examine possible solutions by the spring of 2004. However, no solutions were brought forward in 2005 although the issue was raised once again by the Telecom Policy Review Panel ("TPRP"), who recommended in its Final Report that the foreign ownership restrictions be relaxed. In July 2007, Industry Canada announced the creation of the Competition Policy Review Panel (the "Panel"), which was asked to provide recommendations to the government on how to enhance Canadian competitiveness. The Panel's duties included a sector specific review of the current telecommunications foreign ownership restrictions. We filed a submission in favor of eliminating the restrictions and also presented options to accommodate concerns such as national security. In its report dated June 2008, the Panel supported the TPRP's recommendations regarding the easing of foreign ownership restrictions. Specifically, the Panel recommended a two phase approach to liberalize the Canadian foreign ownership restrictions. The initial phase would permit foreign companies to establish new telecommunication companies or acquire existing telecommunication companies with less than a 10% share of the Canadian telecommunication market. The second phase would consist of further liberalization upon a review of existing cultural policies and an assessment of the impact of the foreign investment resulting from the initial phase. However, despite the support of both the TPRP and the Panel it is premature to predict whether any recommendation to remove the restrictions for telecommunications common carriers will be implemented. In July 2009, the CRTC issued a new Canadian ownership and control review policy which subjects foreign investment arrangements in Canadian carriers to more formal procedures. These new procedures include greater public scruti

European Union

In Europe, the regulation of telecommunications is governed at a supranational level by the European Parliament, the Council of the European Union (the "Council") and the European Commission (the "EU Commission"), each consisting of members including the following countries: Austria, Belgium, Bulgaria, Cyprus, Czech Republic, Denmark, Estonia, Finland, France, Germany, Greece, Hungary, Ireland, Italy, Latvia, Lithuania, Luxembourg, Malta, Netherlands, Poland, Portugal, Romania, Slovakia, Slovenia, Spain, Sweden and the United Kingdom. These institutions are responsible for creating European Union-wide policies and, through legislation, have progressively developed a regulatory framework aimed at ensuring an open, competitive telecommunications market.

In February 2002, the Council agreed to a new European regulatory framework for the communications sector, which was adopted formally on March 7, 2002. The new regime comprises the following legislative texts:

- Directive 2002/21 on a common regulatory framework for electronic communications networks and services ("Framework Directive");
- Directive 2002/20 on the authorization of electronic communications networks and services ("Authorization Directive");
- Directive 2002/19 on access to, and interconnection of, electronic communications networks and associated facilities ("Access Directive"); and
- Directive 2002/22 on universal service and users rights relating to electronic communications networks and services ("Universal Service Directive").

EU member states were obligated to implement these directives by July 25, 2003. As of December 31, 2008, these directives have not materially affected our business operations in Europe.

One of the requirements of the Authorization Directive is that no company seeking to provide electronic communications networks or services is required to obtain prior authorization, although such a company could be required to notify relevant regulatory authorities in the member states where it intends to operate.

This regulatory framework is currently under review by the EU Commission. A number of proposed changes have been adopted, including the desire to create a European level telecom market body to complement national regulator, and assist in opening up markets with low levels of competition. The proposed changes have been adopted by the European Parliament and are awaiting consideration by the Council. Once adopted at the European level, the revised rules have to be incorporated into national law before taking effect. The EU Commission is hopeful that the new framework could be in place from 2010 onwards.

The Regulation of the European Parliament and of the Council of June 27, 2007 on roaming on public mobile telephone networks within the community and amending Directive 2002/21/EC is a regulation by the EU Commission, primarily backed by the EU Commissioner for Information Society and Media, which regulates the charges of mobile phone use when abroad in the EU. The regulation caps the rates operators can charge each other while roaming in the EU and also limits the tariffs an operator can charge from customers. The proposal for a regulation to lower international roaming charges within the EU was published by the EU Commission in July 2006. The regulation entered into force on June 30, 2007. From this date on, mobile phone operators within the EU had one month to inform customers about the new tariffs called "Eurotariff" and provided an offer for switching to the new tariff and the new tariff automatically applied on September 30, 2007, except special roaming package. The Eurotariffs is proposed to gradually decrease over the next three years.

A Eurotariff is available in all 27 member states of the EU. It applies only to voice calls within the EU irrespective of a pre-paid or post-paid status. It does not apply to other mobile phone services, such as short message service ("SMS"), multimedia messaging service ("MMS") or data transfer.

The regulation caps roaming prices for voice services only but it also requires national regulators to monitor developments in the prices of roamed voice, SMS and data services closely. The EU Commission called on mobile operators to demonstrate their willingness to voluntarily reduce the very high roaming charges for SMS and data roaming. Under new rules as of July 1st, 2009, consumers will also benefit from per-second billing after 30 seconds for calls made and per-second billing throughout for calls received to ensure that consumers do not face any hidden costs when they are roaming. This is expected to increase consumer's savings by over 20%.

The regulation is meant to expire in three years from 2007 (in Summer 2010). The EU Commission, together with the National Regulatory Authorities will monitor the development of the prices over the next 18 months. If normal market conditions are established in the market for roaming calls, the regulation will expire after three years. However, the Commission can also propose to continue to regulate the roaming market, if normal market conditions are not working yet. Under the latest EU proposal on September 23, 2008, this regulation would be extended to June 30, 2012.

A regulation on SMS and Data is currently being studied by the EU Commission. The operators were given until July 1, 2008 to lower SMS and data roaming charges. The EU Commission then sent letters to the CEOs of all European mobile operators inquiring about their SMS and data prices. The assessment showed that prices are still too high. The EU Commission then decided to introduce maximum price limits for sending SMS messages while roaming. The European Regulators Group has suggested a level between 11 and 15 eurocents. As of May 2009 a maximum price of SMS send within EU borders on the level of 11 eurocents from July 1, 2009 is almost certain. This would also include a wholesale maximum rate of €0.04.

The price cap for the rates the host operator charges a roaming customer's home operator, calculated on a kilobyte basis is:

- from July 1, 2009: a maximum of €1.00 per megabyte (excluding VAT);
- from July 1, 2010: a maximum of €0.80 per megabyte (excluding VAT); and
- from July 1, 2011: a maximum of €0.50 per megabyte (excluding VAT).

Data and Internet services are not regulated for the moment by the EU at the retail level. A wholesale price cap will be applicable as of July 1, 2009 for 1 EUR/MB. Customers traveling to another member state would also receive an automated message of the charges that apply for data roaming services. On July 1, 2010, operators must provide customers with the opportunity to determine in advance how much they want to spend before the service is cut-off.

A further directive relating to privacy and electronic communications ("Directive 2002/58") was added to the new regime. This directive aims to harmonize national laws regarding personal data protection in the electronic communications age and deals with matters including the confidentiality of billing information, the use of caller identification devices, the use of subscriber directories and unsolicited communications. The directive was supposed to be implemented by EU member states and incorporated into the regulatory regime of each member state by October 31, 2003, but as was also the case with the earlier directives referred to above, a number of member states missed this deadline. Each EU member state in which we currently conduct or plan to conduct our business has historically had a different regulatory regime, and we expect that, even with the adoption of the new EU regulatory regime, differences will continue for the foreseeable future. There may well be differences in the manner in which the new EU regulatory regime is implemented from one member state to another. The requirements for us to obtain necessary approvals have varied considerably from one country to another. We have obtained and will continue to seek to obtain interconnection agreements with other carriers within the EU. While previous EU directives have required that carriers with significant market power offer cost-based and non-discriminatory interconnection to competitors, individual EU member states have implemented this requirement differently and may continue to do so under the new EU regulatory regime. As a result, we may be delayed in obtaining or may not be able to obtain interconnection in certain countries that would allow us to compete effectively.

Further, member states must now introduce domestic legislation to implement the EC Directive on Waste Electrical and Electronic Equipment ("WEEE") and the EC Directive on the Restriction of the Use of Certain Hazardous Substances in Electrical and Electronic Equipment ("RoHS"). The WEEE seeks to implement legislation on electrical and electronic equipment in relation to its composition and levels to which it should be recycled, while making producers responsible for financing most of these activities. Information technology and telecommunications equipment is WEEE, and subject to the WEEE Directive. RoHS seeks to restrict hazardous substances in WEEE.

The Commission also is concerned with services, such as VoIP, and published a working paper applicable to VoIP based services—Commission Staff Working Document of June 14, 2004 on the Treatment of VoIP under the EU Regulatory Framework- an Information and Consultation Document, which identifies various issues that can arise in relation to VoIP. The introduction of 3G mobile broadband services raised new regulatory issues and the Commission published a Communication on Mobile Broadband Services (June 30, 2004-COM (2004) 447), which covers issues, such as spectrum trading.

Employees

The following table summarizes the number of our employees as of December 31, 2009 by region:

Canada	<u>Total</u> 704
Australia	582
United States	158
India	108
Europe Brazil	102
Brazil	47
Total	1,701

We have never experienced a work stoppage. Only some of our employees in Australia are represented by a labor union and covered by a collective bargaining agreement. We believe that our employee relations are positive.

Other Information

Our Internet address is *www.primustel.com*. We make available free of charge through our Internet website our annual reports on Form 10-K, quarterly reports on Form 10-Q, current reports on Form 8-K and amendments to those reports filed or furnished pursuant to the Securities Act of 1934 as soon as reasonably practicable after we electronically file such material with, or furnish it to, the United States Securities and Exchange Commission (SEC).

ITEM 1A. RISK FACTORS

A wide range of factors could materially affect our performance. In addition to factors affecting specific business operations and the financial results of those operations identified elsewhere in this report, the following factors, among others could adversely affect our operations:

The following is not intended as, and should not be construed as, an exhaustive list of relevant risk factors. There may be other risks that are relevant to its own particular circumstances or generally.

RISKS RELATED TO OUR BUSINESS AND INDUSTRY

Continuing global economic conditions could adversely affect our business.

The global economy and capital and credit markets have been experiencing exceptional turmoil and upheaval. Many major economies worldwide entered significant economic recessions in 2007 and continue to experience economic weakness even though economies have begun to show signs of recovery. Ongoing concerns about the systemic impact of potential long-term and widespread recession and potentially prolonged economic recovery, volatile energy costs, geopolitical issues, the availability, cost and terms of credit, consumer and business confidence, substantially increased and increasing unemployment rates and the crisis in the global housing and mortgage markets have all contributed to increased market volatility and diminished expectations for both established and emerging economies, including those in which we operate. In the second half of 2008, added concerns fueled by government interventions in financial systems led to increased market uncertainty and instability in both U.S. and international capital and credit markets. These conditions have contributed to economic uncertainty of unprecedented levels. The availability, cost and terms of credit also have been and may continue to be adversely affected by illiquid markets and wider credit spreads. Concern about the stability of the markets generally and the strength of counterparties specifically has led many lenders and institutional investors to reduce, and in many cases cease to provide, credit to businesses and consumers. These factors have led to a substantial and continuing decrease in spending by businesses and consumers over the past two years, and a corresponding decrease in global infrastructure spending. Continued turbulence in the U.S. and international markets and economies and prolonged declines in business and consumer spending may adversely affect our liquidity and financial condition, and the liquidity and financial condition of our customers, including our ability to refinance maturing debt instruments and to access the capital markets and obtai

- Potential risk in refinancing outstanding debts: Although none of our major debt instruments are scheduled to mature before 2013, at the earliest, if
 the volatility in the global capital markets were to continue, our ability to refinance our existing indebtedness when due could be severely
 constrained. See "—Risks Associated with Our Liquidity Needs and Debt Securities." Any such refinancing could require significantly more
 expensive interest rates and covenants that restrict our operations to a significantly greater extent.
- Negative impacts from increased financial pressures on customers: Uncertainty about current and future global economic conditions and credit markets may cause consumers, business and governments to defer purchases in response to tighter credit, decreased availability of cash and credit, and declining business and consumer confidence, any of which may affect the usage of our services by the customers and the ability of those customers to pay for our services. Accordingly, future demand for our products and services could differ from our current expectations. Similarly, our customers may experience liquidity issues of their own that adversely affect our ability to collect amounts due from them in a timely fashion or at all. In addition, if the global economy and credit markets continue to deteriorate or cease to recover and our future sales decline, our financial condition and results of operations would be adversely impacted.

Strengthening of the United States Dollar ("USD") against certain foreign currencies reduces the amount of USDs generated from foreign currency payments from our foreign operating subsidiaries and may adversely affect our results of operations and our ability to service our debt.

A significant portion of our net revenue (about 82% for the year ended December 31, 2009) is derived from sales and operations outside the U.S. The reporting currency for our consolidated financial statements is the USD. Our foreign operating subsidiaries, including our largest operating subsidiaries in Canada and Australia, generates cash in their respective local currencies and fluctuations in exchange rates can have a material adverse impact on amounts of USDs transferred to U.S. parent entities. In the future, we expect to continue to derive a significant portion of our net revenue (which is a substantial source for servicing our significant debt obligations at the parent entity level, as well as a source for making principal payments) and incur a significant portion of our operating costs outside the U.S., and changes in exchange rates have had and may have a significant, and potentially adverse, effect on our results of operations.

From June 30, 2008 through December 31, 2008, the Canadian and Australian dollars ("CAD" and "AUD," respectively) declined by 17% and 28%, respectively, relative to the USD, and this has had a material adverse impact on amounts of USDs transferred to U.S. parent entities. Conversely, from December 31, 2008 to December 31, 2009, the CAD and AUD increased by 11% and 35%, respectively. Due to the large percentage of our operations conducted outside of the U.S., and the cash transfers from these foreign operating subsidiaries to the U.S. parent, a strengthening of the USD relative to one or more of the foregoing foreign currencies could have an adverse impact on future results of operations and could adversely affect our ability to service or repay our consolidated indebtedness and obligations.

We historically have not engaged in hedging transactions. The exposure of our income from operations to fluctuations in foreign currency exchange rates is reduced in part because a majority of the costs that we incur in connection with our foreign operations are also denominated in local currencies. In addition, the operations of affiliates and subsidiaries in foreign countries have been funded with investments and other advances denominated in foreign currencies. Historically, such investments and advances have been long-term in nature, and we accounted for any adjustments resulting from currency translation as a charge or credit to accumulate other comprehensive loss within the stockholders' deficit section of our consolidated balance sheets. In 2002, agreements with certain subsidiaries were put in place for repayment of a portion of the investments and advances made to those subsidiaries. As we anticipate repayment in the foreseeable future of these amounts, we recognize the unrealized gains and losses in foreign currency transaction gain (loss) on the consolidated statements of operations, and depending upon changes in future currency rates, such gains or losses could have a significant, and potentially adverse, effect on our results of operations.

We are substantially smaller than our major competitors, whose marketing and pricing decisions, and relative size advantage could adversely affect our ability to attract and to retain customers and are likely to continue to cause significant pricing pressures that could adversely affect our net revenues, results of operations and financial condition.

The local, long-distance, Internet, broadband, digital subscriber lines ("DSL"), data and hosting and wireless telecommunications industry is significantly influenced by the marketing and pricing decisions of the larger business participants. Prices in the long-distance industry have continued to decline in recent years and, as competition continues to increase within each of our service segments and each of our product lines, we believe that prices are likely to continue to decrease. The most significant competitors in our primary markets include:

United States: AT&T Inc., Verizon Communications Inc., Qwest Communications International Inc. and other incumbent carriers, cable companies, including Comcast Corporation, Time Warner Cable Inc., Cablevision Systems Corporation and Charter Communications, Inc., other competitive local exchange carriers, including PaeTec Communications, Inc., Time Warner Telecom Inc., XO Communications Services, Inc. and Frontier Communications Corp., independent VoIP providers, including Vonage Holdings Corp and Cbeyond, Inc., wireless carriers in the U.S., including Verizon

- Communications Inc., AT&T Inc., Sprint Corp., T-Mobile USA Inc., MetroPCS Communications, Inc. and Leap Wireless International, Inc., and web based companies, including Skype Technologies S.A. and Google Inc.;
- Australia: Telstra Corporation Limited ("Telstra"), SingTel Optus Pty Limited, Telecom New Zealand Limited, iiNet Limited, SP Telemedia Limited (known as TPG), Macquarie Telecom Group Ltd. and other smaller national and regional service providers and resellers.; and
- Canada: TELUS Corporation ("TELUS"), BCE Inc. ("Bell Canada"), MTS Allstream, Inc., Saskatchewan Telecommunications, wireless providers, including Rogers Communications Inc. ("Rogers"), TELUS, Bell Canada, Bragg Communications Inc., COGECO Inc., Quebecor Inc. and Shaw Communications, Inc., cable companies, and other service providers and resellers including Globalive Communications Corp. in Canada.

Customers frequently change local, long-distance, wireless, broadband providers, and ISPs in response to the offering of lower rates or promotional incentives, increasingly as a result of bundling of various services by competitors. Moreover, competitors' VoIP and broadband product rollouts have added further customer choice and pricing pressure. As a result, customers generally can switch carriers and service offerings at their discretion with little notice to us. Competition in all of our markets is likely to remain intense, or increase in intensity and, as deregulatory influences affect markets outside the U.S., competition in non-U.S. markets is increasing to a level similar to the intense competition in the U.S.

Many of our competitors are significantly larger than us and have substantially greater financial, technical and marketing resources, larger networks, a broader portfolio of service offerings, greater control over network and transmission lines, stronger name recognition and customer loyalty, long-standing relationships with our target customers and lower debt-leverage ratios. As a result, our ability to attract and retain customers may be adversely affected. Many of our competitors enjoy economies of scale that result in low cost structures for transmission and related costs that could cause significant pricing pressures within the industry.

Several long-distance carriers in the U.S., Canada and Australia and the major wireless carriers and cable companies have introduced pricing and product bundling strategies that provide for fixed, low rates or unlimited plans for domestic and international calls. This strategy could have a material adverse effect on our net revenue per minute, results of operations and financial condition if our pricing, which we set to remain competitive, is not offset by similar declines in our costs. We compete on the basis of price, particularly with respect to our sales to other carriers, and also on the basis of customer service and our ability to provide a variety of telecommunications products and services. If such price pressures and bundling strategies intensify, we may not be able to compete successfully in the future, may face quarterly revenue and operating results variability, and may have heightened difficulty in estimating future revenues or results.

Given strong competition in delivering individual and bundled local, wireless, broadband, DSL, VoIP services, we may not be able to operate successfully or expand these parts of our business.

We have accelerated initiatives to become an integrated wireline, wireless and broadband service provider in order to counter competitive pricing pressures initiated by large incumbent providers in certain of the principal markets where we operate and to stem the loss of certain of our wireline voice and dial-up ISP customers to our competitors' bundled wireline, wireless and broadband service offerings. Our primary competitors include incumbent telecommunications providers, cable companies and other ISPs that have a significant national or international presence. Many of these operators have substantially greater resources, capital and operational experience than we do. We are experiencing increased competition from traditional telecommunications carriers, cable companies and other new entrants that have expanded into the market for broadband, VoIP, Internet services, data and hosting and traditional voice services. In addition, regulatory developments may impair our ability to compete. Therefore, future operations involving these individual or bundled services may not succeed in the competitive environment, and we (1) may not be able to expand successfully; (2) may experience margin pressure; (3) may face quarterly revenue and operating results variability; (4) may have limited resources to

develop and to market the new services; and (5) have heightened difficulty in establishing future revenues or results. As a result, there can be no assurance that we will reverse revenue declines in our traditional long-distance voice and dial-up ISP services or maintain or increase revenues or be able to generate sufficient income from operations or net income in the future or on any predictable or timely basis.

Our repositioning in the marketplace and intense domestic and international competition in these services places a significant strain on our resources, and if not managed effectively, could result in operational inefficiencies and other difficulties.

Our repositioning in the marketplace to focus on Growth Services segments of the telecom market, including broadband, IP-based voice, local, wireless, data and data center solutions may place a significant strain on our management, operational and financial resources and increase demand on our systems and controls. However, the local and long-distance telecommunications, data, broadband, Internet, VoIP, data and hosting and wireless industries are intensely competitive, present relatively limited barriers to entry in the more deregulated countries in which we operate and involve numerous entities competing for the same customers. Recent and pending deregulation in various countries may encourage new entrants to compete, including ISPs, wireless companies, and cable television companies, who could offer voice, broadband, Internet access and television services, and electric power utilities, who could offer voice and broadband Internet access. For example, the U.S. and many other countries have committed to open their telecommunications markets to competition pursuant to an agreement under the World Trade Organization which began on January 1, 1998. Further, in the U.S., the major landline incumbent carriers (including AT&T Inc. and Verizon Communications Inc.) have for many years also provided long-distance services, and previously independent long-distance providers (including AT&T Inc. and MCI Inc.) have been acquired by the landline incumbents. In addition, many entities, including large cable television companies (including Comcast Corporation, Time Warner Cable Inc., Cablevision Systems Corporation and Charter Communications, Inc.) and utilities have been allowed to enter both the local service and long-distance telecommunications markets.

To manage our repositioning effectively, we must continue to implement and improve our operational and financial systems and controls, invest in critical network infrastructure to expand its coverage and capacity including the data centers expansion, maintain or improve our service quality levels, purchase and utilize other transmission facilities, evolve our support and billing systems and train and manage our employee base. If we inaccurately forecast the movement of traffic onto our network, we could have insufficient or excessive transmission facilities and disproportionate fixed expenses. As we proceed with our development, operational difficulties could arise from additional demand placed on customer provisioning and support, billing and management information systems, product delivery and fulfillment, support, sales and marketing, administrative resources, network infrastructure, maintenance and upgrading. For instance, we may encounter delays or cost-overruns or suffer other adverse consequences in implementing new systems when required, such as our need to off-shore certain functions. In addition, our operating and financial control systems and infrastructure could be inadequate to ensure timely and accurate financial reporting, which could impact debt covenant compliance.

We have experienced significant historical, and may experience significant future, operating losses and net losses which may hinder our ability to meet our debt service or working capital requirements.

As of June 30, 2009, Predecessor had an accumulated deficit of \$1.06 billion. Predecessor incurred net losses of \$10.6 million in 2004, \$149.2 million in 2005, \$238.0 million in 2006 and \$25.0 million in 2008. During the year ended December 31, 2007, Predecessor recognized net income of \$15.7 million, of which \$32.7 million of revenue was related to the positive impact of foreign currency transaction gains. Even with the elimination of our significant accumulated deficit and the reduction in indebtedness through our recent reorganization under Chapter 11, future losses may continue. We cannot assure you that we will recognize net income in future periods. If we cannot generate net income or sufficient operating profitability, we may not be able to meet our debt service or working capital requirements.

A deterioration in our relationships with facilities-based carriers could have a material adverse effect upon our business.

We primarily connect our customers' telephone calls and data/Internet needs through transmission lines that we lease under a variety of arrangements with other facilities-based local, wireless, broadband, data and long-distance carriers. Many of these carriers are, or may become, our competitors. Our ability to maintain and expand our business depends on our ability to maintain favorable relationships with the facilities-based carriers from which we lease transmission lines. If our relationship with one or more of these carriers were to deteriorate or terminate, for any reason, it could have a material adverse effect upon our cost structure, service quality, network diversity, results of operations, financial condition and cash flows.

Uncertainties and risks associated with international markets and regulatory requirements could adversely impact our international operations.

We have significant international operations and, for the year ended December 31, 2009, derived about 82% of our net revenues by providing services outside of the U.S. In international markets, we are smaller than the principal or incumbent telecommunications carrier that operates in each of the foreign jurisdictions where we operate. In these markets, incumbent carriers: (1) are likely to modify and/or control access to, and pricing of, the local networks; (2) enjoy better brand recognition and brand and customer loyalty; (3) generally offer a wider range of product and services; and (4) have significant operational economies of scale, including a larger backbone network and longer term customer and supplier agreements on preferred and better terms. Moreover, the incumbent carrier may take many months to allow competitors, including us, to interconnect to their switches within our territory, and we are dependent upon their cooperation in migrating customers onto our network. There can be no assurance that we will be able to: (1) obtain the permits and operating licenses required for us to operate in the new service areas; (2) obtain access to local transmission facilities on economically acceptable terms; or (3) market services in international markets.

In addition, operating in international markets generally involves additional risks, including unexpected changes or uncertainties in regulatory requirements, taxes, tariffs, customs and duties. Given the nature of our operations and uncertainties in, or the absence of definitive regulations or interpretations concerning, the taxation of (including value added tax of) certain aspects of our business in certain international jurisdictions in which we conduct (or may be construed by such authorities as conducting or deriving taxable) operations or revenue, we may become subject to assessments for taxes (which may include penalties and interest) which are either unexpected, or have not been accrued for in our historical results of operations or both. This circumstance occurred during March 2008, when we concluded it was probable that assessments would be forthcoming concerning past European prepaid calling services operations, and it is possible that tax uncertainties concerning our international operations could arise in the future. Such developments, in addition to the other uncertainties and risks described above, could have adverse consequences that might result in restatement of prior period results of operations and unanticipated liquidity demands. Additional operating risks and uncertainties in operating in international markets include trade barriers, difficulties in staffing and managing foreign operations, problems in collecting accounts receivable, political risks, fluctuations in currency exchange rates, restrictions associated with the repatriation of funds, technology export and import restrictions, and seasonal reductions in business activity. Our ability to operate and grow our international operations successfully could be adversely impacted by these risks and uncertainties particularly in light of the fact that we derive such a large percentage of our revenues from outside of the U.S.

The telecommunications industry is rapidly changing, and if we are not able to adjust our strategy and resources effectively in the future to meet changing market conditions, we may not be able to compete effectively.

The telecommunications industry is changing rapidly due to deregulation, privatization, consolidation, technological improvements, availability of alternative services such as wireless, broadband, DSL, Internet, VoIP, data and hosting and wireless broadband through use of the fixed wireless spectrum, and the globalization

of the world's economies. In addition, alternative services to traditional fixed wireline services, such as wireless, broadband, Internet and VoIP services, are a substantial competitive threat. As the overall market for domestic and international long-distance and dial-up Internet services continues to decline in favor of Internet-based, wireless, and broadband communications, revenue contribution from our Traditional Services, which for the year ended December 31, 2009 comprised 43% of total revenue, has been consequently declining. If we do not adjust to meet changing market conditions or do not have adequate resources, we may not be able to compete effectively. The telecommunications industry is marked by the introduction of new product and service offerings and technological improvements. Achieving successful financial results will depend on our ability to anticipate, assess and adapt to rapid technological changes, and offer, on a timely and cost-effective basis, services, including the bundling of multiple services that meet evolving industry standards. If we do not anticipate, assess or adapt to such technological changes at a competitive price, maintain competitive services or obtain new technologies on a timely basis or on satisfactory terms, our financial results may be materially and adversely affected.

The rapid enhancement of VoIP and Internet technology may result in increasing levels of traditional domestic and international voice long-distance traffic being transmitted over the Internet, as opposed to traditional telecommunication networks. Currently, there are significant capital investment savings and cost savings associated with carrying voice traffic employing VoIP technology, as compared to carrying calls over traditional networks. Thus, there exists the possibility that the price of traditional long-distance voice services will decrease in order to be competitive with VoIP. Additionally, competition is expected to be intense to switch customers to VoIP product offerings, as is evidenced by numerous recent market announcements in the U.S. and internationally from industry leaders and competitive carriers concerning significant VoIP initiatives. Our ability effectively to retain our existing customer base and generate new customers, either through our traditional network or our own VoIP offerings, may be adversely affected by accelerated competition arising as a result of VoIP initiatives, as well as regulatory developments that may impede our ability to compete, such as restrictions on access to broadband networks owned and operated by others and the requirements to provide enhanced 911 emergency services ("E911"). As competition intensifies as a result of deregulatory, market or technological developments, our results of operations and financial condition could be adversely affected.

If we are not able to operate a cost-effective network, we may not be able to grow our business successfully.

Our long-term success depends on our ability to design, implement, operate, manage, maintain and upgrade a reliable and cost-effective network infrastructure, including data hosting centers. In addition, we rely on third party equipment and service vendors to enable us to expand and manage our global network and to provide local, broadband Internet, data and hosting and wireless services. If we fail to generate additional traffic on our network, if we experience technical or logistical impediments to our ability to develop necessary aspects of our network or to migrate traffic and customers onto our network, or if experience difficulties with our third-party providers, we may not achieve desired economies of scale or otherwise be successful in growing our business.

If we are not able to use and protect intellectual property domestically and internationally, it could have a material adverse effect on our business.

Our ability to compete depends, in part, on our ability to use intellectual property in the U.S. and internationally. We rely on a combination of trade secrets, trademarks and licenses to protect our intellectual property. We are also subject to the risks of claims and litigation alleging infringement of the intellectual property rights of others. The telecommunications industry is subject to frequent litigation regarding patent and other intellectual property rights. We rely upon certain technology, including hardware and software, licensed from third parties. There can be no assurance that the technology licensed by us will continue to provide competitive features and functionality or that licenses for technology currently used by us or other technology that we may seek to license in the future will be available to us on commercially reasonable terms or at all. Although our existing intellectual property licenses are on standard commercial terms made generally available

by the companies providing the licenses and, individually, their costs and terms are not material to our business, the loss of, or our inability to maintain existing licenses, could result in shipment delays or reductions until equivalent technology or suitable alternative products could be developed, identified, licensed and integrated and could cause service disruption to our customers. Such delays or reductions in the aggregate could harm our business. We also generally rely on indemnification provisions in licensing contracts to protect against claims of infringement regarding the licensed technology, which indemnification could be affected by, among other things, the financial strength of the licensor.

The loss of key personnel could have a material adverse effect on our business.

The loss of the services of K. Paul Singh, our Chairman and Chief Executive Officer, or the services of our other key personnel, or our inability to attract and retain additional key management, technical and sales personnel, could have a material adverse effect upon our financial condition and results of operations.

RISKS ASSOCIATED WITH OUR FINANCIAL STATEMENTS

Our disclosure controls and procedures and internal control over financial reporting were determined not to be effective as of December 31, 2006, 2007 and 2008 due to a material weakness that existed in our internal control over accounting for income taxes and as of December 31, 2009 due to a material weakness that existed in our internal control over accounting for foreign currency transaction gain (loss). If we fail to maintain effective internal control over financial reporting at a reasonable assurance level, we may not be able to accurately report our financial results, which could have a material adverse effect on our operations, investor confidence in our business and the trading prices of our securities.

Effective internal controls are necessary for us to provide reliable financial reports. In evaluating the effectiveness of our internal control over financial reporting, our management identified as of December 31, 2006, 2007 and 2008 a control deficiency in our controls and procedures over accounting for income taxes and as of December 31, 2009 a control deficiency over accounting for foreign currency transaction gain (loss) and management concluded in each case that the control deficiency in our internal controls over financial reporting constituted a material weakness. These deficiencies represented a material weakness in internal control over financial reporting on the basis that there is more than a remote likelihood that a material misstatement in our interim or annual financial statements could occur and would not be prevented or detected by our internal control over financial reporting. A material weakness is a deficiency, or combination of deficiencies, in internal control over financial reporting, such that there is a reasonable possibility that a material misstatement of annual or interim financial statements will not be prevented or detected on a timely basis.

Specifically, management's assessment of our internal control over financial reporting as of December 31, 2006, 2007 and 2008 identified a material weakness in internal control related to a lack of documentation, insufficient historical analysis and ineffective reconciliation procedures, primarily caused by lack of personnel with adequate expertise in income tax accounting matters. Since identifying the material weakness concerning accounting for income taxes, we have undertaken the following initiatives to remediate the material weakness – hired a new Corporate Tax Director and a Manager of Taxation in our Canadian operating unit; established functioning procedures for foreign finance personnel to communicate regularly any tax concerns with the Corporate Tax Director; purchased and implemented tax provision preparation software; and developed quarterly tax documentation formats. Accordingly, as of December 31, 2009 our management concluded the controls surrounding accounting for income taxes are effective.

In March 2010, the Company determined that an error existed related to accounting for foreign currency transaction gain (loss) on certain inter-company balances. Specifically, this error related to activity in the third quarter 2009 resulting in the Company amending its Form 10-Q for the quarter ended September 30, 2009. This amendment restated our financial statements in order to correct a non-cash error relating to accounting for unrealized foreign currency transaction losses associated with certain inter-company balances that were

permanent in nature and, therefore, should have been recorded as currency translation adjustment to accumulated other comprehensive income (loss) in the equity section of the balance sheet. Since identifying this, we have undertaken initiatives to remediate this material weakness by (a) performing additional recalculations and analysis of the foreign currency transaction gain (loss) recorded on these intercompany balances; (b) implementing an improved process for assessing the reasonableness of foreign currency transaction gain (loss) recorded on these intercompany balances; and (c) confirming intercompany settlements related to these balances at a transactional level. Notwithstanding such efforts, the material weakness concerning currency transaction will not be remediated until the new controls operate for a sufficient period of time and are tested to enable management to conclude that the controls are effective. As a result, as of December 31, 2009 management concluded that the control deficiency concerning foreign currency transactions represented a material weakness. See "Item 9A. Controls and Procedures" herein.

Our management will consider the design and operating effectiveness of our controls and necessary changes to such controls. However, we can not assure you that additional material weaknesses in our internal control over financial reporting will not be identified in the future. Any failure to maintain or implement required new or improved controls, or any difficulties we encounter in their implementation, could result in additional material weaknesses, and cause us to fail to timely meet our periodic reporting obligations or result in material misstatements in our financial statements. The existence of a material weakness could result in future errors in our financial statements that could result in a restatement of financial statements, cause us to fail to meet our reporting obligations and cause investors to lose confidence in our reported financial information.

Financial information in our future financial statements will not be comparable to our financial information from periods before July 1, 2009 due to our Reorganization and the application of fresh-start accounting to our financial statements.

Upon our Holding Companies' emergence from Chapter 11 on July 1, 2009, we adopted fresh-start accounting in accordance with Accounting Standards Codification ("ASC") 852, Reorganizations, pursuant to which our reorganization value, which represents the fair value of the entity before considering liabilities and approximates the amount a willing buyer would pay for the assets of the entity immediately after the Reorganization, has been allocated to the fair value of assets in conformity with ASC 805, Business Combinations, using the purchase method of accounting for business combinations. We stated liabilities, other than deferred taxes, at a present value of amounts expected to be paid. The amount remaining after allocation of the reorganization value to the fair value of identified tangible and intangible assets is reflected as goodwill, which is subject to periodic evaluation for impairment. In addition, under fresh-start accounting, our accumulated deficit (\$1.06 billion at June 30, 2009) has been eliminated. In addition to fresh-start accounting, our consolidated financial statements reflect all effects of the transactions contemplated by the Plan of Reorganization. Thus, our future consolidated balance sheets and consolidated condensed statements of operations data will not be comparable in many respects to our consolidated balance sheets and consolidated condensed statements of operations data for periods prior to our adoption of fresh-start accounting and prior to accounting for the effects of the Reorganization.

RISKS ASSOCIATED WITH OUR LIQUIDITY NEEDS AND DEBT SECURITIES

We may not be able to generate sufficient cash to service all of our indebtedness, and may be forced to take other actions to satisfy our obligations under our indebtedness that may not be successful.

Our ability to satisfy our debt obligations will depend upon, among other things, our future financial and operating performance, which will be affected by prevailing economic conditions and financial, business, regulatory and other factors, many of which are beyond our control. We cannot assure you that our business will generate cash flow from operations in an amount sufficient to fund our liquidity needs. If our cash flows and capital resources are insufficient to service our indebtedness, we may be forced to reduce or delay capital expenditures, sell assets, seek additional capital or restructure or refinance our indebtedness. These alternative measures may not be successful and may not permit us to meet our scheduled debt service obligations. Our

ability to restructure or refinance our debt will depend on the condition of the capital markets and our financial condition at such time. Any refinancing of our debt could be at higher interest rates and may require us to comply with more onerous covenants, which could further restrict our business operations. In addition, the terms of existing or future debt agreements may restrict us from adopting some of these alternatives. In the absence of such operating results and resources, we could face substantial liquidity problems and might be required to dispose of material assets or operations to meet our debt service and other obligations. We may not be able to consummate those dispositions for fair market value or at all. Furthermore, any proceeds that we could realize from any such disposition may not be adequate to meet our debt service obligations then due.

A significant portion of our future cash flow may need to be committed to repurchasing or redeeming certain outstanding debt prior to its stated maturity, rather than for use in our business operations.

If Group and its "restricted subsidiaries" have Excess Cash Flow (as defined below) for any fiscal year commencing with the fiscal year ending December 31, 2010, then the issuers of the 13% Senior Secured Notes due 2016 (the "13% Notes") are obligated to jointly apply an amount equal to 50% of such Excess Cash Flow for such period (the "Excess Cash Flow Offer Amount") and to make a joint offer to the holders of the 13% Notes to repurchase all or a portion of such notes as Units with an aggregate repurchase price in cash equal to the Excess Cash Flow Offer Amount (an "Excess Cash Flow Offer"). Excess Cash Flow means for any such fiscal year (a) the excess of (1) consolidated EBITDA over (2) the sum, subject to certain exceptions, represented by: (i) capital expenditures; (ii) consolidated interest expense paid in cash; (iii) income and franchise taxes paid in cash; and (iv) reductions in certain indebtedness *minus* (b) the absolute value of negative Excess Cash Flow, if any. See Note 7 to our Consolidated Financial Statements.

Within 110 days after the end of any fiscal year with respect to which an Excess Cash Flow Offer is required, an offer must be sent to each holder of 13% Notes stating the repurchase date, which must be no earlier than 30 days nor later than 60 days from the date such notice is mailed. With respect to each Excess Cash Flow Offer, the Issuers are entitled to reduce the applicable Excess Cash Flow Offer Amount by an amount equal to the sum of (x) the aggregate repurchase price paid for any 13% Notes repurchased by the Issuers in the open market or privately negotiated transaction (and, in each case, cancelled by the Issuers) and (y) the aggregate redemption price paid for any 13% Notes redeemed pursuant to one or more optional redemptions, subject to certain limitations.

If our outstanding 14 ¼% Senior Subordinated Secured Notes due May 2013 (the "14 ¼% Notes") have not been refinanced in accordance with the terms of its indenture on or prior to January 21, 2013, then the Issuers will be required to redeem the 13% Notes in advance of the 14 ¼% Notes scheduled maturity at a price equal to the then applicable optional redemption price.

If such early redemption of the 13% Notes are required or an Excess Cash Flow Offer is made, there can be no assurance that the Issuers will have available funds sufficient to pay for such redemption or for the Excess Cash Flow purchase price for all the 13% Notes that might be delivered by holders seeking to accept the Excess Cash Flow Offer. Any such failure could have a material adverse effect on us.

We must repay or refinance the 14 1/4% Notes prior to the maturity of the 13% Notes. Failure to do so could have a material adverse effect upon us.

The maturity of the 14 ¼% Notes is earlier than the maturity of the 13% Notes. While we expect to refinance the 14 ¼% Notes, we may not be able to do so or refinancing may not be available on commercially reasonable terms. Our ability to complete a refinancing of the 14 ¼% Notes prior to their maturity is subject to a number of conditions beyond our control. For example, if disruption in the financial markets were to occur at the time that we intended to refinance the 14 ¼% Notes, we might be restricted in our ability to refinance that indebtedness. If we are unable to refinance the 14 ¼% Notes our alternatives would consist of negotiating an extension of such indebtedness with the holders and seeking or raising new capital. If we were unsuccessful, the holders of the 14 ¼% Notes could demand repayment of the indebtedness owed to them on May 20, 2013. As a result, our ability to pay the principal of and interest on such indebtedness would be adversely affected.

We may not be able to repurchase the 141/4% Notes or 13% Notes upon a change of control.

Upon the occurrence of certain specific kinds of change of control events, we (or our subsidiaries) will be required to jointly offer to repurchase all outstanding notes at 101% of the principal amount thereof plus, without duplication, accrued and unpaid interest and additional interest, if any, to the date of repurchase. However, it is possible that they will not have sufficient funds at the time of the change of control to make the required repurchase of all notes delivered by holders seeking to exercise their repurchase rights, particularly as that change of control may trigger a similar repurchase requirement for, or result in an event of a default under a debt indenture and may also constitute a cross-default on other indebtedness existing at that time. In addition, certain important corporate events, such as leveraged recapitalization that would increase the level of our indebtedness, would not constitute a "Change of Control" under the indenture.

Our indentures governing our 141/4% Notes and 13% Notes contain significant operating and financial restrictions which may limit our ability and our restricted subsidiaries' ability to operate their business.

The indentures governing our 14 ¼% Notes and 13% Notes contain significant operating and financial restrictions on us and our subsidiaries. These restrictions limit the ability of us and our restricted subsidiaries to, among other things:

- incur additional indebtedness or issue certain preferred shares;
- create liens on certain assets to secure debt;
- · pay dividends or make other equity distributions;
- · purchase or redeem capital stock;
- make certain investments;
- transfer or sell assets;
- agree to restrictions on the ability of restricted subsidiaries to make payments to us or the issuers;
- · consolidate, merge, sell or otherwise dispose of all or substantially all of our or an issuer's assets; and
- · engage in transactions with affiliates.

These restrictions could limit the ability of us and our subsidiaries to finance future operations or capital needs, make acquisitions or pursue available business opportunities. We may be required to take action to reduce their debt or act in a manner contrary to our business objectives to satisfy these covenants. Events beyond our control, including changes in economic and business conditions in the markets in which we operate, may affect our ability to do so. We may not be able to satisfy these covenants. A breach of any of the covenants in our debt could result in a default under such debt, which could lead to that debt becoming immediately due and payable and, if such debt is secured, foreclosure on our assets that secure that obligation. A default under a debt instrument could, in turn, result in a default under other obligations and result in other creditors accelerating the payment of other obligations and foreclosing on asset security such debt, if any. Any such defaults could materially impair our financial conditions and liquidity.

Despite current indebtedness levels and restrictive covenants, we and our subsidiaries may still be able to incur substantial additional debt, which could exacerbate the risks described above.

We may be able to incur additional debt in the future, including debt secured by the collateral that secures our 14¼% Notes and 13% Notes, as well as other assets that do not secure such notes. Although the terms of the agreements governing our indebtedness contain restrictions on our ability to incur additional indebtedness, those restrictions are subject to a number of exceptions, and the indebtedness incurred in compliance with those restrictions could be substantial. In addition, if we are able to designate some of the restricted subsidiaries under the indenture governing the notes as unrestricted subsidiaries, those unrestricted subsidiaries would be permitted to borrow beyond the limitations specified in the indenture and engage in other activities in which restricted

subsidiaries may not engage. If we incur any additional secured debt that ranks equally with the 13% Notes, the holders of that debt will be entitled to share ratably with the holders of the 13% Notes in any proceeds distributed in connection with any bankruptcy, liquidation, reorganization or similar proceedings. Adding new debt to current debt levels could intensify the related risks that we now face.

The issuance of the 13% Notes may subject us to additional currency exchange risks.

The 13% Notes were issued and paid for, and the interest to be paid on those notes will be paid, in U.S. dollars. However, the Canadian issuer of such notes receives revenues primarily in Canadian dollars ("CAD"). As a result, the financial condition of the Canadian issuer might be materially adversely affected if the U.S. dollar appreciates against the CAD. From time to time, if we determine it is appropriate and advisable to do so, we may seek to lessen the effect of exchange rate fluctuations through the use of derivative financial instruments. However, we cannot assure you that we will be successful in these efforts.

ADDITIONAL RISKS RELATED TO REGULATION

We are subject to constantly changing regulation, including the imposition of fees and taxes, the potential adverse effects which may have a material adverse impact on our competitive position, growth and financial performance.

Our operations are subject to constantly changing regulation. There can be no assurance that future regulatory changes will not have a material adverse effect on us, or that domestic, foreign or international regulators or third parties will not raise material issues with regard to our compliance or noncompliance with applicable regulations, any of which could have a material adverse effect upon our competitive position, growth and financial performance. As a multinational telecommunications company, we are subject to varying degrees of regulation in each of the jurisdictions in which we provide our services. Local laws and regulations, and the interpretation of such laws and regulations, differ significantly among the jurisdictions in which we operate. Enforcement and interpretations of these laws and regulations can be unpredictable and are often subject to the informal views of government officials. Potential future regulatory, judicial, legislative, and government policy changes in jurisdictions where we operate could have a material adverse effect on us. Many regulatory actions are underway or are being contemplated by governmental agencies.

In the U.S., the Communications Act of 1934, as amended (the "Communications Act"), and associated Federal Communications Commission ("FCC") regulations, requires that every provider of interstate telecommunications carrier contribute, on an equitable and non-discriminatory basis, to federal universal service mechanisms established by the FCC, which affects our cost of providing services. At present, these contributions are calculated based on contributors' interstate and international revenue derived from U.S. end users for telecommunications or telecommunications services, as those terms are defined under FCC regulations. AT&T Inc. filed a "Petition for Immediate Commission Action" on July 10, 2009, requesting that the FCC adopt a new mechanism for calculating federal universal service fund contribution that would be applicable to all contributors, including us. The specific proposal, which has been pending at the FCC for some time, is to determine contributions to the Universal Service Fund ("USF") based on "assessable telephone numbers" rather than interstate and international revenues. We cannot predict whether the FCC will adopt this or some other contribution methodology, nor can we predict the potential impact on our business at this time. But a revised contribution methodology could increase our contribution obligation, including increasing our contribution disproportionately compared to some of our competitors. In such event, we may need to either raise the total amount of our consumer's bills, potentially making us less competitive with other providers of communications services, or reduce our profit margins.

Increasingly, laws, regulations or rulings that apply to traditional telephone services are being extended to commerce and communications services that utilize Internet Protocol, including VoIP. We are unable to predict the impact, if any, that future legislation, judicial decisions or regulations concerning Internet Protocol products and services may have on our business, financial condition, and results of operations. Regulation may be targeted

towards, among other things, fees, charges, surcharges, and taxation of VoIP services, liability for information retrieved from or transmitted over the Internet, online content regulation, user privacy, data protection, pricing, content, copyrights, distribution, electronic contracts and other communications, filing requirements, consumer protection, public safety issues like E911, the Communications Assistance for Law Enforcement Act ("CALEA"), the provision of online payment services, broadband residential Internet access, and the characteristics and quality of products and services, any of which could restrict our business or increase our cost of doing business. The rules that the FCC has already extended to interconnected VoIP providers include:

- rules with respect to the use of customer proprietary network information ("CPNI") requiring VoIP providers to adhere to particular customer
 approval processes when using CPNI outside of pre-defined limits and when using CPNI for marketing purposes, and requiring VoIP providers to
 take certain steps to verify a customer's identity before releasing any CPNI over the telephone or the Internet, and to report unauthorized disclosures
 of CPNI;
- the disability access requirements of Sections 225 and 255 of the Communications Act, which have been interpreted by the FCC to require interconnected VoIP providers to contribute to the telecommunications relay services fund and offer 711 abbreviated dialing access to relay services, and to ensure that VoIP services are accessible to persons with disabilities, if reasonably achievable;
- rules requiring VoIP providers to configure VoIP networks in a manner that facilitates lawful surveillance under CALEA;
- rules requiring VoIP providers to permit customers to retain their assigned telephone numbers when changing carriers including newly established requirements to process customer carrier changes on an expedited basis;
- rules requiring VoIP providers to provide access to E911 emergency services on terms generally similar to those provided by traditional landline carriers; and
- rules requiring VoIP providers to pay regulatory fees based on reported interstate and international revenues.

In Canada the Canadian Radio-television and Telecommunications Commission ("CRTC") has extended rules to interconnected VoIP providers that are similar to certain of those described above for the U.S., which rules are also subject to change from time to time. In addition, the CRTC is currently conducting public proceedings on whether to recover its operating fees from all telecommunications service providers, including resellers such as us, rather than only from Canadian carriers; and on whether and how to redefine the Basic Service Objectives, for whose subsidization we and other telecom service providers are required to contribute a proportion of our Canadian telecom service revenues.

The increasing growth of the VoIP market and popularity of VoIP products and services heighten the risk that governmental agencies will continue to increase the level of regulation applied to VoIP and the Internet.

Proposed future U.S. federal income tax legislation could impact the Company's effective tax rate.

In May 2009, President Obama's administration announced proposed future tax legislation that could substantially modify the rules governing the U.S. taxation of certain non-U.S. subsidiaries. These potential changes include, but are not limited to: (1) limitations on the deferral of U.S. taxation of foreign earnings; (2) limitations on the ability to claim and utilize foreign tax credits; and (3) deferral of various tax deductions until non-U.S. earnings are repatriated to the U.S. Each of these proposals would be effective for taxable years beginning after December 31, 2010. Many details of the proposal remain unknown, although if any of these proposals are enacted into law they could materially impact our effective tax rate.

The applicability of changes in tax policy to our services will increase their cost to consumers thereby decreasing our competitive price advantage over the competing alternatives available to the customer. Further, we may be subject to liabilities for past taxes, surcharges, fees, penalties and interest.

Unlike those of our competitors who offer traditional landline or wireless services, with respect to our VoIP services, we currently do not collect or remit state or municipal taxes, fees or surcharges on the retail charges we collect from our customers, except where we have determined we are required to do so based on tax law. In some jurisdictions we also did not collect and remit 911 surcharges. In some instances, we have received inquiries or demands from state and municipalities for taxes, fees or surcharges, including, in some instances, 911 fees. Depending on the state, statute or municipal code, we have maintained that these taxes, fees, or surcharges, including 911 fees, do not apply to us. However, recent changes in the law, at the federal, state and local level, may change our legal obligations. Accordingly, some taxes, fees or surcharges, including 911 fees, could apply to us retroactively and we could be subject to penalties and interest.

Other international governmental regulation could limit our ability to provide our services, make them more expensive and may have a material adverse impact on our competitive position, growth and financial performance.

Our international operations are also subject to regulatory risks, including the risk that regulations in some jurisdictions will prohibit us from providing our services cost-effectively or at all, which could limit our growth. We cannot assure you that these conditions will not have a material effect on our revenues and growth in the future. International regulatory considerations that affect or limit our business include:

- ongoing regulatory proceedings regarding efforts by Telstra in Australia to increase prices and charges and to deny access to essential facilities and services needed by us to compete;
- the ultimate outcome of the process launched by the Australian government to help fund the construction of a new national broadband network, including whether and the terms upon which (a) we will have access to such network, and (b) the duration for which the copper wire based last mile infrastructure we use to furnish broadband services using our DSLAM network infrastructure will be continued;
- a regulatory reform package recently announced by the Australian government that, if enacted, will (a) separate Telstra's retail arm from its wholesale business (via either functional or structural separation); and (b) provide the ACCC with greater powers to set access prices;
- general changes in access charges and contribution payments could adversely affect our cost of providing long-distance, wireless, broadband, VoIP, local and other services; and
- regulatory proceedings in Canada determining whether and the extent to which regulation should mandate access to networks and interconnection including intra-exchange transport services which we use to interconnect our DSLAM collocation sites and high speed access to residential and business services.

Any adverse developments implicating the foregoing could materially adversely affect our business, financial condition, result of operations and prospects.

We may be exposed to significant liability resulting from our noncompliance with FCC orders regarding E911 services.

FCC rules require VoIP providers interconnected to the public switched telephone network to provide E911 service in a manner similar to traditional wireline carriers. This requirement took effect as of November 2005. Like many interconnected VoIP providers, Lingo, Inc. ("Lingo"), a subsidiary of ours which sells such services, was able to meet this deadline for some but not all of its customers. We sought a waiver from the FCC asking for additional time to complete deploying our E911 service. The FCC has not yet addressed our waiver petition. As

of December 31, 2009, approximately 99% of our Lingo customers were equipped with E911 service as required by the FCC's rules. If and to the extent that we are determined to be out of compliance with the FCC order regarding E911 services we may be subject to fines, penalties, cease and desist orders prohibiting Lingo from providing service on the federal and state levels or any combination of the foregoing.

The FCC rules also require interconnected VoIP providers to distribute stickers and labels informing customers of the limitations on their emergency services as compared with traditional landline E911 service, as well as to notify and obtain affirmative acknowledgement from customers that they are aware of those limitations. The FCC's Enforcement Bureau released an order providing that the Enforcement Bureau will not pursue enforcement against interconnected VoIP providers that have received affirmative acknowledgement from at least 90% of their subscribers. We have received affirmative acknowledgement from substantially all of our customers, and, therefore, believe that we have effectively satisfied this requirement.

Lingo's current E911 services are more limited than the 911 services offered by traditional wireline telephone companies. These limitations may cause significant delays, or even failures, in callers' receipt of emergency assistance. Despite the fact that we have notified our customers and received affirmative acknowledgement from substantially all of our customers that they understand the differences between the access Lingo provides to emergency services as compared to those available through traditional wireline telephony providers, affected parties may attempt to hold us responsible for any loss, damage, personal injury or death suffered as a result of certain failures to comply with the FCC mandated E911 service for interconnected VoIP providers. Our resulting liability could be significant.

On June 1, 2007, the FCC released Notice of Proposed Rulemaking considering the imposition of additional E911 obligations on interconnected VoIP providers. Specifically, the FCC is considering requiring interconnected VoIP providers to determine automatically the physical location of their customer rather than allowing customers to manually register their location. Moreover, the notice includes a tentative conclusion that all interconnected VoIP service providers that allow customers to use their service in more than one location (nomadic VoIP service providers such as us) must utilize automatic location technology that meets the same accuracy standards applicable to providers of mobile phone service providers. At this time, we are unable to predict the outcome of this proceeding or its impact on us.

In July 2008, the "New and Emerging Technologies 911 Improvement Act of 2008" was signed into law. Previously, interconnected VoIP providers, like us, did not have the same protection from potential liability as applied wireline or wireless 911 emergency calling services. This law provides public safety entities, interconnected VoIP providers and others involved in handling 911 calls the same liability protections when handling 911 calls from interconnected VoIP users as from mobile or wired telephone service users. In October 2008, the FCC issued regulations implementing the provisions of the new law that require other entities involved in the provision of E911 services to make the technical capabilities used in such services available to VoIP providers, like us, on reasonable terms. The applicability of the liability protection to 911 calling services that do not conform to the FCC's rules is unclear. Additionally, any liability associated with 911 call placement and handling prior to the enactment of this new law would not be covered. Therefore, while this law provides significant liability protection to interconnected VoIP providers such as us, we may still face significant and material liability with respect to any past, present or future failures of our E911 service to function properly.

We may be similarly exposed to liability in Canada in connection with emergency services associated with our VoIP services. A description of our regulatory obligations associated with our VoIP services in Canada is set forth under "Regulation."

The rates we pay to interconnected telecommunications carriers in the U.S. may increase, which may reduce our profitability or increase the retail price of our service.

The FCC is considering reform of the methodology that regulated telecommunications carriers use to determine the appropriate payments for the exchange of traffic that is necessary to complete telephone calls to

the traditional telephone network. The result of the FCC's proceeding may be an increase in the rates we pay to such carriers to send traffic to or receive traffic from the traditional telephone network, which would increase our costs. Such a cost increase may result in us increasing the retail price of our service, which may make us less competitive in the communications marketplace, or may reduce our profitability. We cannot predict the outcome of this proceeding.

We may not be able to comply with recent FCC requirements regarding the transfer of telephone numbers to other providers when customers change providers.

In 2008, the FCC clarified that interconnected VoIP providers, such as us, are subject to its rules regarding transferring the telephone numbers of customers that choose to obtain service from other providers, including both interconnected VoIP providers and traditional carriers. In 2009, the FCC released an order that reduces the amount of time within which voice service providers must transfer a telephone number to a new provider. To comply with these new rules, we will likely have to implement new procedures and may have to increase staffing levels. Should this occur, we may need to increase the price of our retail service offering, which may make us less competitive with other providers of communications services or reduce our profitability. Should we not be able to comply with the order, we may be subject to fines, penalties, or cease and desist orders. At this time, we cannot accurately predict the full impact of this order on our business, as the industry is still developing the technical standards that will govern the new expedited number transfer procedures.

The FCC or federal courts may allow states in the U.S. to subject our service to state universal service fund obligations.

Several states have attempted to require nomadic interconnected VoIP providers to contribute to state universal service funds. One state, Nebraska, engaged in litigation with a provider of interconnected VoIP services similar to ours. The U.S. District Court for Nebraska issued a preliminary injunction on March 3, 2008, finding that the Nebraska Public Service Commission did not have jurisdiction to require universal service fund contributions from nomadic interconnected VoIP providers. A panel of the U.S. Circuit Court of Appeals for the Eighth Circuit affirmed the U.S. District court ruling on May 1, 2009. Subsequently, the Nebraska Public Service Commission requested a rehearing that the Court denied on June 5, 2009. On the basis of this litigation, we do not believe that existing law allows states to subject us to state universal service fund contribution obligations.

On July 16, 2009, Kansas and Nebraska filed a petition with the FCC requesting a declaratory ruling that states are not preempted from requiring nomadic interconnected VoIP providers to contribute to state universal service funds. The petition also seeks a retroactive ruling finding that states have been able to collect such contributions for a time period that is not clearly defined in the petition. At this time, we cannot predict the outcome of this proceeding nor its impact on our business. If we were required to pay retroactively into state-level universal service funds, that could have a material negative impact on our earnings.

ITEM 1B. UNRESOLVED STAFF COMMENTS

None.

ITEM 2. PROPERTIES

We currently lease our corporate headquarters facility, which is located in McLean, Virginia. Additionally, we lease administrative, technical and sales office space, as well as space for our switches and data centers, in various locations in the countries in which we operate. Total leased space in the United States, Australia, Canada, Brazil and the United Kingdom, as well as other countries in which we operate, approximates 593,000 square feet and the total annual lease costs are approximately \$15.0 million. The operating leases expire at various times with the longest commitment expiring in 2019. We believe that our present administrative and sales office

facilities are adequate for our anticipated operations and that similar space can be obtained readily as needed, and we further believe that the current leased facilities are adequate to house existing communications equipment.

Certain communications equipment which includes network switches and transmission lines are leased through operating leases, capital leases and vendor financing agreements.

ITEM 3. LEGAL PROCEEDINGS

The Company and its subsidiaries are subject to claims and legal proceedings that arise in the ordinary course of its business. Each of these matters is subject to various uncertainties, and it is possible that some of these matters may be decided unfavorably. The Company believes that any aggregate liability that may result from the resolution of these matters will not have a material adverse effect on the Company's consolidated financial position, results of operations or cash flows.

ITEM 4. (REMOVED AND RESERVED)

PART II

ITEM 5. MARKET FOR REGISTRANT'S COMMON EQUITY, RELATED STOCKHOLDER MATTERS AND ISSUER PURCHASES OF EQUITY SECURITIES

Common Stock

Primus Telecommunications Group, Incorporated ("we" or "us") common stock has traded on the OTC Bulletin Board under the ticker symbol "PMUG" since July 1, 2009, (the "New Common Stock"). Shares of our common stock issued and outstanding immediately prior to July 1, 2009 (the "Old Common Stock") traded on the over-the-counter market, both through listings on the OTC Bulletin Board and in the National Quotation Bureau "Pink Sheets." The ticker symbol "PRTL" was assigned to our Old Common Stock for over the counter quotations. On June 30, 2009 share of our outstanding Old Common Stock were cancelled pursuant to the terms of our Plan of Reorganization and 9,600,000 share of New Common Stock were issued. We have no continuing obligations with respect to the Old Common Stock.

The following table provides the high and low sale prices for our common stock authorized in connection with our emergence from bankruptcy as reported on the over-the-counter market for the period beginning July 1, 2009 and for our Old Common Stock for the periods through June 30, 2009. These prices do not include retail markups, markdowns or commissions.

	Successor					_	Predecessor		
	New Common Stock (1)				_	Old Common St		ck (2)	
<u>Period</u>	High		High I		Low		High	_	Low
2009									
1st Quarter						\$	0.09	\$	0.003
2 nd Quarter						\$	0.04	\$	0.01
3 rd Quarter	\$	7.75	\$	3.20					
4 th Quarter	\$	7.25	\$	5.60					
2008									
1st Quarter						\$	0.46	\$	0.27
2 nd Quarter						\$	0.45	\$	0.24
3 rd Quarter						\$	0.39	\$	0.23
4 th Quarter						\$	0.27	\$	0.05

 ^{9,600,000} Shares of New Common Stock were issued pursuant to the terms of our Plan of Reorganization and trading commenced on July 1, 2009 on a
"when issued" basis. The New Common Stock was listed on the OTC Bulletin Board and trades under the ticker symbol "PMUG".

Dividend Policy

We have not paid any cash dividends on our common stock to date. The payment of dividends, if any, in the future is within the discretion of the Board of Directors and will depend on our earnings, our capital requirements and financial condition. Dividends are currently restricted by the senior secured notes and the senior subordinated secured notes indentures, and may be restricted by other credit arrangements entered into in the future by us. It is the present intention of the Board of Directors to retain all earnings, if any, for use in our business operations, and accordingly, the Board of Directors does not expect to declare or pay any dividends in the foreseeable future.

⁽²⁾ All outstanding shares of the Old Common Stock were cancelled pursuant to the terms of our Plan of Reorganization and no further trading occurred after June 30, 2009.

Contingent Value Rights Distribution Agreement

Pursuant to the terms of the Plan, Group issued to holders of Old Common Stock contingent value rights ("Contingent Value Rights" or "CVRs") to receive up to an aggregate of 2,665,000 shares (the "CVR Shares") of New Common Stock. In connection with the issuance of the Contingent Value Rights, Group entered into a Contingent Value Rights Distribution Agreement (the "CVR Agreement"), in favor of holders of CVRs thereunder, dated as of the Effective Date.

The CVRs may not be transferred by the holder thereof except in certain limited circumstances. Subject to the terms of the CVR Agreement, holders of CVRs will receive their pro rata share of up to 2,665,000 CVR Shares. A distribution of CVR Shares is required to be made by Group if, as of any determination date (described below), Group's equity value (assuming cash exercise in full on such date of in-the-money warrants and options of Group) divided by the sum of the number of shares of New Common Stock then issued and outstanding plus the number of shares of New Common Stock underlying warrants, options and similar securities of Group (other than CVRs) that are then in-the-money exceeds \$35.95. The aggregate number of such shares of New Common Stock is referred to as the "Applicable Shares;" the price per share of \$35.95, subject to adjustment as described below, is referred to as the "CVR Strike Price;" and the per share amount of any such excess over the CVR Strike Price is referred to as the "Excess Equity Value Per Share." If such a distribution is required, the number of CVR Shares to be distributed by Group equals the product of Excess Equity Value Per Share multiplied by the number of Applicable Shares divided by the CVR Strike Price. Such product of Excess Equity Value Per Share is referred to as the "Excess Equity Value."

Group will determine if and to the extent a distribution of CVR Shares is required on January 1 and July 1 of each year, commencing on the first such date (but in no event later than July 1, 2013) on which data is available to confirm that Group's adjusted EBITDA for the immediately preceding four fiscal quarters is equal to at least \$100 million, and upon a change of control of Group. Distributions of CVR Shares (if any) will be made within 45 calendar days of a determination by Group that a distribution is required. Notwithstanding the foregoing, no distribution of CVR Shares is required to be made by Group unless Excess Equity Value exceeds \$1 million as of any determination date.

The number of CVR Shares and the CVR Strike Price will be adjusted from time to time in connection with any stock dividend or distribution, or subdivision, split, combination, reclassification or recapitalization of the New Common Stock. In addition, if Group distributes to holders of New Common Stock any of its assets (including but not limited to cash), securities or rights to purchase securities of Group (other than any regular cash dividend not to exceed in any fiscal year 45% of the consolidated net income of Group for the immediately preceding fiscal year), then the number of CVR Shares will be increased and the CVR Strike Price will be decreased, in each case pursuant to the terms of the CVR Agreement. Additionally, in case of any reclassification, merger, consolidation, capital reorganization or other change in the capital stock of Group (other than in connection with a change of control) in which all or substantially all of the outstanding shares of New Common Stock are converted into or exchanged for stock, other securities or other property, Group shall make appropriate provision so that the holders of Contingent Value Rights shall thereafter be entitled to receive, at such time such holder would have otherwise been entitled to receive a distribution under the CVR Agreement, the kind and amount of stock and other securities and property having a value substantially equivalent to the value of New Common Stock that the holders of Contingent Value Rights would have been entitled to receive in connection with a distribution of CVR Shares immediately prior to such reclassification, merger, consolidation, reorganization or other change in the capital stock of Group at a CVR Strike Price that, in each case, is reasonably determined by the board of directors of Group after consultation with an independent valuation advisor to preserve, to the extent practicable, the intrinsic value of such CVR immediately prior to such event.

The Contingent Value Rights will expire and the CVR Agreement will terminate upon the earliest to occur of: (1) the date upon which no further CVR Shares are available for distribution, (2) the consummation of a change of control (subject to any potential distribution of CVR Shares as a result thereof), and (3) July 1, 2019.

The foregoing description of the CVR Agreement does not purport to be complete and is qualified in its entirety by reference to the full text of such agreement, a copy of which is attached as an exhibit to Group's Form 8-A filed with the SEC on July 1, 2009 and such exhibit is incorporated herein by reference.

Holders of Common Stock

As of February 28, 2010, we had one holder of record of our common stock. The Company believes that this represents approximately 700 beneficial owners.

Recent Sales of Unregistered Securities

There are no unregistered sales of securities for 2009, other than the transactions that have been previously reported in our quarterly report on Form 10-Q for the period ended June 30, 2009, which description is incorporated herein by reference.

ITEM 6. SELECTED FINANCIAL DATA

The selected consolidated financial data set forth below should be read in conjunction with Item 7 entitled "Management's Discussion and Analysis of Financial Condition and Results of Operations Overview" and (ii) our consolidated audited annual financial statements and the notes thereto, each of which are contained in Item 8 entitled "Financial Statements and Supplementary Data."

As of July 1, 2009, the Company adopted fresh-start accounting in accordance with Accounting Standards Codification (ASC) No. 852. The adoption of fresh-start accounting resulted in the Company becoming a new entity for financial reporting purposes. Accordingly, the financial statements on or prior to July 1, 2009 are not comparable with the financial statements for periods after July 1, 2009. The consolidated condensed statements of operations and any references to "Successor" or "Successor Company" for the six months ended December 31, 2009, show the operations of the reorganized Company from and including July 1, 2009, the Effective Date, through December 31, 2009. References to "Predecessor" or "Predecessor Company" refer to the operations of the Company prior to July 1, 2009, except for Predecessor's July 1, 2009 statements of operations which reflect only the effect of the plan adjustments and fresh-start accounting as of such date and do not reflect any operating results.

In accordance with ASC No. 805, the preliminary allocation of the reorganization value is subject to additional adjustment until the Company has completed its analysis, but not to exceed one year after emergence from bankruptcy, to provide the Company with the time to complete the valuation of its assets and liabilities. The final determination of the fair value of individual assets and liabilities and the final allocation of the reorganization value may differ from the amounts included in the consolidated financial statements and there can be no assurance that such adjustments will not be material.

PRIMUS TELECOMMUNICATIONS GROUP, INCORPORATED

CONSOLIDATED CONDENSED STATEMENTS OF OPERATIONS

(in thousands, except per share amounts)

	Succes	sor					Pı	edecessor				
	Six Mor Ende Decembe 2009	nths d er 31,	E Ji	Months Inded uly 1, 2009	Dec	Year Ended cember 31, 2008	I Dece	Year Ended ember 31, 2007		Year Ended ember 31, 2006	Dec	Year Ended ember 31, 2005
NET REVENUE	\$ 42	4,334	\$	391,216	\$	895,863	\$	896,029	\$	993,034	\$	1,153,253
OPERATING EXPENSES												
Cost of revenue (exclusive of depreciation included below)		8,417		255,288		569,865		551,303		653,905		766,963
Selling, general and administrative		2,911		95,836		260,430		281,010		278,951		369,666
Depreciation and amortization	3	9,530		12,346		32,791		30,529		47,002		85,952
(Gain) loss on sale or disposal of assets		181		(43)		(6,028)		1,463		14,158		13,364
Asset impairment write-down		_								206,139		
Total operating expenses	42	1,039		363,427		857,058		864,305		1,200,155		1,235,945
INCOME FROM OPERATIONS		3,295		27,789		38,805		31,724		(207,121)		(82,692)
INTEREST EXPENSE	(1	7,323)		(14,135)		(53,888)		(61,347)		(54,128)		(53,394)
ACCRETION ON DEBT PREMIUM, net		(3)		189		583		(449)		(1,732)		
CHANGE IN FAIR VALUE OF DERIVATIVES EMBEDDED WITHIN												
CONVERTIBLE DEBT		_		_		_		_		5,373		
GAIN (LOSS) ON EARLY EXTINGUISHMENT OR RESTRUCTURING OF DEBT		4,146)		_		36,872		(7,652)		7,409		(1,693)
GAIN (LOSS) FROM CONTINGENT VALUE RIGHTS VALUATION	(2,804)		_		_		_		_		_
INTEREST INCOME AND OTHER INCOME (EXPENSE)		564		396		3,279		5,665		2,584		2,098
FOREIGN CURRENCY TRANSACTION GAIN (LOSS)	1	8,384		21,121		(47,563)		32,699		10,668		(12,285)
LOSS FROM CONTINUING OPERATIONS BEFORE REORGANIZATION ITEMS												
AND INCOME TAXES	(2,033)		35,360		(21,912)		640		(236,947)		(147,966)
REORGANIZATION ITEMS, net		(EC2)		440,094								
		(562)			_				_		_	
INCOME (LOSS) FROM CONTINUING OPERATIONS BEFORE INCOME TAXES		2,595)		475,454		(21,912)		640		(236,947)		(147,966)
INCOME TAX BENEFIT (EXPENSE)		0,041		(3,907)		366		9,232	_	(4,863)		(3,809)
INCOME (LOSS) FROM CONTINUING OPERATIONS		7,446		471,547		(21,546)		9,872		(241,810)		(151,775)
LOSS FROM DISCONTINUED OPERATIONS, net of tax		(271)		(676)		(326)		(268)		(4,673)		2,157
GAIN (LOSS) FROM SALE OF DISCONTINUED OPERATIONS, net of tax		(110)		251				6,132		7,415		
NET INCOME (LOSS)		7,065		471,122		(21,872)		15,736		(239,068)		(149,618)
Less: Net loss attributable to the noncontrolling interest		(333)		32		(3,159)				1,110		381
NET INCOME (LOSS) ATTRIBUTABLE TO PRIMUS TELECOMMUNICATIONS												
GROUP, INCORPORATED	\$	6,732	\$	471,154	\$	(25,031)	\$	15,736	\$	(237,958)	\$	(149,237)
BASIC INCOME (LOSS) PER COMMON SHARE:					_						_	
Income (loss) from continuing operations attributable to Primus												
Telecommunications Group, Incorporated	\$	0.74	\$	3.30	\$	(0.17)	\$	0.07	\$	(2.14)	\$	(1.59)
Gain (loss) from discontinued operations	Ψ	(0.03)	Ψ		Ψ	(0.01)	Ψ		Ψ	(0.05)	Ψ	0.03
Gain (loss) from sale of discontinued operations		(0.01)		_		(0.01)		0.05		0.07		-
Net income (loss) attributable to Primus Telecommunications Group,		(0.02)	_									
Incorporated	¢	0.70	¢	3.30	\$	(0.18)	\$	0.12	¢	(2.12)	¢	(1.56)
	Φ	0.70	Ψ	3.30	Ψ	(0.10)	Ψ	0.12	Ψ	(2.12)	Ψ	(1.50)
DILUTED LOSS PER COMMON SHARE:												
Income (loss) from continuing operations attributable to Primus		. ==		0.00		(0.45)		0.05	_	(0.4.1)	_	/4 m=:
Telecommunications Group, Incorporated	\$	0.73	\$	2.72	\$	(0.17)	\$	0.06	\$	(2.14)	\$	(1.59)
Gain (loss) from discontinued operations		(0.03)		_		(0.01)				(0.05)		0.03
Gain (loss) from sale of discontinued operations		(0.01)			_			0.03	_	0.07		
Net income (loss) attributable to Primus Telecommunications Group,												
Incorporated	\$	0.69	\$	2.72	\$	(0.18)	\$	0.09	\$	(2.12)	\$	(1.56)

		cessor Aonths	Predecessor Six Months Year Year Year							Year	
	Er Decen	nded nber 31, 009		Ended July 1, 2009		Ended ember 31, 2008		Ended ember 31, 2007		Ended cember 31, 2006	Ended cember 31, 2005
WEIGHTED AVERAGE COMMON SHARES OUTSTANDING											
Basic		9,600	_	142,695	_	142,643		128,771	_	112,366	 95,384
Diluted		9,800		173,117		142,643		196,470		112,366	95,384
AMOUNTS ATTRIBUTABLE TO COMMON SHAREHOLDERS OF PRIMUS TELECOMMUNICATIONS GROUP, INCORPORATED	<u>-</u>										
Income (loss) from continuing operations, net of tax	\$	7,113	\$	471,579	\$	(24,705)	\$	9,872	\$	(240,700)	\$ (151,394)
Gain (loss) from discontinued operations		(271)		(676)		(326)		(268)		(4,673)	2,157
Gain (loss) from sale of discontinued operations		(110)	l _	251				6,132		7,415	
Net income (loss)	\$	6,732	\$	471,154	\$	(25,031)	\$	15,736	\$	(237,958)	\$ (149,237)

Balance Sheet Data:

	:	Successor	Predecessor							
	As of December 31,					As of Dec	embe	r 31,		
(in thousands)		2009	_	2008		2007		2006		2005
Total assets	\$	558,914	\$	330,444	\$	460,403	\$	392,250	\$	641,089
Total long-term obligations (including current portion)	\$	257,516	\$	604,837	\$	673,903	\$	644,074	\$	635,212
Total liabilities	\$	459,005	\$	789,169	\$	907,943	\$	860,506	\$	877,423
Total Primus Telecommunications Group, Inc. stockholders' equity (deficit)	\$	99,909	\$	(461,539)	\$	(447,540)	\$	(468,255)	\$	(236,334)

Discontinued Operations Data:

	Successor	Predecessor									
	As of December 31.	Six Months Ended July 1,		As of Dec	ecember 31,						
(in thousands)	2009	2009	2008	2007	2006	2005					
Net revenue	\$ —	\$ 300	\$3,851	\$ 9,961	\$24,086	\$34,143					
Operating expenses	\$ 38	\$ 995	\$4,610	\$10,233	\$28,640	\$31,372					
Income (loss) from operations	\$ (38)	\$ (695)	\$ (759)	\$ (272)	\$ (4,554)	\$ 2,771					
Interest expense	\$ —	\$ —	\$ —	\$ (26)	\$ (47)	\$ (47)					
Interest income and other income (expense)	\$ (233)	\$ 19	\$ 60	\$ 39	\$ (2)	\$ (120)					
Foreign currency transaction gain (loss)	\$	\$ —	\$ 376	\$ (6)	\$ 10	\$ (258)					
Income (loss) before income tax	\$ (271)	\$ (676)	\$ (323)	\$ (265)	\$ (4,593)	\$ 2,346					
Income tax expense	<u>\$</u>	\$ —	\$ (3)	\$ (3)	\$ (80)	\$ (189)					
Income (loss) from discontinued operations	\$ (271)	\$ (676)	\$ (326)	\$ (268)	\$ (4,673)	\$ 2,157					

ITEM 7. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

You should read the following discussion and analysis of our financial condition and results of operations together with our consolidated annual audited financial statements and the notes thereto, each of which are contained in Item 8 entitled "Financial Statements and Supplementary Data," and other financial information incorporated by reference. Some of the information contained in this discussion and analysis includes forward-looking statements that involve risks and uncertainties. You should review the "Risk Factors" section for a discussion of important factors that could cause actual results to differ materially from the results described in or implied by the forward-looking statements contained in the following discussion and analysis.

Introduction and Overview of Operations

We are an integrated facilities-based communications services provider offering a portfolio of international and domestic voice, wireless, Internet, VoIP, data and data center services to customers located primarily in the United States Australia, Canada, Brazil, the United Kingdom and western Europe. Our primary markets are Australia and Canada where we have deployed significant network infrastructure. We classify our services into three categories: Growth Services, Traditional Services and Wholesale Services. Our focus is on expanding our Growth Services, which includes our broadband, IP-based voice, local, wireless, data and data center services, to fulfill the demand for high quality, competitively priced communications services. This demand is being driven, in part, by the globalization of the world's economies, the global trend toward telecommunications deregulation and the migration of communication traffic to the Internet. We manage our Traditional Services, which includes our domestic and international long-distance voice, prepaid cards, dial-up Internet services and Australian offnetwork local services for cash flow generation that we reinvest to develop and market our Growth Services, particularly in our primary markets of Australia and Canada. We provide our wholesale voice termination services to other telecommunications carriers and resellers requiring IP or time-division multiplexing access.

Generally, we price our services competitively with the major carriers and service providers operating in our principal service regions. We seek to generate net revenue through sales and marketing efforts focused on customers with significant communications needs, including SMEs, multinational corporations, residential customers, and other telecommunications carriers and resellers.

Industry trends have shown that the overall market for domestic and international long-distance voice, prepaid phone cards and dial-up internet services has declined in favor of Internet-based, wireless and broadband communications. Our challenge concerning net revenue in recent years has been to overcome declines in long-distance voice minutes of use per customer as more customers are using wireless devices and the Internet as alternatives to the use of wireline phones. Also, product substitution (*e.g.*, wireless/Internet for fixed line voice) has resulted in revenue declines in our long-distance voice services. Additionally, we believe that because deregulatory influences have begun to affect telecommunications markets outside the United States, the deregulatory trend is resulting in greater competition from the existing wireline and wireless competitors and from more recent entrants, such as cable companies and VoIP companies, which could continue to affect adversely our net revenue per minute, as well as minutes of use. More recently, adverse global economic conditions have resulted in a contraction of spending by business and residential customers generally which, we believe, has had an adverse affect on our net revenues.

In order to manage our network transmission costs, we pursue a flexible approach with respect to the management of our network capacity. In most instances, we (1) optimize the cost of traffic by using the least expensive cost routing, (2) negotiate lower variable usage based costs with domestic and foreign service providers, (3) negotiate additional and lower cost foreign carrier agreements with the foreign incumbent carriers and others, and (4) continue to expand/reduce the capacity of our network when traffic volumes justify such actions.

Our overall margin may fluctuate based on the relative volumes of international versus domestic long-distance services; carrier services versus business and residential long-distance services; prepaid services versus

traditional post-paid voice services; Internet, VoIP and data services versus fixed line voice services; the amount of services that are resold; and the proportion of traffic carried on our network versus resale of other carriers' services. Our margin is also affected by customer transfer and migration fees. We generally pay a charge to install and transfer a new customer onto our network and to migrate broadband and local customers. However, installing and migrating customers to our network infrastructure, enables us to increase our margin on such services as compared to resale of services using other carriers' networks.

Selling, general and administrative expenses are comprised primarily of salaries and benefits, commissions, occupancy costs, sales and marketing expenses, advertising, professional fees, and other administrative costs. All selling, general and administrative expenses are expensed when incurred. Emphasis on cost containment and the shift of expenditures from non-revenue producing expenses to sales and marketing expenses has been heightened since growth in net revenue has been under pressure.

Emergence from Voluntary Reorganization Under Chapter 11 Proceedings

On March 16, 2009, the Holding Companies filed Chapter 11 Cases in the Bankruptcy Court for reorganization relief under Chapter 11 of the Bankruptcy Code. Subsequently, the Holding Companies sought and received an order directing the joint administration of the Chapter 11 Cases under the caption In re: Primus Telecommunications Group, Incorporated, et al., Debtors, Case No. 09-10867. On April 24, 2009, an unsecured creditors' committee was appointed by the United States Trustee.

On April 27, 2009, the Bankruptcy Court approved the Holding Companies' use of a disclosure statement dated April 27, 2009 (the "Disclosure Statement") to solicit votes on the Plan of Reorganization. The Disclosure Statement was distributed to holders of record (as of April 27, 2009) of claims against, and interests in, the Holding Companies who were entitled to vote on the Plan of Reorganization (the "Record Date").

The Plan was confirmed by the Bankruptcy Court on June 12, 2009 (the "Confirmation Date"). On the Effective Date, the Holding Companies consummated their reorganization under the Bankruptcy Code and the Plan of Reorganization became effective. See "Note 2—Emergence from Voluntary Reorganization under Chapter 11 of the Bankruptcy Code" to our consolidated financial statements for further details.

As of July 1, 2009, we adopted fresh-start accounting in accordance with Financial Accounting Standards Board ASC No. 852, "Reorganizations." The adoption of fresh-start accounting resulted in the Company becoming a new entity for financial reporting purposes. Accordingly, the financial statements on or prior to July 1, 2009 are not comparable with the financial statements for periods after July 1, 2009. See "Note 4—Fresh-Start Accounting" to our consolidated financial statements for further detail.

Impact of Reorganization on Our Capital Structure, Long-Term Debt and Financial Statements.

Upon our emergence under the Plan of Reorganization on July 1, the Holding Companies' principal debt was reduced by \$316 million, or over 50%, interest payments were reduced by over 50% and certain debt maturities were extended. The significant features of the Plan included the developments summarized below:

- Holding's \$96 million Term Loan facility due February 2011 was reinstated and amended;
- IHC's \$173 million of outstanding 14 \(^1/4\)% Senior Secured Notes due 2011 were cancelled and the holders thereof received their pro rata portion of \$123.5 million of aggregate principal amount of 14 \(^1/4\)% Senior Subordinated Secured Notes due May 20, 2013 and 4,800,000 shares of the new Common Stock of Group (the "New Common Stock");
- the \$209 million in aggregate outstanding 5% Exchangeable Senior Notes and 8% Senior Notes issued by

- Holding (collectively, the "Holding Senior Notes") were cancelled, and the holders thereof received 4,800,000 shares of the New Common Stock and Class A warrants to purchase up to an aggregate of 3,000,000 shares of New Common Stock;
- the 3 ³/4 % Senior Notes due September 2010, 12 ³/4% Senior Notes due October 2009 and Step Up Convertible Subordinated Debentures due August 2009 issued by Group (collectively, the "Group Notes") were cancelled, and the holders thereof received Class B warrants to purchase up to an aggregate of 1,500,000 shares of the New Common Stock;
- all existing shares of common stock outstanding prior to the Effective Date (the "Old Common Stock") were cancelled on the Effective Date, and holders thereof received their pro rata share of contingent value rights ("Contingent Value Rights" or "CVRs") to acquire up to 2,665,000 shares of New Common Stock; and
- all outstanding equity incentive grants of Group were cancelled on the Effective Date, and the Primus Telecommunications Group, Incorporated Management Compensation Plan (the "Management Compensation Plan") became effective. As of the Effective Date, 400,000 restricted stock units, 400,000 service-based stock options and 100,000 performance-based stock options were granted to certain employees and executive officers under the Management Compensation Plan.

In accordance with the Plan, Group's certificate of incorporation and bylaws were amended and restated in their entirety. Group's Second Amended and Restated Certificate of Incorporation (the "Amended Certificate Incorporation") and Amended and Restated By-Laws (the "Amended By-Laws") both became effective on the Effective Date. The Amended Certificate of Incorporation provides for 80,000,000 shares of authorized New Common Stock and 20,000,000 shares of authorized new preferred stock, of which 9,600,000 shares of New Common Stock were issued on the Effective Date. A further description of the key provisions of the Amended Certificate of Incorporation and the Amended By-Laws is included in Group's registration statement on Form 8-A under "Description of Capital Stock" filed with the SEC on July 1, 2009, which description is incorporated herein by reference. This description is qualified in its entirety by reference to the full text of these documents, which are attached as exhibits to such Form 8-A and such text is incorporated herein by reference.

Prior to the Effective Date, Group had approximated 142,695,390 shares of Old Common Stock issued and outstanding, and all of these shares were cancelled in connection with the effectiveness of the Plan. On the Effective Date, Group issued 9,600,000 shares of New Common Stock and had an additional 7,165,000 shares of New Common Stock reserved for future issuance in respect of claims and interests filed and allowed under the Plan, consisting of (1) 3,000,000 shares of new common stock reserved for issuance upon exercise of Class A warrants, (2) 1,500,000 shares of new common stock reserved for issuance upon exercise of Class B warrants, and (3) 2,665,000 shares of new common stock reserved for distribution on account of contingent value rights. The total number of shares of New Common Stock issued and reserved for issuance under the Management Compensation Plan, the total number of shares of New Common Stock reserved for issuance under the Management Compensation Plan, the total number of shares of New Common Stock issued and reserved for issuance under the Plan was 17,765,000.

Foreign Currency

Foreign currency can have a major impact on our financial results. During 2009, approximately 82% of our net revenue was derived from sales and operations outside the U.S. The reporting currency for our consolidated financial statements is the United States dollar ("USD"). The local currency of each country is the functional currency for each of our respective entities operating in that country. In the future, we expect to continue to derive the majority of our net revenue and incur a significant portion of our operating costs from outside the U.S., and therefore changes in exchange rates have had and may continue to have a significant, and potentially adverse, effect on our results of operations. Our primary risk of loss regarding foreign currency exchange rate risk is caused primarily by fluctuations in the following exchange rates: USD/Canadian dollar ("CAD"), USD/Australian dollar ("AUD"), USD/British pound ("GBP"), and USD/Euro ("EUR"). Due to the large percentage of

our revenue derived outside of the U.S., changes in the USD relative to one or more of the foregoing currencies could have an adverse impact on our future results of operations. We have agreements with certain subsidiaries for repayment of a portion of the investments and advances made to these subsidiaries. As we anticipate repayment in the foreseeable future, we recognize the unrealized gains and losses in foreign currency transaction gain (loss) on the consolidated statements of operations. The exposure of our income from operations to fluctuations in foreign currency exchange rates is reduced in part because a majority of the costs that we incur in connection with our foreign operations are also denominated in local currencies.

We are exposed to financial statement gains and losses as a result of translating the operating results and financial position of our international subsidiaries. We translate the local currency statements of operations of our foreign subsidiaries into USD using the average exchange rate during the reporting period. Changes in foreign exchange rates affect the reported profits and losses and cash flows and may distort comparisons from year to year. By way of example, when the USD strengthens compared to the EUR, there could be a negative or positive effect on the reported results for Europe, depending upon whether Europe is operating profitably or at a loss. It takes more profits in EUR to generate the same amount of profits in USD and a greater loss in EUR to generate the same amount of loss in USD. The opposite is also true. For instance, when the USD weakens there is a positive effect on reported profits and a negative effect on the reported losses for Europe.

In the year ended December 31, 2009, as compared to the year ended December 31, 2008, the USD was stronger on average as compared to the CAD and AUD; the USD was weaker on average as compared to the GBP and the EUR. The following tables demonstrate the impact of currency fluctuations on our net revenue for the years ended December 31, 2009 and 2008:

Net Revenue by Location—in USD (in thousands)

		For the Year Ended December 31,								
	2009	2008								
	Net Revenue	Net Revenue	Variance	Variance %						
Canada	\$ 224,397	\$ 260,834	\$(36,437)	(14)%						
Australia	\$ 243,158	\$ 276,413	\$(33,254)	(12)%						
United Kingdom	\$ 93,078	\$ 87,706	\$ 5,372	6%						
Europe	\$ 89,182	\$ 87,623	\$ 1,559	2%						

Net Revenue by Location—in Local Currencies (in thousands)

		For the Year Ended December 31,								
	2009 Net Revenue	2008 Net Revenue	Variance	Variance %						
Canada (in CAD)	255,674	276,294	(20,621)	(7)%						
Australia (in AUD)	308,068	324,724	(16,655)	(5)%						
United Kingdom (in GBP)	59,919	48,335	11,583	24%						
Europe* (in EUR)	63,633	59,640	3,993	7%						

^{*} Europe includes only subsidiaries whose functional currency is the EUR.

In the year ended December 31, 2008, as compared to the year ended December 31, 2007, the USD was weaker on average as compared to the CAD, AUD and EUR, notwithstanding a significant strengthening of the USD relative to such currencies between June 30, 2008 and December 31, 2008, and stronger on average as compared to the GBP. The following tables demonstrate the impact of currency fluctuations on our net revenue for the year ended December 31, 2008 and 2007:

Net Revenue by Location—in USD (in thousands)

		For the Year Ended December 31,							
	2008	2007							
	Net Revenue	Net Revenue	Variance	Variance %					
Canada	\$ 260,834	\$ 262,412	\$(1,578)	(1)%					
Australia	\$ 276,413	\$ 285,963	\$(9,550)	(3)%					
United Kingdom	\$ 87,706	\$ 89,363	\$(1,657)	(2)%					
Europe	\$ 87,623	\$ 81,891	\$ 5,732	7%					

Net Revenue by Location—in Local Currencies (in thousands)

	<u></u>	For the Year Ended December 31,								
	2008	2007								
	Net Revenue	Net Revenue	Variance	Variance %						
Canada (in CAD)	276,294	281,419	(5,125)	(2)%						
Australia (in AUD)	324,724	341,361	(16,637)	(5)%						
United Kingdom (in GBP)	48,335	44,773	3,562	8%						
Europe* (in EUR)	59,640	59,771	(131)	0%						

^{*} Europe includes only subsidiaries whose functional currency is the EUR.

Critical Accounting Policies

To aid in the understanding of our financial reporting, our most critical accounting policies are described below. These policies have the potential to have a more significant impact on our financial statements, either because of the significance of the financial statement item to which they relate, or because they require judgment and estimation due to the uncertainty involved in measuring, at a specific point in time, events which are continuous in nature.

Revenue Recognition and Deferred Revenue—Net revenue is derived from carrying a mix of business, residential and carrier long-distance traffic, data and Internet traffic, and also from the provision of local, data center and wireless services. For voice and wholesale VoIP, net revenue is earned based on the number of minutes during a call and is recorded upon completion of a call, adjusted for allowance for doubtful accounts receivable, service credits and service adjustments. Revenue for a period is calculated from information received through our network switches. Customized software has been designed to track the information from the switch and analyze the call detail records against stored detailed information about revenue rates. This software provides us the ability to do a timely and accurate analysis of revenue earned in a period. Separate prepaid services software is used to track additional information related to prepaid service usage such as activation date, monthly usage amounts, fees and charges, and expiration date. Revenue on these prepaid services is recognized as service is provided until expiration, when all unused minutes, which are no longer available to the customers, are recognized as revenue.

Net revenue is also earned on a fixed monthly fee basis for unlimited local and long-distance plans and for the provision of data/Internet (including retail VoIP) and hosting services. Data/Internet and hosting services include monthly fees collected for the provision of dedicated and dial-up access at various speeds and bandwidths and collocation services. These fees are recognized as access and collocation is provided on a monthly basis. Additionally, service activation and installation fees are deferred and amortized over the longer of the average customer life or the contract term. We record payments received in advance for prepaid services and services to be provided under contractual agreements, such as Internet broadband and dial-up access, as deferred revenue until such related services are provided.

A portion of revenue, representing less than 1% of total revenue, is earned from the sale of wireless handsets and VoIP routers. We apply the provisions of ASC No. 605-25, "Revenue Recognition—Multiple Element Arrangements," which provides guidance on when and how an arrangement involving multiple deliverables should be divided into separate units of accounting. We have concluded that ASC No. 605-25 requires us to account for the sale of wireless handsets and VoIP routers and the related cost of handset and router revenues as a separate unit of accounting when title to the handset or router passes to the customer. Revenue recognized is the portion of the activation fees allocated to the router or handset unit of accounting in the statement of operations when title to the router or handset passes to the customer. We defer the portion of the activation fees allocated to the service unit of accounting and recognize such deferred fees on a straight-line basis over the contract life in the statement of operations.

Net revenue represents gross revenue, net of allowance for doubtful accounts receivable, service credits and service adjustments.

Allowance for doubtful accounts receivable—Determining our allowance for doubtful accounts receivable requires significant estimates. Due to the large number of customers that we serve, it is impractical to review the creditworthiness of each of our customers, although a credit review is performed for larger carrier and retail business customers. We consider a number of factors in determining the proper level of the allowance, including historical collection experience, current economic trends, the aging of the accounts receivable portfolio and changes in the creditworthiness of our customers. Systems to detect fraudulent call activity are in place within our network, but if these systems fail to identify such activity, we may realize a higher degree of uncollectible accounts.

Cost of revenue—Cost of revenue is comprised primarily of costs incurred from other domestic and foreign telecommunications carriers to originate, transport and terminate calls. The majority of our cost of revenue is variable, based upon the number of minutes of use, with transmission and termination costs being the most significant expense. Call activity is tracked and analyzed with customized software that analyzes the traffic flowing through our network switches and calculates the variable cost of revenue with predetermined contractual rates. If the domestic or foreign telecommunications carriers have tracked and invoiced the volume of minutes at levels different than what our activity shows or have invoiced at different rates, we will dispute the charges invoiced. There is no guarantee that we will prevail in such disputes. We use significant estimates to determine the level of success in dispute resolution and consider past historical experience, basis of disputes, financial status, and current relationships with vendors and aging of prior disputes in quantifying our estimates.

Valuation of long-lived assets—We review long-lived assets whenever events or changes indicate that the carrying amount of an asset may not be recoverable. In making such evaluations, we compare the expected undiscounted future cash flows to the carrying amount of the assets. If the total of the expected undiscounted future cash flows is less than the carrying amount of the assets, we are required to make estimates of the fair value of the long-lived assets in order to calculate the impairment loss equal to the difference between the fair value and carrying value of the assets.

We make significant assumptions and estimates in this process regarding matters that are inherently uncertain, such as determining asset groups and estimating future cash flows, remaining useful lives, discount rates and growth rates. The resulting undiscounted cash flows are projected over an extended period of time, which subjects those assumptions and estimates to an even larger degree of uncertainty. While we believe that our estimates are reasonable, different assumptions could materially affect the valuation of the long-lived assets.

We have concluded that we have one asset group: the network. This is due to the nature of our telecommunications network which utilizes all of the POPs, switches, cables and various other components throughout the network to seamlessly form the telecommunications gateway over which our products and services are carried for any given customer's phone call or data or Internet transmission. Furthermore, outflows to many of the external network providers are not separately assignable to revenue inflows for any phone call or service plan.

We make assumptions about the remaining useful life of our long-lived assets. The assumptions are based on the average life of our historical capital asset additions, our historical asset purchase trend and the fact that our primary assets, our network switches, have an 8-year life. Because of the nature of our industry, we also assume that the technology changes in the industry render all equipment obsolete with no salvage value after their useful lives. In certain circumstances in which the underlying assets could be leased for an additional period of time, we have included such estimated cash flows in our estimate.

The estimate of the appropriate discount rate to be used to apply the present value technique in determining fair value was our weighted average cost of capital, which is based on the effective rate of our long-term debt obligations at the current market values as well as the current volatility and trading value of our common stock.

Valuation of goodwill—Goodwill and other intangible assets—Under ASC No.350 "Intangibles—Goodwill and Other", goodwill and indefinite lived intangible assets are not amortized but are reviewed annually for impairment, or more frequently, if impairment indicators arise. Intangible assets that have finite lives are amortized over their useful lives and are subject to the impairment provisions of ASC 360, "Property Plant and Equipment."

Goodwill impairment is tested annually using a two-step process that begins with an estimation of the fair value of each reporting unit as of October 1. The first step is a screen for potential impairment by comparing the fair value of a reporting unit with its carrying amount. The second step measures the amount of impairment loss, if any, by comparing the implied fair value of the reporting unit goodwill with its carrying amount. The implied fair value of goodwill shall be determined in the same manner as the amount of goodwill recognized in a business combination is determined, that is through an allocation of the fair value of a reporting unit to all of the assets and liabilities of that unit as if the reporting unit had been acquired in a business combination. If the carrying amount of the reporting unit goodwill exceeds the implied fair value of that goodwill, an impairment loss shall be recognized in an amount equal to that excess.

Our reporting units are the same as our operating segments as each segment's components have been aggregated and deemed a single reporting unit because they have similar economic characteristics. Each component is similar in that they each provide telecommunications services for which all of the resources and costs are drawn from the same pool, and are evaluated using the same business factors by management.

Intangible assets not subject to amortization consist of trade names. Such indefinite lived intangible assets are tested for impairment annually, or more frequently if events or changes in circumstances indicate that the asset might be impaired. The impairment test shall consist of a comparison of the fair value of an intangible asset with its carrying amount. If the carrying amount of the intangible asset exceeds its fair value, an impairment loss shall be recognized in an amount equal to the excess.

Intangible assets subject to amortization consist of certain trade names and customer relationships. These finite lived intangible assets are amortized based on their useful lives. Such assets are subject to the impairment provisions of ASC 360, wherein impairment is recognized and measured only if there are events and circumstances that indicate that the carrying amount may not be recoverable. The carrying amount is not recoverable if it exceeds the sum of the undiscounted cash flows expected to result from the use of the asset group. An impairment loss is recorded if after determining that it is not recoverable, the carrying amount exceeds the fair value of the asset.

As disclosed in Note 4—"Fresh Start Accounting" to the Company's consolidated financial statements, on July 1, 2009, the Company adopted fresh-start accounting in accordance with ASC No. 852, "Reorganizations." Fresh-start accounting reflects the value of the Company as determined in the confirmed Plan. Under fresh-start accounting, the Company's asset and liability values are re-measured and allocated in conformity with ASC No. 805, "Business Combinations." The excess of reorganization value over the fair value of tangible and identifiable intangible assets is recorded as goodwill in the accompanying consolidated balance sheet. Fresh-start accounting also requires that all liabilities, other than deferred taxes, should be stated at fair value.

The carrying values by reporting unit of the goodwill and other indefinite-lived intangible assets are disclosed in Note 6—"Goodwill and Other Intangible Assets," to the Company's Consolidated Financial Statements.

To facilitate the first step of the goodwill impairment analysis the Company first determines the fair value of each reporting unit as of October 1, the goodwill measurement date, and compares those amounts to the carrying value of the respective reporting units as of October 1, 2009. To the extent the fair value of the reporting unit exceeds the book value, no additional analysis is required and no impairment is recognized. To facilitate the Company's calculation of the fair value of the reporting units, the valuation methods included (i) a discounted cash flow analysis, considering a range of between 12.5% and 17% weighted average costs of capital and market-based multiples of projected earnings of between 2.8 and 5.1 times for its terminal value. A key assumption in the fair value calculations is the Company's future operating performance and resulting cash flow which is inherently subject to significant uncertainties and contingencies, and are based on management's best estimates at the date of measurement. Potential events which could have a negative effect on the key assumptions include competitive actions, technological developments, regulatory actions and currency movements. For additional risk factors which could affect the assumptions see "Item 1A—Risk Factors" and "Item 7—Special Note Regarding Forward Looking Statements." Accordingly, the Company cannot provide assurance that the estimates, assumptions, and values reflected in the valuations will be realized, and actual results could vary materially. The goodwill impairment test date of October 1, 2009, was three months after the initial recording of the fair value of our assets under fresh start accounting. After performing step one of the goodwill impairment test, no impairment was identified, as the fair value was greater than the carrying value of each reporting unit. In the reporting units with significant goodwill balances which are the United States and Canada, the Canada reporting unit is not at risk of failing step one of the test, but the United States reporting unit's fair value exceeded the carrying value of the United States reporting unit by 15%. The increase in fair value for the Canadian Reporting Unit is due primarily to foreign currency translation as the United States dollar has weakened considerably and cost reduction initiatives were implemented in the second half of the year. Additionally, no impairment was recognized based on the analysis of the fair value of the intangible assets not subject to amortization as compared to their carrying value.

In addition to the forgoing, the Company reviews its goodwill and intangible assets for possible impairment whenever events or circumstances indicate that certain carrying amounts of assets may not be recoverable. The factors that the Company considers important, and which could trigger an impairment review, include, but are not limited to: a significant decline in the market value of our common stock or debt securities for a sustained period; a material adverse change in economic, financial market, industry or sector trends; a material failure to achieve operating results relative to historical levels or projected future levels; and significant changes in operations or business strategy.

Accounting for income taxes—We recognize deferred tax assets and liabilities for the expected future tax consequences of transactions and events. Under this method, deferred tax assets and liabilities are determined based on the difference between the financial statement bases and the tax bases of assets and liabilities using enacted tax rates in effect for the year in which the differences are expected to reverse. If necessary, deferred tax assets are reduced by a valuation allowance to an amount that is determined to be more likely than not recoverable. We must make significant estimates and assumptions about future taxable income and future tax consequences when determining the amount of the valuation allowance. The additional guidance provided by ASC No. 740 "Income Taxes, clarifies the accounting for uncertainty in income taxes recognized in the financial statements. At January 1, 2007, its implementation resulted in adjustments to increase our total unrecognized tax benefits by \$106.4 million. Expected outcomes of current or anticipated tax examinations, refund claims and tax-related litigation and estimates regarding additional tax liability (including interest and penalties thereon) or refunds resulting therefrom will be recorded based on the guidance provided by ASC No. 740 to the extent applicable.

At present, our subsidiaries in the major jurisdictions in which we operate have significant deferred tax assets resulting from tax loss carryforwards. With the exception of our Canadian companies, these deferred tax assets are fully offset with valuation allowances. The appropriateness and amount of these valuation allowances

are based on our assumptions about the future taxable income of each affiliate or available tax planning strategies. Except in the case of our Canadian companies, if our assumptions have significantly underestimated future taxable income with respect to a particular affiliate, all or part of the valuation allowance for the affiliate would be reversed and additional income could result. With the exception of our Canadian affiliates, if our assumptions have significantly overestimated future taxable income with respect to a particular affiliate, there would be no change in the net value of the deferred tax asset and no additional income or tax expense would result. If our assumptions with respect to our Canadian affiliates have significantly overestimated future taxable income, a full or partial valuation allowance would be applied to the corresponding deferred tax assets and additional tax expense would result.

In accordance with ASC No. 852, for periods including and subsequent to the filing of the Chapter 11 petition, prior to the emergence from bankruptcy, all pre-petition liabilities subject to compromise were segregated in the Consolidated Condensed Balance Sheets and classified as liabilities subject to compromise, at management's estimate of the amount of allowable claims. Liabilities not subject to compromise were separately classified as current and non-current in the Consolidated Condensed Balance Sheet. Revenues, expenses, realized gains and losses, and provisions for losses that result from the reorganization were reported separately as reorganization items, net, in the Consolidated Condensed Statements of Operations. Net cash used for reorganization items was disclosed separately in the Consolidated Condensed Statements of Cash Flows.

After the emergence from bankruptcy on July 1, 2009, the amounts reported on our subsequent financial statements materially changed. We adopted the "fresh start" provisions of ASC No. 852, which requires that all assets and liabilities except deferred taxes be restated to their fair value. Deferred tax balances have been established as a result of the differences in the basis adjustments from fresh-start accounting. Certain of these fair values differ materially from the values recorded on the Predecessor Consolidated Condensed Balance Sheets. Our emergence from reorganization resulted in a new reporting entity that had no retained earnings or accumulated deficit before the Effective Date. Additionally, we must also adopt any changes in GAAP that are otherwise required to be adopted within twelve months of such date. For all of these reasons, our Successor's financial statements are not comparable to our Predecessor's. See "Note 8—Income Taxes," to the Company's consolidated financial statements for further details.

Discontinued Operations

In the first quarter 2009, the Company sold certain assets of its Japan retail operations. The sale price was \$0.4 million (40 million Japanese yen), which included \$0.2 million (20 million Japanese yen) in cash and \$0.2 million (20 million Japanese yen) receivable. The Company recorded a \$0.3 million gain from sale of assets. The Company reported Japan retail operations as a discontinued operation.

In the second quarter 2008, the Company determined it would sell its German retail operations. However, buyers were not found; therefore the Company decided to cease operations of the German retail business during the first quarter of 2009.

In August 2007, the Company sold its 51% interest in its German telephone installation system subsidiaries. The sale price was \$0.8 million (0.6 million Euros), which included \$0.5 million (0.4 million Euros) in cash and \$0.3 million (0.2 million Euros) for payment of outstanding intercompany debt. For the intercompany debt payment, the Company received \$0.1 million (0.1 million Euros) in cash at closing. The balance owing is represented by a note receivable and will be paid in fifteen equal monthly installment payments. As a result, the Company recorded a \$0.2 million gain from sale of assets. Net assets held for sale were \$0.6 million at the closing date.

In February 2007, the Company sold its Australian domain name registry and web hosting subsidiary, Planet Domain. The sale price was \$6.5 million (\$8.3 million AUD). The Company received \$5.5 million in net cash proceeds from the transaction after closing adjustments. As a result, the Company recorded a \$6.0 million gain from sale of assets. The net assets of Planet Domain were \$0.2 million at the closing date.

As a result of these events, the Company's consolidated financial statements for all periods presented reflect the Japan retail operations, German retail operations, discontinued German subsidiary and Planet Domain operations as discontinued operations for the six months ended December 31, 2009, six months ended July 1, 2009, years ended December 31, 2008 and 2007. Accordingly, revenue, costs, and expenses of the discontinued operations have been excluded from the respective captions in the consolidated statements of operations. The net operating results of the discontinued operations have been reported, net of applicable income taxes as loss from discontinued operations.

Summarized operating results of the discontinued operations for the year ended December 31, 2009, 2008 and 2007 are as follows:

	Successor			
(in thousands)	Six Months Ended December 31, 2009	Six Months Ended July 1, 2009	Year Ended	December 31, 2007
Net revenue	\$ —	\$ 300	\$ 3,851	\$ 9,961
Operating expenses	\$ 38	\$ 995	\$ 4,610	\$ 10,233
Income (loss) from operations	\$ (38)	\$ (695)	\$ (759)	\$ (272)
Interest expense	\$ —	\$ —	\$ —	\$ (26)
Interest income and other income (expense)	\$ (233)	\$ 19	\$ 60	\$ 39
Foreign currency transaction gain (loss)	\$ —	\$ —	\$ 376	\$ (6)
Income (loss) before income tax	\$ (271)	\$ (676)	\$ (323)	\$ (265)
Income tax expense	\$ —	\$ —	\$ (3)	\$ (3)
Income (loss) from discontinued operations	\$ (271)	\$ (676)	\$ (326)	\$ (268)

Explanatory Preliminary Note

In the following presentations and narratives within this Management's Discussion and Analysis of Financial Condition and Results of Operations, we combine Successor's results of operations for the period from July 2, 2009 to December 31, 2009 (the "Successor Period") with Predecessor's results of operations for the period from January 1, 2009 to July 1, 2009 and we compare these combined results of operations with Predecessor's results of operations for the year ended December 31, 2008. We also present detailed changes in results, excluding currency impacts, since a large portion of our revenues are derived outside of the U.S., and currency changes can influence or mask underlying changes in foreign operating unit performance. We believe that the comparison of the combined financial results provide management and investors with a meaningful analysis of our performance and trends for comparative purposes. In addition, it should be noted that the application of fresh-start accounting will have a significant non-cash impact on our future results of operations, but will have no impact on the underlying cash flows of the Company.

We have combined the results of operations and the sources and uses of cash for the six months ended June 30, 2009 of the Predecessor and the six months ended December 31, 2009 for the Successor, and compared these combined results with the corresponding periods in 2008 and 2007 to provide a meaningful perspective on our financial and operational performance and trends in order to supplement GAAP data that would not otherwise have been available if we had not combined the results of operations and sources and uses of cash of the Predecessor and the Successor in this manner. The presentation of these combined results for the twelve months ended December 31, 2009 are intended to supplement investors' understanding of our operating performance and liquidity. However, the combined presentation of the twelve months ended December 31, 2009 are not intended to replace net income (loss), cash flows, financial position or comprehensive income (loss), as determined in accordance with U.S. GAAP for the Predecessor and Successor periods. The Consolidated Financial Statements on or after July 1, 2009 are not comparable to the Consolidated Financial Statements prior to that date. For purposes of calculating constant currency rates between periods in connection with presentations

that describe changes in values "excluding currency effects" herein, we have taken results from foreign operations for a given year (that were computed in accordance with GAAP using local currency) and converted such amounts utilizing the same U.S. dollar to applicable local currency exchange rates that were used for purposes of calculating corresponding preceding year GAAP presentations.

In reviewing the results and narratives below, it is important to note that there were significant effects resulting from the adoption of fresh-start accounting that affected our historical presentations and that will impact future results compared to pre-Reorganization results, including significant changes in:

- debt balances and associated interest expense;
- taxes and the adverse cash flow effects of our obligation to pay additional taxes compared to prior periods, given the termination of significant net operating loss carry-forward credits in connection with the Reorganization; and
- depreciation and amortization, as triggered by our requirement to institute a new capital structure and fully re-measure our tangible and identifiable intangible assets.

Results of Operations

The following information for the years ended December 31, 2009, 2008 and 2007 reflects all the items included in consolidated statements of operations as a percentage of net revenue:

	Year		
	2009	2008	2007
NET REVENUE	100.0%	100.0%	100.0%
OPERATING EXPENSES			
Cost of revenue (exclusive of depreciation included below)	65.4%	63.6%	61.5%
Selling, general and administrative	24.4%	29.1%	31.4%
Depreciation and amortization	6.4%	3.7%	3.4%
(Gain) loss on sale or disposal of assets	0.0%	-0.7%	0.2%
Total operating expenses	96.2%	95.7%	96.5%
INCOME FROM OPERATIONS	3.8%	4.3%	3.5%
INTEREST EXPENSE	-3.9%	-6.0%	-6.8%
ACCRETION (AMORTIZATION) ON DEBT PREMIUM/DISCOUNT, net	0.0%	0.1%	-0.1%
GAIN (LOSS) ON EARLY EXTINGUISHMENT OR RESTRUCTURING OF DEBT	-0.5%	4.1%	-0.9%
GAIN (LOSS) FROM CONTINGENT VALUE RIGHTS VALUATION	-0.3%	0.0%	0.0%
INTEREST AND OTHER INCOME	0.1%	0.4%	0.6%
FOREIGN CURRENCY TRANSACTION GAIN (LOSS)	4.8%	-5.3%	3.6%
LOSS FROM CONTINUING OPERATIONS BEFORE REORGANIZATION ITEMS AND INCOME			
TAXES	4.1%	-2.4%	0.1%
REORGANIZATION ITEMS, net	53.9%	0.0%	0.0%
INCOME (LOSS) FROM CONTINUING OPERATIONS BEFORE INCOME TAXES	58.0%	-2.4%	0.1%
INCOME TAX BENEFIT (EXPENSE)	0.7%	0.0%	1.0%
INCOME (LOSS) FROM CONTINUING OPERATIONS	58.7%	-2.4%	1.1%
LOSS FROM DISCONTINUED OPERATIONS, net of tax	-0.1%	0.0%	0.0%
GAIN (LOSS) FROM SALE OF DISCONTINUED OPERATIONS, net of tax	0.0%	0.0%	0.7%
NET INCOME (LOSS)	58.6%	-2.4%	1.8%

The following information reflects net revenue by product line for the years ended December 31, 2009, 2008 and 2007 (in thousands, except percentages) and is provided for informational purposes and should be read in conjunction with the Consolidated Financial Statements and Notes.

(in thousands)	2009	%	2008	%	2007	%
Retail Voice	\$393,722	48%	\$461,817	52%	\$481,160	54%
Wholesale	212,961	26%	197,278	22%	170,450	19%
Data/Internet	160,040	20%	182,891	20%	196,198	22%
Retail VOIP	48,827	6%	53,877	6%	48,221	5%
Total	\$815,550	100%	\$895,863	100%	\$896,029	100%

Results of operations for the year ended December 31, 2009 as compared to the year ended December 31, 2008

Net revenue: Net revenue, exclusive of the currency effect, decreased \$19.3 million, or 2.2%, to \$876.5 million for the year ended December 31, 2009 from \$895.9 million for the year ended December 31, 2008. Inclusive of the currency effect, which accounted for a decrease of \$61.0 million, net revenue decreased \$80.3 million to \$815.6 million for the year ended December 31, 2009 from \$895.9 million for the year ended December 31, 2008.

								Inclusiv	e of
		Exclusive of Currency Effect						Currency	Effect
		Year Ended						Year En	ded
	December 3	December 31, 2009		December 31, 2008		Year-over-Year		December 31, 2009	
	Net	% of	Net	% of			Currency	Net	% of
(in thousands)	Revenue	Total	Revenue	Total	Variance	Variance %	Effect	Revenue	Total
Canada	241,603	27.6%	260,834	29.1%	(19,231)	-7.4%	(17,206)	224,397	27.5%
Australia	262,917	30.0%	276,413	30.9%	(13,496)	-4.9%	(19,759)	243,158	29.9%
Wholesale	230,285	26.3%	197,278	22.0%	33,007	16.7%	(17,323)	212,962	26.1%
United States	66,146	7.5%	86,578	9.7%	(20,432)	-23.6%	_	66,146	8.1%
Europe	59,532	6.8%	64,629	7.2%	(5,097)	-7.9%	(5,338)	54,194	6.6%
Brazil	16,040	1.8%	10,131	1.1%	5,909	58.3%	(1,347)	14,693	1.8%
Total Revenue	\$876,523	100.0%	\$895,863	100.0%	\$(19,340)	-2.2%	\$(60,973)	\$815,550	100.0%

Canada: Canada net revenue, exclusive of the currency effect, decreased \$19.2 million, or 7.4%, to \$241.6 million for the year ended December 31, 2009 from \$260.8 million for the year ended December 31, 2008. The net revenue decrease is primarily attributable to a decrease of \$16.3 million in retail voice services, a decrease of \$3.8 million in prepaid voice services, a decrease of \$1.1 million in wireless services and a decrease of \$0.3 million in local services offset, in part, by an increase of \$2.3 million in Internet, data and hosting services. Inclusive of the currency effect, which accounted for a \$17.2 million decrease, net revenue decreased \$36.4 million to \$224.4 million for the year ended December 31, 2009 from \$260.8 million for the year ended December 31, 2008.

Australia: Australia net revenue, exclusive of the currency effect, decreased \$13.5 million, or 4.9%, to \$262.9 million for the year ended December 31, 2009 from \$276.4 million for the year ended December 31, 2008. The net revenue decrease is primarily attributable to a decrease of \$13.5 million in Internet services, a decrease of \$3.1 million in residential voice and a decrease of \$2.6 million in broadband services offset, in part, by increases of \$3.9 million in business voice services and \$1.9 million in wireless services. Inclusive of the currency effect, which accounted for a \$19.8 million decrease, net revenue decreased \$33.2 million to \$243.2 million for the year ended December 31, 2009 from \$276.4 million for the year ended December 31, 2008.

Wholesale: Wholesale net revenue, exclusive of the currency effect, increased \$33.0 million, or 16.7%, to \$230.3 million for the year ended December 31, 2009 from \$197.3 million for the year ended December 31,

2008. The net revenue increase is primarily a result of expanded routed capabilities and enhanced traffic volumes from existing customers and the obtainment of new customers. Inclusive of the currency effect which accounted for a \$17.3 million decrease, net revenue increased \$15.7 million to \$213.0 million for the year ended December 31, 2009, from \$197.3 million for the year ended December 31, 2008.

United States: United States net revenue decreased \$20.4 million, or 23.6%, to \$66.1 million for the year ended December 31, 2009 from \$86.6 million for the year ended December 31, 2008. The decrease is primarily attributable to a decrease of \$13.1 million in retail voice services (for residential and small businesses), a decrease of \$5.6 million in VoIP services and a decrease of \$1.8 million in Internet services.

Europe: Europe net revenue, exclusive of the currency effect, decreased \$5.1 million, or 7.9%, to \$59.5 million for the year ended December 31, 2009 from \$64.6 million for the year ended December 31, 2008. The decrease is primarily attributable to a decline in retail voice services. Inclusive of the currency effect, which accounted for a \$5.3 million decrease, net revenue decreased \$10.4 million to \$54.2 million for the year ended December 31, 2009 from \$64.6 million for the year ended December 31, 2008.

Brazil: Brazil net revenue, exclusive of the currency effect, increased \$5.9 million, or 58.4%, to \$16.0 million for the year ended December 31, 2009 from \$10.1 million for the year ended December 31, 2008. The revenue increase is due primarily to an increase in VoIP services. Inclusive of the currency effect, which accounted for a \$1.3 million decrease, net revenue increased \$4.6 million to \$14.7 million for the year ended December 31, 2009 from \$10.1 million for the year ended December 31, 2008.

Cost of revenue: Cost of revenue, exclusive of the currency effect, increased \$5.3 million to \$575.2 million, or 65.6% of net revenue, for the year ended December 31, 2009 from \$569.9 million, or 63.6% of net revenue, for the year ended December 31, 2008. Inclusive of the currency effect, which accounted for a \$41.5 million decrease, cost of revenue decreased \$36.2 million to \$533.7 million for the year ended December 31, 2009 from \$569.9 million for the year ended December 31, 2008.

Canada: Canada cost of revenue, exclusive of the currency effect, decreased \$8.4 million to \$106.0 million, or 43.9% of net revenue, for the year ended December 31, 2009 from \$114.4 million, or 43.9% of net revenue, for the year ended December 31, 2008. The decrease is primarily attributable to a decrease in net revenue of \$19.2 million. Inclusive of the currency effect, which accounted for a \$7.5 million decrease, cost of revenue decreased \$15.9 million to \$98.5 million for the year ended December 31, 2008.

Australia: Australia cost of revenue, exclusive of the currency effect, decreased \$4.0 million to \$165.4 million, or 62.9% of net revenue, for the year ended December 31, 2009 from \$169.4 million, or 61.3% of net revenue, for the year ended December 31, 2008. The decrease is primarily attributable to a \$13.5 million decrease in net revenue, offset in part by a \$5.8 million reduction to cost of revenue in 2008 as the result of rulings by the Australian regulatory authority. Inclusive of the currency effect, which accounted for a \$12.4 million decrease, cost of revenue decreased \$16.4 million to \$153.0 million for the year ended December 31, 2009 from \$169.4 million for the year ended December 31, 2008.

Wholesale: Wholesale cost of revenue, exclusive of the currency effect, increased \$30.4 million to \$220.6 million, or 95.8% of net revenue, for the year ended December 31, 2009 from \$190.2 million, or 96.4% of net revenue, for the year ended December 31, 2008. The increase is primarily attributable to an increase in net revenue of \$33.0 million offset, in part, by lower costs, as a percentage of net revenues. Inclusive of the currency effect, which accounted for a \$16.7 million decrease, cost of revenues decreased \$13.7 million to \$203.9 million for the year ended December 31, 2009 from \$190.2 million for the year ended December 31, 2008.

United States: United States cost of revenue decreased \$15.5 million to \$29.9 million, or 45.2% of net revenue, for the year ended December 31, 2009 from \$45.4 million, or 52.5% of net revenue, for the year ended December 31, 2008. The decrease is attributable to a decrease in net revenue of \$20.4 million, a cost recovery of \$1.5 million from a dispute settlement and lower costs resulting from our cost control initiatives.

Europe: Europe cost of revenue, exclusive of the currency effect, decreased by \$2.6 million to \$41.5 million, or 69.7% of net revenue, for the year ended December 31, 2009 from \$44.1 million, or 68.3% of net revenue, for the year ended December 31, 2008. The decrease is primarily due to a decrease in net revenue of \$5.1 million offset, in part, by slightly higher costs, as a percentage of net revenue. Inclusive of the currency effect, which accounted for a \$3.8 million decrease, cost of revenue decreased \$6.4 million to \$37.7 million for the year ended December 31, 2009 from \$44.1 million for the year ended December 31, 2008.

Brazil: Brazil cost of revenue, exclusive of the currency effect, increased \$5.3 million to \$11.7 million, or 73.1% of net revenue, for the year ended December 31, 2009 from \$6.4 million, or 62.8% of net revenue, for the year ended December 31, 2008. The increase is primarily attributable to an increase in net revenue of \$5.9 million and a shift in the revenue product mix to lower margin products. Inclusive of the currency effect, which accounted for a \$1.0 million decrease, cost of revenue increased \$4.4 million to \$10.8 million for the year ended December 31, 2009 from \$6.4 million for the year ended December 31, 2008.

Selling, general and administrative expenses: Selling, general and administrative expenses, exclusive of the currency effect, decreased \$48.8 million to \$211.6 million, or 24.1% of net revenue, for the year ended December 31, 2009 from \$260.4 million, or 29.1% of net revenue, for the year ended December 31, 2008. Inclusive of the currency effect, which accounted for a \$12.9 million decrease, selling, general and administrative expenses decreased \$61.7 million to \$198.7 million for the year ended December 31, 2009 from \$260.4 million for the year ended December 31, 2008.

Canada: Canada selling, general and administrative expense, exclusive of the currency effect, decreased \$12.9 million to \$85.0 million, or 35.2% of net revenue, for the year ended December 31, 2009 from \$97.9 million, or 37.5% of net revenue, for the year ended December 31, 2008. The decrease is attributable to a decrease of \$7.0 million in salaries and benefits, a decrease of \$7.1 million in sales and marketing expenses and a decrease of \$0.7 million in general and administrative expenses offset, in part, by an increase of \$0.8 million in advertising expenses and \$1.1 million of other expenses. Inclusive of the currency effect, which accounted for a \$5.7 million decrease, selling, general and administrative expenses decreased \$18.6 million to \$79.3 million for the year ended December 31, 2008 from \$97.9 million for the year ended December 31, 2008.

Australia: Australia selling, general and administrative expense, exclusive of the currency effect, decreased \$18.9 million to \$63.3 million, or 24.1% of net revenue, for the year ended December 31, 2009 from \$82.2 million, or 29.7% of net revenue, for the year ended December 31, 2008. The decrease is attributable to a decrease of \$6.1 million in salaries and benefits, a decrease of \$8.4 million in sales and marketing expense, a decrease of \$2.0 million in advertising expenses, a decrease of \$1.7 million in general and administrative and a decrease of \$0.7 million in all other expenses. Inclusive of the currency effect, which accounted for a \$4.7 million decrease, selling, general and administrative expense decreased \$23.6 million to \$58.6 million for the year ended December 31, 2009 from \$82.2 million for the year ended December 31, 2008.

Wholesale: Wholesale selling, general and administrative expense, exclusive of the currency effect, decreased \$1.2 million to \$7.2 million, or 3.1% of net revenue, for the year ended December 31, 2009 from \$8.4 million, or 4.3% of net revenue, for the year ended December 31, 2008. Inclusive of the currency effect, which accounted for a \$0.5 million decrease, selling, general and administrative expense decreased \$1.7 million to \$6.7 million for the year ended December 31, 2009 from \$8.4 million for the year ended December 31, 2008.

United States: United States selling, general and administrative expense decreased \$10.1 million to \$25.1 million, or 37.9% of net revenue, for the year ended December 31, 2009 from \$35.2 million, or 40.6% of net revenue, for the year ended December 31, 2008. The decrease is attributable to decreases of \$4.0 million in salaries and benefits, \$2.1 million in general and administrative expenses, \$1.6 million in advertising expenses, \$1.1 million in sales and marketing expense, \$0.7 million in occupancy, and \$0.6 million in other expenses.

Europe: Europe selling, general and administrative expense, exclusive of the currency effect, decreased \$4.6 million to \$17.2 million, or 28.8% of net revenue, for the year ended December 31, 2009 from \$21.8 million, or

33.7% of net revenue, for the year ended December 31, 2008. The decrease is attributable to a decrease of \$1.8 million in salaries and benefits and \$1.1 million of general and administrative and \$1.7 million in other expenses. Inclusive of the currency effect, which accounted for a \$1.7 million decrease, selling, general and administrative expense decreased \$6.3 million to \$15.5 million for the year ended December 31, 2009 from \$21.8 million for the year ended December 31, 2008.

Brazil: Brazil selling, general and administrative expense, exclusive of the currency effect, decreased \$0.5 million to \$3.5 million, or 21.6% of net revenue, for the year ended December 31, 2009 from \$4.0 million, or 39.8% of net revenue, for the year ended December 31, 2008. The decrease is attributable to a \$0.6 million decrease in salaries and benefits and a \$0.3 million decrease in sales and marketing offset in part by a \$0.3 million increase in all other expenses. Inclusive of the currency effect, which accounted for a \$0.3 million decrease, selling, general and administrative expense decreased \$0.9 million to \$3.1 million for the year ended December 31, 2009 from \$4.0 million for the year ended December 31, 2008.

Corporate: Corporate selling, general and administrative expense decreased \$0.3 million to \$10.8 million for the year ended December 31, 2009 from \$11.1 million for the year ended December 31, 2008. The decrease is primarily due to a decrease in salaries and benefits.

Depreciation and amortization expense: Depreciation and amortization expense was \$51.9 and \$32.8 million for the years ended December 31, 2009 and 2008, respectively. This change occurred as a result of the fair valuing of fixed and intangible assets per Fresh-Start accounting which was implemented effective July 1, 2009. See "Note 4—Fresh-Start Accounting" in our consolidated financial statements for further detail.

Interest expense and accretion (amortization) on debt discount/premium: Interest expense and accretion (amortization) on debt discount/premium, net decreased \$22.0 million to \$31.3 million for the year ended December 31, 2009 from \$53.3 million for the year ended December 31, 2008. The decrease was due to the lower debt levels primarily resulting from the cancellation of certain indebtedness through the Plan of Reorganization, effective as of July 1, 2009.

Gain (loss) from contingent value rights valuation. The value of the contingent value rights decreased \$2.8 million during 2009 due to the change of the fair market value. The Company determined these contingent value rights to be derivative instruments to be accounted for as liabilities and were marked to fair value, (and in future periods will be marked to fair value), at each balance sheet date. Upon issuance of the contingent value rights, the Company recorded a liability of \$2.6 million in other liabilities as part of fresh-start accounting, and we will adjust this liability quarterly to its then estimated fair value, (which in future periods potentially could be substantially greater than the initial recorded liability balance). The change in fair value of the liability is reflected in our Statements of Operations as other income (expense). Estimates of fair value represent the Company's best estimates based on a Black-Scholes pricing model.

Interest income and other (expense). Interest income and other (expense) decreased \$2.3 million to \$1.0 million for the year ended December 31, 2009 from \$3.3 million expense for the year ended December 31, 2008.

Foreign currency transaction gain (loss): Foreign currency transaction gain increased \$87.1 million to a gain of \$39.5 million for the year ended December 31, 2009 from a loss of \$47.6 million for the year ended December 31, 2008. The gains are attributable to the impact of foreign currency exchange rate changes on intercompany debt balances and on receivables and payables denominated in a currency other than the subsidiaries' functional currency.

Reorganization items, net. Reorganization items reflect the favorable impact of \$439.5 million for the year ended December 31, 2009 due to fresh-start accounting and revaluation of assets and liabilities (\$188.6 million), gain on extinguishment of debt (\$243.2 million), and reversal of future interest payments recorded as long-term obligations (\$20.4 million), offset in part principally by the incurrence of professional fees as a result of the Chapter 11 Cases (\$12.6 million).

Income tax benefit (expense): Income tax benefit (expense) is a \$6.1 million benefit for the year ended December 31, 2009 compared to a \$0.4 million benefit for the year ended December 31, 2008. The benefit includes expenses consisting of foreign withholding tax on intercompany interest, the release of certain "ASC 740" liabilities as a result of the expiration of the statute of limitations and charges for uncertain tax positions under ASC No. 740, "Income Taxes." The foreign tax expense was offset by a significant benefit from the release of deferred tax liabilities related to amortization of certain fresh-start adjustments to fixed assets and intangibles as discussed in "Note 8—Income Taxes," to our consolidated audited financial statements.

Results of operations for the year ended December 31, 2008 as compared to the year ended December 31, 2007

Net revenue: Net revenue, exclusive of the currency effect, decreased \$4.2 million, or 0.5%, to \$891.8 million for the year ended December 31, 2008 from \$896.0 million for the year ended December 31, 2007. Inclusive of the currency effect, which accounted for an increase of \$4.1 million, net revenue decreased \$0.1 million to \$895.9 million for the year ended December 31, 2008 from \$896.0 million for the year ended December 31, 2007.

	Exclusive of Currency Effect							Incl	Inclusive of Currency Effect	
	Year Ended							Year Ended		
	December 31, 2008		December 31, 2007		Year-over-Year				December 31, 2008	
(in thousands)	Net Revenue	% of Total	Net Revenue	% of Total	Variance	Variance %	Currency Effect	R	Net evenue	% of Total
Canada	\$258,066	28.9%	\$262,200	29.3%	\$ (4,134)	-1.6%	\$ 2,768	\$	260,834	29.1%
Australia	272,090	30.5%	284,935	31.8%	(12,845)	-4.5%	4,323		276,414	30.9%
Wholesale	200,729	22.5%	170,450	19.0%	30,279	17.8%	(3,451)		197,278	22.0%
United States	86,578	9.7%	101,583	11.3%	(15,005)	-14.8%	_		86,579	9.7%
Europe	64,680	7.3%	68,943	7.7%	(4,263)	-6.2%	(51)		64,629	7.2%
Brazil	9,662	1.1%	7,918	0.9%	1,744	22.0%	469		10,129	1.1%
Total Revenue	\$891,805	100.0%	\$896,029	100.0%	\$ (4,224)	-0.5%	\$ 4,058	\$	895,863	100.0%

Canada: Canada net revenue, exclusive of the currency effect, decreased \$4.1 million, or 1.6%, to \$258.1 million for the year ended December 31, 2008 from \$262.2 million for the year ended December 31, 2007. The revenue decrease is primarily attributable to a decrease of \$10.4 million in retail voice services, and a decrease of \$1.1 million in prepaid voice services offset, in part, by an increase of \$3.2 million in local service, an increase of \$3.2 million in Internet, data and hosting, an increase of \$0.7 million in VoIP and an increase of \$0.2 million in wireless. Inclusive of the currency effect, which accounted for a \$2.8 million increase, net revenue decreased \$1.4 million to \$260.8 million for the year ended December 31, 2008 from \$262.2 million for the year ended December 31, 2007.

Australia: Australia net revenue, exclusive of the currency effect, decreased \$12.8 million, or 4.5%, to \$272.1 million for the year ended December 31, 2008 from \$284.9 million for the year ended December 31, 2007. The revenue decrease is primarily attributable to a decrease of \$16.9 million in Internet services, a decrease of \$2.5 million in residential voice and, a decrease of \$0.6 million in broadband services offset, in part, by increases of \$5.4 million in business services, \$0.9 million in wireless services and \$0.9 million in VoIP and all other. Inclusive of the currency effect, which accounted for a \$4.3 million increase, net revenue decreased \$8.5 million to \$276.4 million for the year ended December 31, 2007.

Wholesale: Wholesale net revenue, exclusive of the currency effect, increased \$30.2 million, or 17.8%, to \$200.7 million for the year ended December 31, 2008 from \$170.5 million for the year ended December 31, 2007. Inclusive of the currency effect which accounted for a \$3.5 million decrease, net revenue increased \$26.8 million to \$197.3 million for the year ended December 31, 2008, from \$170.5 million for the year ended December 31, 2007.

United States: U.S. net revenue decreased \$15.0 million, or 14.8%, to \$86.6 million for the year ended December 31, 2008 from \$101.6 million for the year ended December 31, 2007. The decrease is primarily attributable to a decrease of \$13.4 million in retail voice services (for residential and small businesses), a decrease of \$1.5 million in Internet services and a decrease of \$0.1 million in VoIP services.

Europe: Europe net revenue, exclusive of the currency effect, decreased \$4.2 million, or 6.2%, to \$64.7 million for the year ended December 31, 2008 from \$68.9 million for the year ended December 31, 2007. The decrease is primarily attributable to a decrease of \$3.9 million in retail voice, a decrease of \$1.5 million in wireless services and a decrease of \$0.6 million in other services, offset in part by an increase of \$1.8 million in VoIP services. Inclusive of the currency effect, which had no impact, net revenue decreased \$4.3 million to \$64.6 million for the year ended December 31, 2008 from \$68.9 million for the year ended December 31, 2007.

Brazil: Brazil net revenue, exclusive of the currency effect, increased \$1.7 million, or 21.9%, to \$9.7 million for the year ended December 31, 2008 from \$7.9 million for the year ended December 31, 2007. The revenue increase is due primarily to an increase in VoIP services. Inclusive of the currency effect which accounted for a \$0.5 million increase, net revenue increased \$2.2 million to \$10.1 million for the year ended December 31, 2008 from \$7.9 million for the year ended December 31, 2007.

Cost of revenue: Cost of revenue, exclusive of the currency effect, increased \$19.1 million to \$570.4 million, or 64.0% of net revenue, for the year ended December 31, 2008 from \$551.3 million, or 61.5% of net revenue, for the year ended December 31, 2007. Inclusive of the currency effect, which accounted for a \$0.6 million decrease, cost of revenue decreased \$18.6 million to \$569.9 million for the year ended December 31, 2008 from \$551.3 million for the year ended December 31, 2007.

Canada: Canada cost of revenue, exclusive of the currency effect, decreased \$2.7 million to \$113.6 million, or 44.0% of net revenue, for the year ended December 31, 2008 from \$116.3 million, or 44.3% of net revenue, for the year ended December 31, 2007. The decrease is primarily attributable to a decrease in net revenue of \$4.3 million. Inclusive of the currency effect, which accounted for a \$0.8 million increase, cost of revenue decreased \$1.9 million to \$114.4 million for the year ended December 31, 2007.

Australia: Australia cost of revenue, exclusive of the currency effect, decreased \$7.8 million to \$167.4 million, or 61.5% of net revenue, for the year ended December 31, 2008 from \$175.2 million, or 61.5% of net revenue, for the year ended December 31, 2007. The decrease is primarily attributable to a \$12.8 million decrease in net revenue. Inclusive of the currency effect, which accounted for a \$2.0 million increase, cost of revenue decreased \$5.8 million to \$169.4 million for the year ended December 31, 2008 from \$175.2 million for the year ended December 31, 2007.

Wholesale: Wholesale cost of revenue, exclusive of the currency effect, increased \$29.7 million to \$193.6 million, or 96.4% of net revenue, for the year ended December 31, 2008 from \$163.9 million, or 96.3% of net revenue, for the year ended December 31, 2007. The increase is primarily attributable to an increase in net revenue of \$30.5 million. Inclusive of the currency effect, which accounted for a \$3.4 million decrease, cost of revenues increased \$26.3 million to \$190.2 million for the year ended December 31, 2007.

United States: United States cost of revenue decreased \$1.2 million to \$45.4 million, or 52.5% of net revenue, for the year ended December 31, 2008 from \$46.6 million, or 45.8% of net revenue, for the year ended December 31, 2007. The decrease was primarily due to a \$1.9 million decrease in retail voice services and a \$0.5 million decrease in Internet services partially offset by a \$1.3 million increase in retail VoIP services.

Europe: Europe cost of revenue, exclusive of the currency effect, decreased by \$1.0 million to \$44.4 million, or 68.6% of net revenue, for the year ended December 31, 2008 from \$45.4 million, or 65.8% of net revenue, for

the year ended December 31, 2007. The decrease is primarily due to a decrease in net revenue of \$4.3 million offset, in part, by slightly lower costs, as a percentage of net revenue. Inclusive of the currency effect, which accounted for a \$0.2 million decrease, cost of revenue decreased \$1.3 million to \$44.1 million for the year ended December 31, 2008 from \$45.4 million for the year ended December 31, 2007.

Brazil: Brazil cost of revenue, exclusive of the currency effect, increased \$2.1 million to \$6.1 million, or 63.2% of net revenue, for the year ended December 31, 2008 from \$4.0 million, or 50.7% of net revenue, for the year ended December 31, 2007. The increase is primarily attributable to an increase in net revenue of \$1.7 million and higher costs as a percentage of net revenue. Inclusive of the currency effect, which accounted for a \$0.3 million increase, cost of revenue increased \$2.4 million to \$6.4 million for the year ended December 31, 2008 from \$4.0 million for the year ended December 31, 2007.

Selling, general and administrative expenses: Selling, general and administrative expenses, exclusive of the currency effect, decreased \$23.6 million to \$257.4 million, or 28.9% of net revenue, for the year ended December 31, 2008 from \$281.0 million, or 31.4% of net revenue, for the year ended December 31, 2007. Inclusive of the currency effect, which accounted for a \$3.1 million increase, selling, general and administrative expenses decreased \$20.6 million to \$260.4 million for the year ended December 31, 2008 from \$281.0 million for the year ended December 31, 2007.

Canada: Canada selling, general and administrative expense, exclusive of the currency effect, decreased \$3.5 million to \$96.7 million, or 37.5% of net revenue, for the year ended December 31, 2008 from \$100.2 million, or 38.2% of net revenue, for the year ended December 31, 2007. The decrease is attributable to a decrease of \$6.2 in advertising costs and \$0.9 million in salaries and benefits offset, in part, by increases of \$2.5 million in sales and marketing expenses and \$1.2 million in general and administrative and all other expenses. Inclusive of the currency effect, which accounted for a \$1.2 million increase, selling, general and administrative expenses decreased \$2.3 million to \$97.9 million for the year ended December 31, 2008 from \$100.2 million for the year ended December 31, 2007.

Australia: Australia selling, general and administrative expense, exclusive of the currency effect, decreased \$1.6 million to \$80.3 million, or 29.5% of net revenue, for the year ended December 31, 2008 from \$81.9 million, or 28.7% of net revenue, for the year ended December 31, 2007. The decrease is attributable to a decrease of \$1.9 million in salaries and benefits, a decrease a decrease of \$1.2 million in professional fees and a decrease of \$0.2 million in general and administrative and all other expenses offset in part by increases of \$1.2 million in advertising expenses and \$0.5 million in occupancy costs. Inclusive of the currency effect, which accounted for a \$1.9 million increase, selling, general and administrative expense increased \$0.3 million to \$82.2 million for the year ended December 31, 2008 from \$81.9 million for the year ended December 31, 2007.

Wholesale: Wholesale selling, general and administrative expense, exclusive of the currency effect, decreased \$0.4 million to \$8.5 million, or 4.2% of net revenue, for the year ended December 31, 2008 from \$8.9 million, or 5.2% of net revenue, for the year ended December 31, 2007. Inclusive of the currency effect, which accounted for a \$0.1 million decrease, selling, general and administrative expense decreased \$0.5 million to \$8.4 million for the year ended December 31, 2008 from \$8.9 million for the year ended December 31, 2007.

United States: United States selling, general and administrative expense decreased \$5.3 million to \$35.2 million, or 40.6% of net revenue, for the year ended December 31, 2008 from \$40.5 million, or 39.8% of net revenue, for the year ended December 31, 2007. The decrease is attributable to a decrease of \$2.5 million in salaries and benefits expense due to staff reduction efforts, a decrease of \$1.7 million professional fees as the first quarter 2007 included significant expenses for litigation defense and the FIN No. 48 implementation, a decrease of \$0.9 million in sales and marketing expenses and a decrease of \$1.2 million in general and administrative expense offset, in part, by an increase of \$1.0 million in advertising.

Europe: Europe selling, general and administrative expense, exclusive of the currency effect, decreased \$10.2 million to \$22.0 million, or 34.0% of net revenue, for the year ended December 31, 2008 from \$32.2 million, or 46.7% of net revenue, for the year ended December 31, 2007. The decrease is attributable to decreases of \$4.1 in sales and marketing, \$2.6 in occupancy costs, \$1.1 million in salaries and benefits, and \$2.4 million of general and administrative and all other expenses. Inclusive of the currency effect, which accounted for a \$0.2 million decrease, selling, general and administrative expense decreased \$10.4 million to \$21.8 million for the year ended December 31, 2008 from \$32.2 million for the year ended December 31, 2007.

Brazil: Brazil selling, general and administrative expense, exclusive of the currency effect, increased \$0.8 million to \$3.8 million, or 38.9% of net revenue, for the year ended December 31, 2008 from \$3.0 million, or 37.7% of net revenue, for the year ended December 31, 2007. Inclusive of the currency effect, which accounted for a \$0.3 million decrease, selling, general and administrative expense increased \$1.0 million to \$4.0 million for the year ended December 31, 2008 from \$3.0 million for the year ended December 31, 2007.

Corporate: Corporate selling, general and administrative expense decreased \$3.5 million to \$11.1 million for the year ended December 31, 2008 from \$14.6 million for the year ended December 31, 2007. The decrease is primarily due to a decrease in salaries and benefits.

Depreciation and amortization expense: Depreciation and amortization expense increased \$2.3 million to \$32.8 million for the year ended December 31, 2008 from \$30.5 million for the year ended December 31, 2007. The increase consists of an increase in depreciation expense of \$2.9 million for assets placed in service as we enhanced our network infrastructure and expanded our data centers.

(Gain) loss on sale or disposal of assets: (Gain) loss on sale or disposal of assets was a \$6.0 million gain for the year ended December 31, 2008 compared to a \$1.5 million loss for the year ended December 31, 2007. In the year ended December 31, 2008, we recognized a gain of \$0.8 million associated with the sale of certain surplus fiber assets in the U.S., a gain of \$1.7 million associated with a sale of a minority equity investment in a Japanese entity, and a gain of \$4.6 million associated with the sale of primarily WiMAX spectrum assets, offset by a loss of \$0.9 million associated with disposal of equipment owned by a minority equity investment in Canada.

Interest expense and accretion (amortization) on debt discount/premium: Interest expense and accretion on debt premium (discount), net decreased \$8.5 million to \$53.3 million for the year ended December 31, 2008 from \$61.8 million for the year ended December 31, 2007. There was an increase of \$3.4 million mainly resulting from issuance and exchange of our Original Second Lien Notes, offset by a \$10.9 million decrease mainly resulting from reductions in principal outstanding balances of Group's 12 3/4% senior notes due 2009 (the "12 3/4% Senior Notes"), Group's step up convertible subordinated debentures due 2009 (the "Debentures"), the U.S. issuer's 8% senior notes due 2014 (the "8% Senior Notes"), the 5% Exchangeable Senior Notes, and Group's 3 3/4% convertible senior notes due 2010 (the "3 3/4% Convertible Senior Notes") and from a decreased interest rate on our variable rate Term Loan Facility.

Gain (loss) on early extinguishment or restructuring of debt: Gain (loss) on early extinguishment or restructuring of debt was a \$36.9 million gain for the year ended December 31, 2008 compared to a \$7.7 million loss for the year ended December 31, 2007. In the second quarter of 2008, we exchanged \$49.0 million principal amount of our 8% Senior Notes, \$33.0 million principal amount of our 5% Exchangeable Senior Notes, \$43.1 million principal amount of our 3³/4% Convertible Senior Notes, \$5.3 million principal amount of our 12³/4% Senior Notes for \$67.1 million principal amount of our newly issued Original Second Lien Notes and \$4.7 million of cash, resulting in a gain on restructuring of debt of \$32.3 million including the expensing of related financing costs. We made open market purchases of \$0.8 million principal amount of our 12³/4% Senior Notes, \$13.8 million principal amount of our Debentures, and \$2.1 million principal amount of our Original Second Lien Notes, resulting in a \$4.2 million gain on early extinguishment of debt including the write-off of related deferred financing costs.

In 2007, we converted \$5.0 million principal amount of our Debentures to 6.0 million shares of our common stock, resulting in an induced debt conversion expense of \$2.3 million, which includes deferred financing cost and discount write-offs. In first quarter of 2007, we issued in a private transaction \$57.2 million principal amount of the Original Second Lien Notes in exchange for \$40.7 million principal amount of the outstanding 12 ³/₄% Senior Notes and \$23.6 million in cash. This exchange was deemed a debt modification, resulting in a \$5.1 million loss on restructuring of debt for financing costs incurred. The remaining \$0.9 million of expense in the three months ended June 30, 2007, resulted from costs related to the early retirement of a Canadian credit facility. The losses were offset by our open market purchases of \$10.5 million principal amount of our 12 ³/₄% Senior Notes resulting in a \$0.3 million gain on early extinguishment of debt including the write-off of related deferred financing costs and a \$0.5 million gain on forgiveness of equipment financing in Brazil.

Interest and other income: Interest and other income of \$0.1 million for the year ended December 31, 2008 was mainly interest income, offset by minority interest expense recognized for the gain from the sale of the Canadian WiMAX spectrum assets.

Foreign currency transaction gain (loss): Foreign currency transaction gain (loss) was \$47.6 million loss for the year ended December 31, 2008 as compared to \$32.7 million gain for the year ended December 31, 2007. The gain and loss are attributable to the impact of foreign currency exchange rate changes on intercompany debt balances and on receivables and payables denominated in a currency other than the subsidiaries' functional currency.

Income tax benefit (expense): Income tax benefit (expense) was a benefit of \$0.4 million for year ended December 31, 2008 as compared to a benefit of \$9.2 million for the year ended December 31, 2007. In 2008, we closed certain federal and provincial income tax examinations in Canada for the years 2000 through 2005 (federal) and 2001 through 2005 (provincial) with all assessments being received. We also concluded an examination in the Netherlands for the years 2002, 2003, 2004, 2005 and 2006. Based on the final settlement, we reversed our remaining liability in the fourth quarter of 2008.

Liquidity and Capital Resources

Important Long-Term Capital Structure Developments:

In the fourth quarter 2009, we completed a private offering of \$130 million in 13% Senior Secured Notes due 2016. Net proceeds, after deducting underwriting and transactional expenses, of \$123.5 million were used principally to fully repay and terminate our \$94.8 million senior secured term loan facility and \$27.0 million Canadian credit facility, each of which were due in 2011. In connection with the July 2009 Reorganization and our December 2009 issuance of 13% Notes, we significantly adjusted our capital structure, our debt agreements' covenants and terms, our debt service obligations and our long-term debt maturities. We believe these efforts have better aligned our capital structure with our business's cash flow and liquidity expectations.

Changes in Cash Flows

Our principal liquidity requirements arise from cash used in operating activities, purchases of network equipment including switches, related transmission equipment and capacity, development of back-office systems, expansion of data center facilities, interest and principal payments on outstanding debt and other obligations and income taxes. We have financed our growth and operations to date through public offerings and private placements of debt and equity securities, vendor financing, capital lease financing and other financing arrangements.

Net cash provided by operating activities was \$29.0 million for the year ended December 31, 2009 as compared to net cash provided by operating activities of \$8.8 million for the year ended December 31, 2008. For the year ended December 31, 2009, net income, net of non-cash operating activity, provided \$61.6 million of cash. In addition, cash was increased by a reduction in accounts receivable of \$12.5 million, a reduction in

prepaid expenses and other current assets of \$2.3 million, a reduction in other assets of \$4.3 million and an increase in accrued interest of \$0.4 million. For the year ended December 31, 2009, we used, \$20.5 million to reduce our accounts payable, \$9.0 million to reduce our accrued income taxes, \$2.6 million to reduce our accrued expenses, deferred revenue, other current liabilities and other liabilities, net, and \$7.8 million to reduce our accrued interconnection costs. Cash effect of reorganization items was \$12.2 million. Net cash provided by operating activities was \$8.8 million for the year ended December 31, 2008 as compared to net cash provided by operating activities of \$11.5 million for the year ended December 31, 2007. For the year ended December 31, 2008, net loss, net of non-cash operating activity, provided \$34.1 million of cash. In addition, cash was increased by a reduction in prepaid expenses and other current assets of \$10.2 million, a reduction in other assets of \$1.9 million, and an increase in accrued interconnection costs of \$2.2 million. In 2008, \$15.7 million was used by an increase to accounts receivable. We used \$5.9 million to reduce accounts payable, \$10.6 to reduce our accrued income tax, \$5.6 million to reduce deferred revenue, accrued expenses, and other liabilities and \$1.8 million to reduce accrued interest.

Net cash used in investing activities was \$15.1 million for the year ended December 31, 2009 compared to \$18.7 million for the year ended December 31, 2008. Net cash used in investing activities during the year ended December 31, 2009 included \$15.1 million of capital expenditures. Net cash used in investing activities was \$18.7 million for the year ended December 31, 2008 compared to \$39.5 million for the year ended December 31, 2007. Net cash used in investing activities during the year ended December 31, 2008 included \$25.4 million of capital expenditures and \$0.6 million for a customer list acquisition in Australia, offset by \$4.9 million from the sale of certain primarily rural Canadian WiMAX spectrum assets (the cash proceeds from which can be used for operations within the variable interest entity but requires unanimous shareholder consent for a dividend distribution), \$1.7 million net cash proceeds from the disposition of a minority equity investment in Japan and \$0.8 million from the disposition of surplus fiber assets. Net cash used in investing activities during the year ended December 31, 2007 included \$44.7 million of capital expenditures, \$0.7 million to increase restricted cash, and \$0.2 million to acquire an additional 39% of a subsidiary, offset by \$6.1 million net cash proceeds from the disposition of our Australian Planet Domain subsidiary and a German subsidiary.

Net cash used in financing activities was \$12.7 million for the year ended December 31, 2009 compared to \$28.1 million for the year ended December 31, 2008. During the year ended December 31, 2009, \$35.0 million was used to retire in full our Canadian Credit Facility, \$96.3 million was used to retire in full the Senior-Secured Term Loan Facility, \$4.9 million was used to reduce the principal amounts outstanding on capital leases, leased fiber capacity, financing facilities and other long-term obligations. Net cash proceeds received from the issuance of the Senior Secured Notes was \$123.5 million. Net cash used in financing activities was \$28.1 million for the year ended December 31, 2008 as compared to \$41.5 million provided by financing activities for the year ended December 31, 2007. During the year ended December 31, 2008, \$11.5 million was used to purchase and retire \$0.8 million principal amount of our 12 ³/4% Senior Notes, \$13.8 million principal amount of our Debentures and \$2.1 million principal amount of our Original Second Lien Notes. We also used \$16.5 million for payments on capital leases, leased fiber capacity, financing facilities and other long-term obligations. During the year ended December 31, 2007, net cash provided by financing activities consisted of \$102.7 million from the issuance of debt, net of issuance costs, which was comprised of \$75.2 million principal amount of Original Second Lien Notes for \$69.2 million in net cash, and \$35.0 million from a credit facility with a financial institution (less \$1.5 million principal amount of our 2000 convertible subordinated debentures, the retirement of \$10.5 million principal amount of our 12 ³/4% Senior Notes, the repayment in full of a \$29.9 million Canadian Credit Facility and a \$4.9 million capital lease financing, and \$12.4 million principal payments of capital leases, leased fiber capacity, financing facilities and other long-term obligations.

Short- and Long-Term Liquidity Considerations and Risks

As of December 31, 2009, we had \$42.5 million of unrestricted cash and cash equivalents. We believe that our existing cash and cash equivalents will be sufficient to fund our debt service requirements, other fixed obligations (such as capital leases, vendor financing and other long-term obligations), and other cash needs for our operations for at least the next twelve months.

As of December 31, 2009, we have \$24.8 million in future minimum purchase obligations, \$64.2 million in future operating lease payments and \$259.5 million of indebtedness. At December 31, 2009, approximately \$89.7 million of unrecognized tax benefits have been recorded as liabilities in accordance with ASC No. 740; however, we are uncertain as to if or when such amounts may be settled, so we have not included these amounts in the table below.

The obligations reflected in the table below reflect the contractual payments of principal and interest that existed as of December 31, 2009:

(in thousands) Year Ending December 31,	Vendor Financing <u>and Other</u>	13% Senior Secured <u>Notes</u>	14 ¹ /4% Senior Subordinated Secured Notes	Purchase Obligations	Operating Leases	Total
2010	\$ 4,815	\$ 16,665	\$ 17,595	\$ 17,719	\$17,028	\$ 73,822
2011	1,239	16,900	17,595	4,305	14,231	54,270
2012	329	16,900	17,595	2,761	12,546	50,131
2013	102	16,900	132,269	_	9,060	158,331
2014	28	16,900	_	_	3,727	20,655
Thereafter	20	163,847			7,578	171,445
Total Minimum Principal & Interest Payments	6,533	248,112	185,054	24,785	64,170	528,654
Less: Amount						
Representing Interest	(546)	(118,112)	(61,582)	_	_	(180,240)
Total Long-Term Obligations	\$ 5,987	\$ 130,000	\$ 123,472	\$ 24,785	\$64,170	\$ 348,414

The foregoing table assumes that the 14 1/4% Notes are refinanced before January 21, 2013 and holders of 13% Notes do not accept any Excess Cash Flow Offer to purchase 13% Notes. In this regard, the Company must extend an offer to repurchase to the holders of the 13% Notes an applicable amount, (equal to 50% of Excess Cash Flow), of the 13% Notes at par, in the event the Company and certain subsidiaries have excess cash flow (as defined in Note 7 — Long Term Obligations.) for any fiscal year commencing with the fiscal year ending December 31, 2010. See Item 1A. "Risks Associated with our Liquidity Needs and Debt Securities." for certain risks and uncertainties related thereto.

We have contractual obligations to utilize network facilities from certain carriers with terms greater than one year. We generally do not purchase or commit to purchase quantities in excess of normal usage or amounts that cannot be used within the contract term. We have minimum annual purchase obligations of \$17.7 million, \$4.3 million, and \$2.8 million remaining in 2010, 2011 and 2012, respectively.

Newly Adopted Accounting Principles

Effective July 1, 2009, the Company adopted ASC No. 105, "Generally Accepted Accounting Principles." ASC No. 105 establishes the FASB Accounting Standards Codification as the source of accounting principles and the framework for selecting the principles to be used in the preparation of financial statements. The adoption did not have a material impact on its results of operations, financial position and cash flows.

Effective April 1, 2009, the Company adopted ASC No. 855, "Subsequent Events." ASC No. 855 establishes principles and requirements for evaluating and reporting subsequent events and distinguishes which subsequent events should be recognized in the financial statements versus which subsequent events should be disclosed in the financial statements. The adoption did not have a material effect on the Company's results of operations, financial position or cash flows.

Effective January 1, 2009, the Company adopted ASC No. 810, "Consolidation." This statement changes the presentation of outstanding noncontrolling interests in one or more subsidiaries or the deconsolidation of those subsidiaries. The adoption did not have a material effect on the Company's results of operations, financial position or cash flows.

Effective January 1, 2009, the Company adopted ASC No. 470-20, "Debt with Conversion and Other Options." ASC No. 470-20 requires issuers of convertible debt securities to separate securities into a debt component and an equity component, resulting in the debt component being recorded at fair value without consideration given to the conversion feature. Issuance costs are also allocated between the debt and equity components. ASC No. 470-20 requires that convertible debt within its scope reflect a company's nonconvertible debt borrowing rate when interest expense is recognized. The adoption did not have a material effect on the Company's results of operations, financial position or cash flows.

Effective January 1, 2009, the Company adopted ASC No. 805, "Business Combinations." ASC No. 805 establishes principles and requirements for how the acquirer of a business recognizes and measures in its financial statements the identifiable assets acquired, the liabilities assumed, and any noncontrolling interest in the acquiree. The statement also provides guidance for recognizing and measuring the goodwill acquired in the business combination and determines what information to disclose to enable users of the financial statement to evaluate the nature and financial effects of the business combination. The Company applied this standard on July 1, 2009 to the Fresh Start Balance Sheet, (See Note 4 "Fresh Start Accounting" for further detail).

In February 2008, the FASB issued ASC No. 820, "Fair Value Measurements." The provisions of ASC No. 820 provide guidance for, among other things, the definition of fair value and the methods used to measure fair value, were adopted January 1, 2008 for financial instruments. The provisions adopted in 2008 did not have a material impact on the Company's financial statements. ASC No. 820 delayed the effective date for all nonrecurring fair value measurements of nonfinancial assets and liabilities (except for those that are recognized or disclosed at fair value in the financial statements on a recurring basis) until financial years beginning after November 15, 2008, and changed the scope of ASC No. 820. On January 1, 2009, the Company adopted the remaining provisions of ASC No. 820 for nonrecurring fair value measurements of nonfinancial assets and liabilities. The provisions adopted in the first quarter 2009 did not have an impact on the Company's financial statements as the Company did not have any fair value measurements of nonfinancial assets and liabilities for the year ended December 31, 2009.

The valuation techniques required by ASC No. 820 are based upon observable and unobservable inputs. Observable inputs reflect market data obtained from independent sources, while unobservable inputs reflect our market assumptions. These two types of inputs create the following fair value hierarchy:

- Level 1: Quoted prices for identical instruments in active markets.
- Level 2: Quoted prices for similar instruments in active markets; quoted prices for identical or similar instruments in markets that are not active; and model-derived valuations whose inputs are observable or whose significant value drivers are observable.
 - Level 3: Significant inputs to the valuation model are unobservable.

The Company had a cross-currency principal and interest rate agreement with Lehman Brothers Special Financing, Inc., who entered bankruptcy in October 2008 and ceased performing on the agreement. The Company estimated the value to be zero, requiring a write-off of \$1.2 million in 2008, and moved the instrument from Level 2 to Level 3 because the counter party's credit risk is not observable.

In March 2008, the FASB issued ASC No. 815, "Derivatives and Hedging." ASC No. 815 amends and expands the disclosure requirements with the intent to provide users of financial statements with an enhanced

understanding of the use of derivative instruments, how derivative instruments and related hedged items are accounted for, and how derivative instruments and related hedged items affect an entity's financial statements. ASC No. 815 is effective for financial statements issued for fiscal years interim periods beginning after November 15, 2008. In addition the Company accounted for the Contingent Value Rights Distribution Agreement in accordance with ASC No. 815, "Derivatives and Hedging" as well as related interpretations of this standard. The change in value of the Contingent Value Rights Distribution Agreement is reflected in our Statements of Operations as other income (expense), (see Note 2—Emergence from Voluntary Reorganization Under Chapter 11 of the Bankruptcy Code," for further detail). The adoption of ASC No. 815 on January 1, 2009 did not have a material impact on the Company's results of operations, financial position and cash flows.

New Accounting Pronouncements

In June 2009, the FASB issued ASC No. 810-10, "Consolidation," which amends the definition of the primary beneficiary of a variable interest entity and will require the Company to assess each reporting period if any of the Company's variable interests give it a controlling financial interest in the applicable variable interest entity. The provisions of ASC No. 810-10 will become effective for financial statements issued for fiscal years and interim periods begin after November 15, 2009. The Company is currently evaluating the impact the provisions of ASC No. 810-10 on its financial statements.

In May 2008, the FASB issued ASC No. 470-20, "Debt with Conversion and Other Options." ASC No. 470-20 requires issuers of convertible debt securities to separate securities into a debt component and an equity component, resulting in the debt component being recorded at fair value without consideration given to the conversion feature. Issuance costs are also allocated between the debt and equity components. ASC No. 470-20 requires that convertible debt within its scope reflect a company's nonconvertible debt borrowing rate when interest expense is recognized. ASC No. 470-20 is effective for financial statements issued for fiscal years and interim periods beginning after December 15, 2008. The Company is currently evaluating the impact of ASC No. 470-20 on its financial statements, but does not believe there will be a material impact.

In April 2008, the FASB issued ASC No. 852-10, "Reorganizations". ASC No. 852-10 nullifies certain requirements regarding changes in accounting principles that will be applicable to the financial statements of an entity emerging from bankruptcy. Any changes in accounting principles required within the twelve months following the implementation of fresh start accounting by such an entity are no longer required to be adopted at the time fresh start accounting is implemented. Entities emerging from bankruptcy that implement fresh start accounting should only follow accounting standards in effect at the date fresh start accounting is implemented, including any standards eligible for early adoption.

Special Note Regarding Forward Looking Statements

Certain statements in this Report on Form 10-K and elsewhere concerning strategic objectives and initiatives, prospects, future liquidity, cost savings initiatives and related matters constitute "forward-looking statements" within the meaning of Section 27A of the Securities Act of 1933 and Section 21E of the Securities Exchange Act of 1934. Such statements are based on current expectations, and are not strictly historical statements. In some cases, you can identify forward-looking statements by terminology such as "if," "may," "should," "believe," "anticipate," "future," "forward," "potential," "estimate," "reinstate," "opportunity," "goal," "objective," "exchange," "growth," "outcome," "could," "expect," "intend," "plan," "strategy," "provide," "commitment," "result," "seek," "pursue," "ongoing," "include" or in the negative of such terms or comparable terminology. These forward-looking statements inherently involve certain risks and uncertainties, although they are based on our current plans or assessments which are believed to be reasonable as of the date of this filing. Forward-looking statements include, without limitation, statements set forth in this document and elsewhere regarding, among other things:

- our financial condition, financing requirements, prospects and cash flow;
- expectations of future growth, creation of shareholder value, revenue, foreign revenue contributions and net income, as well as income from operations, margins, earnings per share, cash flow and cash sufficiency levels, working capital, network development, customer migration and related costs, spending on and success with growth products, including broadband Internet, VOIP, wireless, local, data and hosting services, traffic development, capital expenditures, selling, general and administrative expenses, income tax and withholding tax expense, fixed asset and goodwill impairment charges, service introductions, cash requirements and potential asset sales;
- increased competitive pressures, declining usage patterns, and our growth products, bundled service offerings, the pace and cost of customer migration onto our networks, the effectiveness and profitability of the growth products;
- financing, refinancing, debt extension, de-leveraging, restructuring, exchange or tender plans or initiatives, and potential dilution of existing equity holders from such initiatives;
- liquidity and debt service forecast;
- assumptions regarding currency exchange rates;
- timing, extent and effectiveness of cost reduction initiatives and management's ability to moderate or control discretionary spending;
- management's plans, goals, expectations, guidance, objectives, strategies, and timing for future operations, acquisitions, asset dispositions, product plans, performance and results;
- · management's assessment of market factors and competitive developments, including pricing actions and regulatory rulings; and
- · ability to generate net cash proceeds from the disposition of selective assets without material impairment to profitability.

Factors and risks that could cause actual results or circumstances to differ materially from those set forth or contemplated in forward looking statements include those set forth in "Risk Factors" as well as, without limitation:

- the occurrence of a default or event of default under our indentures or other financing agreements;
- an inability to fully fund and repurchase holder acceptances of offers to repurchase 13% Notes that we are obligated to make annually, subject to certain limitations, in connection with Excess Cash Flow Offers;

- an inability to fully fund and repurchase holder acceptances of offers to repurchase debt securities that we may be obligated to make following certain change in control developments affecting Group and certain of its subsidiaries;
- customer, vendor, carrier and third-party responses to our completed Reorganization;
- changes in business conditions causing changes in the business direction and strategy by management;
- heightened competitive pricing and bundling pressures in the markets in which we operate;
- the ability to service substantial indebtedness;
- accelerated decrease in minutes of use on wireline phones;
- fluctuations in the exchange rates of currencies, particularly of the USD relative to foreign currencies of the countries where we conduct our foreign operations;
- adverse interest rate developments affecting our variable interest rate debt;
- difficulty in maintaining or increasing customer revenues and margins through our product initiatives and bundled service offerings, and difficulties in migrating and provisioning broadband and local customers to digital subscriber line (DSL) networks;
- · inadequate financial resources to promote and to market product initiatives, whether due to acceptances of Excess Cash Flow Offers or otherwise;
- fluctuations in prevailing trade credit terms or revenues due to the adverse impact of, among other things, further telecommunications carrier bankruptcies or adverse bankruptcy related developments affecting our large carrier customers;
- the possible inability to raise additional capital when needed, on attractive terms, or at all;
- possible claims under our existing debt instruments which could impose constraints and limit our flexibility;
- the inability to service substantial indebtedness and to reduce, refinance, extend, exchange, tender for or restructure debt significantly, or in amounts sufficient to conduct regular ongoing operations;
- · the impact of the delisting of our common stock from the Nasdaq Capital Market which may impair our ability to raise capital;
- further changes in the telecommunications or Internet industry, including rapid technological changes, regulatory and pricing changes in our principal markets and the nature and degree of competitive pressure that we may face;
- adverse tax or regulatory rulings from applicable authorities;
- · enhanced broadband, DSL, Internet, wireless, VOIP, date and hosting and local and long distance voice telecommunications competition;
- changes in financial, capital market and economic conditions;
- changes in service offerings or business strategies, including the need to modify business models if performance is below expectations;
- difficulty in retaining existing long distance wireline and dial-up ISP customers;
- · difficulty in migrating or retaining customers associated with acquisitions of customer bases, or integrating other assets;
- difficulty in selling new services in the marketplace;
- difficulty in providing broadband, DSL, local, VOIP, data and hosting or wireless services;

- changes in the regulatory schemes or requirements and regulatory enforcement in the markets in which we operate;
- restrictions on our ability to execute certain strategies or complete certain transactions as a result of our inexperience with new products, or limitations imposed by available cash resources, our capital structure or debt covenants;
- risks associated with our limited DSL, Internet, VOIP, data and hosting and wireless experience and expertise, including effectively utilizing new
 marketing channels such as interactive marketing employing the Internet;
- entry into developing markets;
- aggregate margin contribution from the new products is not sufficient in amount or timing to offset the margin decline in our legacy long distance voice and dial-up ISP businesses;
- · the possible inability to hire and/or retain qualified executive management, sales, technical and other personnel;
- risks and costs associated with our effort to locate certain activities and functions off-shore;
- risks associated with international operations;
- dependence on effective information and billing systems;
- · possible claims for patent infringement on products or processes employed in providing our services;
- dependence on third parties for access to their networks to enable us to expand and manage our global network and operations and to offer broadband, DSL, local, VOIP and wireless services, including dependence upon the cooperation of incumbent carriers relating to the migration of customers:
- dependence on the performance of our global standard asynchronous transfer mode and Internet-based protocol (ATM+IP) communications network; risks associated with maintaining and upgrading networks; and
- adverse regulatory rulings or actions affecting our operations, including the imposition of taxes and fees, the imposition of obligations upon VOIP providers to provide enhanced 911 (E911) services and restricting access to broadband networks owned and operated by others, including the development of a national broadband network in Australia.

As such, actual results or circumstances may vary materially from such forward looking statements or expectations. Readers are also cautioned not to place undue reliance on these forward looking statements which speak only as of the date these statements were made. We are not obligated to update or revise any forward looking statements, whether as a result of new information, future events or otherwise.

ITEM 7A. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK

Our primary market risk exposures relate to changes in foreign currency exchange rates, valuations of derivatives and to changes in interest rates.

Foreign currency can have a major impact on our financial results. Currently about 82% of our net revenue is derived from sales and operations outside the U.S. The reporting currency for our consolidated financial statements is the USD. The local currency of each country is the functional currency for each of our respective entities operating in that country. In the future, we expect to continue to derive the majority of our net revenue and incur a significant portion of our operating costs from outside the U.S., and therefore changes in exchange rates have had and may continue to have a significant, and potentially adverse, effect on our results of operations. Our primary risk of loss regarding foreign currency exchange rate risk is caused primarily by fluctuations in the following exchange rates: USD/CAD, USD/AUD, USD/GBP, and USD/EUR. Due to the large percentage of our revenue derived outside of the U.S., changes in the USD relative to one or more of the foregoing currencies could have an adverse impact on our future results of operations. We have agreements with certain subsidiaries for repayment of a portion of the investments and advances made to these subsidiaries. As we anticipate repayment in the foreseeable future, we recognize the unrealized gains and losses in foreign currency transaction gain (loss) on the consolidated statements of operations. We historically have not engaged in hedging transactions.

We are exposed to financial statement gains and losses as a result of translating the operating results and financial position of our international subsidiaries. We translate the local currency statements of operations of our foreign subsidiaries into USD using the average exchange rate during the reporting period. Changes in foreign exchange rates affect the reported profits and losses and cash flows and may distort comparisons from year to year. By way of example, when the USD strengthens compared to the EUR, there could be a negative or positive effect on the reported results for Europe, depending upon whether Europe is operating profitably or at a loss. It takes more profits in EUR to generate the same amount of loss in USD and a greater loss in EUR to generate the same amount of loss in USD. The opposite is also true. For instance, when the USD weakens there is a positive effect on reported profits and a negative effect on reported losses for Europe.

In the year ended December 31, 2009, as compared to the year ended December 31, 2008, the USD was stronger on average as compared to the AUD, CAD, and GBP; and weaker on average to the EUR. As a result, our revenue of the subsidiaries whose local currency is AUD, CAD, GBP and EUR increased (decreased) (5)%, (7)%, 24% and 7% in local currency compared to the year ended December 31, 2008, but increased (decreased) (12)%, (14)%, 6% and 2% in USD, respectively.

In the year ended December 31, 2008, as compared to the year ended December 31, 2007, the USD was weaker on average as compared to the AUD, CAD, and EUR, and stronger on average as compared to the GBP. As a result, our revenue of the subsidiaries whose local currency is AUD, CAD, GBP and EUR increased (decreased) (5)%, (2)%, 8% and (0)% in local currency compared to the year ended December 31, 2007, but increased (decreased) (3)%, (1)%, (2)% and 7% in USD, respectively.

Interest rates—The Senior Secured Notes and Senior Subordinated Secured Notes are at a fixed interest rate of 13.00% and 14.25%, respectively. We are exposed to interest rate risk as debt refinancing may be required. Our primary exposure to market risk stems from fluctuations in interest rates.

The interest rate sensitivity table below summarizes our market risks associated with fluctuations in interest rates for the year ended December 31, 2009 in USD, which is our reporting currency. The table presents principal cash flows and related weighted average interest rates by year of expected maturity for our Senior Secured Notes, Senior Subordinated Secured Notes, leased fiber capacity, and other long-term obligations in effect at December 31, 2009.

		Year of Maturity						
	2010	2011	2012	2013	2014	Thereafter	Total	Fair Value
				(in thousands, e	xcept percent	ages)		
Fixed Rate	\$4,472	\$1,142	\$307	\$123,566	\$ 28	\$130,005	\$259,519	\$250,729
Average Interest Rate	12.4%	10.8%	8.8%	14.2%	4.0%	13.0%	13.6%	

ITEM 8. FINANCIAL STATEMENTS AND SUPPLEMENTARY DATA

	Page
Management's Report on Internal Control Over Financial Reporting	F-2
Report of Independent Registered Public Accounting Firm	F-3
Consolidated Financial Statements	
Consolidated Statements of Operations for the years ended December 31, 2009, 2008 and 2007	F-4
Consolidated Balance Sheets as of December 31, 2009 and 2008	F-5
Consolidated Statements of Stockholders' Equity (Deficit) for the years ended December 31, 2009, 2008 and 2007	F-6
Consolidated Statements of Cash Flows for the years ended December 31, 2009, 2008 and 2007	F-7
Consolidated Statements of Comprehensive Income (Loss) for the years ended December 31, 2009, 2008 and 2007	F-8
Notes to the Consolidated Financial Statements	F-9

ITEM 9. CHANGES IN AND DISAGREEMENTS WITH ACCOUNTANTS ON ACCOUNTING AND FINANCIAL DISCLOSURE

None.

ITEM 9A. CONTROLS AND PROCEDURES

Evaluation of Disclosure Controls and Procedures.

Our management evaluated, with the participation of our Chief Executive Officer and Chief Financial Officer, the effectiveness of our disclosure controls and procedures as of the end of the period covered by this report. Based on this evaluation, as a result of the material weakness described below, our principal executive officer and our principal financial officer have concluded that, as of the end of the period covered by this report, our disclosure controls and procedures were not effective. Disclosure controls and procedures mean our controls and other procedures that are designed to ensure that information required to be disclosed by us in our reports that we file or submit under the Securities Exchange Act of 1934 is recorded, processed, summarized and reported within the time periods specified in the SEC's rules and forms. Disclosure controls and procedures include, without limitation, controls and procedures designed to ensure that information required to be disclosed by us in our reports that we file or submit under the Securities Exchange Act of 1934 is accumulated and communicated to our management, including our principal executive officer and principal financial officer, as appropriate to allow timely decisions regarding required disclosure.

Evaluation of Internal Control Over Financial Reporting.

As part of our compliance efforts relative to Section 404 of Sarbanes-Oxley Act of 2002, management assessed the effectiveness of internal control over financial reporting as of December 31, 2009. In making this assessment, management used the criteria set forth by the Committee of Sponsoring Organizations of the Treadway Commission ("COSO") in Internal Control-Integrated Framework. Based on this assessment, management identified a material weakness in our internal controls over accounting for foreign currency transaction gain (loss) on inter-company balances. In March 2010, the Company determined that an error existed related to accounting for foreign currency transaction gain (loss) on certain inter-company balances. Specifically, this error related to activity in the third quarter 2009 resulting in the Company amending its Form 10-Q for the quarter ended September 30, 2009. Management concluded that, as of December 31, 2009, our internal controls over financial reporting were not effective based on the criteria set forth by COSO.

Management's report on internal control over financial reporting as of December 31, 2009 appears on page F-2 and is incorporated herein by reference. This annual report does not include an attestation report of our registered public accounting firm regarding internal control over financial reporting. Management's report was not subject to attestation by our registered public accounting firm pursuant to temporary rules of the Securities and Exchange Commission that permit us to provide only management's report in this annual report.

Changes in Internal Control.

Our accounting for income taxes has significant complexity due to the nature of our business, our varied types of refinancing and debt transactions, the significant number of foreign subsidiary legal entities and various tax planning strategies. To address this complexity, on May 27, 2009, we re-hired our former Corporate Tax Director to provide oversight of the domestic, foreign and consolidated income tax responsibilities and on July 6, 2009 a new Manager of Taxation in our Canadian operating unit. Our Corporate Tax Director originally held this position from October 1, 2007—September 5, 2008. Additionally, we established procedures for foreign finance personnel to communicate regularly any tax concerns with the Corporate Tax Director, we purchased and implemented tax provision software and we developed quarterly tax documentation formats. We believe that the personnel hired have the appropriate knowledge, experience and skills related to accounting for income taxes and we have implemented

and maintained the proper controls over accounting for income taxes. As a result of these changes in control, our management concluded as of December 31, 2009 that the controls surrounding accounting for income taxes are effective. See Item 1A, "Risk Factors; Risks Associated with our Financial Statements."

As a result of the Company's determination that the controls in place over the process of translating certain inter-company balances did not operate effectively as of December 31, 2009, the Company has designed new procedures and is implementing new controls to address the control failure that occurred, including: (a) performing additional recalculations and analysis of the foreign currency transaction gain (loss) recorded on inter-company balances; (b) implementing an improved process for assessing the reasonableness of foreign currency transaction gain (loss) recorded on these inter-company balances; and (c) confirming inter-company settlements related to these balances at a transactional level. Management believes that these corrective actions, taken as a whole, will successfully mitigate the material weakness described above, and the Company will continue to perform the enhanced procedures as part of the normal accounting process.

Other than the changes in accounting for income taxes and foreign currency transactions noted above, there have been no changes in our internal control over financial reporting or in other factors that could significantly affect internal controls over financial reporting, that occurred during the year ended December 31, 2009, that have materially affected, or is reasonably likely to affect materially, our internal control over financial reporting.

ITEM 9B. OTHER INFORMATION

None.

PART III

Certain of the information required by Part III will be provided in our definitive proxy statement for our 2010 annual meeting of stockholders, which definitive proxy statement will be filed pursuant to Regulation 14A not later than April 30, 2010 ("2010 Proxy Statement"), and is incorporated herein by this reference.

ITEM 10. DIRECTORS, EXECUTIVE OFFICERS AND CORPORATE GOVERNANCE

Information relating to our directors is set forth under the caption entitled "Election of Directors" in our 2010 Proxy Statement and is incorporated herein by reference. Information relating to our executive officers is set forth in our 2010 Proxy Statement under the caption "Executive Officers, Directors and Key Employees" and is incorporated herein by reference.

Section 16(a) Beneficial Ownership Reporting Compliance

Our executive officer, directors and 10% stockholders are required under the Securities Exchange Act of 1934 to file with the Securities and Exchange Commission reports of ownership and changes in ownership in their holdings in our stock. Based solely on an examination of these reports, all such reports have been timely filed.

Code of Ethics

We have adopted a Code of Ethics applicable to all directors, officers and employees, including the chief executive officer, senior financial officers and other persons performing similar functions. The Code of Ethics is a statement of business practices and principles of behavior that support our commitment to conducting business while maintaining the highest standards of business conduct and ethics. Our Code of Ethics covers topics including, but not limited to, compliance resources, conflicts of interest, compliance with laws, rules and regulations, internal reporting of violations and accountability for adherence to the Code. A copy of the Code of Ethics is available on our website at www.primustel.com. Any amendment of the Code of Ethics or any waiver of its provisions for a director, executive officer or senior financial officer must be approved by the Board of Directors. We will publicly disclose any such waivers or amendments pursuant to applicable SEC and the Over-the-Counter Bulletin Board regulations.

ITEM 11. EXECUTIVE COMPENSATION

The information regarding compensation of our officers and directors is set forth under the caption entitled "Executive Compensation" in our 2010 Proxy Statement and is incorporated herein by reference.

ITEM 12. SECURITY OWNERSHIP OF CERTAIN BENEFICIAL OWNERS AND MANAGEMENT AND RELATED STOCKHOLDER MATTERS

Information regarding ownership of certain of our securities is set forth under the captions entitled "Security Ownership of Certain Beneficial Owners" and "Security Ownership of Management" in our 2010 Proxy Statement and is incorporated herein by reference.

ITEM 13. CERTAIN RELATIONSHIPS AND RELATED TRANSACTIONS, AND DIRECTOR INDEPENDENCE

Information regarding certain relationships and related transactions with us is set forth under the caption entitled "Certain Relationships and Related Transactions" in our 2010 Proxy Statement and is incorporated herein by reference.

ITEM 14. PRINCIPAL ACCOUNTANT FEES AND SERVICES

The following table summarizes the aggregate fees paid by us to Deloitte & Touche LLP, the member firms of Deloitte Touche Tohmatsu, and their respective affiliates (in thousands):

	2009	2008
Audit fees (a)	\$ 4,432.8	\$ 2,902.5
Audit-related fees	63.9	_
Tax fees (b)	595.7	629.4
All other fees (c)	40.3	9.9
Total	\$ 5,132.7	\$ 3,541.8

- (a) Fees for audit services include audit of annual financial statements, as required by the Sarbanes-Oxley Act of 2002, Section 404, reviews of quarterly financial statements, statutory and regulatory audits, comfort letters, consents and other matters related to SEC filings.
- (b) Fees for tax services include corporate tax compliance and tax planning and advice.
- (c) Fees for other services include fees billed for permitted non-audit services.

In considering the nature of the services provided by Deloitte & Touche LLP, the Audit Committee determined that such services are compatible with the provision of independent audit services. The Audit Committee discussed these services with the independent registered public accounting firm and management to determine that they are permitted under the rules and regulations concerning auditor independence promulgated by the SEC to implement the Sarbanes-Oxley Act of 2002, as well as the American Institute of Certified Public Accountants.

Pre-Approval Policy

The services performed by the independent registered public accounting firm in 2009 were pre-approved in accordance with the pre-approval policy and procedures adopted by the Audit Committee at its February 4, 2004 meeting, as amended at its February 9, 2005 meeting. This policy describes the permitted audit, audit-related, tax, and other services (collectively, the "Disclosure Categories") that the independent auditor may perform.

On a quarterly basis, the Audit Committee reviews a description of services (the "Service List") expected to be performed by the independent registered public accounting firm in each of the Disclosure Categories, the status of services and fees incurred year-to-date against the original Service List and pre-approval limits and the forecast of remaining services and fees for the fiscal year.

Services provided by the independent registered public accounting firm during the year and included in the Service List were pre-approved in accordance with the policies and procedures of the Audit Committee.

Any requests for audit, audit-related, tax, and other services contemplated on the Service List must be submitted to the Audit Committee for specific preapproval and cannot commence until such approval has been granted. Normally, pre-approval is provided at regularly scheduled meetings. However, the authority to grant specific pre-approval between meetings, as necessary, has been delegated to the Chairman of the Audit Committee. The Chairman must update the Audit Committee at the next regularly scheduled meeting of any services that were granted specific pre-approval.

PART IV

ITEM 15. EXHIBITS AND FINANCIAL STATEMENT SCHEDULE

a) Financial Statements and Schedules

The financial statements as set forth under Item 8 of this report on Form 10-K are incorporated herein.

Financial Statement Schedules:

(II) Valuation and Qualifying Accounts

All other financial statement schedules have been omitted since they are either not required, not applicable, or the information is otherwise included.

b) Exhibit listing

Exhibit Number

Number 2.11	<u>Description</u> Joint Plan of Reorganization of Primus Telecommunications Group, Incorporated and its Affiliate Debtors, as filed by the Debtors with the Bankruptcy Court on June 12, 2009.
3.12	Second Amended and Restated Certificate of Incorporation of Primus Telecommunications Group, Incorporated
3.22	Amended and Restated By-Laws of Primus Telecommunications Group, Incorporated
4.13	Class A Warrant Agreement, dated as of July 1, 2009, by and between Primus Telecommunications Group, Incorporated and StockTrans, Inc., as Warrant Agent
4.23	Class B Warrant Agreement, dated as of July 1, 2009, by and between Primus Telecommunications Group, Incorporated and StockTrans, Inc., as Warrant Agent
4.33	Contingent Value Rights Distribution Agreement of Primus Telecommunications Group, Incorporated
4.44	Indenture, dated as of December 22, 2009, by and among Primus Telecommunications Holding, Inc., Primus Telecommunications Canada Inc., the Guarantors party thereto, The Bank of New York Mellon, as Trustee, U.S. Bank National Association, as U.S. Collateral Trustee, and Computershare Trust Company of Canada, as Canadian Collateral Trustee, relating to Units, each Unit comprised of \$653.85 principal amount of 13% Senior Secured Notes due 2016 issued by Primus Telecommunications Holding, Inc. (the "13% Holding Notes") and \$346.15 principal amount of 13% Senior Secured Notes due 2016 issued by Primus Telecommunications Canada Inc. (the "13% Canadian Notes").
4.54	Form of Unit comprised of \$653.85 principal amount of 13% Senior Secured Notes due 2016 issued by Primus Telecommunications Holding, Inc. and \$346.15 principal amount of 13% Senior Secured Notes due 2016 issued by Primus Telecommunications Canada Inc. (included in Exhibit 4.1).
4.64	Form of 13% Senior Secured Note due 2016 issued by Primus Telecommunications Holding, Inc. (included in Exhibit 4.4).
4.74	Form of 13% Senior Secured Notes due 2016 issued by Primus Telecommunications Canada Inc. (included in Exhibit 4.4).
4.83	Specimen of Common Stock
4.93	Supplemental Indenture, dated as of July 1, 2009, by and among Primus Telecommunications IHC, Inc., the Guarantors thereto and U.S. Bank National Association, as Trustee

Exhibit Number 4.10 ⁵	<u>Description</u> Intercreditor Agreement, dated as of February 26, 2007 (as amended), by and among Primus Telecommunications IHC, Inc., Primus Telecommunications Group, Incorporated, Primus Telecommunications Holding, Inc., and the parties thereto
4.115	Collateral Agreement pursuant to the $14\frac{1}{4}$ % Notes Indenture, dated as of February 26, 2007 (as amended), by and among Primus Telecommunications IHC, Inc., and the parties thereto
4.12*	Collateral Agreement, dated as of December 22, 2009, made by Primus Telecommunications Holding, Inc. and certain of its Affiliates in favor of U.S. Bank National Association, as Collateral Trustee in respect of the 13% Holding Notes
4.13*	Canadian Collateral Agreement, dated as of December 22, 2009, made by Primus Telecommunications Canada Inc. and certain of its Affiliates in favor of Computershare Trust Company of Canada, as Canadian Collateral Trustee in respect of the 13% Canadian Notes
4.14*	New Debt Notice and Joinder Agreement, dated as of December 22, 2009
10.13	Separation Agreement by and between John F. DePodesta and Primus Telecommunications Group, Incorporated, dated as of July 1, 2009
10.23	Separation Agreement by and between Thomas R. Kloster and Primus Telecommunications, Inc., dated as of July 1, 2009
10.33	Primus Telecommunications Group, Incorporated Management Compensation Plan
10.43	Form of Restricted Stock Unit Agreement between Grantee and Primus Telecommunications Group, Incorporated, dated as of July 1, 2009
10.53	Form of Nonqualified Stock Option Agreement between Grantee and Primus Telecommunications Group, Incorporated, dated as of July 1, 2009
10.6^{3}	Form of Performance Nonqualified Stock Option Agreement between Grantee and Primus Telecommunications Group, Incorporated, dated as of July 1, 2009
10.7^{6}	Employment Agreement, dated April 26, 2007, between Primus Telecommunications Group, Incorporated and K. Paul Singh
10.87	401(k) Plan, as amended
12.1*	Ratio of Earnings to Fixed Charges
21.1*	Subsidiaries of the Registrant
23.1*	Consent of Independent Registered Public Accounting Firm
31*	Certifications.
328	Certification.
1 .	

Incorporated by reference to the Registrant's Form 8-K, filed with the Commission on June 15, 2009.

Incorporated by reference to the Registrant's Registration Statement on Form 8-A, filed with the Commission on July 1, 2009.

Incorporated by reference to the Registrant's Form 8-K, filed with the Commission on July 1, 2009.

Incorporated by reference to the Registrant's Form 8-K, filed with the Commission on December 24, 2009.

Incorporated by reference to the Registrant's Form 8-K/A dated March 16, 2007.

Incorporated by reference to the Registrant's incorporated by reference to Registrant's annual report on Form 10-K for the year ended December 31, 2007. Incorporated by reference to Registrant's Registration Statement on Form S-8 (No. 333-35005) and as amended on Post Effective Amendment No. 1 to Form S-8 filed with the SEC on March 26, 2003.

- Furnished Herewith. This certification is being "furnished" and will not be deemed "filed" for purposes of Section 18 of the Securities Exchange Act (15 U.S.C. 78r) and will not be deemed to be incorporated by reference into any filing under the Securities Act of 1933 or the Securities Exchange Act of 1934, except to the extent that the registrant specifically incorporates it by reference.
- * Filed herewith.

SIGNATURES

Pursuant to the requirements of the Section 13 or 15(d) of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized on April 5, 2010.

PRIMUS TELECOMMUNICATIONS GROUP, INCORPORATED

By: /s/ K. PAUL SINGH

K. Paul Singh

Chairman of the Board, President and
Chief Executive Officer

KNOW ALL MEN BY THESE PRESENTS, that each person whose signature appears below constitutes and appoints K. Paul Singh, Thomas R. Kloster and James C. Keeley, and each of them, his true and lawful attorneys-in-fact and agents, with full power of substitution and resubstitution, for him and in his name, place and stead, in any and all capacities to sign any and all amendments to this Form 10-K of the Securities and Exchange Commission for the fiscal year of Primus Telecommunications Group, Incorporated ended December 31, 2009, and to file the same, with all exhibits thereto, and other documents in connection therewith, with authority to do and perform each and every act and thing requisite and necessary to be done in and about the premises as fully to all intents and purposes as he or she might or could do in person, hereby ratifying and confirming all that said attorneys-in-fact and agents, or either of them, or their or his substitute or substitutes, may lawfully do or cause to be done by virtue hereof.

Pursuant to the requirements of the Securities Exchange Act of 1934, this report has been signed below by the following persons on behalf of the Registrant and in the capacities and on the date indicated.

Signature	<u>Title</u>	<u>Date</u>
/s/ K. PAUL SINGH K. Paul Singh	Chairman, President and Chief Executive Officer (Principal Executive Officer) and Director	April 5, 2010
/s/ THOMAS R. KLOSTER Thomas R. Kloster	Chief Financial Officer (Principal Financial Officer)	April 5, 2010
/s/ JAMES C. KEELEY James C. Keeley	Vice President, Corporate Controller (Principal Accounting Officer)	April 5, 2010
/s/ PETER D. AQUINO Peter D. Aquino	Director	April 5, 2010
/s/ JOHN B. SPIRTOS John B. Spirtos	Director	April 5, 2010
/s/ NEIL S. SUBIN Neil S. Subin	Director	April 5, 2010

INDEX TO FINANCIAL STATEMENTS AND SCHEDULE

	Page
Management's Report on Internal Control Over Financial Reporting	F-2
Report of Independent Registered Public Accounting Firm	F-3
Consolidated Financial Statements:	
Consolidated Statements of Operations for the six months ended December 31, 2009, and July 1, 2009, and for the years ended December 31, 2008	
and 2007	F-4
Consolidated Balance Sheets as of December 31, 2009 and 2008	F-5
Consolidated Statements of Stockholders' Equity (Deficit) for the six months ended December 31, 2009, and July 1, 2009, and for the years ended	
<u>December 31, 2008 and 2007</u>	F-6
Consolidated Statements of Cash Flows for the six months ended December 31, 2009, and July 1, 2009, and for the years ended December 31,	
2008 and 2007	F-7
Consolidated Statements of Comprehensive Income (Loss) for the six months ended December 31, 2009, and July 1, 2009, and for the years ended	
<u>December 31, 2008 and 2007</u>	F-8
Notes to Consolidated Financial Statements	F-9
Consolidated Financial Statement Schedule:	
Schedule II. Valuation and Qualifying Accounts	S-1

MANAGEMENT'S REPORT ON INTERNAL CONTROL OVER FINANCIAL REPORTING

The management of Primus Telecommunications Group, Incorporated ("Primus" or the "Company") is responsible for establishing and maintaining adequate internal control over financial reporting as defined in Rules 13a-15(f) and 15d-15(f) under the Securities Exchange Act of 1934. The Company's internal control system over financial reporting is designed to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles. The Company's internal control over financial reporting includes those policies and procedures that:

- 1. pertain to the maintenance of records that, in reasonable detail, accurately and fairly reflect the transactions and dispositions of the assets of the Company;
- 2. provide reasonable assurance that transactions are recorded as necessary to permit preparation of financial statements in accordance with United States generally accepted accounting principles, and that receipts and expenditures of the Company are being made only in accordance with authorizations of management and directors of the Company; and
- 3. provide reasonable assurance regarding prevention or timely detection of unauthorized acquisition, use or disposition of the Company's assets that could have a material effect on the financial statements.

Because of inherent limitations, internal control over financial reporting may not prevent or detect misstatements. Therefore, even those systems determined to be effective can provide only reasonable assurance with respect to financial statement preparation and presentation. Also, projections of any evaluation of effectiveness to future periods are subject to risk that controls may become inadequate because of changes in condition, or that the degree of compliance with the policies or procedures may deteriorate.

The Company's management assessed the effectiveness of its internal control over financial reporting as of December 31, 2009. In making this assessment, management used the criteria set forth by the Committee of Sponsoring Organizations of the Treadway Commission (COSO) in Internal Control-Integrated Framework. Based on management's assessment and those criteria, management concludes that Primus did not maintain effective internal control over financial reporting as of December 31, 2009, due to the material weakness in the Company's internal control over accounting for foreign currency translation gain (loss) (details provided in Item 9A, Controls and Procedures of the Company's Annual Report on Form 10-K for the period ended December 31, 2009). This annual report does not include an attestation report of the Company's registered public accounting firm regarding internal control over financial reporting. Management's report was not subject to attestation by the Company's registered public accounting firm pursuant to temporary rules of the Securities and Exchange Commission that permit the Company to provide only management's report in this annual report.

/s/ K. PAUL SINGH	April 5, 2010
K. Paul Singh Chairman, President and Chief Executive Officer and Director	•
/s/ Thomas R. Kloster	April 5, 2010
Thomas R. Kloster Chief Financial Officer (Principal Financial Officer)	•

REPORT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

To the Board of Directors and Stockholders of Primus Telecommunications Group, Incorporated and subsidiaries McLean, VA

We have audited the accompanying consolidated balance sheet of Primus Telecommunications Group, Incorporated and subsidiaries (the "Company" and "Successor") as of December 31, 2009 and the related consolidated statements of operations, stockholders' equity (deficit), cash flows, and comprehensive income (loss) for the six months ended December 31, 2009. We have also audited the accompanying balance sheet of Primus Telecommunications Group, Incorporated and subsidiaries (the "Company" and "Predecessor") as of December 31, 2008 and the related consolidated statements of operations, stockholders' equity (deficit), cash flows and comprehensive income (loss) for the six months ended July 1, 2009 and for the years ended December 31, 2008 and 2007. Our audits also included the financial statement schedule listed in the Index at Item 15. These financial statements and financial statement schedule are the responsibility of the Company's management. Our responsibility is to express an opinion on the financial statements and financial statement schedule based on our audits.

We conducted our audits in accordance with the standards of the Public Company Accounting Oversight Board (United States). Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. The Company is not required to have, nor were we engaged to perform, an audit of its internal control over financial reporting. Our audits included consideration of internal control over financial reporting as a basis for designing audit procedures that are appropriate in the circumstances, but not for the purpose of expressing an opinion on the effectiveness of the Company's internal control over financial reporting. Accordingly, we express no such opinion. An audit also includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements, assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion.

In our opinion, such consolidated financial statements present fairly, in all material respects, the financial position of the Successor as of December 31, 2009 and the results of its operations and its cash flows for the six months ended December 31, 2009 and the financial position of the Predecessor as of December 31, 2008 and the results of its operations and its cash flows for the six months ended July 1, 2009 and for the years ended December 31, 2008 and 2007, in conformity with accounting principles generally accepted in the United States of America. Also, in our opinion, such financial statement schedule, when considered in relation to the basic consolidated financial statements taken as a whole, presents fairly in all material respects the information set forth therein.

As discussed in Notes 2 and 4 to the consolidated financial statements, during 2009 the Company and certain subsidiaries filed for reorganization under Chapter 11 of the United States bankruptcy code, and emerged from bankruptcy under a plan of reorganization. As a consequence, the Company adopted fresh-start accounting in accordance with Accounting Standards Codification ("ASC") No. 852, *Reorganizations*.

As discussed in Notes 3 and 8 to the consolidated financial statements, in 2007 the Company changed its method of accounting for uncertain tax positions to conform to ASC No. 740, *Income Taxes* (formerly Financial Accounting Standards Board Interpretation No. 48).

/s/ Deloitte & Touche LLP McLean, Virginia April 5, 2010

PRIMUS TELECOMMUNICATIONS GROUP, INCORPORATED

CONSOLIDATED STATEMENTS OF OPERATIONS

(in thousands, except per share amounts)

	Successor		Predecessor	
	Six Months Ended December 31, 2009	Six Months Ended July 1, 2009	Year Ended December 31, 2008	Year Ended December 31, 2007
NET REVENUE	\$ 424,334	\$ 391,216	\$ 895,863	\$ 896,029
OPERATING EXPENSES				
Cost of revenue (exclusive of depreciation included below)	278,417	255,288	569,865	551,303
Selling, general and administrative	102,911	95,836	260,430	281,010
Depreciation and amortization	39,530	12,346	32,791	30,529
(Gain) loss on sale or disposal of assets	181	(43)	(6,028)	1,463
Total operating expenses	421,039	363,427	857,058	864,305
INCOME FROM OPERATIONS	3,295	27,789	38,805	31,724
INTEREST EXPENSE	(17,323)	(14,135)	(53,888)	(61,347)
ACCRETION (AMORTIZATION) ON DEBT PREMIUM/DISCOUNT, net	(3)	189	583	(449)
GAIN (LOSS) ON EARLY EXTINGUISHMENT OR RESTRUCTURING OF DEBT	(4,146)	_	36,872	(7,652)
LOSS FROM CONTINGENT VALUE RIGHTS VALUATION	(2,804)	_	_	
INTEREST AND OTHER INCOME	564	396	3,279	5,665
FOREIGN CURRENCY TRANSACTION GAIN (LOSS)	18,384	21,121	(47,563)	32,699
INCOME (LOSS) FROM CONTINUING OPERATIONS BEFORE REORGANIZATION ITEMS AND INCOME				
TAXES	(2,033)	35,360	(21,912)	640
REORGANIZATION ITEMS, net	<u>(562</u>)	440,094		
INCOME (LOSS) FROM CONTINUING OPERATIONS BEFORE INCOME TAXES	(2,595)	475,454	(21,912)	640
INCOME TAX BENEFIT (EXPENSE)	10,041	(3,907)	366	9,232
INCOME (LOSS) FROM CONTINUING OPERATIONS	7,446	471,547	(21,546)	9,872
LOSS FROM DISCONTINUED OPERATIONS, net of tax	(271)	(676)	(326)	(268)
GAIN (LOSS) FROM SALE OF DISCONTINUED OPERATIONS, net of tax	(110)	251	<u>—</u> .	6,132
NET INCOME (LOSS)	7,065	471,122	(21,872)	15,736
Less: Net (income) loss attributable to the noncontrolling interest	(333)	32	(3,159)	_
NET INCOME (LOSS) ATTRIBUTABLE TO PRIMUS TELECOMMUNICATIONS GROUP, INCORPORATED	\$ 6,732	\$ 471,154	\$ (25,031)	\$ 15,736
BASIC INCOME (LOSS) PER COMMON SHARE:				
Income (loss) from continuing operations attributable to Primus Telecommunications Group, Incorporated	\$ 0.74	\$ 3.30	\$ (0.17)	\$ 0.07
Loss from discontinued operations	(0.03)	(0.00)	(0.01)	(0.00)
Gain (loss) from sale of discontinued operations	(0.01)	0.00	(0.01)	0.05
Net income (loss) attributable to Primus Telecommunications Group, Incorporated	\$ 0.70	\$ 3.30	\$ (0.18)	\$ 0.12
	\$ 0.70	\$ 3.30	\$ (0.10)	5 0.12
DILUTED LOSS PER COMMON SHARE:				
Income (loss) from continuing operations attributable to Primus Telecommunications Group, Incorporated	\$ 0.73	\$ 2.72	\$ (0.17)	\$ 0.06
Loss from discontinued operations	(0.03)	(0.00)	(0.01)	(0.00)
Gain (loss) from sale of discontinued operations	(0.01)	0.00		0.03
Net income (loss) attributable to Primus Telecommunications Group, Incorporated	\$ 0.69	\$ 2.72	\$ (0.18)	\$ 0.09
WEIGHTED AVERAGE COMMON SHARES OUTSTANDING				
Basic	9,600	142,695	142,643	128,771
Diluted	9,800	173,117	142,643	196,470
AMOUNTS ATTRIBUTABLE TO COMMON SHAREHOLDERS OF PRIMUS TELECOMMUNICATIONS GROUP. INCORPORATED	3,000		142,040	130,470
Income (loss) from continuing operations, net of tax	\$ 7.113	\$ 471,579	\$ (24,705)	\$ 9,872
Loss from discontinued operations	(271)	(676)	(326)	(268)
Gain (loss) from sale of discontinued operations	(110)	251	_	6,132
Net income (loss)	\$ 6,732	\$ 471,154	\$ (25,031)	\$ 15,736
i co meome (1000)	9 0,752	Ψ 471,104	<u> </u>	<u> </u>

PRIMUS TELECOMMUNICATIONS GROUP, INCORPORATED

CONSOLIDATED BALANCE SHEETS

(in thousands, except share amounts)

	Successor December 31, 2009	Predecessor December 31, 2008
ASSETS		
CURRENT ASSETS:		
Cash and cash equivalents	\$ 42,538	\$ 37,000
Accounts receivable (net of allowance for doubtful accounts receivable of \$8,163 and \$9,710)	89,342	99,483
Prepaid expenses and other current assets	15,147	15,846
Total current assets	147,027	152,329
RESTRICTED CASH	10,438	8,133
PROPERTY AND EQUIPMENT—Net	147,606	112,152
GOODWILL	64,220	32,688
OTHER INTANGIBLE ASSETS—Net	178,807	746
OTHER ASSETS	10,816	24,396
TOTAL ASSETS	\$ 558,914	\$ 330,444
LIABILITIES AND STOCKHOLDERS' EQUITY (DEFICIT)		
CURRENT LIABILITIES:		
Accounts payable	\$ 45,819	\$ 58,671
Accrued interconnection costs	37,561	41,422
Deferred revenue	13,882	13,303
Accrued expenses and other current liabilities	49,704	42,440
Accrued income taxes	10,629	18,213
Accrued interest	1,985	10,248
Current portion of long-term obligations	4,274	564,797
Total current liabilities	163,854	749,094
LONG-TERM OBLIGATIONS	253,242	40,040
DEFERRED TAX LIABILITY	36,052	_
OTHER LIABILITIES	5,857	35
Total liabilities	459,005	789,169
COMMITMENTS AND CONTINGENCIES (See Note 10.)		
CTOCKHOLDEDC: FOLHER (DEFICIE)		
STOCKHOLDERS' EQUITY (DEFICIT):		
Predecessor Preferred stock: Not Designated, \$0.01 par value—1,410,050 shares authorized; none issued and		
outstanding; Series A and B, \$0.01 par value—485,000 shares authorized; none issued and outstanding; Series C,		
\$0.01 par value—559,950 shares authorized; none issued and outstanding Predecessor Common stock, \$0.01 par value—300,000,000 shares authorized; 142,695,390 shares issued and	-	_
outstanding		1,427
Successor Preferred stock, \$0.001 par value—20,000,000 shares authorized; none issued or outstanding	_	1,427
Successor Common stock, \$0.001 par value—20,000,000 shares authorized; none issued or outstanding Successor Common stock, \$0.001 par value—80,000,000 shares authorized; 9,600,000 shares issued or outstanding	10	_
Additional paid-in capital	85,533	718,956
Accumulated earnings (deficit)	6,732	(1,099,809)
Accumulated other comprehensive income (loss)	4,064	(82,113)
Total stockholders' equity (deficit) before noncontrolling interest	96,339	(461,539)
Noncontrolling interest	3,570	2.814
Total stockholders' equity (deficit)	99,909	(458,725)
TOTAL LIABILITIES AND STOCKHOLDERS' EQUITY (DEFICIT)	\$ 558,914	\$ 330,444
		l ====

PRIMUS TELECOMMUNICATIONS GROUP, INCORPORATED CONSOLIDATED STATEMENT OF STOCKHOLDERS' EQUITY (DEFICIT) (in thousands)

	PREDECESSOR PREDECESSOR						
	Commo	on Stock		Accumulated	Accumulated Other		Stockholders'
	Common Shares	Common Stock	Additional Paid- In-Capital	Earnings (Deficit)	Comprehensive Income (Loss)	Noncontrolling Interest	Equity (Deficit)
BALANCE AS OF JANUARY 1, 2007	113,849	\$ 1,138	\$ 692,941	\$ (1,082,853)	\$ (79,481)	\$ 43	\$ (468,212)
Common shares issued in private equity offering	22,500	225	19,124	_	_	_	19,349
Offering cost for sale of stock	_	_	(179)	_	_	_	(179)
Common shares issued for compensation	284	3	(3)	_	_	_	
Common shares issued in exchange for the Company's debentures	6,000	60	6,566	_	_	_	6,626
Share based compensation expense	_	_	246	_	_	_	246
Adjustment from implementation of FIN No. 48	_	_	_	(7,661)	_	_	(7,661)
Foreign currency translation adjustment	_	_	_	_	(13,402)	8	(13,394)
Net Income (Loss)				15,736			15,736
BALANCE AS OF DECEMBER 31, 2007	142,633	1,426	718,695	(1,074,778)	(92,883)	51	(447,489)
Common shares issued for restricted stock units	62	1	(1)	· · · · · · · · · · · · · · · · · · ·	` — ′	_	` _ ′
Share based compensation expense	_	_	262	_	_	_	262
Foreign currency translation adjustment	_	_	_	_	10,770	(396)	10,374
Net Income (Loss)				(25,031)		3,159	(21,872)
BALANCE AS OF DECEMBER 31, 2008	142,695	1,427	718,956	(1,099,809)	(82,113)	2,814	(458,725)
Share based compensation expense	_	_	27	· · · · · · ·	` _ ′	_	27
Foreign currency translation adjustment	_	_	_	_	(7,103)	149	(6,954)
Net Income (Loss)				39,357		(32)	39,325
BALANCE AS OF JUNE 30, 2009	142,695	1,427	718,983	(1,060,452)	(89,216)	2,931	(426,327)
Plan of Reorganization and fresh-start adjustments	(133,095)	(1,417)	(634,601)	1,060,452	89,216	_	513,650
BALANCE AS OF JULY 1, 2009	9,600	10	84,382			2,931	87,323
				SUCCESSO	R		
BALANCE AS OF JULY 1, 2009	9,600	10	84,382	_	_	2,931	87,323
Share based compensation expense	_	_	1,151	_	_	_	1,151
Foreign currency translation adjustment	_	_	-	_	4,064	306	4,370
Net Income				6,732		333	7,065
BALANCE AS OF DECEMBER 31, 2009	9,600	\$ 10	\$ 85,533	\$ 6,732	\$ 4,064	\$ 3,570	\$ 99,909

PRIMUS TELECOMMUNICATIONS GROUP, INCORPORATED

CONSOLIDATED STATEMENTS OF CASH FLOWS (in thousands)

	Successor		Predecessor	
	Six Months Ended December 31 2009	Ended	Year Ended December 31, 2008	Year Ended December 31, 2007
CASH FLOWS FROM OPERATING ACTIVITIES:				
Net income (loss)	\$ 7,065	5 \$ 471,122	\$ (21,872)	\$ 15,736
Adjustments to reconcile net income (loss) to net cash provided by operating activities: Reorganization items, net	562	2 (440,094)		
Provision for doubtful accounts receivable	5,014		11,940	10,493
Stock compensation expense	1.15		262	246
Depreciation and amortization	39,530		32,798	30,594
(Gain) loss on sale or disposal of assets	29:			(4,668)
Accretion (amortization) of debt premium/discount, net	3	3 (189)		449
Change in fair value of Contingent Value Rights	2,804		`— ´	_
Deferred income taxes	(3,89		5,838	(12,463)
(Gain) loss on early extinguishment of debt	2,189		(36,872)	7,652
Unrealized foreign currency transaction (gain) loss on intercompany and foreign debt	(20,476	6) (20,702)	48,588	(34,862)
Changes in assets and liabilities, net of acquisitions:	4 = 0		(4= 5=0)	
(Increase) decrease in accounts receivable	4,72		(15,658)	5,275
(Increase) decrease in prepaid expenses and other current assets	1,880 1,80		10,191 1,877	(1,556)
Decrease in other assets Decrease in accounts payable	(7,680			2,309 (2,652)
Increase (decrease) in accrued interconnection costs	(2,478			(6,244)
Increase (decrease) in accrued interconnection costs Increase (decrease) in accrued expenses, deferred revenue, other current liabilities and other liabilities, net	(3,92)		(5,640)	(2,029)
Increase (decrease) in accrued income taxes	(11,080		(10,625)	4,352
Increase (decrease) in accrued interest	1,965		(1,750)	(1,165)
Net cash provided by operating activities before cash reorganization items	19,450		8,779	11,467
Cash effect of reorganization items	(7,615		0,773 —	11,407
Net cash provided by operating activities	11.835		8,779	11,467
CASH FLOWS FROM INVESTING ACTIVITIES:	11,000	17,143	0,773	11,407
Purchase of property and equipment	(9,390	6) (5,660)	(25,441)	(44,745)
Sale of property and equipment and intangible assets	12		5,756	(,,)
Cash from disposition of business, net of cash disposed	(110		1,676	6,140
Cash used in business acquisitions, net of cash acquired	_	(199)		(200)
Increase in restricted cash	(19	9) (146)	(102)	(668)
Net cash used in investing activities	(9,513	3) (5,594)	(18,694)	(39,473)
CASH FLOWS FROM FINANCING ACTIVITIES:				
Purchase of the Company's debt securities	_	_	(11,534)	_
Proceeds from issuance of long-term obligations	128,063	3 —	``	109,275
Deferred financing costs	(4,569	9) —	_	(6,570)
Principal payments on long-term obligations	(127,87	1) (8,292)	(16,545)	(80,415)
Proceeds from sale of common stock, net of issuance costs				19,170
Net cash provided by (used in) financing activities	(4,37)	7) (8,292)	(28,079)	41,460
EFFECTS OF EXCHANGE RATE CHANGES ON CASH AND CASH EQUIVALENTS	3,132	1,202	(6,288)	3,511
NET CHANGE IN CASH AND CASH EQUIVALENTS	1,07	7 4,461	(44,282)	16,965
CASH AND CASH EQUIVALENTS, BEGINNING OF PERIOD	41,46	1 37,000	81,282	64,317
CASH AND CASH EQUIVALENTS, END OF PERIOD	\$ 42,538		\$ 37,000	\$ 81,282
SUPPLEMENTAL CASH FLOW INFORMATION:				
Cash paid for interest	\$ 14,609	9 \$ 14,909	\$ 53,442	\$ 59,933
Cash paid for taxes	\$ 3,018		\$ 1,908	\$ 1,056
Non-cash investing and financing activities:	Ψ 5,010	φ 302	ψ 1,500	Ψ 1,050
Capital lease additions	\$ 409	9 \$ 1,882	\$ 775	\$ 2,418
Accrued deferred financing costs	\$ 1,74	. , ,	\$ —	\$ —
Leased fiber capacity additions	\$	\$ —	\$ —	\$ 1,786
Property and equipment, accrued in current liabilities	\$ —	\$ —	\$ —	\$ 883
Settlement of outstanding debt with issuance of common stock	\$ —	\$ —	\$ —	\$ 6,626
Settlement of outstanding debt with issuance of new senior secured debt	\$ —	\$ —	\$ (133,159)	\$ —
Issuance of new senior secured debt in exchange for outstanding debt	\$ —	\$ —	\$ 88,794	\$ —
Business disposition proceeds in note receivable	\$ —	\$ —	\$ —	\$ 845
Business acquisition financed with note payable	\$ —	\$ —	\$ 247	\$ —

PRIMUS TELECOMMUNICATIONS GROUP, INCORPORATED CONSOLIDATED STATEMENTS OF COMPREHENSIVE INCOME (LOSS) (in thousands)

	Successor Six Months Ended December 31, 2009	Six Months Ended July 1, 2009	Predecessor Year Ended December 31, 2008	Year Ended December 31, 2007	
NET INCOME (LOSS)	\$ 7,065	\$471,122	\$ (21,872)	\$ 15,736	
OTHER COMPREHENSIVE INCOME (LOSS)					
Foreign currency translation adjustment	4,370	(6,954)	10,374	(13,394)	
Fresh-start adjustment		89,216			
COMPREHENSIVE INCOME (LOSS)	11,435	553,384	(11,498)	2,342	
Less: Comprehensive (income) loss attributable to the noncontrolling interest	(639)	(117)	(2,763)	(8)	
COMPREHENSIVE INCOME (LOSS) ATTRIBUTABLE TO PRIMUS TELECOMMUNICATIONS GROUP, INCORPORATED	\$ 10,796	\$553,267	\$ (14,261)	\$ 2,334	

PRIMUS TELECOMMUNICATIONS GROUP, INCORPORATED NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

1. ORGANIZATION AND BUSINESS

Primus Telecommunications Group, Incorporated, ("Primus" or the "Company") is a facilities based telecommunications services provider offering a portfolio of international and domestic voice, wireless, Internet, voice-over-Internet protocol (VOIP), data and data center services to business and residential retail customers and other carriers located primarily in the United States, Australia, Canada, Brazil, the United Kingdom and western Europe. The Company's focus is to service the demand for high quality, competitively priced communications services that is being driven by the globalization of the world's economies, the worldwide trend toward telecommunications deregulation and the growth of broadband, Internet, VOIP, wireless and data traffic.

The Company targets customers with significant telecommunications needs, including small- and medium-sized enterprises (SMEs), multinational corporations, residential customers, and other telecommunications carriers and resellers. The Company provides services over its global network, which consists of:

- 10 data centers in 8 cities in Canada, Australia and Brazil;
- 18 carrier-grade international gateway and domestic switching systems (the hardware/software devices that direct the voice traffic across the network), internet routers and media gateways in the United States, Canada, Europe and the Asia-Pacific region;
- approximately 500 interconnection points to the Company's network, or points of presence (POPs), which includes digital subscriber line access (DSLAM), which is equipment that allows digital traffic to flow over copper wiring, within its service regions and other markets;
- undersea and land-based fiber optic transmission line systems that the Company owns or leases and that carry voice and data traffic across the network:
- · a global network that use a high-bandwidth network standard (asynchronous transfer mode) and Internet-based protocol (ATM+IP); and
- a global VOIP network is based on routers and gateways with an open network architecture which connects the Company's partners in over 150 countries.

The Company is incorporated in the state of Delaware and operates as a holding company of wholly-owned operating subsidiaries primarily in the United States, Canada, Australia, Brazil, the United Kingdom and western Europe.

2. EMERGENCE FROM VOLUNTARY REORGANIZATION UNDER CHAPTER 11 OF THE BANKRUPTCY CODE

On March 16, 2009, Primus Telecommunications Group, Incorporated ("Group" or "PTGI") and three of its subsidiaries, Primus Telecommunications Holding, Inc. ("Holding" or "PTHI"), Primus Telecommunications International, Inc. ("PTII") and Primus Telecommunications IHC, Inc., ("IHC" and together with Group, Holding and PTII, collectively, the "Debtors") each filed a voluntary petition (the "Chapter 11 Cases") in the United States Bankruptcy Court for the District of Delaware (the "Bankruptcy Court") for reorganization relief ("Reorganization") under chapter 11 of title 11 of the United States Code, 11 U.S.C. §§ 101 *et seq.*, as amended (the "Bankruptcy Code"). Subsequently, the Debtors sought and received an order directing the joint administration of the Chapter 11 Cases under the caption In re: Primus Telecommunications Group, Incorporated, et al., Debtors, Case No. 09-10867. On April 8, 2009, April 20, 2009, and April 24, 2009, filings were made by the Debtors in the Bankruptcy Court concerning amended Disclosure Statements and Joint Plans of Reorganization of Primus Telecommunications Group, Incorporated and its Affiliate Debtors. On April 24, 2009, an unsecured creditors' committee was appointed by the United States Trustee.

On April 27, 2009, the Bankruptcy Court approved the Debtors' use of a disclosure statement dated April 27, 2009 (the "Disclosure Statement") to solicit votes on the Joint Plan of Reorganization of Primus Telecommunications Group, Incorporated and its Affiliate Debtors attached thereto (the "Plan"). The Disclosure Statement was distributed to holders of record (as of April 27, 2009) of claims against, and interests in, the Debtors who are entitled to vote on the Plan (the "Record Date").

The order approving the Disclosure Statement also (i) established the Record Date and a voting deadline of June 5, 2009, (ii) established June 5, 2009 as the last date and time for filing and serving objections to confirmation of the Plan (and related requirements and procedures set forth in such order), and (iii) fixed June 1, 2009 as the deadline for claimants and interest holders to file and serve motions under Bankruptcy Rule 3018(a) requesting temporary allowance of the movant's claim or interest for purposes of voting.

The Plan was confirmed by the Bankruptcy Court on June 12, 2009 (the "Confirmation Date"). On July 1, 2009 (the "Effective Date"), the Debtors consummated their reorganization under the Bankruptcy Code and the Plan became effective.

The Plan provided for a plan of reorganization of the Debtors on terms that are summarized below:

- · Holding's Term Loan facility due February 2011 was reinstated and amended (see "Amended Term Loan Facility" below);
- IHC's 14 1/4% Senior Secured Notes were cancelled and the holders thereof received (a) their pro rata portion of \$123.5 million of aggregate principal amount of 14 1/4% Senior Subordinated Secured Notes due May 20, 2013 (see "Amendment of IHC's 14 1/4% Senior Secured Notes Indenture" below), (b) 4,800,000 shares of the new Common Stock of Group (the "New Common Stock"), and (c) all reasonable fees, expenses and disbursements of their counsel;
- the 5% Exchangeable Senior Notes and 8% Senior Notes issued by Holding (collectively, the "Holding Senior Notes") were cancelled, and the holders thereof received (a) 4,800,000 shares of the New Common Stock, (b) Class A warrants to purchase up to an aggregate of 3,000,000 shares of New Common Stock on terms described below under "Warrant Agreements," and (c) all reasonable fees, expenses and disbursements of their counsel;
- the 3³/₄ % Senior Notes due September 2010, 12³/₄% Senior Notes due October 2009 and Step Up Convertible Subordinated Debentures due August 2009 issued by Group (collectively, the "Group Notes") were cancelled, and the holders thereof received Class B warrants to purchase up to an aggregate of 1,500,000 shares of the New Common Stock on terms described below under "Warrant Agreements;"
- all existing shares of common stock outstanding prior to the Effective Date (the "Old Common Stock") were cancelled on the Effective Date, and holders thereof received their pro rata share of contingent value rights ("Contingent Value Rights" or "CVRs") to acquire up to 2,665,000 shares of New Common Stock on terms described below under "Contingent Value Rights Distribution Agreement;"
- all outstanding equity incentive grants of Group were cancelled on the Effective Date, and the Primus Telecommunications Group, Incorporated Management Compensation Plan (the "Management Compensation Plan") became effective. As of the Effective Date, 400,000 restricted stock units, 387,403 service-based stock options and 100,000 performance-based stock options were granted to certain employees and executive officers under the Management Compensation Plan.

The following table summarizes the effect of the Plan of Reorganization adjustments with respect to long-term obligations after giving effect to the July 1, 2009 emergence from bankruptcy.

	p. 1	Plan of Reorganization	
Long-Term Obligations	Predecessor	Adjustments	Successor
Obligations under capital leases and other	\$ 5,056	\$ —	\$ 5,056
Leased fiber capacity	2,531	_	2,531
Senior secured term loan facility	95,750	_	95,750
Canadian credit facility	29,500	_	29,500
Senior subordinated secured notes	_	123,472	123,472
Subtotal	132,837	123,472	256,309
Less: Current portion of long-term obligations	(107,097)	91,100	(15,997)
Total long-term obligations	\$ 25,740	\$ 214,572	\$240,312
Liabilities Subject to Compromise			
Senior secured notes	\$ 173,157	\$ (173,157)	\$ —
Senior notes	200,186	(200,186)	_
Exchangeable senior notes	23,369	(23,369)	_
Convertible senior notes	34,200	(34,200)	
Step up convertible subordinated debentures	8,641	(8,641)	_
Accrued interest	11,497	(11,497)	
Total liabilities subject to compromise	\$ 451,050	\$ (451,050)	\$ —

In December 2009, the Company issued \$130.0 million principal amount of 13% Senior Secured Notes due 2016 subject to certain conditions. The proceeds from the issuance were used to retire the Senior Secured Term Loan Facility and the Canadian Credit Facility. See "Note 6—Goodwill and Other Intangible Assetes," for further detail.

Warrant Agreements

As of the Effective Date, Group issued Class A warrants to purchase up to an aggregate of 3,000,000 shares of New Common Stock to holders of the Holding Notes. The Class A warrants consist of 1,000,000 each of Class A-1 warrants, Class A-2 warrants and Class A-3 warrants. In connection with the issuance of the Class A warrants, Group entered into a warrant agreement, dated as of the Effective Date (the "Class A Warrant Agreement"), with StockTrans, Inc., as warrant agent. Subject to the terms of the Class A Warrant Agreement, Class A-1 warrant holders are entitled to purchase up to 1,000,000 shares of New Common Stock at an initial exercise price of \$12.22 per share, Class A-2 warrant holders are entitled to purchase up to 1,000,000 shares of New Common Stock at an initial exercise price of \$16.53 per share, and Class A-3 warrant holders are entitled to purchase up to 1,000,000 shares of New Common Stock at an initial exercise price of \$20.50 per share. The Class A warrants have a five-year term and will expire on July 1, 2014. A holder may exercise Class A warrants by paying the applicable exercise price in cash. In addition, a holder may exercise Class A warrants on a cashless basis in connection with a change of control (as defined in the Class A Warrant Agreement), in connection with a transaction pursuant to an effective registration statement covering the sale of New Common Stock underlying such Class A warrants, or if the exercise occurs on a date when the daily volume-weighted average price of the New Common Stock for the immediately preceding 10 trading days exceeds 150% of the exercise price applicable to such Class A warrants. The Class A warrants are freely transferrable by the holder thereof.

As of the Effective Date, Group issued Class B warrants to purchase up to an aggregate of 1,500,000 shares of New Common Stock to holders of the Group Notes. In connection with the issuance of the Class B warrants, Group entered into a warrant agreement, dated as of the Effective Date (the "Class B Warrant Agreement"), with

StockTrans, Inc., as warrant agent. Subject to the terms of the Class B Warrant Agreement, Class B warrant holders are entitled to purchase 1,500,000 shares of New Common Stock at an initial exercise price of \$26.01 per share. The Class B warrants have a five-year term and will expire at on July 1, 2014. A holder may exercise Class B warrants by paying the applicable exercise price in cash. In addition, a holder may exercise Class B warrants on a cashless basis in connection with a change of control (as defined in the Class B Warrant Agreement), in connection with a transaction pursuant to an effective registration statement covering the sale of New Common Stock underlying such Class B warrants, or if the exercise occurs on a date when the daily volume-weighted average price of the New Common Stock for the immediately preceding 10 trading days exceeds 150% of the exercise price applicable to the Class B warrants. The Class B warrants are freely transferrable by the holder thereof.

The number of shares of New Common Stock issuable upon exercise of the Class A warrants and Class B warrants (together, the "Warrants") and the exercise prices of the Warrants will be adjusted in connection with any dividend or distribution of New Common Stock, assets or cash (other than any regular cash dividend not to exceed in any fiscal year 45% of the consolidated net income of Group), or any subdivision or combination of the New Common Stock. In addition, the number of shares of New Common Stock issuable upon exercise of the Warrants and the exercise prices of the Warrants are also subject to adjustment in connection with any issuance, grant or sale to any person of (A) rights, warrants, options, exchangeable securities or convertible securities entitling such person to subscribe for, purchase or otherwise acquire shares of New Common Stock at a price per share less than the fair market value of the New Common Stock on the trading day immediately prior to such issuance, sale or grant, subject to certain exceptions, or (B) shares of New Common Stock at a price per share less than the fair market value of the New Common Stock on the trading day immediately prior to such issuance, sale or grant. Additionally, if any transaction or event occurs in which all or substantially all of the outstanding New Common Stock is converted into, exchanged for, or the holders thereof are otherwise entitled to receive on account thereof stock, other securities, cash or assets (each, a "Fundamental Change Transaction") the holder of each Warrant outstanding immediately prior to the occurrence of such Fundamental Change Transaction shall have the right to receive upon exercise of the applicable Warrant the kind and amount of stock, other securities, cash and/or assets that such holder would have received if such Warrant had been exercised.

The Company recorded \$1.0 million of equity related to the issuance of the warrants.

Contingent Value Rights Distribution Agreement

Pursuant to the terms of the Plan, Group issued to holders of Group's Old Common Stock Contingent Value Rights to receive up to an aggregate of 2,665,000 shares (the "CVR Shares") of New Common Stock. In connection with the issuance of the Contingent Value Rights, Group entered into a Contingent Value Rights Distribution Agreement (the "CVR Agreement"), in favor of holders of CVRs thereunder, dated as of the Effective Date.

The CVRs may not be transferred by the holder thereof except in certain limited circumstances. Subject to the terms of the CVR Agreement, holders of CVRs will receive their pro rata share of up to 2,665,000 CVR Shares. A distribution of CVR Shares is required to be made by Group if, as of any determination date (described below), Group's equity value (assuming cash exercise in full on such date of in-the-money warrants and options of Group) divided by the sum of the number of shares of New Common Stock then issued and outstanding plus the number of shares of New Common Stock underlying warrants, options and similar securities of Group (other than CVRs) that are then in-the-money exceeds \$35.95. The aggregate number of such shares of New Common Stock is referred to as the "Applicable Shares;" the price per share of \$35.95, subject to adjustment as described below, is referred to as the "CVR Strike Price;" and the per share amount of any such excess over the CVR Strike Price is referred to as the "Excess Equity Value Per Share." If such a distribution is required, the number of CVR Shares to be distributed by Group equals the product of Excess Equity Value Per Share multiplied by the number of Applicable Shares divided by the CVR Strike Price. Such product of Excess Equity Value Per Share is referred to as the "Excess Equity Value."

Group will determine if and to the extent a distribution of CVR Shares is required on January 1 and July 1 of each year, commencing on the first such date (but in no event later than July 1, 2013) on which data is available to confirm that Group's adjusted EBITDA for the immediately preceding four fiscal quarters is equal to at least \$100 million, and upon a change of control of Group. Distributions of CVR Shares (if any) will be made within 45 calendar days of a determination by Group that a distribution is required.

Notwithstanding the foregoing, no distribution of CVR Shares is required to be made by Group unless Excess Equity Value exceeds \$1 million as of any determination date.

The number of CVR Shares and the CVR Strike Price will be adjusted from time to time in connection with any stock dividend or distribution, or subdivision, split, combination, reclassification or recapitalization of the New Common Stock. In addition, if Group distributes to holders of New Common Stock any of its assets (including but not limited to cash), securities or rights to purchase securities of Group (other than any regular cash dividend not to exceed in any fiscal year 45% of the consolidated net income of Group for the immediately preceding fiscal year), then the number of CVR Shares will be increased and the CVR Strike Price will be decreased, in each case pursuant to the terms of the CVR Agreement. Additionally, in case of any reclassification, merger, consolidation, capital reorganization or other change in the capital stock of Group (other than in connection with a change of control) in which all or substantially all of the outstanding shares of New Common Stock are converted into or exchanged for stock, other securities or other property, Group shall make appropriate provision so that the holders of Contingent Value Rights shall thereafter be entitled to receive, at such time such holder would have otherwise been entitled to receive a distribution under the CVR Agreement, the kind and amount of stock and other securities and property having a value substantially equivalent to the value of New Common Stock that the holders of Contingent Value Rights would have been entitled to receive in connection with a distribution of CVR Shares immediately prior to such reclassification, merger, consolidation, reorganization or other change in the capital stock of Group at a CVR Strike Price that, in each case, is reasonably determined by the board of directors of Group after consultation with an independent valuation advisor to preserve, to the extent practicable, the intrinsic value of such CVR immediately prior to such event.

The Contingent Value Rights will expire and the CVR Agreement will terminate upon the earliest to occur of: (1) the date upon which no further CVR Shares are available for distribution, (2) the consummation of a change of control (subject to any potential distribution of CVR Shares as a result thereof), and (3) July 1, 2019.

Due to the nature of CVR, the Company accounted for the instrument in accordance with ASC No. 815, "Derivatives and Hedging", as well as related interpretations of these standards. The Company recorded a liability of \$2.6 million in other liabilities as part of fresh-start accounting, and will adjust the liability to its estimated fair value. The change in value is reflected in our Statements of Operations as other income (expense). The Company recognized \$2.8 million of expense during the six months ended December 31, 2009 (the "Successor Period").

3. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

Principles of Consolidation—The consolidated financial statements include the Company's accounts, its wholly-owned subsidiaries and all other subsidiaries over which the Company exerts control. The Company owns 45.6% of Globility Communications Corporations (GCC) through direct and indirect ownership structures. The results of GCC and its subsidiary are consolidated with the Company's results based on guidance from the Financial Accounting Standards Board ("FASB") Accounting Standards Codification ("ASC") No. 810 "Consolidation." All intercompany profits, transactions and balances have been eliminated in consolidation.

In December 2007, the FASB issued ASC 810 (formerly SFAS No. 160), *Noncontrolling Interests in Consolidated Financial Statements, an Amendment of ARB 51* ("ASC 810"). ASC 810 amends ARB 51 to establish new standards that will govern the accounting and reporting of noncontrolling interests in partially owned consolidated subsidiaries and the loss of control of subsidiaries. Also, ASC 810 requires that: (1) noncontrolling

interest, previously referred to as minority interest, be reported as part of equity in the consolidated financial statements; (2) losses be allocated to the noncontrolling interest even when such allocation might result in a deficit balance, reducing the losses attributed to the controlling interest; (3) changes in ownership interests be treated as equity transactions if control is maintained; and, (4) upon a loss of control, any gain or loss on the interest sold be recognized in earnings. ASC 810 is effective for financial statements issued for fiscal years, beginning on or after December 15, 2008, except for the presentation and disclosure requirements, which will be applied retrospectively. Implementation of the standard did not have a material effect on the Company's results from operations or financial position.

Discontinued Operations—In the first quarter of 2009, the Company sold certain assets of its Japan retail operations. The Company intended and had the authority to sell certain assets of its Japan retail operations since the fourth quarter of 2008, and therefore, reported this unit as discontinued operations. In the second quarter 2008, the Company intended and had the authority to sell certain assets of its German retail operations, and therefore, reported this unit as discontinued operations. In March 2008, the Company sold its minority equity interest in Bekkoame Internet, Inc. ("Bekko"). The Company used the equity method of accounting for its investment in Bekko.

Reorganization Costs—In accordance with Financial Accounting Standard Board ("FASB") Accounting Standards Codification ("ASC") No. 852, "Reorganizations," for periods including and subsequent to the filing of the Chapter 11 petition through the bankruptcy emergence date of July 1, 2009, all revenues, expenses, realized gains and losses, and provisions for losses that result from the reorganization were reported separately as reorganization items, net, in the Consolidated Statements of Operations. Net cash used for reorganization items was disclosed separately in the Consolidated Statements of Cash Flows.

Fresh Start—As of July 1, 2009, the Company adopted fresh-start accounting in accordance with ASC No. 852. The adoption of fresh-start accounting resulted in the Company becoming a new entity for financial reporting purposes. Accordingly, the financial statements on or prior to July 1, 2009 are not comparable with the financial statements for periods after July 1, 2009. The consolidated statements of operations, comprehensive income (loss) and any references to "Successor" or "Successor Company" for the six months ended December 31, 2009, show the operations of the reorganized Company from and including July 1, 2009, the Effective Date, through December 31, 2009. References to "Predecessor" or "Predecessor Company" refer to the operations of the Company prior to July 1, 2009, except for Predecessor's July 1, 2009 statements of operations and comprehensive income (loss), which reflect only the effect of the plan adjustments and fresh-start accounting as of such date and do not reflect any operating results. See "Note 4—Fresh Start Accounting," for further details.

Revenue Recognition and Deferred Revenue—Net revenue is derived from carrying a mix of business, residential and carrier long distance traffic, data and Internet traffic, and also from the provision of hosting, local and wireless services.

For certain voice services, net revenue is earned based on the number of minutes during a call and is recorded upon completion of a call. Revenue for a period is calculated from information received through the Company's network switches. Customized software has been designed to track the information from the switch and analyze the call detail records against stored detailed information about revenue rates. This software provides the Company the ability to do a timely and accurate analysis of revenue earned in a period. Separate prepaid services software is used to track additional information related to prepaid service usage such as activation date, monthly usage amounts and expiration date. Revenue on these prepaid services is recognized as service is provided until expiration, when all unused minutes, which are no longer available to the customers, are recognized as revenue.

Net revenue is also earned on a fixed monthly fee basis for unlimited local and long distance voice plans and for the provision of data/Internet services (including retail VOIP), hosting, and collocation. Data/Internet services

include monthly fees collected for the provision of dedicated and dial-up access at various speeds and bandwidths. These fees are recognized as access is provided on a monthly basis. Additionally, service activation and installation fees are deferred and amortized over the longer of the average customer life or the contract term. The Company records payments received in advance for services and services to be provided under contractual agreements, such as Internet broadband, dial-up access, hosting, and collocation, as deferred revenue until such related services are provided.

A portion of revenue, representing less than 1% of total revenue, is earned from the sale of wireless handsets and VOIP routers. The Company applies the provisions of ASC 605-25, "Revenue Recognition—Multiple Element Arrangements," which provides guidance on when and how an arrangement involving multiple deliverables should be divided into separate units of accounting. The Company has concluded to account for the sale of wireless handsets and VOIP routers and the related cost of handset and router revenues as a separate unit of accounting when title to the handset or router passes to the customer. Revenue recognized is the portion of the activation fees allocated to the router or handset unit of accounting in the statement of operations when title to the router or handset passes to the customer. The Company defers the portion of the activation fees allocated to the service unit of accounting, and recognizes such deferred fees on a straight-line basis over the contract life in the statement of operations.

Net revenue represents gross revenue, net of allowance for doubtful accounts receivable, service credits and service adjustments.

Presentation of Taxes Collected—The Company reports any tax assessed by a governmental authority that is directly imposed on a revenue-producing transaction between the Company and a customer (including sales, use, value-added and some excise taxes) on a net basis (excluded from revenues).

Cost of Revenue—Cost of revenue includes network costs that consist of access, transport and termination costs. A portion of cost of revenue, representing less than 1% of total cost of revenue, consists of the product cost of wireless handsets and VOIP routers. The majority of the Company's cost of revenue is variable, primarily based upon minutes of use, with transmission and termination costs being the most significant expense. Such costs are recognized when incurred in connection with the provision of telecommunications services.

Foreign Currency Transaction—Foreign currency transactions are transactions denominated in a currency other than a subsidiary's functional currency. A change in the exchange rates between a subsidiary's functional currency and the currency in which a transaction is denominated increases or decreases the expected amount of functional currency cash flows upon settlement of the transaction. That increase or decrease in expected functional currency cash flows is reported by the Company as a foreign currency transaction gain (loss). The primary component of the Company's foreign currency transaction gain (loss) is due to written agreements in place with certain subsidiaries in foreign countries regarding intercompany loans. The Company anticipates repayment of these loans in the foreseeable future, and recognizes the realized and unrealized gains or losses on these transactions that result from foreign currency changes in the period in which they occur as foreign currency transaction gain (loss).

Income Taxes—The Company recognizes income tax expense for financial reporting purposes following the asset and liability approach for computing deferred income taxes. Under this method, the deferred tax assets and liabilities are determined based on the difference between financial reporting and tax bases of assets and liabilities based on enacted tax rates. Deferred tax assets are reduced by a valuation allowance when, in the opinion of management, it is more likely than not that some portion or all of the deferred tax assets will not be realized. On January 1, 2007, the Company adopted the uncertain tax position-related provisions of ASC No 740 "Income Taxes" (formerly FIN No. 48, "Accounting for Uncertainty in Income Taxes"). These provisions clarify the accounting for uncertainty in income taxes recognized in the financial statements in accordance with ASC No. 740.

Foreign Currency Translation—The assets and liabilities of the Company's foreign subsidiaries are translated at the exchange rates in effect on the reporting date. The net effect of such translation gains and losses are reflected within accumulated other comprehensive income (loss) in the stockholders' equity (deficit) section of the balance sheet. Income and expenses are translated at the average exchange rate during the period.

Cash and Cash Equivalents—Cash and cash equivalents are comprised principally of amounts in money market accounts, operating accounts, certificates of deposit, and overnight repurchase agreements with original maturities of three months or less.

Restricted Cash—Restricted cash consists of bank guarantees and certificates of deposit utilized to support letters of credit and contractual obligations.

Advertising Costs—In accordance with ASC No. 720-35, "Other Expenses—Advertising," costs for advertising are expensed as incurred. Advertising expense for the six months ended December 31, 2009, and July 1, 2009 and for the years ended December 31, 2008 and 2007 was \$8.8 million, \$5.6 million, \$18.0 million and \$21.5 million, respectively.

Property and Equipment—Property and equipment is recorded at cost less accumulated depreciation, which is provided on the straight-line method over the estimated useful lives of the assets. Cost includes major expenditures for improvements and replacements which extend useful lives or increase capacity of the assets as well as expenditures necessary to place assets into readiness for use. Expenditures for maintenance and repairs are expensed as incurred. The estimated useful lives of property and equipment are as follows: network equipment—5 to 8 years, fiber optic and submarine cable—8 to 25 years, furniture and equipment—5 years, leasehold improvements and leased equipment—shorter of lease or useful life. Costs for internal use software that are incurred in the preliminary project stage and in the post-implementation stage are expensed as incurred. Costs incurred during the application development stage are capitalized and amortized over the estimated useful life of the software.

Fiber Optic and Submarine Cable Arrangements—The Company obtains capacity on certain fiber optic and submarine cables under three types of arrangements. The Indefeasible Right of Use ("IRU") basis provides the Company the right to use a cable for the estimated economic life of the asset according to the terms of the IRU agreement with most of the rights and duties of ownership. The Minimum Assignable Ownership Units ("MAOU") basis provides the Company an ownership interest in the fiber optic cable with certain rights to control and to manage the facility. The Company accounts for both IRU and MAOU agreements under network equipment and depreciates the recorded asset over the term of the agreement which is generally 25 years. The Company also enters into shorter-term arrangements with other carriers which provide the Company the right to use capacity on a cable but without any rights and duties of ownership. Under these shorter-term arrangements, the costs are expensed in the period the services are provided.

Goodwill and Other Intangible Assets—Under ASC No. 350 ("ASC 350"), "Intangibles—Goodwill and Other," goodwill and indefinite lived intangible assets are not amortized but are reviewed annually for impairment, or more frequently, if impairment indicators arise. Intangible assets that have finite lives are amortized over their useful lives and are subject to the provisions of ASC No. 360, "Property, Plant and Equipment."

Goodwill impairment is tested annually (October 1 for the Company) using a two-step process that begins with an estimation of the fair value of each reporting unit. The first step is a screen for potential impairment by comparing the fair value of a reporting unit with its carrying amount. The second step measures the amount of impairment loss, if any, by comparing the implied fair value of the reporting unit goodwill with its carrying amount. The implied fair value of goodwill shall be determined in the same manner as the amount of goodwill recognized in a business combination is determined, that is through an allocation of the fair value of a reporting unit to all of the assets and liabilities of that unit as if the reporting unit had been acquired in a business combination. If the carrying amount of the reporting unit goodwill exceeds the implied fair value of that goodwill, an impairment loss shall be recognized in an amount equal to that excess.

Our reporting units are the same as our operating segments as each segment's components have been aggregated and deemed a single reporting unit because they have similar economic characteristics. Each component is similar in that they each provide telecommunications services for which all of the resources and costs are drawn from the same pool, and are evaluated using the same business factors by management.

Intangible assets not subject to amortization consist of trade names. Such indefinite lived intangible assets are tested for impairment annually, or more frequently if events or changes in circumstances indicate that the asset might be impaired. The impairment test shall consist of a comparison of the fair value of an intangible asset with its carrying amount. If the carrying amount of the intangible asset exceeds its fair value an impairment loss shall be recognized in an amount equal to the excess.

Intangible assets subject to amortization consist of certain trade names and customer relationships. These finite lived intangible assets are amortized based on their useful lives. Such assets are subject to the impairment provisions of ASC No. 360, wherein impairment is recognized and measured only if there are events and circumstances that indicate that the carrying amount may not be recoverable. The carrying amount is not recoverable if it exceeds the sum of the undiscounted cash flows expected to result from the use of the asset group. An impairment loss is recorded if after determining that it is not recoverable, the carrying amount exceeds the fair value of the asset.

Valuation of Long-Lived Assets—The Company reviews long-lived assets whenever events or changes indicate that the carrying amount of an asset may not be recoverable. In making such evaluations, the Company compares the expected undiscounted future cash flows to the carrying amount of the assets. If the total of the expected undiscounted future cash flows is less than the carrying amount of the assets, the Company is required to make estimates of the fair value of the long-lived assets in order to calculate the impairment loss equal to the difference between the fair value and carrying value of the assets.

The Company makes significant assumptions and estimates in this process regarding matters that are inherently uncertain, such as determining asset groups and estimating future cash flows, remaining useful lives, discount rates and growth rates. The resulting undiscounted cash flows are projected over an extended period of time, which subjects those assumptions and estimates to an even larger degree of uncertainty. While the Company believes that its estimates are reasonable, different assumptions could materially affect the valuation of the long-lived assets. The Company derives future cash flow estimates from its historical experience and its internal business plans, which include consideration of industry trends, competitive actions, technology changes, regulatory actions, available financial resources for marketing and capital expenditures and changes in its underlying cost structure.

The Company has concluded that it has one asset group; the network. This is due to the nature of its telecommunications network which utilizes all of the points of presence (POPs), switches, cables and various other components throughout the network to form seamlessly the telecommunications gateway over which its products and services are carried for any given customer's phone call or data or Internet transmission. Furthermore, outflows to many of the external network providers are not separately assignable to revenue inflows for any phone call or service plan.

The Company makes assumptions about the remaining useful life of its long-lived assets. The assumptions are based on the average life of its historical capital asset additions, its historical asset purchase trend and that its primary assets, its network switches, have an 8-year life. Because of the nature of its industry, the Company also assumes that the technology changes in the industry render all equipment obsolete with no salvage value after their useful lives. In certain circumstances in which the underlying assets could be leased for an additional period of time, the Company includes such estimated cash flows in its estimate.

The estimate of the appropriate discount rate to be used to apply the present value technique in determining fair value was the Company's weighted average cost of capital which is based on the effective rate of its long-term debt obligations at the current market values as well as the current volatility and trading value of the Company's common stock.

Deferred Financing Costs—Deferred financing costs incurred in connection with financing arrangements are reflected within other assets and are amortized over the life of the respective financing arrangements using the effective interest method. As the Company completes debt transactions, corresponding amounts of deferred financing costs are written-off in determining the gain or loss on early extinguishment or restructuring of debt.

Derivative Instruments—In March 2007, the Company entered into a Senior Secured Credit Agreement ("Canadian Credit Facility") with a financial institution, to refinance a Canadian credit facility. The Canadian Credit Facility provided for a \$35.0 million non-amortizing loan bearing interest at a rate of LIBOR plus 425 basis points and maturing in 2012. In October 2007, the Company entered into a cross-currency principal and interest rate swap agreement, which fixed the interest rate at 9.21% starting from October 31, 2007. The Canadian Credit Facility required that the Company, at all times that the loan amounts were outstanding, maintain a hedging agreement to hedge the full amount of its currency rate exposures with respect to the aggregate principal amount outstanding under this Canadian Credit Facility at any time. The cross-currency principal and interest rate swap agreement's counter party was Lehman Brothers Special Financing, Inc. ("Lehman SFI"). Lehman SFI entered into bankruptcy in early October 2008 following its ultimate parent entering bankruptcy in mid-September 2008. The Company had an outstanding cross-currency principal and interest rate agreement with Lehman Brothers SFI, who entered bankruptcy in October 2008 and ceased performing on the agreement. The Company estimated the value of the cross-currency principal and interest rate agreement to be zero, requiring a write-off of \$1.2 million in the third quarter of 2008. Despite the counterparty to the swap agreement entering bankruptcy in October 2008, management believed no breach or event of default had occurred in relation to the Canadian Credit Facility. As of March 10, 2009, under the Waiver and Amendment Agreement described below in "Note 7—Long-Term Obligations," the Lenders under the Canadian Credit Facility waived any such possible breach or event of default. The Company has estimated the fair value of the outstanding swap agreement with Lehman Brothers Special Financing, Inc. to be zero at December 31, 2009, 2008 and December 31, 2007.

Stock-Based Compensation—The Company accounts for stock-based compensation under ASC No. 718, "Stock Compensation", (former SFAS No. 123(R), "Share-Based Payments"), which addresses the accounting for stock-based payment transactions whereby an entity receives employee services in exchange for equity instruments, including stock options and restricted stock units. ASC No. 718 generally requires that stock-based compensation be accounted for using a fair-value based method. The Company elected the modified prospective transition method which requires that stock-based compensation expense be recorded for all new and unvested stock options that are ultimately expected to vest as the requisite service is rendered beginning on January 1, 2006. The Company issues new shares of common stock upon the exercise of stock options.

In November 2005, the FASB issued FASB Staff Position No. FAS 123R-3, "Transition Election Related to Accounting for Tax Effects of Share-Based Payment Awards." The Company elected to adopt the alternative transition method provided in the FASB Staff Position for calculating the tax effects of share-based compensation. The alternative transition method includes simplified methods to determine the beginning balance of the additional paid in capital ("APIC") pool related to the tax effects of share-based compensation and to determine the subsequent impact on the APIC pool and the statement of cash flows of the tax effects of share-based award that were fully vested and outstanding upon the adoption of ASC No. 718.

The Company uses a Black-Scholes option valuation model to determine the fair value of stock-based compensation under ASC No. 718. The Black-Scholes model incorporates various assumptions including the expected term of awards, volatility of stock price, risk-free rates of return and dividend yield. The expected term of an award is no less than the option vesting period and is based on the Company's historical experience. Expected volatility is based upon the historical volatility of the Company's stock price. Because of the short trading history of the Company's new Common Stock, the Company calculates the expected volatility by averaging the historical volatility of the Company's New Common Stock price and the historical volatility of the stock price of a peer group in the telecommunications industry with similar market capitalization. The risk-free interest rate is approximated using rates available on U.S. Treasury securities with a remaining term similar to the option's expected life. The Company uses a dividend yield of zero in the Black-Scholes option valuation model as it does not anticipate paying cash dividends in the foreseeable future.

The weighted average fair value at date of grant for options granted during the six month period ended December 31, 2009 was \$4.00 dollars. Options granted prior to July 1, 2009 were cancelled in conjunction with the cancellation of the Company's Old Common Stock pursuant to the terms of our Plan of Reorganization and we have no continuing obligations with respect to the options issued prior to July 1, 2009. The fair value of each option grant is estimated on the date of grant using the Black-Scholes option-pricing model with the following assumptions:

	Successor	Predecessor	
	2009	2008	2007
Expected dividend yield	0%	0%	0%
Expected stock price volatility	42%	96%	95%
Risk-free interest rate	2%	2.3%	4.6%
Expected option term	4 years	4 years	4 years

Use of Estimates—The preparation of consolidated financial statements in conformity with accounting principles generally accepted in the United States of America requires management to make estimates and assumptions. These estimates and assumptions affect the reported amounts of assets and liabilities and disclosures of contingent assets and liabilities at the date of the consolidated financial statements and the reported amounts of net revenue and expenses during the reporting period. Actual results may differ from these estimates. Significant estimates include allowance for doubtful accounts receivable, accrued interconnection cost disputes, the fair value of embedded derivatives, market assumptions used in estimating the fair values of certain assets and liabilities such as marketable securities and long-term obligations, the calculation used in determining the fair value of the Company's stock options required by ASC No. 718, various tax contingencies, asset impairment write-downs, and fair values of assets and liabilities in connection with fresh-start accounting.

Concentration of Credit Risk—Financial instruments that potentially subject the Company to concentration of credit risk principally consist of trade accounts receivable. The Company performs ongoing credit evaluations of its larger carrier and retail business customers but generally does not require collateral to support customer receivables. The Company maintains its cash with high quality credit institutions, and its cash equivalents are in high quality securities.

Income (*Loss*) *Per Common Share*—Basic income (loss) per common share is computed using the weighted average number of shares of common stock outstanding during the year. Diluted income (loss) per common share is computed using the weighted average number of shares of common stock, adjusted for the dilutive effect of potential common stock and related income. Potential common stock, computed using the treasury stock method or the if-converted method, includes options, restricted stock units, warrants and convertible debt securities.

Reclassification—Certain previous year amounts have been reclassified to conform with current year presentations, as related to the reporting of our discontinued operations.

Newly Adopted Accounting Principles

Effective July 1, 2009, the Company adopted ASC No. 105, "Generally Accepted Accounting Principles." ASC No. 105 establishes the FASB Accounting Standards Codification as the source of accounting principles and the framework for selecting the principles to be used in the preparation of financial statements. The adoption did not have a material impact on its results of operations, financial position and cash flows.

Effective April 1, 2009, the Company adopted ASC No. 855, "Subsequent Events." ASC No. 855 establishes principles and requirements for evaluating and reporting subsequent events and distinguishes which subsequent events should be recognized in the financial statements versus which subsequent events should be disclosed in the financial statements. The adoption did not have a material effect on the Company's results of operations, financial position or cash flows.

Effective January 1, 2009, the Company adopted ASC No. 810, "Consolidation." Reconciliations at the beginning and the end of the period of the total equity, equity attributable to the Company and equity attributable to the noncontrolling interest for Successor's six months ended December 31, 2009 and Predecessor's six months ended July 1, 2009 and year ended December 2008 are included in the Company's consolidated statement of stockholders' equity (deficit).

Effective January 1, 2009, the Company adopted ASC No. 470-20, "Debt with Conversion and Other Options." ASC No. 470-20 requires issuers of convertible debt securities to separate securities into a debt component and an equity component, resulting in the debt component being recorded at fair value without consideration given to the conversion feature. Issuance costs are also allocated between the debt and equity components. ASC No. 470-20 requires that convertible debt within its scope reflect a company's nonconvertible debt borrowing rate when interest expense is recognized. The adoption did not have a material effect on the Company's results of operations, financial position or cash flows.

Effective January 1, 2009, the Company adopted ASC No. 805, "Business Combinations." ASC No. 805 establishes principles and requirements for how the acquirer of a business recognizes and measures in its financial statements the identifiable assets acquired, the liabilities assumed, and any noncontrolling interest in the acquiree. The statement also provides guidance for recognizing and measuring the goodwill acquired in the business combination and determines what information to disclose to enable users of the financial statement to evaluate the nature and financial effects of the business combination. The Company applied this standard on July 1, 2009 to the Fresh Start Balance Sheet, (See Note 4 "Fresh Start Accounting," for further detail).

In February 2008, the FASB issued ASC No. 820, "Fair Value Measurements." The provisions of ASC No. 820 provide guidance for, among other things, the definition of fair value and the methods used to measure fair value, were adopted January 1, 2008 for financial instruments. The provisions adopted in 2008 did not have a material impact on the Company's financial statements. ASC No. 820 delayed the effective date for all nonrecurring fair value measurements of nonfinancial assets and liabilities (except for those that are recognized or disclosed at fair value in the financial statements on a recurring basis) until financial years beginning after November 15, 2008, and changed the scope of ASC No. 820. On January 1, 2009, the Company adopted the remaining provisions of ASC No. 820 for nonrecurring fair value measurements of nonfinancial assets and liabilities. The provisions adopted in the first quarter 2009 were applied by the Company in the fair value measurements as part of the fresh start accounting adoption upon emergence from bankruptcy on July 1, 2009 and the goodwill impairment analysis performed as of October 1, 2009.

The valuation techniques required by ASC No. 820 are based upon observable and unobservable inputs. Observable inputs reflect market data obtained from independent sources, while unobservable inputs reflect our market assumptions. These two types of inputs create the following fair value hierarchy:

- Level 1: Quoted prices for identical instruments in active markets.
- Level 2: Quoted prices for similar instruments in active markets; quoted prices for identical or similar instruments in markets that are not active; and model-derived valuations whose inputs are observable or whose significant value drivers are observable.
 - Level 3: Significant inputs to the valuation model are unobservable.

The Company had a cross-currency principal and interest rate agreement with Lehman Brothers Special Financing, Inc., who entered bankruptcy in October 2008 and ceased performing on the agreement. The Company estimated the value to be zero, requiring a write-off of \$1.2 million in 2008, and moved the instrument from Level 2 to Level 3 because the counter party's credit risk is not observable.

In March 2008, the FASB issued ASC No. 815, "Derivatives and Hedging." ASC No. 815 amends and expands the disclosure requirements with the intent to provide users of financial statements with an enhanced

understanding of the use of derivative instruments, how derivative instruments and related hedged items are accounted for, and how derivative instruments and related hedged items affect an entity's financial statements. ASC No. 815 is effective for financial statements issued for fiscal years interim periods beginning after November 15, 2008. In addition the Company accounted for the Contingent Value Rights Distribution Agreement in accordance with ASC No. 815, "Derivatives and Hedging" as well as related interpretations of this standard. The change in value of the Contingent Value Rights Distribution Agreement is reflected in our Statements of Operations as other income (expense), (see "Note 2—Emergence from Voluntary Reorganization Under Chapter 11 of the Bankruptcy Code," for further detail). The adoption of ASC No. 815 on January 1, 2009 did not have a material impact on the Company's results of operations, financial position and cash flows.

New Accounting Pronouncements

In June 2009, the FASB issued ASC No. 810-10, "Consolidation," which amends the definition of the primary beneficiary of a variable interest entity and will require the Company to assess each reporting period if any of the Company's variable interests give it a controlling financial interest in the applicable variable interest entity. The provisions of ASC No. 810-10 will become effective for financial statements issued for fiscal years and interim periods begin after November 15, 2009. The Company is currently evaluating the impact the provisions of ASC No. 810-10 on its financial statements.

In May 2008, the FASB issued ASC No. 470-20, "Debt with Conversion and Other Options." ASC No. 470-20 requires issuers of convertible debt securities to separate securities into a debt component and an equity component, resulting in the debt component being recorded at fair value without consideration given to the conversion feature. Issuance costs are also allocated between the debt and equity components. ASC No. 470-20 requires that convertible debt within its scope reflect a company's nonconvertible debt borrowing rate when interest expense is recognized. ASC No. 470-20 is effective for financial statements issued for fiscal years and interim periods beginning after December 15, 2008. The adoption of ASC No. 470-20 did not have a material impact on the Company's results of operations, financial position and cash flows.

In April 2008, the FASB issued ASC No. 852-10, "Reorganizations." ASC No. 852-10 nullifies certain requirements regarding changes in accounting principles that will be applicable to the financial statements of an entity emerging from bankruptcy. Any changes in accounting principles required within the twelve months following the implementation of fresh start accounting by such an entity are no longer required to be adopted at the time fresh start accounting is implemented. Entities emerging from bankruptcy that implement fresh start accounting should only follow accounting standards in effect at the date fresh start accounting is implemented, including any standards eligible for early adoption.

4. FRESH START ACCOUNTING

On July 1, 2009, the Company adopted fresh-start accounting in accordance with ASC No. 852, "Reorganizations." Fresh-start accounting results in the Company becoming a new entity for financial reporting purposes. Accordingly, the Successor Company's consolidated financial statements are not comparable to consolidated financial statements of the Predecessor Company.

Under ASC No. 852, the Successor Company must determine a value to be assigned to the equity of the emerging company as of the date of adoption of fresh-start accounting. To facilitate this calculation the Company first determined the enterprise value of the Successor Company. The valuation methods included (i) a discounted cash flow analysis, considering a range of the weighted average cost of capital between 14.0% and 16.0% and multiples of projected earnings of between 4.5 and 5.0 times for its terminal value, and (ii) a market multiples analysis. This analysis resulted in an estimated enterprise value of between \$320 million and \$360 million, and with the midpoint of \$340 million chosen for purposes of applying fresh-start accounting.

The estimated enterprise value, and corresponding equity value, is highly dependent upon achieving the future financial results set forth in the financial projections included in the Company's Plan, as filed with the Bankruptcy Court. These projections were limited by the information available to the Company as of the date of the preparation of the projections and reflected numerous assumptions concerning anticipated future performance and prevailing and anticipated market and economic conditions that were and continue to be beyond the Company's control and that may not materialize. Projections are inherently subject to uncertainties and to a wide variety of significant business, economic and competitive risks. Therefore variations from the projections may be material.

Fresh-start accounting reflects the value of the Company as determined in the confirmed Plan. Under fresh-start accounting, the Company's asset values are remeasured and allocated in conformity with ASC No. 805, "Business Combinations." The excess of reorganization value over the fair value of tangible and identifiable intangible assets is recorded as goodwill in the accompanying consolidated balance sheet. Fresh-start accounting also requires that all liabilities, other than deferred taxes and pension and other postretirement benefit obligations, should be stated at fair value.

Estimates of fair value included in the Successor Company financial statements, in conformity with ASC No. 820, represent the Company's best estimates and valuations developed with the assistance of independent appraisers and, where the foregoing have not yet been completed or are not available, industry data and trends and by reference to relevant market rates and transactions. The foregoing estimates and assumptions are inherently subject to significant uncertainties and contingencies beyond the control of the Company. Accordingly, the Company cannot provide assurance that the estimates, assumptions, and values reflected in the valuations will be realized, and actual results could vary materially. In accordance with ASC No. 805, the allocation of the reorganization value is subject to additional adjustment until the Company has completed its analysis, but not to exceed one year after emergence from bankruptcy, to provide the Company with the time to complete the valuation of its assets and liabilities. The final determination of the fair value of individual assets and liabilities and the final allocation of the reorganization value may differ from the amounts included in the consolidated financial statements and there can be no assurance that such adjustments will not be material.

The following fresh-start Consolidated Condensed Balance Sheet presents the financial effects on the Company of the implementation of the Plan and the adoption of fresh-start accounting. The effect of the consummation of the transactions contemplated in the Plan includes the settlement of liabilities and the issuance of common stock as described in more detail in "Note 2—Emergence from Reorganization Under Chapter 11of the Bankruptcy Code."

The effects of the Plan and fresh-start reporting on the Company's Consolidated Condensed Balance Sheet are as follows:

Agistation Agi
CURRENT ASSETS: Cash and cash equivalents
Cash and cash equivalents \$ 41,461 \$ — \$ 41,461 Accounts receivable Accounts receivable Prepaid expenses and other current assets 16,955 — — 9,38.26 Prepaid expenses and other current assets 16,955 — — — 152,242 RESTRICTED CASH 9,467 — — — 9,467 RODERITY AND EQUIPMENT—Net 117,840 — — 25,947 d, b 61,298 OTHER INTANGIBLE ASSETS—Net 482 — — 184,318 d 184,800 OTHER ASSETS 19,155 — — \$244,024 — \$578,561 CURRENT LIABILITIES AND STOCKHOLDERS' EQUITY (DEFICIT) CURRENT LIABILITIES: Accounts payable \$ 50,890 \$ — \$ — \$ 50,890 Accounts payable \$ 50,890 \$ — \$ — \$ 50,890 Accounts payable \$ 50,890 \$ — \$ — \$ 50,890 Accounts payable \$ 50,890 \$ — \$ — \$ 50,890
Accounts receivable Prepaid expenses and other current assets 163,955 —
Prepaid expenses and other current assets 16,955 — — 16,955 Total current assets 152,242 — — 152,242 RESTRICTED CASH 9,467 — — — 9,242 PROPERTY AND EQUIPMENT—Net 117,840 — — 32,298 d, both 150,138 GOODWILL 35,351 — — 184,318 d, both 26,138 OTHER INTANGIBLE ASSETS—Net 482 — 1,461 d, both 20,616 TOTAL ASSETS 19,155 — 1,461 d, both 20,616 TOTAL ASSETS 5334,537 5 244,022 > \$78,561 LIABILITIES — — 4,61 2,61 3,676 — \$ 2,44,022 — \$ 50,890 Accounts payable \$ 50,890 \$ — \$ — — \$ 50,890 Accrued interconnection costs 33,778 — — — \$ 2,3225 Accrued expenses and other current liabilities 53,982 — — — 1,932
Total current assets
RESTRICTED CASH 9,467 — — 9,467 PROPERTY AND EQUIPMENT—Net 117,840 — 32,298 d 150,138 GOODWILL 35,351 — 184,318 d, b 161,298 OTHER INTANGIBLE ASSETS—Net 482 — 184,318 d, b 20,616 TOTAL ASSETS 19,155 — 1,461 d, b 20,616 TOTAL ASSETS \$334,537 \$— \$24,024 \$78,561 LABILITIES CURRENT LIABILITIES: Account jayable \$50,890 \$— \$— \$5,089 Accruel interconnection costs 38,778 — — \$5,089 Accruel expense and other current liabilities 53,982 — — — \$2,098 Accrued incerte taxes 20,986 — — — 19,992 Accrued interest 19 — — — 15,997 Total current portion of long-term obligations 284,07 91,100 (1,767) 15,997
PROPERTY AND EQUIPMENT—Net 117,840 — 32,298 d 150,138 GOODWILL 35,351 — 25,947 d, h 61,298 OTHER INTANGIBLE ASSETS—Net 482 — 184,318 d 18,018 d 18,016 d 18,018 d 18,018 d 18,018 d 18,018 d 18,018 d 18,016 d 2,018 d 2,018 d 2,029 d 2,029 d 2,029
OTHER INTANGIBLE ASSETS—Net 482 — 184,318 d 184,800 OTHER ASSETS 19,155 — 1,461 d, h 20,616 TOTAL ASSETS \$334,537 \$— \$244,024 \$578,561 LIABILITIES AND STOCKHOLDERS' EQUITY (DEFICIT) S S — \$ \$ \$50,890 Accounts payable \$50,890 \$— \$— \$ \$50,890 Accrued interconnection costs 38,778 — — \$50,890 Accrued interconnection costs 38,778 — — \$50,890 Accrued expenses and other current liabilities 53,982 — — — \$12,322 Accrued interest 20,986 — — — 19 — — — 19,920 Accrued interest 20,986 — — — 19 — — — 19,920 Current portion of long-term obligations 107,097 (91,100) g — — 19,920 LON
OTHER ASSETS 19,155 — 1,461 d, 20,616 TOTAL ASSETS \$ 334,537 \$ — \$ 244,024 \$ 578,561 LIABILITIES AND STOCKHOLDERS' EQUITY (DEFICIT) CURRENT LIABILITIES: Accounts payable \$ 50,890 \$ — \$ 50,890 Accrued interconnection costs 38,778 — — — 38,778 Deferred revenue 12,322 — — — 12,322 Accrued expenses and other current liabilities 53,982 — — — 12,322 Accrued income taxes 20,986 — — — 20,986 Accrued interest 19 — — 19 — — 19,997 Total current liabilities 284,071 (91,00) © (1,767) 191,209 LONG-TERM OBLIGATIONS 25,740 214,572 e,g — 2,937 OTHER LIABILITIES SUBJECT TO COMPROMISE 309,814 126,029 55,395 491,238 LIABILITIES SUBJECT TO COMPROM
TOTAL ASSETS
CURRENT LIABILITIES S
CURRENT LIABILITIES:
Accounts payable \$ 50,890 \$ — \$ — \$ 50,890 Accrued interconnection costs 38,778 — — 38,778 Deferred revenue 12,322 — — 12,322 Accrued expenses and other current liabilities 53,982 — — (1,767) d 52,215 Accrued income taxes 20,986 — — — 20,986 Accrued interest 19 — — — 19 Current portion of long-term obligations 107,097 (91,100) g — — 15,997 Total current liabilities 284,074 (91,100) g — — 19,207 LONG-TERM OBLIGATIONS 25,740 214,572 e, g — — 240,312 OTHER LIABILITIES — 2,557 b 57,162 h 59,719 Total liabilities not subject to compromise 309,814 126,029 55,395 491,238 COMMITMENTS AND CONTINGENCIES — — —
Accrued interconnection costs 38,778 — — 38,778 Deferred revenue 12,322 — — 12,322 Accrued expenses and other current liabilities 53,982 — (1,767) d 52,215 Accrued income taxes 20,986 — — 20,986 Accrued interest 19 — — — 19 Current portion of long-term obligations 107,097 (91,100) g — — 15,997 Total current liabilities 284,074 (91,100) (1,767) 191,207 LONG-TERM OBLIGATIONS 25,740 214,572 e, g — 240,312 OTHER LIABILITIES — 2,557 b 57,162 h 59,719 Total liabilities not subject to compromise 309,814 126,029 55,395 491,238 LIABILITIES SUBJECT TO COMPROMISE 451,050 (451,050) a — — — — — — — — — — — — — — — — — —
Deferred revenue 12,322
Accrued expenses and other current liabilities 53,982 — (1,767) d 52,215 Accrued income taxes 20,986 — — 20,986 Accrued interest 19 — — 19 Current portion of long-term obligations 107,097 (91,100) g — 15,997 Total current liabilities 284,074 (91,100) (1,767) 191,207 LONG-TERM OBLIGATIONS 25,740 214,572 e, g — 240,312 OTHER LIABILITIES — 2,557 b 57,162 h 59,719 Total liabilities not subject to compromise 309,814 126,029 55,395 491,238 LIABILITIES SUBJECT TO COMPROMISE 451,050 (451,050) a — — Total Liabilities 760,864 (325,021) 55,395 491,238 COMMITMENTS AND CONTINGENCIES STOCKHOLDERS' EQUITY (DEFICIT): Primus Telecommuncations Group, Incorporated Stockholders' Equity (Deficit): Predecessor Common stock, \$0.01 par value—300,000,000 shares
Accrued income taxes 20,986 20,986 Accrued interest 19 19 19 19 19 19
Accrued interest 19
Current portion of long-term obligations 107,097 (91,100) g — 15,997 Total current liabilities 284,074 (91,100) (1,767) 191,207 LONG-TERM OBLIGATIONS 25,740 214,572 e, g — 240,312 OTHER LIABILITIES — 2,557 b 57,162 h 59,719 Total liabilities not subject to compromise 309,814 126,029 55,395 491,238 LIABILITIES SUBJECT TO COMPROMISE 451,050 (451,050) a — — Total Liabilities 760,864 (325,021) 55,395 491,238 COMMITMENTS AND CONTINGENCIES STOCKHOLDERS' EQUITY (DEFICIT): Primus Telecommuncations Group, Incorporated Stockholders' Equity (Deficit): Predecessor Common stock, \$0.01 par value—300,000,000 shares
Total current liabilities
LONG-TERM OBLIGATIONS 25,740 214,572 e, g — 240,312 OTHER LIABILITIES — 2,557 b 57,162 h 59,719 Total liabilities not subject to compromise 309,814 126,029 55,395 491,238 LIABILITIES SUBJECT TO COMPROMISE 451,050 (451,050) a — — Total Liabilities 760,864 (325,021) 55,395 491,238 COMMITMENTS AND CONTINGENCIES STOCKHOLDERS' EQUITY (DEFICIT): Primus Telecommuncations Group, Incorporated Stockholders' Equity (Deficit): Predecessor Common stock, \$0.01 par value—300,000,000 shares
OTHER LIABILITIES — 2,557 b 57,162 h 59,719 Total liabilities not subject to compromise 309,814 126,029 55,395 491,238 LIABILITIES SUBJECT TO COMPROMISE 451,050 (451,050) a — — Total Liabilities 760,864 (325,021) 55,395 491,238 COMMITMENTS AND CONTINGENCIES STOCKHOLDERS' EQUITY (DEFICIT): Primus Telecommuncations Group, Incorporated Stockholders' Equity (Deficit): Predecessor Common stock, \$0.01 par value—300,000,000 shares
Total liabilities not subject to compromise LIABILITIES SUBJECT TO COMPROMISE 451,050 451,050 451,050) a — — Total Liabilities 760,864 COMMITMENTS AND CONTINGENCIES STOCKHOLDERS' EQUITY (DEFICIT): Primus Telecommuncations Group, Incorporated Stockholders' Equity (Deficit): Predecessor Common stock, \$0.01 par value—300,000,000 shares
LIABILITIES SUBJECT TO COMPROMISE Total Liabilities 760,864 (325,021) 55,395 491,238 COMMITMENTS AND CONTINGENCIES STOCKHOLDERS' EQUITY (DEFICIT): Primus Telecommuncations Group, Incorporated Stockholders' Equity (Deficit): Predecessor Common stock, \$0.01 par value—300,000,000 shares
Total Liabilities 760,864 (325,021) 55,395 491,238 COMMITMENTS AND CONTINGENCIES STOCKHOLDERS' EQUITY (DEFICIT): Primus Telecommuncations Group, Incorporated Stockholders' Equity (Deficit): Predecessor Common stock, \$0.01 par value—300,000,000 shares
COMMITMENTS AND CONTINGENCIES STOCKHOLDERS' EQUITY (DEFICIT): Primus Telecommuncations Group, Incorporated Stockholders' Equity (Deficit): Predecessor Common stock, \$0.01 par value—300,000,000 shares
STOCKHOLDERS' EQUITY (DEFICIT): Primus Telecommuncations Group, Incorporated Stockholders' Equity (Deficit): Predecessor Common stock, \$0.01 par value—300,000,000 shares
Primus Telecommuncations Group, Incorporated Stockholders' Equity (Deficit): Predecessor Common stock, \$0.01 par value—300,000,000 shares
Primus Telecommuncations Group, Incorporated Stockholders' Equity (Deficit): Predecessor Common stock, \$0.01 par value—300,000,000 shares
(Deficit): Predecessor Common stock, \$0.01 par value—300,000,000 shares
Predecessor Common stock, \$0.01 par value—300,000,000 shares
authorized; 142,695,390 shares issued and outstanding 1,427 (1,427) c — —
Successor Common stock, \$0.001 par value—80,000,000 shares
authorized; 9,600,000 shares issued or outstanding — 10 a — 10
Predecessor Additional paid-in capital 718,983 (1,129) c, b (717,854) f —
Successor Additional paid-in capital — 84,382 a — 84,382
Accumulated income (deficit) (1,060,452) 243,185 a 817,267 d, f —
Accumulated other comprehensive income (loss) (89,216) — 89,216 f —
Total Primus Telecommunications Group, Incorporated stockholders' income (deficit) (429,258) 325,021 188,629 84,392
Noncontrolling interest 2,931 — 2,931
Total stockholders' income (deficit) (426,327) 325,021 188,629 87,323
TOTAL LIABILITIES AND STOCKHOLDERS' EQUITY (DEFICIT) \$ 334,537 \$ — \$ 244,024 \$ 578,561

Notes to Plan of Reorganization and fresh-start accounting adjustments:

- (a) This adjustment reflects the discharge of \$451.1 million of liabilities subject to compromise (see "Liabilities Subject to Compromise" below), of which includes \$123.5 million Senior Subordinated Secured Notes reclassed to long-term obligations, in accordance with the terms of the Plan and the issuance of 4.8 million shares of Successor Company common stock to the holders of each of the Senior Subordinated Secured Notes and the Holding Senior Notes.
- (b) To record the issuance of Contingent Value Rights to the holders of the Old Common Stock.
- (c) To record the cancellation of the Old Common Stock.
- (d) To record assets and liabilities at their estimated fair values per fresh-start accounting. These amounts include adjustments to the estimated fair values from what was originally reported in the quarter ending September 30, 2009.
- (e) To reclass Term Loan from current portion of long-term obligations to long-term obligations and record the issuance of the Senior Subordinated Secured Notes.
- (f) To reset additional paid-in capital, accumulated other comprehensive loss and accumulated deficit to zero.
- (g) To reclass long-term portion of the Term Loan to long-term obligations.
- (h) To record the deferred tax attributes related to fresh-start accounting.

In the fourth quarter of 2009, the Company further adjusted the fair value of certain assets and liabilities to reflect the excess of the reorganization value of the Successor over the fair values of tangible and identifiable intangible assets, net of liabilities, from the adoption of fresh start reporting. Property, plant and equipment, net and Other intangibles, net, were reduced by \$1.9 million and \$3.2 million, respectively; Goodwill and Other assets increased by \$1.9 million and \$1.3 million, respectively; and Accrued expenses and other current liabilities and Other liabilities were reduced by \$1.8 million and \$0.1 million, respectively.

5. PROPERTY AND EQUIPMENT

Property and equipment consisted of the following (in thousands):

	Successor As of December 31, 2009	Predecessor As of December 31, 2008
Network equipment	\$ 158,206	\$ 163,475
Furniture and equipment	5,029	6,128
Leasehold Improvements	7,742	5,028
Construction in Progress	2,708	3,682
Subtotal	173,685	178,313
Less: accumulated depreciation	(26,079)	(66,161)
Total property and equipment, net	\$ 147,606	\$ 112,152

Successor

Depreciation and amortization expense for property and equipment for the six months ended December 31, 2009, including equipment under capital leases and vendor financing obligations, was \$25.5 million.

At December 31, 2009, total equipment under capital lease and vendor financing obligations consisted of \$12.9 million of network and peripheral equipment, with accumulated depreciation of \$1.3 million.

Predecessor

Depreciation and amortization expense for property and equipment including equipment under capital leases and vendor financing obligations for the six months ended July 1, 2009 and the years ended December 31, 2008 and 2007 was \$11.8 million, \$30.8 million and \$27.9 million, respectively.

At December 31, 2008, the total equipment under capital lease and vendor financing obligations consisted of \$27.3 million of network equipment and \$0.2 million of administrative equipment, with accumulated depreciation of \$19.1 million and \$0.1 million, respectively.

6. GOODWILL AND OTHER INTANGIBLE ASSETS

Goodwill reflects the excess of the reorganization value of Successor over the fair value of tangible and identifiable intangible assets as determined upon the adoption of fresh-start accounting. The Company recorded goodwill of \$61.3 million upon emergence from bankruptcy as well as intangible assets of \$184.8 million, which includes \$81.6 million of indefinite-lived trade names, \$99.2 million of amortizable customer relationships, and \$4.0 million of amortizable trade names.

The intangible assets not subject to amortization consisted of the following (in thousands):

	Successor As of December 31,	Predecessor As of December 31,	
Trade names	2009 \$ 81,372	<u>2008</u>	
Goodwill	\$ 64,220	\$ 32,688	

The Company allocated goodwill to all of its reporting units as part of fresh-start accounting, excluding the wholesale reporting unit which had nominal value relative to the total value of the Company. The changes in the carrying amount of goodwill and trade names by reporting unit for Successor's balance as of December 31, 2009 and Predecessor's balance as of December 31, 2008 are as follows (in thousands):

Goodwill

	Predecessor					
	United States	Canada	Australia	Europe	Brazil	Total
Balance as of January 1, 2008	\$ 194,256	\$108,151	\$ 32,796	\$ 110,229	\$ 3,935	\$ 449,367
Accumulated impairment losses	(194,256)	(80,864)	(20,157)	(110,229)	(3,727)	(409,233)
Net goodwill	_	27,287	12,639	_	208	40,134
Effect of change in foreign currency exchange rates	_	(5,374)	(2,021)	_	(51)	(7,446)
Balance as of December 31, 2008	\$ —	\$ 21,913	\$ 10,618	\$ —	\$ 157	\$ 32,688
Effect of change in foreign currency exchange rates		\$ 1,266	\$ 1,365	\$ —	\$ 32	\$ 2,663
Fresh-start adjustments	29,960	4,313	(10,437)	2,262	(151)	25,947
Balance as of July 1, 2009	29,960	27,492	1,546	2,262	38	61,298
			Succe	essor		
Balance as of July 1, 2009	\$ 29,960	\$ 27,492	\$ 1,546	\$ 2,262	\$ 38	\$ 61,298
Effect of change in foreign currency exchange rates		2,793	168	(45)	6	2,922
Balance as of December 31, 2009	\$ 29,960	\$ 30,285	\$ 1,714	\$ 2,217	\$ 44	\$ 64,220

Trade Names

	Predecessor					
	United States	Canada	Australia	Europe	Brazil	Total
Fresh-start adjustments	76,200			5,400		81,600
Balance as of July 1, 2009	\$ 76,200	<u>\$ —</u>	<u>\$</u>	\$5,400	<u>\$—</u>	\$81,600
			Succes	sor		
Balance as of July 1, 2009	\$ 76,200	\$ —	\$ —	\$5,400	\$—	\$81,600
Effect of change in foreign currency exchange rates				(228)		(228)
Balance as of December 31, 2009	\$ 76,200	s —	\$	\$5,172	\$	\$81,372

Acquired intangible assets subject to amortization consisted of the following (in thousands):

		Successor			Predecessor	
	As	of December 31, 200	09	A	As of December 31, 20	008
	Gross Carrying Amount	Accumulated Amortization	Net Book value	Gross Carrying Amount	Accumulated Amortization	Net Book value
Trade names	\$ 4,057	\$ (203)	\$ 3,854	\$ 1,665	\$ (1,575)	\$ 90
Customer relationships	107,612	(14,032)	93,580	3,806	(3,150)	656
Total	\$ 111,669	\$ (14,235)	\$97,434	\$ 5,471	\$ (4,725)	\$ 746

Successor

Amortization expense for customer lists and other intangible assets for the six months ended December 31, 2009 was \$14.0 million. The Company expects amortization expense for customer lists and other intangible assets for the fiscal years ended December 31, 2010, 2011, 2012, 2013, 2014 and thereafter to be approximately \$23.2 million, \$18.6 million, \$13.1 million, \$9.6 million, \$7.2 million and \$26.1 million, respectively.

The Company tested the intangible assets for impairment as of October 1, 2009 in accordance with the Company's policy and ASC No. 350. As a result no impairment was recognized. The intangible assets subject to amortization include trade names and customer relationships. The useful lives of the trade names are 10 years; the useful lives of the customer relationships range from 11 to 22 years. The amortizable trade names are amortized using the straight-line method over the 10 year life, and the customer relationships are amortized using the pattern of benefits method over their useful lives, resulting in accelerated amortization in the early periods of the useful life.

Predecessor

Amortization expense for customer lists and other intangible assets for the six months ended June 30, 2009, and the years ended December 31, 2008 and 2007 was \$0.5 million, \$2.0 million and \$2.6 million, respectively.

7. LONG-TERM OBLIGATIONS

Long-term obligations consisted of the following (in thousands):

Obligations under capital leases and other \$ 3,178 \$ 4,851 Leased fiber capacity 2,809 2,172 Senior secured term loan facility — 96,250 Canadian credit facility — 35,000 Senior notes — 200,186 Senior secured notes 128,057 — Senior subordinated secured notes 123,472 198,960 Exchangeable senior notes — 25,122 Convertible senior notes — 34,006 Step up convertible subordinated debentures — 8,290 Subtotal \$ 257,516 604,837 Less: Current portion of long-term obligations (4,274) (564,797)		Successor December 31, 2009	Predecessor December 31, 2008
Senior secured term loan facility — 96,250 Canadian credit facility — 35,000 Senior notes — 200,186 Senior secured notes 128,057 — Senior subordinated secured notes 123,472 198,960 Exchangeable senior notes — 34,006 Step up convertible senior notes — 34,006 Step up convertible subordinated debentures — 8,290 Subtotal \$ 257,516 604,837 Less: Current portion of long-term obligations (4,274) (564,797)	Obligations under capital leases and other	\$ 3,178	\$ 4,851
Canadian credit facility — 35,000 Senior notes — 200,186 Senior secured notes 128,057 — Senior subordinated secured notes 123,472 198,960 Exchangeable senior notes — 25,122 Convertible senior notes — 34,006 Step up convertible subordinated debentures — 8,290 Subtotal \$ 257,516 604,837 Less: Current portion of long-term obligations (4,274) (564,797)	Leased fiber capacity	2,809	2,172
Senior notes — 200,186 Senior secured notes 128,057 — Senior subordinated secured notes 123,472 198,960 Exchangeable senior notes — 25,122 Convertible senior notes — 34,006 Step up convertible subordinated debentures — 8,290 Subtotal \$ 257,516 604,837 Less: Current portion of long-term obligations (4,274) (564,797)	Senior secured term loan facility	_	96,250
Senior secured notes 128,057 — Senior subordinated secured notes 123,472 198,960 Exchangeable senior notes — 25,122 Convertible senior notes — 34,006 Step up convertible subordinated debentures — 8,290 Subtotal \$ 257,516 604,837 Less: Current portion of long-term obligations (4,274) (564,797)	Canadian credit facility	_	35,000
Senior subordinated secured notes 123,472 198,960 Exchangeable senior notes — 25,122 Convertible senior notes — 34,006 Step up convertible subordinated debentures — 8,290 Subtotal \$ 257,516 604,837 Less: Current portion of long-term obligations (4,274) (564,797)	Senior notes	-	200,186
Exchangeable senior notes — 25,122 Convertible senior notes — 34,006 Step up convertible subordinated debentures — 8,290 Subtotal \$ 257,516 604,837 Less: Current portion of long-term obligations (4,274) (564,797)	Senior secured notes	128,057	_
Convertible senior notes—34,006Step up convertible subordinated debentures—8,290Subtotal\$ 257,516604,837Less: Current portion of long-term obligations(4,274)(564,797)	Senior subordinated secured notes	123,472	198,960
Step up convertible subordinated debentures—8,290Subtotal\$ 257,516604,837Less: Current portion of long-term obligations(4,274)(564,797)	Exchangeable senior notes	_	25,122
Subtotal \$ 257,516 604,837 Less: Current portion of long-term obligations (4,274) (564,797)	Convertible senior notes	_	34,006
Less: Current portion of long-term obligations (4,274) (564,797)	Step up convertible subordinated debentures	_	8,290
	Subtotal	\$ 257,516	604,837
# 0FD 040	Less: Current portion of long-term obligations	(4,274)	(564,797)
Total long-term obligations \$ 253,242 \$ 40,040	Total long-term obligations	\$ 253,242	\$ 40,040

Successor

The following table reflects the contractual payments of principal and interest for Successor's long-term obligations as of December 31, 2009 as follows (in thousands):

Successor: Year Ending December 31,	Vendor Financing and Other	Senior Secured Notes	Senior Subordinated Secured Notes	Total
2010	\$ 4,815	\$ 16,665	\$ 17,595	\$ 39,075
2011	1,239	16,900	17,595	35,734
2012	329	16,900	17,595	34,824
2013	102	16,900	132,269	149,271
2014	28	16,900	_	16,928
Thereafter	20	163,847	_	163,867
Total Minimum Principal & Interest Payments	6,533	248,112	185,054	439,699
Less: Amount Representing Interest	(546)	(118,112)	(61,582)	(180,240)
Total Long Term Obligations	\$ 5,987	\$ 130,000	\$ 123,472	\$ 259,459

13% Senior Secured Notes

In December 2009, wholly owned subsidiaries of the Company, Primus Telecommunications Holding, Inc. (the "U.S. Issuer") and Primus Telecommunications Canada, Inc. (the "Canadian Issuer"), (jointly, the "Issuers"), issued \$130 million of aggregate principal amount of units (the "Units"), consisting of \$85 million of 13% Senior Secured Notes due 2016 issued by the U.S. Issuer (the "U.S. Notes") and \$45 million of 13% Senior Secured Notes due 2016 issued by the Canadian Issuer (the "Canadian Notes") (the U.S. Notes and, together with the Canadian Notes, the "13% Senior Secured Notes"). Accrued interest will be paid each June 15 and December 15, beginning June 15, 2010 to holders of record on the preceding June 1 and December 1, respectively. Net cash proceeds from the 13% Senior Secured Notes issuance, after giving effect to expenses, discounts and fees was \$123.5 million.

The U.S. Notes are guaranteed by the Company and certain of the U.S. Issuer's domestic subsidiaries and the Canadian Notes are guaranteed by its parent and subsidiaries as well as the Company. The U.S. Notes and related guarantees are secured by a first lien security interest in substantially all of the assets of the U.S. Issuer and 13% Senior Secured guarantors, including a first priority pledge of all of the capital stock held by the U.S. Issuer and the 13% Senior Secured guarantors, which pledge in the case of capital stock of Primus Telecommunications International, Inc. and each non-U.S. subsidiary shall be limited to 65% of such capital stock. The Canadian Notes and related guarantees are secured by a first lien security interest in substantially all of the assets of the Canadian Issuer and Canadian guarantors.

In the event the Company and certain subsidiaries have excess cash flow (as defined below) for any fiscal year commencing with the fiscal year ending December 31, 2010 then the Issuers must jointly offer to the holders of the 13% Senior Secured Notes to repurchase an applicable amount, (equal to 50% of such Excess Cash Flow), of the Notes at par.

In the event the 14.25% Senior Secured Notes have not been refinanced in accordance with the terms of the 13% Senior Secured Notes indenture by January 21, 2013, then the Issuers will be required to redeem the full principal of the 13% Senior Secured Notes at a price equal to the then applicable optional redemption price on such date.

"Excess Cash Flow" means with respect to any Person, for any period, (a) the excess of (i) Consolidated "EBITDA," (as defined in the 13% Senior Secured Notes indenture), for such period over (ii) the sum of (A) the aggregate amount of capital expenditures by the Company and its restricted subsidiaries during such period (other than any such capital expenditures made with Net Proceeds from an asset sale (without giving effect to the threshold set forth in the definition thereof) or insurance or condemnation proceeds), (B) the cash portion of consolidated interest expense paid by the Company and its restricted subsidiaries during such period, (C) the aggregate amount (without duplication) of all income franchise taxes paid in cash by the Company and its restricted subsidiaries during such period (D) any reduction in the principal amount of Indebtedness resulting from principal payments made thereon (other than payments on the 13% Senior Secured Notes, any Junior Lien Indebtedness, any unsecured Indebtedness or any Indebtedness that is subordinated to the 13% Senior Secured Notes or any 13% Senior Secured Note guarantee except, in each case, any required amortization payment or payment at the Stated Maturity thereof) during such period (provided that (i) such Indebtedness has been incurred in accordance with the Note's Indenture and (ii) to the extent such Indebtedness is revolving in nature, such payment shall have been accompanied by a concurrent corresponding permanent reduction in the revolving commitment relating thereto) minus (b) to the extent that Excess Cash Flow for any prior period was less than \$0 (dollars) and such amount has not been deducted in the calculation of the calculation of Excess Cash Flow for another prior period, the absolute value of such negative number.

141/4% Senior Subordinated Secured Notes

In February 2007, IHC issued in a private transaction \$57.2 million principal amount of the 14 ½% Senior Secured Notes in exchange for \$40.7 million principal amount of the Company's outstanding 12 ½% Senior Notes and \$23.6 million in cash. This exchange was accounted for as a modification of debt with a portion deemed to be a troubled debt restructuring. In March 2007, IHC also issued for cash in private transactions an additional \$51.0 million principal amount of 14 ½% Senior Secured Notes with a \$0.3 million discount. Net cash proceeds from the 14 ½% Senior Secured Notes issuance, after giving effect to expenses, discounts and fees related to all of the foregoing transactions (including the amendment of the Facility) was \$69.2 million. The Company recorded \$5.1 million in costs associated with this issuance of the 14 ½% Senior Secured Notes, which have been recorded as a loss on restructuring of debt.

In May 2008, IHC issued \$67.1 million principal amount of the 14 1/4% Senior Secured Notes and paid \$4.7 million in cash in exchange for \$49.0 million principal amount of the Company's 8% Senior Notes, \$33.0 million principal amount of the Company's 5% exchangeable senior notes due June 2010 ("5% Exchangeable Senior Notes"), \$43.1 million principal amount of the Company's 3 3/4% Convertible Senior Notes, and \$5.3 million principal amount of the Company's 12 3/4% Senior Notes. All exchanges were deemed troubled debt

restructurings, and accordingly, were accounted for as modifications of debt, with scheduled future cash interest payments of \$26.4 million recorded in long-term obligations. The Company recognized a gain on restructuring of debt of \$32.2 million in connection with this exchange, including the expensing of \$0.5 million of financing costs.

The 14 ½ % Senior Secured Notes were to mature on May 20, 2011 with early redemption at a premium to par at IHC's option at any time after February 2008. During specified periods, IHC may redeem at par up to 35% of the aggregate principal amount of the 14 ½ % Senior Secured Notes with the net cash proceeds of certain equity offerings of the Company. Accrued interest will be paid each May 31st and November 30th, beginning May 31st, 2007. The effective interest rate for the 14 ½ % Senior Secured Notes at December 31, 2008 was 12.4% for those amounts not related to the troubled debt restructuring discussed above. (see Note 22—"Guarantor/Non-Guarantor Consolidating Condensed Financial Information.")

In December 2008, the Company made open market purchases of \$2.1 million principal amount of its $14\frac{1}{4}$ % Senior Secured Notes, resulting in a \$2.0 million gain on early extinguishment of debt including the write-off of related deferred financing costs. The notes were recorded as a reduction of long-term obligations.

On the Effective Date, IHC, Group, Holding, the other Guarantors party thereto and U.S. Bank National Association, as trustee, entered into a supplemental indenture (the "Supplemental Indenture") to the indenture governing IHC's 14 ½ % Senior Secured Notes due 2011 (the "Original Indenture"). The Supplemental Indenture amended the Original Indenture to provide for the issuance of the "Senior Subordinated Secured Notes". The maturity of the Senior Subordinated Secured Notes was extended by two years to May 20, 2013 beyond the maturity of the notes issued under the Original Indenture. At the option of IHC, prior to the earlier of (1) the extension of the maturity of or the repayment in full of the indebtedness outstanding pursuant to the Amended Term Loan and the Canadian Credit Facility, and (2) June 1, 2011, up to 4.25% per annum of the interest on the Senior Subordinated Secured Notes may be paid in kind. The Supplemental Indenture also modified covenants in the Original Indenture to prevent subsidiary guarantors from incurring debt to refinance indebtedness of non-guarantors and to limit the incurrence of indebtedness of restricted persons that is secured by a lien on the assets of IHC, any subsidiary guarantor or other restricted persons, as defined under the modified indenture.

On the Effective Date, IHC entered into a First Amendment to the Intercreditor Agreement (as amended through the date hereof, the "Amended Intercreditor Agreement"), with Group, Holding, The Bank of New York Mellon, as First Lien Collateral Agent, and U.S. Bank National Association, as Second Lien Collateral Agent. Pursuant to the Amended Intercreditor Agreement, the Senior Subordinated Secured Notes are subordinated in right of payment to the prior indefeasible payment in cash in full of all obligations under the Amended Term Loan.

Also on the Effective Date, IHC, each of the Grantors party thereto and U.S. Bank National Association, as collateral agent, entered into a First Amendment to the Collateral Agreement (as amended through the date hereof, the "Amended Collateral Agreement"), to provide that the obligations of both IHC and PTII, an indirect wholly owned subsidiary of Group, are secured by PTII's assets, including 65% of the voting stock of foreign subsidiaries owned by PTII. In addition, on the Effective Date, Group and Holding entered into an Assumption Agreement in favor of U.S. Bank National Association, as collateral agent, pursuant to which each of Group and Holding became party to the Amended Collateral Agreement. As a result, Group and Holding's existing guarantees of the Senior Subordinated Secured Notes are secured by a lien on the property of Group and Holding, respectively.

In addition, as a result of the Plan of Reorganization, IHC's \$173.2 million of outstanding 14 \(^1/4\)% Senior Secured Notes due 2011 were cancelled and the holders thereof received their pro rata portion of \$123.5 million of aggregate principal amount of 14 \(^1/4\)% Senior Subordinated Secured Notes due May 20, 2013 (see "Amendment of IHC's 14 \(^1/4\)% Senior Secured Notes Indenture"), and 4,800,000 shares of the new Common Stock of Group (the "New Common Stock").

Senior Secured Term Loan Facility

In February 2005, a direct wholly-owned subsidiary of the Company, Primus Telecommunications Holding, Inc. ("Holding"), entered into a six-year, \$100 million senior secured term loan facility (the "Facility"). Each borrowing made under the Facility may be, at the election of Holding at the time of the borrowing, a London Inter-Bank Offered Rate (LIBOR) loan (which will bear interest at a rate equal to LIBOR + 6.50%), or a base rate loan (which will bear interest at a rate equal to the greater of the prime rate plus 5.50% or the federal funds effective rate plus 6.00%). The Facility contained no financial maintenance covenants. The Company borrowed \$100 million under this Facility in February 2005.

The Facility was to be repaid in 24 quarterly installments, which began on June 30, 2005, at a rate of one percent of the original principal per year over the next five years and nine months, and the remaining balance repaid on the sixth anniversary date of the Facility, with early redemption at a premium to par at Holding's option at any time after February 18, 2006. The Facility is guaranteed by the Company and certain of Holding's domestic subsidiaries and is secured by certain assets of Holding and its guarantor subsidiaries and by partial stock pledges of certain foreign subsidiaries.

In February 2007, the Company received unanimous consent to an amendment of its existing \$100 million Facility. This amendment enabled Primus Telecommunications IHC, Inc. (IHC), a wholly-owned indirect subsidiary of the Company, to issue and have at any one time outstanding up to \$200 million of existing authorized indebtedness in the form of newly authorized secured notes with a second lien security position. On February 26, 2007, an Intercreditor Agreement was entered into between the 14 1/4% Senior Secured Notes and the lenders of the Facility. Pursuant to this authorization, the Company has issued certain 14 1/4% Senior Secured Notes. The amendment allowed for an increase of 1/4% to the interest rate of the Facility and adjusted the early call features. The effective interest rate for the Facility at December 31, 2008 was 10.2%.

The debt under the Facility was in default as a result of the bankruptcy filings on March 16, 2009. On the Effective Date, Group and Holding entered into a Third Amendment to the Term Loan Agreement (the "Amended Term Loan"). In accordance with the Amended Term Loan, Holding's Term Loan facility due February 2011 was reinstated and amended in certain respects, including: (i) at the option of Holding, interest rates are now (A) LIBOR + 9.00% with a LIBOR floor of 3.00% (or LIBOR + 11.00% with 4.00% to be paid in kind) or (B) Prime Rate + 8.00% with a Prime Rate floor of 4.00% (or Prime Rate + 10.00% with 4.00% to be paid in kind); (ii) The Bank of New York Mellon has been appointed as successor Administrative Agent; (iii) amortization payments have been increased; (iv) mandatory prepayments are required from (A) 25% of the net proceeds of certain equity issuances (including 25% of the cash of businesses acquired in exchange for equity), (B) 100% of the net proceeds from debt issuances (other than as permitted under the limitation of indebtedness covenant), and (C) 80% of net cash proceeds from asset sales or insurance recoveries not otherwise reinvested within 180 days or committed to reinvestment within 270 days of such asset sales; (v) Group or its affiliates are able to purchase annually up to \$5 million in principal amount of loans at less than par without being subject to the pro-rata provisions of the Term Loan facility (or purchases in excess of such annual amount by way of an offer to all lenders), any such purchased loans deemed immediately cancelled; and (vi) certain covenants have been modified, including restrictions on the ability to incur additional debt and the addition of a minimum EBITDA covenant, a maximum indebtedness covenant and a maximum capital expenditure covenant. In addition, the Debtors have agreed to pay all reasonable fees, expenses and disbursements of the counsel and financial advisors to the Term Loan lenders.

In December 2009, the Company used proceeds from the issuance of the 13% Senior Secured Notes to retire in full the Senior Secured Term Loan Facility.

Canadian Credit Facility

In March 2007, an indirect wholly-owned subsidiary of the Company, Primus Telecommunications Canada, Inc. ("Primus Canada") entered into a Senior Secured Credit Agreement ("Canadian Credit Facility") with a financial institution to refinance an existing Canadian credit facility. The Canadian Credit Facility provided for a

\$35.0 million non-amortizing loan bearing interest at a rate of LIBOR plus 425 basis points and matured in March 2012. The loan proceeds were used to refinance the existing Canadian credit facility, including certain costs related to the transaction, and to finance certain capital expenditures. The Canadian Credit Facility was secured by the assets of Primus Canada and certain guarantees.

In October 2007, Primus Canada entered into a cross-currency principal and interest rate swap agreement, a portion of which was required by the Canadian Credit Facility, which fixed the interest rate at 9.21% starting from October 31, 2007. The cross-currency principal and interest rate swap agreement's counter party was Lehman Brothers Special Financing, Inc. ("Lehman SFI"). Lehman SFI entered into bankruptcy in early October 2008 following its ultimate parent entering bankruptcy in mid-September 2008. Since September 2008 month end interest rate swap payments were not made by Lehman SFI to Primus Canada nor, correspondingly, were payments made from Primus Canada to Lehman SFI. While the covenant language is arguably ambiguous, the Company believed that the swap agreement with Lehman SFI continued to be in force with respect to the requirements under the Canadian Credit Facility and, accordingly, that no breach or event of default had occurred. The Waiver and Amendment Agreement (described below), eliminated any currency swap requirements of the Canadian Credit Facility.

On August 25, 2009, Primus Canada provided notice of Early Termination of the swap agreement, effective August 26, 2009. On September 16, 2009, Primus Canada filed Proofs of Claim in the bankruptcy proceedings of Lehman SFI and Lehman Brothers Holdings Inc. ("LBHI"), which was the guarantor of Lehman SFI's obligations under the swap agreement and which also is in bankruptcy. Lehman SFI and LBHI have stated they reserve all rights to dispute the Proof of Claim, including the effectiveness of Primus Canada's termination notice. The eventual outcome of this matter is unknown. As of December 31, 2009, the Company has not maintained any balances concerning the Lehman cross-currency principal and interest rate swap agreement.

On March 10, 2009, Primus Canada, 3082833 Nova Scotia Company and certain affiliate guarantors entered into a Waiver and Amendment Agreement (the "Waiver and Amendment") to their \$35 million Canadian Credit Facility with Guggenheim Corporate Funding, LLC, as Administrative Agent and Collateral Agent.

The lenders under the Waiver and Amendment waived events constituting events of default and potential events of default under the Canadian Credit Facility, subject to the terms and conditions of the Waiver and Amendment. In connection with the Waiver and Amendment, the applicable margin under the Canadian Credit Facility was increased to LIBOR +4.50%, with a 2.50% LIBOR floor, and the maturity date was changed to May 21, 2011.

In December 2009, the Company used proceeds from the issuance of the 13% Senior Secured Notes to retire in full the Canadian Credit Facility.

Leased Fiber Capacity

In December 2000, the Company entered into a financing arrangement to purchase fiber optic capacity in Australia for 51.1 million Australian dollars (AUD) (\$28.5 million at December 31, 2000) from Optus Networks Pty. Limited. As of December 31, 2001, the Company had fulfilled the total purchase obligation. The Company signed a promissory note payable over a four-year term ending in April 2005 bearing interest at a rate of 14.31%. During the three months ended June 30, 2003, the Company renegotiated the payment terms extending the payment schedule through March 2007, and lowering the interest rate to 10.2%. In October 2006, the Company renegotiated the payment terms of its promissory note payable to Optus Networks Pty. Limited to defer principal payments from April 2006 through December 2006 and was obligated to pay the remaining balance in three equal monthly principal payments in the first quarter 2007. In February 2007, the Company again renegotiated the payment terms of its \$8.8 million (10.1 million AUD) promissory note payable to Optus Networks Pty. Limited to extend the payment schedule through December 2008 in 24 equal monthly payments. During the third

quarter 2008, the payment terms were again renegotiated to extend payment of the principal balance of \$2.7 million (3.1 million AUD) to June 2010 with monthly payments of interest at a rate of 13.5%. Certain volume purchase targets were not met by September 30, 2009. Accordingly, Optus Networks Pty. Limited may give, but has not given, 30 days notice requiring full payment of the principal balance. At December 31, 2009 and December 31, 2008, the Company had a liability recorded in the amount of \$2.8 million (3.1 million AUD). Also, \$2.8 million (3.1 million AUD) and \$0 million were classified as current portion of long-term obligations at December 31, 2009 and December 31, 2008, respectively. (See "Note 23—Subsequent Events").

Equipment Financing and Other Long-Term Obligations

In November 2005, Primus Australia entered into a financing arrangement for network equipment. Payments are made over a five-year term ending October 2010. In June 2009, the Company entered into an amendment which removed certain financial covenants and required a principal payment of \$1.2 million (1.5 million AUD) in the same month. In September 2009, the Company made a principal payment of \$1.6 million (1.8 million AUD) and completed the full repayment of the debt.

Predecessor

In accordance with the Plan, on the Effective Date all of the obligations of Group and its subsidiaries with respect to the following indentures were terminated and the respective notes and debentures issued under each such indenture were cancelled:

- Indenture, dated as of January 16, 2004, by and among Holding, as Issuer, Group, as Guarantor, and Wachovia Bank N.A., as Trustee, relating to the 8% Senior Notes due 2014;
- Indenture, dated as of June 28, 2006, by and among Holding, as Issuer, Group, as Guarantor, and U.S. Bank N.A., as Trustee, relating to the 5% Exchangeable Senior Notes due 2009;
- Indenture, dated as of September 15, 2003, by and between Group and Wachovia Bank, N.A., as Trustee, relating to the 3³/₄% Convertible Senior Notes due 2010;
- Indenture, dated as of February 27, 2006, by and between Group and U.S. Bank N.A., as Trustee, relating to the Step-up Convertible Subordinated Debentures due 2009; and
- Indenture, dated as of October 15, 1999, by and between Group and First Union National Bank, as Trustee, relating to the 12 ³/₄% Senior Notes due 2009.

8. INCOME TAXES

The total provision (benefit) for income taxes for the six months ended December 31 and July 1, 2009 and the years ended December 31, 2008 and 2007 is as follows (in thousands):

	Successor		Predecessor	
	Six Months Ended December 31, 2009	Six Months Ended July 1, 2009	Year Ended	December 31,
Current: Federal	\$ 0	\$ 0	\$ 59	86
State	287	56	63	154
Foreign	(3,691)	2,198	(6,852)	2,052
Subtotal Current	(3,404)	2,254	(6,730)	2,292
Deferred: Federal	_	_	162	_
State	_	_	_	_
Foreign	(6,637)	1,653	6,202	(11,524)
Subtotal Deferred	(6,637)	1,653	6,364	(11,524)
Total Tax Provision (Benefit)	\$ (10,041)	\$ 3,907	\$ (366)	\$ (9,232)

The US and foreign components of income from continuing operations before income taxes for the six months ended December 31 and July 1, 2009 and the years ended December 31, 2008 and 2007 are as follows (in thousands):

	Successor		Predecessor		
	Six Months Ended December 31,	Six Months Ended July 1,	Year Ended	December 31,	
	2009	2009	2008	2007	
US	\$ (3,423)	\$359,427	\$ (43,097)	\$ (13,514)	
Foreign	828	116,027	18,026	14,154	
Income from continuing operations before income taxes	\$ (2,595)	\$475,454	\$(25,071)	\$ 640	

The provision for (benefit from) income taxes differed from the amount computed by applying the federal statutory income tax rate to income (loss) before income taxes due to the following for the six months ended December 31 and July 1, 2009 and the years ended December 31, 2008 and 2007 (in thousands):

	Successor		Predecessor	
	Six Months Ended December 31, 2009	Six Months Ended July 1, 2009	Year Ended	December 31,
Tax provision (benefit) at federal statutory rate	\$ (1,158)	\$166,271	\$ (8,524)	\$ 218
Permanent differences	3,534	1,531	(1,940)	(29,299)
State tax (net of federal)	(178)	8,887	(1,043)	1,129
Effect of statutory tax rate change	106	(5,320)	_	_
Effect of foreign tax rate change	-	_	(1,241)	(3,278)
Foreign withholding taxes (net of Federal)	(3,554)	774	(3,159)	2,518
ASC 740-10 items	(1,324)	3,608	(5,014)	(3,525)
Foreign taxes	_	_	3,180	8,432
Increase (decrease) in valuation allowance	(6,869)	(72,906)	15,710	12,579
Tax – book fresh-start valuation difference	_	13,673		_
Capital loss disposal of subsidiary	_	(13,283)	_	_
Reorganization	_	4,415	_	_
Cancellation of debt allocated to foreign basis	_	(58,005)	_	_
Goodwill	_	10,227	_	_
Taxes charged to goodwill	-	(54,512)	_	_
Other	(598)	(1,453)	1,665	1,994
Income tax (benefit) expense	\$ (10,041)	\$ 3,907	\$ (366)	\$ (9,232)

Deferred income taxes are recognized to account for the tax consequences in future years of differences between the tax basis of assets and liabilities and their financial reporting amounts of each period-end, based on enacted tax laws and statutory income tax rates applicable to periods in which the differences are expected to affect taxable income. Deferred income taxes reflect the net income tax effect of temporary differences between the basis of assets and liabilities for financial reporting purposes and for income tax purposes. Net deferred tax balances are comprised of the following as of December 31, 2009, 2008 and 2007 (in thousands):

	Successor	Predec	essor
	As of December 31,	As of Dece	ember 31,
	2009	2008	2007
Deferred tax assets	\$ 154,812	\$ 274,000	\$ 268,659
Valuation allowance	(118,520)	(227,798)	(239,680)
Deferred tax liabilities	(76,211)	(32,183)	(4,881)
Net deferred taxes	\$ (39,919)	\$ 14,019	\$ 24,098

Certain fixed and intangible assets were marked to their fair market values, as a result of our emergence from bankruptcy and the associated fresh-start accounting, (see Note 4), which resulted in a substantial decrease to our deferred tax asset, net operating loss carry-forwards, the discharge of certain tax liabilities, and tax attribute reduction, as prescribed by Internal Revenue Code Section 108.

The significant components of the Company's deferred tax assets and liabilities are as follows as of December 31, 2009, 2008 and 2007 (in thousands):

	Successor As of December 31,	Predece As of Decer	
	2009	2008	2007
Current			
Allowance for bad debt	\$ 1,024	\$ 2,301	\$ 2,443
Basis difference on intangibles	(3,873)	_	_
Other	4,056	3,683	4,424
Valuation allowance	(5,081)	(5,133)	(5,187)
Total Current	\$ (3,874)	\$ 851	\$ 1,680
Non-current			
Basis difference in intangibles	\$ (46,908)	\$ 27,915	\$ 30,872
Basis difference in fixed assets impairment	_	57,066	64,672
Other basis difference		_	450
Bond related adjustments	11,863	(30,940)	
Capital loss carryforwards	46,171	27,491	34,305
Net operating loss carryforwards	50,929	139,664	111,168
Basis difference in fixed assets	25,201	8,615	13,375
Unrealized foreign exchange gains	(11,963)	3,256	(4,723)
Accrued withholding tax	(907)	3,283	6,062
Minimum tax credit carryforward	_	726	888
Other	3,008	(1,243)	(158)
Valuation allowance	(113,439)	(222,665)	(234,493)
Total Non-current	\$ (36,045)	\$ 13,168	\$ 22,418

As of December 31, 2009, the Company had foreign operating loss carryforwards of approximately \$205 million of which \$47.6 million expire periodically from 2010 through 2029, and the remainder of which carryforward without expiration.

At December 31, 2009, the Company projects no United States operating loss carryforwards available to reduce future United States taxable income after attribute reduction. These amounts are subject to adjustment based upon the finalization of attribute reduction after the filing of the Company's 2009 Federal income tax return. With the filing of the 2009 return, the mix of reduced attributes may shift between net operating losses and other tax attributes resulting in the survival of some net operating losses. Any net operating losses which survive attribute reduction will expire periodically between 2014 through 2029. Any surviving net operating loss carryforwards not related to the second half of 2009 are subject to limitations in the future, in accordance with Section 382 of the Internal Revenue Code. The net operating loss carryforward amounts above reflect all ASC 740-10 (previously FIN No. 48) changes for both foreign and domestic net operating losses.

Pursuant to Section 382 of the Internal Revenue Code, the Company believes that it underwent an ownership change for tax purposes (i.e., a more than 50% change in stock ownership) on the July 1, 2009 emergence date. As a result, the use of any of the Company's federal and state NOL carryforwards and tax credits generated prior to the ownership change (that are not reduced pursuant to the provisions discussed in the preceding paragraph) will be subject to an annual limitation of approximately \$1.7 million. The annual limitation will be determined based upon an Internal Revenue Code section that allows corporations emerging from bankruptcy to determine their section 382 limitation based upon the post emergence stock value. The Company has prepared its financial statements assuming the annual limitation will apply. However, Section 382 provides that a taxpayer emerging from bankruptcy can elect out of the annual limitation. If the Company elects not to

apply the limitation, there are adverse consequences if an ownership change occurs before July 1, 2011. The determination to make the election is not required to be made until the extended due date of the 2009 return, which is September 15, 2010. The company has reviewed its 13-G filings, as filed with the United States Securities Exchange Commission, subsequent to emergence from bankruptcy and believes that a change in ownership has not occurred during this period of July 1, 2009 to December 31, 2009.

The Company is required to determine whether it had a net unrealized built-in gain (NUBIG) or net unrealized built-in loss (NUBIL) at the 2009 emergence date. The Company believes that there was a NUBIL at the time of the 2009 emergence date. As a result, certain depreciation and loss deductions recognized during the five-year period beginning on the 2009 emergence date would be subject to the section 382 limitation.

As a result of the fresh start accounting, the company has increased its deferred tax liabilities and goodwill by \$55.7 million to reflect the increase in book basis without any increase in the tax basis of certain assets which were adjusted for the fresh start. \$31.2 million of the deferred tax liabilities relates to the fresh start increase in the value of certain indefinite lived intangibles. These deferred tax liabilities will remain on the books indefinitely subject to reduction only if the related indefinite lived intangibles which gave rise to the deferred tax liabilities are impaired. The reduction of the deferred tax liability due to impairment of the underlying indefinite lived intangibles would provide a tax benefit for the Company which would be offset by any future impairment expense of the indefinite lived intangibles. The remaining \$24.5 million of deferred tax liability will be a tax benefit which will be offset when its underlying fresh start adjustment is released. The relief of the deferred tax liability will not result in any payment of cash by the company.

The Company incurred \$(2.8) million, \$(3.1) million and \$3.9 million of expense in 2009, 2008 and 2007, respectively, related to foreign withholding tax on intercompany interest and royalties owed to our United States subsidiary.

On December 15, 2008 the U.S. Treasury Department announced that the Fifth Protocol ("Protocol") to the United States – Canada Income Tax Treaty entered into force. The Protocol is generally effective for tax years beginning on or after January 1, 2009, however certain provisions of the Protocol have different effective dates. The elimination of withholding tax on interest paid or credited between related parties is phased in over a 3-year period, between 2008 through 2010. The reduced withholding tax rates are 7% for 2008, 4% for 2009, and 0% for 2010 and thereafter. Effective January 1, 2010, interest paid by or to certain hybrid entities will not qualify for treaty benefits. The changes made by the Protocol have substantially reduced the accrued income tax liability by \$5.4 million, thereby providing for an overall tax benefit during 2008.

During the fourth quarter of 2009, the company restructured its Canadian hybrid entity such that the Company now qualifies for an exemption from Canadian withholding tax on interest from Canadian subsidiaries. This restructuring also resulted in the release of accrued withholding tax resulting in a tax benefit of \$2.8 million during 2009.

No provision, with the exception of \$0.1 million of accrued withholding tax, was made in 2009 for United States income taxes on the undistributed earnings of the foreign subsidiaries as it is the Company's intention to utilize those earnings in the foreign operations for an indefinite period of time or to repatriate such earnings only when it is tax effective to do so.

On January 1, 2007, the Company adopted the provision of ASC No. 740-10, "Income Taxes". ASC No. 740-10 clarifies the accounting for uncertainty in income taxes recognized in the financial statements in accordance with FASB Statement No. 109, "Accounting for Income Taxes." This Interpretation prescribes a comprehensive model for how a company should recognize, measure, present, and disclose in its financial statements uncertain tax positions that the Company has taken or expects to take on a tax return.

The Company is subject to challenge from various taxing authorities relative to certain tax planning strategies, including certain intercompany transactions as well as regulatory taxes. Prior to the implementation of

ASC No. 740-10 on January 1, 2007, the Company accrued income tax contingencies under ASC No. 450, "Contingencies," when it was probable that a liability to a taxing authority had been incurred and the amount of the contingency could be reasonably estimated, based on past experience. The Company's tax contingency reserve was adjusted for changes in circumstances and additional uncertainties, such as significant amendments to existing tax law, both legislated and concluded through various jurisdictions' tax court systems. For 2009 the reserve is recorded under ASC No. 740.

Reconciliations of the successor period July 2, 2009 to December 31, 2009 and predecessor periods of January 1, 2009 to July 1, 2009 and January 1, 2008 to December 31, 2008 and January 1, 2007 to December 31, 2007 balances of unrecognized tax benefits are as follows (in thousands):

	Liability for Unrecognized Tax Benefits					
	Successor		Predecessor			
	Six Months Ended	Six Months Ended				
	December 31,	July 1,	Year Ended l	December 31,		
	2009	2009	2008	2007		
Balance at January 1, (July 2, for Successor period)	\$ 88,953	\$ 90,413	\$116,454	\$111,426		
Foreign currency adjustments	1,138	3,492	(11,215)	4,934		
Statute expiration	(4,514)	_	(441)	(4,269)		
Gross increases (decreases) of tax positions in prior period	1,622	(6,479)	(501)	3,075		
Audit resolution	-	_	(16,977)	(9,343)		
Gross increases of tax positions in current period	2,301	1,069	3,042	7,961		
Penalties and interest	216	458	51	2,670		
Balance at December 31, (July 1, for 2009 Predecessor period)	\$ 89,716	\$ 88,953	\$ 90,413	\$116,454		

The total unrecognized tax benefits as of December 31 2009 were \$89.7 million. Total unrecognized tax benefits of \$6.4 million, if recognized, would affect the effective tax rate and includes penalties and interest of \$2.4 million. The Company does not expect its unrecognized tax benefits to significantly increase or decrease over the next 12 months, with the possible exception of its Canada operating segment, which is currently involved in an ongoing international tax planning audit; the outcome of the tax planning audit is uncertain at this time.

The following table summarizes the open tax years for each major jurisdiction:

Jurisdiction	Open Tax Years
United States Federal	2000, 2002-2009
Australia	2002-2009
Canada	2003-2009
United Kingdom	2004-2009
Netherlands	2007-2009

The Company has closed certain federal and provincial income tax examinations in Canada for the years 2000 through 2005 (federal) and 2001 through 2005 (provincial) with all assessments being received. The Company has concluded an examination in the Netherlands for the years 2002, 2003, 2004, 2005 and 2006. Based on the final settlement, the Company reversed its remaining liability in the fourth quarter of 2008 and paid those liabilities during 2009. The Company is also currently under examination in other foreign tax jurisdictions, none of which are individually material.

9. FAIR VALUE OF FINANCIAL INSTRUMENTS AND DERIVATIVES

The carrying amounts reported in the consolidated balance sheets for cash and cash equivalents, restricted cash, accounts receivable and accounts payable approximate fair value due to relatively short periods to maturity. The estimated aggregate fair value of the Successor Company's 13% Senior Secured Notes and 14 \(^{1}/4\)% Senior Subordinated Secured Notes and the Predecessor Company's senior secured term loan facility, 14 \(^{1}/4\)% Senior Subordinated Secured Notes, 5% Exchangeable Senior Notes, Step Up Convertible Subordinated Debentures, 8% Senior Notes, 12 \(^{3}/4\)% Senior Notes and 3 \(^{3}/4\)% Convertible Senior Notes (face value of \(^{2}/3\) million and \(^{2}/3\) million and \(^{2}/3\) million, at December 31, 2009 and 2008, respectively), based on quoted market prices for all but the senior secured term loan facility, was \(^{2}/4\) million and \(^{2}/4\) million, respectively, at December 31, 2009 and 2008. For the senior secured term loan facility, fair value determination presented as of December 31, 2008, was based on a ASC No. 820, "Fair Value Measurements," Level 2 input, as the market is inactive with no trades since June 2008, and was based on a current offering sheet.

Contingent Value Rights Distribution Agreement (CVR)

Pursuant to the terms of the Reorganization Plan, (see "Note 2—Emergence from Voluntary Reorganization Under Chapter 11 of the Bankruptcy Code"), Group issued to holders of Group's Old Common Stock Contingent Value Rights to receive up to an aggregate of 2,665,000 shares (the "CVR Shares") of New Common Stock. In connection with the issuance of the Contingent Value Rights, Group entered into a Contingent Value Rights Distribution Agreement (the "CVR Agreement"), in favor of holders of CVRs thereunder, dated as of the Effective Date.

Due to the nature of CVR, the company accounted for the instrument in accordance with ASC No. 815, "Derivatives and Hedging," as well as related interpretations of this standard. The Company determined these CVRs to be derivative instruments to be accounted for as liabilities and marked to fair value at each balance sheet date. Upon issuance, the Company recorded a liability of \$2.6 million in other liabilities as part of fresh-start accounting, and will adjust the liability quarterly to its estimated fair value. The change in value is reflected in our Statements of Operations as other income (expense). Estimates of fair value represent the Company's best estimates based on a Black-Scholes pricing model. The Company recognized \$2.8 million of expense during the six months ended December 31, 2009.

See table below for summary of the Company's financial instruments accounted for at fair value on a recurring basis:

		Fa	ir Value as of December 31, 2009, using:	
	December 31, 2009	Quoted prices in Active Markets for Identical Assets (Level 1)	Significant Other Observable Inputs (Level 2)	Significant Unobservable Inputs (Level 3)
Liabilities:				
Contingent Value Rights	\$ 5,362		\$ 5,362	
Total	\$ 5,362		\$ 5,362	

10. COMMITMENTS AND CONTINGENCIES

Future minimum lease payments under capital leases and leased fiber capacity financing ("Vendor Financing"), purchase obligations and non-cancelable operating leases as of December 31, 2009 are as follows (in thousands):

	Vendor		
Year Ending December 31,	Financing an		Operating
	Other	Obligations	Leases
2010	\$ 4,81	5 17,719	17,028
2011	1,23	9 4,305	14,231
2012	32	9 2,761	12,546
2013	10	2 —	9,060
2014	2	8 —	3,727
Thereafter	2	0 —	7,578
Total minimum lease payments	6,53	3 24,785	64,170
Less: Amount representing interest	(54	6)	
Total	\$ 5,98	7 \$ 24,785	\$64,170

The Company has contractual obligations to utilize network facilities from certain carriers with terms greater than one year. The Company does not purchase or commit to purchase quantities in excess of normal usage or amounts that cannot be used within the contract term or at rates below or above market value. The Company made purchases under purchase commitments of \$26.5 million, \$33.6 million and \$30.3 million for the years ended December 31, 2009, 2008 and 2007, respectively.

Successor

Rent expense under operating leases was \$8.7 million for the six months ended December 31, 2009.

Predecessor

Rent expense under operating leases was \$6.1 million, \$16.6 million and \$17.0 million for the six months ended July 1, 2009 and the years ended December 31, 2008 and 2007, respectively.

Litigation

Legal Proceedings

Group and its subsidiaries are subject to claims and legal proceedings that arise in the ordinary course of its business. Each of these matters is subject to various uncertainties, and it is possible that some of these matters may be decided unfavorably. The Company believes that any aggregate liability that may result from the resolution of the legal proceedings will not have a material adverse effect on the Company's consolidated financial position, results of operations or cash flows.

11. STOCKHOLDERS' EQUITY (DEFICIT)

Successor

On the Effective Date of the Plan of Reorganization, all existing shares of Old Common Stock were cancelled and holders thereof received their pro rata share of contingent value rights to acquire up to 2,665,000 shares of New Common Stock on terms described under the Contingent Value Rights Distribution Agreement in Note 2. Holders of IHC's 14 1/4% Senior Secured Notes and the Holding Senior Notes received 9,600,000 shares of New Common Stock. As of December 31, 2009 there are 9,600,000 shares of common stock outstanding. In addition, the Company recorded \$1.0 million of equity related to the issuance of the warrants.

Predecessor

During the year ended December 31, 2008, 100,000 restricted stock units were fully vested upon involuntary termination of an employee without cause. The fair market value of the stock units at the grant date was \$0.40 per share. The employee withholding tax was netted against the share issuance, and the Company issued 62,850 shares of common stock to the employee.

During the year ended December 31, 2007, the Company exchanged 6,000,000 shares of the Company's common stock for the extinguishment of \$5.0 million in principal amount of the Step Up Convertible Subordinated Debentures (see Note 7— "Long-Term Obligations"). Also, the Company sold 22,500,000 shares of the Company's common stock to certain investors pursuant to a subscription agreement for \$19.3 million cash. The Company also granted 284,000 shares to an executive employee as a stock award.

12. SHARE-BASED COMPENSATION

Successor

On the Effective Date, pursuant to the Plan, the Predecessor's employee stock compensation plan (the "Equity Incentive Plan") was cancelled and the Group's Management Compensation Plan became effective. The Management Compensation Plan provides that awards may be granted for up to 1,000,000 shares of New Common Stock, subject to adjustment in the case of certain changes in capitalization of reorganized Group, including, among other things, a reorganization, merger, consolidation, spin-off, combination, repurchase, share exchange, or other similar corporate transaction or event that affects the New Common Stock. The maximum aggregate number of shares of New Common Stock that may be granted during any fiscal year to any single individual who is likely to be a "covered employee" within the meaning of section 162(m)(3) of the Internal Revenue Code is 600,000, in the case of stock options or stock appreciation rights, and 400,000, in the case of restricted stock or other stock-based awards (other than stock appreciation rights).

The Compensation Committee (the "Committee") of the Board of Directors of Group administers the Management Compensation Plan. The Committee has broad authority to administer, construe and interpret the Management Compensation Plan; however, it may not take any action with respect to an award that would be treated, for accounting purposes, as a "repricing" of an award unless the action is approved by the shareholders of Group.

The Management Compensation Plan provides for the grant of incentive stock options, nonqualified stock options, restricted stock, restricted stock units, and other stock-based or cash-based performance awards (collectively, "awards").

Grants of Awards under the Management Compensation Plan—Emergence Awards

Restricted Stock Unit Award Agreements

On July 2, 2009, 400,000 restricted stock units (RSUs) were granted to certain employees and executive officers pursuant to the terms and conditions set forth in the form of restricted stock unit agreement (the "Emergence RSU Agreement").

Pursuant to the Emergence RSU Agreement, RSUs will vest in two equal tranches if reorganized Group attains at least ninety percent (90%) of the specified Adjusted EBITDA Targets for any fiscal year during the ten-year term of the Emergence RSU Agreement. Under certain circumstances specified in the Emergence RSU Agreement, if a participant's employment is terminated during any fiscal year in which the applicable percentage of the Adjusted EBITDA Target for such fiscal year is met, a pro rata portion of such participant's RSUs will vest based on the number of days such participant was employed during such fiscal year. The aggregate grant date fair value of the RSUs issued upon emergence was \$1.6 million.

As of December 31, 2009, 200,000 such RSUs vested with common shares to be issued in the first quarter of 2010, upon approval by the Compensation Committee.

Stock Option Award Agreements

On July 1, 2009, 387,403 service-based stock options were granted to certain employees and executive officers pursuant to the terms and conditions set forth in the form of stock option agreement (the "Emergence Option Agreement") and 100,000 performance-based stock options were granted to certain employees and executive officers pursuant to the terms and conditions set forth in the form of performance stock option agreement (the "Emergence Performance Option Agreement").

Emergence Option Agreement

The Emergence Option Agreement provides for the grant of a nonqualified stock option to purchase shares of New Common Stock at an exercise price per share equal to \$12.22. The option vests ratably over two years, with 25% of the option vesting every six months after the date of grant, July 1, 2009. The stock option expires on the tenth anniversary of the Effective Date (the "Expiration Date"). The aggregate grant date fair value of the emergence stock options granted was \$94 thousand.

Emergence Performance Option Agreement

The Emergence Performance Option Agreement provides for the grant of a nonqualified stock option to purchase shares of New Common Stock at an exercise price per share equal to \$12.22. The performance option vests in two equal tranches if at least 115% of the specified Adjusted EBITDA Target is attained for any fiscal year during the term of the Emergence Performance Option Agreement. Under certain circumstances specified in the Emergence Performance Option Agreement, if a participant's employment is terminated during any fiscal year in which the applicable percentage of the Adjusted EBITDA Target for such fiscal year is met, a pro rata portion (based on the number of days such participant was employed during such fiscal year) of such participant's performance options will vest and remain exercisable until the earlier of one year following the date on which the Committee determines the applicable percentage of such Adjusted EBITDA Target has been attained (the "Measurement Date") and the tenth anniversary of the Effective Date (the "Expiration Date"). Any unvested portion of the performance option which is not vested and exercisable as of the Measurement Date will thereafter terminate. The performance option expires on the Expiration Date. The aggregate grant date fair value of the emergence performance stock options granted was \$28 thousand.

Non-Employee Directors' Grants

Unless otherwise provided by the Committee, immediately following each annual meeting of the shareholders of Group during the term of the Management Compensation Plan commencing with the 2010 annual meeting of shareholders, each non-employee director will be granted a nonqualified stock option to purchase 10,000 shares of New Common Stock with an exercise price per share equal to the fair market value of a share of New Common Stock on the date of grant. Each stock option so granted will vest one third upon grant, one third on the first anniversary and one third on the second anniversary, such that 100% of the option will be vested and exercisable on the second anniversary of the grant date (subject to continued service as a non-employee director through each applicable vesting date). All other terms and conditions of the grants will be established by the Committee and set forth in the non-employee director's award agreement.

A summary of Successor's stock option activity during the six months ended December 31, 2009 is as follows:

	Successor Six Months Ended December 31, 2009	
	Shares	Weighted Average Exercise Price
Outstanding—July 1, 2009		\$ —
Granted	487,403	\$ 12.22
Exercised	_	\$ —
Forfeitures	(9,204)	\$ 12.22
Outstanding—December 31, 2009	478,199	\$ 12.22
Eligible for exercise	144,105	\$ 12.22

The weighted average fair value at date of grant for options granted during the six months ended December 31, 2009 was \$0.12 per option. The fair value of each option grant was estimated on the date of grant using the Black-Scholes option-pricing model with the following assumptions:

	Successor
	Six Months Ended
	December 31, 2009
Expected dividend yield	0%
Expected stock price volatility (1)	42%
Risk-free interest rate	2.0%
Expected option term	4 years

Expected volatility is based upon the historical volatility of the Company's stock price. Because of the short trading history of the Company's new Common Stock, the Company calculates the expected volatility by averaging the historical volatility of the Company's New Common Stock price from July 13, 2009 to December 31, 2009 at a rate of 36.4% and the historical volatility of the stock price of a peer group in the telecommunications industry with similar market capitalization from October 1, 2004 to September 30, 2009 at a rate of 48.3%.

The following table summarizes information about Successor's stock options outstanding at December 31, 2009:

		Options Outstanding				Options Exe	rcisable	
		Weighted			·	Weighted		
		Average	Weighted			Average	Weighted	
		Remaining	Average			Remaining	Average	
	Total	Life in	Exercise	Intrinsic	Total	Life in	Exercise	Intrinsic
Range of Option Prices	Outstanding	Years	Price	Value	Exercisable	Years	Price	Value
\$12.22	478,199	9.5	\$ 12.22	\$ —	144,105	9.5	\$ 12.22	\$ —

For Emergence Performance Option and RSU compensation expense calculation, the Company assumes that it will meet the specified Adjusted EBITDA Targets in the next two consecutive fiscal years; therefore, according to the Plan, the two equal tranches of the options and RSUs will vest in Year 1 and Year 2, consecutively.

As of December 31, 2009, Successor had 0.5 million unvested awards outstanding of which \$0.6 million of compensation expense will be recognized over the weighted average remaining vesting period of 1.0 years. The number of unvested awards expected to vest is 0.5 million awards, with a weighted average remaining life of 9.5 years, a weighted average exercise price of \$12.22, and an intrinsic value of \$0.

The Successor Company recorded \$1.2 million stock-based compensation expenses for the six months ended December 31, 2009, under guidance in ASC No. 718.

Predecessor

Under the Plan of Reorganization, all stock options granted under the Predecessor's Equity Incentive Plan were cancelled as of the Effective Date and \$64 thousand of share-based compensation expense was recognized due to the accelerated recognition of the cancelled options.

The Predecessor Company recorded \$27 thousand, \$0.3 million and \$0.2 million stock-based compensation expenses for the six months ended July 1, 2009 and the years ended December 31, 2008 and 2007, respectively, under guidance in ASC No. 718.

13. EMPLOYEE BENEFIT PLANS

The Company sponsors a 401(k) employee benefit plan (the "401(k) Plan") that covers substantially all United States based employees. Employees may contribute amounts to the 401(k) Plan not to exceed statutory limitations. The 401(k) Plan provides an employer matching contribution in cash of 50% of the first 6% of employee annual salary contributions capped at \$2,000 which are subject to three-year cliff vesting.

Successor

The matching contribution made during the six months ended December 31, 2009 was \$87 thousand.

Predecessor

The matching contribution made for the six months ended July 1, 2009, and the years ended 2008 and 2007 was \$94 thousand, \$168 thousand and \$262 thousand, respectively.

14. RELATED PARTIES

There were no material related party transactions during 2009, 2008 or 2007.

15. OPERATING SEGMENT AND RELATED INFORMATION

The Company has six reportable operating segments based on management's organization of the enterprise into geographic areas—United States, Canada, Europe, Australia, Brazil and the wholesale business from the United States and Europe managed as a separate global segment. During the quarter ended June 30, 2009, management identified its Brazil operating segment as a reportable segment due to management's increased focus on it as a separate market and operations. The Company has appropriately presented its Brazil segment for the current periods and prior periods presented. The Company evaluates the performance of its segments and allocates resources to them based upon net revenue and income (loss) from operations. The accounting policies of the segments are the same as those described in the summary of significant accounting policies. Net revenue by geographic segment is reported on the basis of where services are provided. The Company has no single customer representing greater than 10% of its revenues. Corporate assets, capital expenditures and property and equipment-net are included in the United States segment, while corporate expenses are presented separately in Income (loss) from operations. The wholesale business' assets are indistinguishable from the respective geographic segments. Therefore, any reporting related to the wholesale business for assets, capital expenditures or other balance sheet items is impractical.

Summary information with respect to the Company's segments is as follows:

	Successor		Predecessor	
	Six Months Ended December 31, 2009	Six Months Ended July 1, 2009	Year Ended December 31, 2008	Year Ended December 31, 2007
Net Revenue by Geographic Region				
United States	\$ 71,315	\$ 78,060	\$ 170,552	\$ 166,799
Canada	116,091	108,306	260,834	262,412
Europe	95,824	88,102	177,760	172,938
Australia	132,656	110,502	276,588	285,962
Brazil	8,448	6,246	10,129	7,918
Total	\$ 424,334	\$391,216	\$ 895,863	\$ 896,029
Net Revenue by Segment				
United States	\$ 31,132	\$ 35,013	\$ 86,579	\$ 101,583
Canada	116,091	108,306	260,834	262,200
Europe	27,527	26,667	64,629	68,943
Australia	132,656	110,502	276,414	284,935
Wholesale	108,480	104,482	197,278	170,450
Brazil	8,448	6,246	10,129	7,918
Total	\$ 424,334	\$391,216	\$ 895,863	\$ 896,029
Provision for Doubtful Accounts Receivable				
United States	\$ 1,051	\$ 1,456	\$ 2,830	\$ 1,919
Canada	1,077	1,106	2,708	2,979
Europe	448	210	765	817
Australia	1,837	1,738	3,948	3,509
Wholesale	441	516	1,092	958
Brazil	160	114	597	168
Total	\$ 5,014	\$ 5,140	\$ 11,940	\$ 10,350
Income (Loss) from Operations				
United States	\$ 2,564	\$ 4,304	\$ 2,284	\$ 11,421
Canada	3,127	18,738	41,393	37,138
Europe	(2,144)	(309)	(3,968)	(13,947)
Australia	3,073	10,123	10,408	15,594
Wholesale	971	1,372	(1,407)	(3,097)
Brazil	(206)	231	(437)	(832)
Total From Operating Segments	7,385	34,459	48,273	46,277
Corporate	(4,090)	(6,670)	(9,468)	(14,553)
Total	\$ 3,295	\$ 27,789	\$ 38,805	\$ 31,724
Capital Expenditures				
United States	\$ 43	\$ 73	\$ 1,579	\$ 1,819
Canada	5,261	3,127	10,394	23,434
Europe	40	174	839	3,418
Australia	3,795	1,997	12,376	15,860
Brazil	257	289	253	214
Total	\$ 9,396	\$ 5,660	\$ 25,441	\$ 44,745
				

The above capital expenditures exclude assets acquired in business combinations and under terms of capital lease and vendor financing obligations.

	Successor December 31, 2009	Predecessor December 31, 2008	
Property and Equipment—Net			
United States	\$ 10,760	\$ 15,590	
Canada	58,927	44,234	
Europe	4,955	6,542	
Australia	71,682	45,282	
Brazil	1,282	504	
Total	\$ 147,606	\$ 112,152	

	<u>Successor</u> December 31, 2009	Predecessor December 31, 2008		
Assets				
United States	\$ 133,276	\$ 28,230		
Canada	194,600	121,105		
Europe	84,587	75,196		
Australia	138,988	101,650		
Brazil	7,463	4,263		
Total	\$ 558,914	\$ 330,444		

The Company offers four main products—retail voice, wholesale voice, data/Internet and VOIP. Summary net revenue information with respect to the Company's products is as follows (in thousands):

	Successor		Predecessor	
	Six Months Ended December 31, 2009	Six Months Ended July 1, 2009	Twelve Months Ended December 31, 2008	Twelve Months Ended December 31, 2007
Retail voice	\$ 200,054	\$193,669	\$ 461,817	\$ 481,160
Wholesale	108,480	104,482	197,278	170,450
Data/Internet	87,061	72,979	182,891	196,198
Retail VOIP	28,741	20,086	53,877	48,221
Total	\$ 424,334	\$391,216	\$ 895,863	\$ 896,029

16. QUARTERLY RESULTS OF OPERATIONS (UNAUDITED)

The following is a tabulation of the unaudited quarterly results of operations for the years ended December 31, 2009 and 2008.

		For the Quarter Ended				
		(in thousands, except per share amounts)				
	Predec	cessor	Successor			
	March 31, 2009	July 1, 2009	September 30, 2009	December 31, 2009		
Net revenue	\$194,474	\$196,742	\$ 207,947	\$ 216,387		
Cost of revenue (exclusive of depreciation)	\$129,374	\$125,914	\$ 136,076	\$ 142,341		
Income from operations	\$ 13,627	\$ 14,162	\$ 620	\$ 2,675		
Income (loss) from continuing operations	\$ 13,997	\$457,550	\$ 2,485	\$ 4,961		
Income (loss) from discontinued operations	\$ (393)	\$ (283)	\$ —	\$ (271)		
Net income (loss)	\$ 13,991	\$457,163	\$ 2,165	\$ 4,567		
Basic income (loss) per common share:						
Income (loss) from continuing operations	\$ 0.10	\$ 3.21	\$ 0.24	\$ 0.50		
Loss from discontinued operations	-	(0.01)	(0.01)	(0.03)		
Net income (loss)	\$ 0.10	\$ 3.20	\$ 0.23	\$ 0.47		
Diluted income (loss) per common share:						
Income (loss) from continuing operations	\$ 0.08	\$ 2.64	\$ 0.24	\$ 0.49		
Loss from discontinued operations	-	_	(0.01)	(0.03)		
Net income (loss)	\$ 0.08	\$ 2.64	\$ 0.23	\$ 0.46		

	For the Quarter Ended (in thousands, except per share amounts) Predecessor			
	March 31, 2008	June 30, 2008	September 30, 2008	December 31, 2008
Net revenue	\$225,434	\$235,897	\$ 231,256	\$ 203,276
Cost of revenue (exclusive of depreciation)	\$141,484	\$142,495	\$ 149,834	\$ 136,052
Income from operations	\$ 9,713	\$ 15,227	\$ 7,136	\$ 6,729
Income (loss) from continuing operations	\$ (2,954)	\$ 46,545	\$ (32,784)	\$ (35,512)
Income (loss) from discontinued operations	\$ (45)	\$ (21)	\$ (436)	\$ 176
Net income (loss)	\$ (2,999)	\$ 46,524	\$ (33,220)	\$ (35,336)
Basic income (loss) per common share:				
Income (loss) from continuing operations	\$ (0.02)	\$ 0.33	\$ (0.23)	\$ (0.25)
Loss from discontinued operations	(0.00)	(0.00)	(0.00)	(0.00)
Net income (loss)	\$ (0.02)	\$ 0.33	\$ (0.23)	\$ (0.25)
Diluted income (loss) per common share:				
Income (loss) from continuing operations	\$ (0.02)	\$ 0.25	\$ (0.23)	\$ (0.25)
Loss from discontinued operations	(0.00)	(0.00)	(0.00)	(0.00)
Net income (loss)	\$ (0.02)	\$ 0.25	\$ (0.23)	\$ (0.25)

Quarterly and year-to-date computations of per share amounts are made independently; therefore, the sum of per share amounts for the quarters may not agree with per share amounts for the year.

References to "Predecessor" or "Predecessor Company" refer to the operations of the Company prior to July 1, 2009, except for Predecessor's July 1, 2009 statements of operations which reflect only the effect of the plan of reorganization adjustments and fresh-start accounting as of such date and do not reflect any operating results. (See Note 4 "Fresh-Start Accounting" for further detail).

17. LOSS ON SALE OR DISPOSAL OF ASSETS

Successor

During the six months ended December 31, 2009, the Company recognized a loss of \$0.2 million associated with the sale or disposal of specific long-lived assets which were taken out of service. The assets disposed or sold were comprised of switch and peripheral and other network equipment.

Predecessor

During the six months ended July 1, 2009, the company recognized a gain of \$43 thousand associated with the sale or disposal of specific long-lived assets which were taken out of service. The assets disposed or sold were comprised of switch and peripheral and other network equipment.

During the year ended December 31, 2008, the Company recognized a gain of \$6.0 million associated with the sale or disposal of specific long-lived assets, as detailed below.

In the third quarter of 2008, a consolidated, variable interest entity in Canada, of which the Company currently owns 45.6% of the equity, sold certain primarily rural WIMAX spectrum (spectrum for transmission of sound, data, and video) assets (representing approximately 10% of the entity's spectrum population coverage) for cash consideration of \$4.9 million (\$5.0 million CAD). The gain on the sale was \$4.6 million; the noncontrolling interest on this gain was \$2.5 million and was included in interest income and other income (expense). Total noncontrolling interest is \$2.9 million included in other long-term liabilities. The cash proceeds from the sale of \$4.9 million can be used for operations within the variable interest entity, but requires unanimous shareholder consent for a dividend distribution. Note that upon the adoption of the noncontrolling interest provisions of ASC No. 810, the presentation within the financial statements for these items has changed in accordance with these provisions. See "Note 3—Summary Of Significant Accounting Policies; *Newly Adopted Accounting Principles.*"

Additionally, a \$0.8 million gain was recognized for the sale of certain surplus fiber assets in the United States; a \$1.7 million gain was recognized for the sale of a minority equity investment in a Japanese entity; and a \$0.9 million loss was recognized on the disposal of certain equipment related to the WIMAX operations in Canada.

During the year ended December 31, 2007, the Company recognized a charge of \$1.5 million for the sale or disposal of specific long-lived assets which were taken out of service. The assets disposed or sold were comprised of switch and peripheral equipment and other network equipment.

18. GAIN (LOSS) ON EARLY EXTINGUISHMENT OR RESTRUCTURING OF DEBT

In December 2009, the Company issued \$130.0 million principal amount of 13% Senior Secured Notes due 2016. The proceeds from the offering were used to pay-off the Senior Secured Term Loan Facility and the Canadian Credit Facility. The Company recorded a \$4.1 million loss on the early extinguishment of debt for the six months ended December 31, 2009 which included the write-off of related deferred financing costs, a prepayment premium and professional fees.

In 2008, the Company exchanged \$49.0 million principal amount of the 8% Senior Notes, \$33.0 million principal amount of the 5% Exchangeable Senior Notes, \$43.1 million principal amount of the 3 ³/4% Convertible Senior Notes, \$5.3 million principal amount of the 12 ³/4% Senior Notes for \$67.1 million principal amount of its

newly issued 14 ¹/4% Senior Secured Notes and \$4.7 million of cash, resulting in a gain on restructuring of debt of \$32.2 million including the expensing of related financing costs, which was adjusted to \$32.3 million in the third quarter. The Company also made open market purchases of \$0.8 million principal amount of the 12 ³/4 % Senior Notes, \$13.8 million principal amount of the Step Up Convertible Subordinated Debentures and \$2.1 million principal amount of the 14 ¹/4% Senior Secured Notes, resulting in a \$0.1 million, \$2.1 million and \$2.0 million gain, respectively, on early extinguishment of debt including the write-off of related deferred financing costs, discount and effective interest.

In 2007, the Company issued in a private transaction \$57.2 million principal amount of the 14 \(^1/4\%\) Senior Secured Notes, in exchange for \$40.7 million principal amount of the Company's outstanding 12 \(^3/4\%\) Senior Notes and \$23.6 million in cash. This exchange has been accounted for as a modification of debt with a portion deemed to be a troubled debt restructuring. The Company recorded \$5.1 million in costs associated with the issuance of the 14\(^1/4\%\) Senior Secured Notes, which have been recorded as loss on restructuring of debt. The Company refinanced an existing Canadian credit facility and recognized a \$0.9 million loss on early extinguishment of debt for pre-payment penalties and the write-off of related deferred financing costs. The Company also exchanged 6,000,000 shares of the Company's common stock for the extinguishment of \$5.0 million principal amount of its Step Up Convertible Subordinated Debentures resulting in a loss on early extinguishment of debt of \$2.3 million, including the write-off of related deferred financing costs and debt discount. These losses were offset by a \$0.5 million gain on early extinguishment of debt on forgiveness of equipment financing and a \$0.3 million gain on open market purchases of \$10.5 million principal amount of the 12 \(^3/4\%\) Senior Notes including the write-off of related deferred financing costs.

19. DISCONTINUED OPERATIONS

In the first quarter 2009, the Company sold certain assets of its Japan retail operations. The sale price was \$0.4 million (40 million Japanese yen), which included \$0.2 million (20 million Japanese yen) in cash and \$0.2 million (20 million Japanese yen) receivable. The Company recorded a \$0.3 million gain from sale of assets. The Company reported Japan retail operations as a discontinued operation beginning in the fourth quarter of 2008.

In the second quarter 2008, the Company determined it would sell its German retail operations. However, buyers were not found; therefore the Company decided to cease operations of the German retail business during the first quarter of 2009.

In August 2007, the Company sold its 51% interest in its German telephone installation system subsidiaries. The sale price was \$0.8 million (0.6 million Euros), which included \$0.5 million (0.4 million Euros) in cash and \$0.3 million (0.2 million Euros) for payment of outstanding intercompany debt. For the intercompany debt payment, the Company received \$0.1 million (0.1 million Euros) in cash at closing. The balance owing is represented by a note receivable and will be paid in fifteen equal monthly installment payments. As a result, the Company recorded a \$0.2 million gain from sale of assets. Net assets held for sale were \$0.6 million at the closing date.

In February 2007, the Company sold its Australian domain name registry and web hosting subsidiary, Planet Domain. The sale price was \$6.5 million (\$8.3 million AUD). The Company received \$5.5 million in net cash proceeds from the transaction after closing adjustments. As a result, the Company recorded a \$6.0 million gain from sale of assets. The net assets of Planet Domain were \$0.2 million at the closing date.

As a result of these events, the Company's consolidated financial statements for all periods presented reflect the Japan retail operations, German retail operations, discontinued German subsidiary and Planet Domain operations as discontinued operations for the six months ended December 31, 2009, six months ended July 1, 2009, years ended December 31, 2008 and 2007. Accordingly, revenue, costs, and expenses of the discontinued operations have been excluded from the respective captions in the consolidated statements of operations. The net operating results of the discontinued operations have been reported, net of applicable income taxes as loss from discontinued operations.

Summarized operating results of the discontinued operations for the six months ended December 31, 2009, six months ended July 1, 2009 and years ended December 31, 2008 and 2007 are as follows (in thousands):

Successor		Predecessor	
		Year Ended 1	Dagamban 21
December 31,		rear Ended	December 51,
2009	2009	2008	2007
\$ —	\$ 300	\$ 3,851	\$ 9,961
\$ 38	\$ 995	\$ 4,610	\$ 10,233
\$ (38)	\$ (695)	\$ (759)	\$ (272)
\$ —	\$ —	\$ —	\$ (26)
\$ (233)	\$ 19	\$ 60	\$ 39
\$	<u>\$</u>	\$ 376	\$ (6)
\$ (271)	\$ (676)	\$ (323)	\$ (265)
\$	\$ —	\$ (3)	\$ (3)
\$ (271)	\$ (676)	\$ (326)	\$ (268)
	Six Months Ended December 31, 2009	Six Months Ended December 31, 2009 Six Months Ended July 1, 2009 \$ — \$ 300 \$ 38 \$ 995 \$ (38) \$ (695) \$ — \$ — \$ (233) \$ 19 \$ — \$ — \$ (271) \$ (676) \$ — \$ —	Six Months Ended December 31, 2009 Six Months Ended July 1, 2009 Year Ended 1 \$ 300 \$ 3,851 \$ 38 \$ 995 \$ 4,610 \$ (38) \$ (695) \$ (759) \$ — \$ — \$ — \$ (233) \$ 19 \$ 60 \$ — \$ 376 \$ (271) \$ (676) \$ (323) \$ — \$ (3)

20. BASIC AND DILUTED INCOME (LOSS) PER COMMON SHARE

Basic income (loss) per common share is calculated by dividing income (loss) attributable to common stockholders by the weighted average common shares outstanding during the period. Diluted income per common share adjusts basic income per common share for the effects of potentially dilutive common share equivalents.

Successor

Potentially dilutive common shares for Successor include the dilutive effects of common shares issuable under the Successor's Management Compensation Plan, including stock options, stock warrants and restricted stock units (RSUs), using the treasury stock method, as well as contingent value rights (CVRs).

For the six months ended December 31, 2009, the following could potentially dilute income per common share in the future but was excluded from the calculation of diluted income per common share due to its antidilutive effect:

- 0.7 million shares issuable upon exercise of stock options and RSUs under the Successor's Management Compensation Plan;
- 4.5 million shares issuable upon exercise of stock warrants; and
- 2.7 million shares issuable upon exercise of CVRs.

Predecessor

Potentially dilutive common shares for Predecessor primarily included the dilutive effects of common shares issuable under Predecessor's Equity Incentive Plan computed using the treasury stock method and the dilutive effects of shares issuable upon conversion of its 5% Exchangeable Senior Notes, the Step Up Convertible Subordinated Debentures, the 3 ³/₄% Convertible Senior Notes and the 2000 Convertible Subordinated Debentures.

For the six months ended July 1, 2009, the following was potentially dilutive but was excluded from the calculation of diluted income per common share due to its antidilutive effect:

• 7.8 million shares issuable under the Predecessor's Equity Incentive Plan.

The Company had no dilutive common share equivalents during the year ended December 31, 2008, due to the results of operations being a net loss. For the year ended December 31, 2008, the following were potentially dilutive but were excluded from the calculation of diluted loss per common share due to their antidilutive effects:

- 7.8 million shares issuable under the Company's stock option compensation plans,
- 19.5 million shares issuable upon the conversion of the 5% Exchangeable Senior Notes,
- 7.3 million shares issuable upon the conversion of the Step Up Convertible Subordinated Debentures, and
- 3.7 million shares issuable upon conversion of the 3 3/4% Convertible Senior Notes.

For the year ended December 31, 2007, the following could potentially dilute income per common share in the future but were excluded from the calculation of diluted income per common share due to their antidilutive effects:

- 7.4 million shares issuable under the Company's stock option compensation plans,
- 8.3 million shares issuable upon conversion of the 3 3/4% Convertible Senior Notes, and
- 0.1 million shares issuable upon the conversion of the 2000 Convertible Subordinated Debentures.

A reconciliation of basic income (loss) per common share to diluted income (loss) per common share is below (in thousands, except per share amounts):

	Successor	Predecessor		
	Six Months Ended December 31, 2009	Six Months Ended July 1, 2009	Year Ended December 31, 2008	Year Ended December 31, 2007
Income (loss) from continuing operations	\$ 7,113	\$471,579	\$ (24,705)	\$ 9,872
Loss from discontinued operations, net of tax	(271)	(676)	(326)	(268)
Gain from sale of discontinued operations, net of tax	(110)	251		6,132
Net income (loss) attributable to common stockholders—basic	6,732	471,154	(25,031)	15,736
Adjustment for interest expense on Step Up Convertible Subordinated Debentures	_	210	_	1,846
Adjustment for interest expense on 3 3/4% Convertible Senior Notes	_	332	_	_
Income attributable to common stockholders—diluted	\$ 6,732	\$471,696	\$ (25,031)	\$ 17,582
Weighted average common shares outstanding—basic	9,600	142,695	142,643	128,771
Restricted Stock Units (1)	200	_	_	
5% Exchangeable Senior Notes		19,474	_	46,936
Step Up Convertible Subordinated Debentures	_	7,280	_	20,751
3 ³ /4% Convertible Senior Notes	_	3,668	_	
In-the-money options exercisable under stock option compensation plans	<u> </u>			12
Weighted average common shares outstanding—diluted	9,800	173,117	142,643	196,470
Basic income per common share:				
Income from continuing operations attributable to common stockholders	\$ 0.74	\$ 3.30	\$ (0.17)	\$ 0.07
Loss from discontinued operations	(0.03)	(0.00)	(0.01)	(0.00)
Gain from sale of discontinued operations	(0.01)	0.00	_	0.05
Net income (loss) attributable to common stockholders	\$ 0.70	\$ 3.30	\$ (0.18)	\$ 0.12
Diluted income per common share:				
Income from continuing operations attributable to common				
stockholders	\$ 0.73	\$ 2.72	\$ (0.17)	\$ 0.06
Loss from discontinued operations	(0.03)	(0.00)	(0.01)	(0.00)
Gain from sale of discontinued operations	(0.01)	0.00	<u> </u>	0.03
Net income (loss) attributable to common stockholders	\$ 0.69	\$ 2.72	\$ (0.18)	\$ 0.09

⁽¹⁾ Restricted Stock Units that vested in 2009 and common stock that was issuable subsequent to December 31, 2009

21. REORGANIZATION ITEMS, NET

Reorganization items, net, represents amounts incurred as a direct result of the Chapter 11 filings and is presented separately in the Consolidated Statements of Operations. The following describes the components of reorganization items, net (in thousands):

	Six I	ccessor Months Ended ber 31, 2009	<u>Predecessor</u> Six Months Ended July 1, 2009
Professional Fees	\$	(562)	\$ (12,067)
Gain on Extinguishment of debt		_	243,185
Revaluation of assets and liabilities		_	188,612
Debt Premium, Discount and Deferred Financing Costs Write-off		_	(91)
Reversal of Future Interest Payments Recorded as Long Term Obligations		_	20,453
Interest Income		_	2
Reorganization Items, net	\$	(562)	\$ 440,094

Professional fees include financial, legal and other services directly associated with the reorganization process. Payments for reorganization expense for the six months ended December 30, 2009 were \$7.6 million. Payments for reorganization expense for the six months ended July 1, 2009 were \$4.6 million. In accordance with ASC No. 852, the Company ceased amortization of debt premiums, discounts and deferred financing costs related to the liabilities subject to compromise on the Petition Date. The \$3.5 million of unamortized debt premiums and discounts have been written off and recorded as a gain, offset by the expensing of \$3.6 million of unamortized deferred financing costs, as an adjustment to the net carrying value of the pre-petition debt. Long term debt was further reduced by \$20.5 million of future interest payable that previously had been recorded as a portion of long-term obligations for the 14 1/4% Senior Secured Notes and 5% Exchangeable Senior Notes as the issuance of these notes had been deemed troubled debt restructurings. Upon emergence from bankruptcy, Predecessor debt in the amount of \$439.6 million was written off along with the accrued interest related to this debt in the amount of \$11.5 million. In addition, the Successor Company issued \$123.5 million of new debt upon emergence and issued 4.8 million shares of Successor Company common stock to the former holders of the Company's debt.

22. GUARANTOR/NON-GUARANTOR CONSOLIDATING CONDENSED FINANCIAL INFORMATION

Primus Telecommunications IHC, Inc.'s 14 1/4% Senior Secured Notes were fully, unconditionally, jointly and severally guaranteed by Group on a senior basis and by Holding, Primus Telecommunications, Inc., TresCom International Inc., Least Cost Routing, Inc., TresCom U.S.A., Inc., iPRIMUS USA, Inc., and iPRIMUS.com, Inc., all 100% indirectly owned subsidiaries of Group (collectively, the "Other Guarantors"). Group has a 100% ownership in Holding and no direct subsidiaries other than Holding.

On the Effective Date, IHC, each of the Grantors party and U.S. Bank National Association, as collateral agent, entered into a First Amendment to the Collateral Agreement (the "Amended Collateral Agreement"), to provide that the obligations of both IHC and PTII, an indirect wholly owned subsidiary of Group, was secured by PTII's assets, including 65% of the voting stock of foreign subsidiaries owned by PTII. In addition, on the Effective Date, Group and Holding entered into an Assumption Agreement in favor of U.S. Bank National Association, as collateral agent, pursuant to which each of Group and Holding became party to the Amended Collateral Agreement. As a result, Group and Holding's existing guarantees of the Senior Subordinated Secured Notes are secured by a lien on the property of Group and Holding, respectively.

Accordingly, the following consolidating condensed financial information for the six months ended December 31, 2009 for Successor and for the six months ended July 1, 2009, and years ended December 31, 2008 and 2007 are included for (a) Group on a stand-alone basis; (b) Primus Telecommunications IHC, Inc. (IHC) on a stand-alone basis; (c) the Other Guarantor subsidiaries on a combined basis and (e) Group on a consolidated basis. The plan and fresh-start accounting adjustments reflected in Predecessor's Consolidated Condensed Statements of Operations on July 1, 2009 are not presented separately in this presentation.

Investments in subsidiaries are accounted for using the equity method for purposes of the consolidating presentation. The principal elimination entries eliminate investments in subsidiaries, intercompany balances and intercompany transactions.

	Successor								
			For the Six Mont		per 31, 2009				
	-			Non					
			Guarantor	Guarantor					
	PTGI	IHC	Subsidiaries	Subsidiaries	Eliminations	Consolidated			
NET REVENUE	\$ —	\$ —	\$ 59,751	\$ 364,583	\$ —	\$ 424,334			
OPERATING EXPENSES									
Cost of revenue (exclusive of depreciation included below)	_	_	48,017	230,400	_	278,417			
Selling, general and administrative	266	1	13,091	89,553	_	102,911			
Depreciation and amortization	_	_	2,334	37,196	_	39,530			
(Gain) loss on sale or disposal of assets			11	170		181			
Total operating expenses	266	1	63,453	357,319		421,039			
INCOME (LOSS) FROM OPERATIONS	(266)	(1)	(3,702)	7,264	_	3,295			
INTEREST EXPÉNSE	`— `	(8,797)	(6,196)	(2,330)	_	(17,323)			
ACCRETION (AMORTIZATION) ON DEBT PREMIUM/ DISCOUNT, net	_		(3)		_	(3)			
LOSS ON EARLY EXTINGUISHMENT OR RESTRUCTURING OF DEBT	_	_	(2,728)	(1,418)	_	(4,146)			
LOSS FROM CONTINGENT VALUE RIGHTS VALUATION	(2,804)	_	· — ·	· — ·	_	(2,804)			
INTEREST AND OTHER INCOME	_	_	135	429	_	564			
FOREIGN CURRENCY TRANSACTION GAIN	13	8,989	6	9,376	_	18,384			
INTERCOMPANY INTEREST	(616)	7,698	(5,118)	(1,964)	_	_			
MANAGEMENT FEE	_	_	2,070	(2,070)	_	_			
ROYALTY FEE		6,314		(6,314)					
INCOME (LOSS) BEFORE REORGANIZATION ITEMS, INCOME TAXES AND EQUITY IN NET INCOME OF SUBSIDIARIES	(3,673)	14,203	(15,536)	2,973	_	(2,033)			
REORGANIZATION ITEMS—NET	(515)		95	(142)	_	(562)			
INCOME (LOSS) BEFORE INCOME TAX AND EQUITY IN NET INCOME OF SUBSIDIARIES	(4,188)	14,203	(15,441)	2,831		(2,595)			
INCOME TAX BENEFIT (EXPENSE)	64	916	34	9,027	_	10,041			
INCOME ITA BENEFIT (EATEROE)				3,027		10,041			
INCOME (LOSS) BEFORE EQUITY IN NET INCOME (LOSS) OF SUBSIDIARIES	(4,124)	15,119	(15,407)	11,858	_	7,446			
EQUITY IN NET INCOME (LOSS) OF SUBSIDIARIES	10,856		26,671		(37,527)				
INCOME (LOSS) FROM CONTINUING OPERATIONS	6,732	15,119	11,264	11,858	(37,527)	7,446			
LOSS FROM DISCONTINUED OPERATIONS, net of tax		_		(271)	`	(271)			
LOSS FROM SALE OF DISCONTINUED OPERATIONS, net of tax	_	_	_	(110)	_	(110)			
NET (LOSS) INCOME	6,732	15,119	11,264	11,477	(37,527)	7,065			
Less: Net income attributable to the noncontrolling interest	_	_		(333)		(333)			
NET INCOME (LOSS) ATTRIBUTABLE TO PRIMUS TELECOMMUNICATIONS GROUP,									
INCORPORATED	\$ 6,732	\$15,119	\$ 11,264	\$ 11,144	\$ (37,527)	\$ 6,732			
AMOUNTS ATTRIBUTABLE TO COMMON SHAREHOLDERS OF PRIMUS									
TELECOMMUNICATIONS GROUP, INCORPORATED	# C #DD	* 4 = 440		A 44 505	A (0==0=)				
Income (loss) from continuing operations, net of tax	\$ 6,732	\$15,119	\$ 11,264	\$ 11,525	\$ (37,527)	\$ 7,113			
Loss from discontinued operations	_			(271)		(271)			
Gain from sale of discontinued operations				(110)		(110)			
Net income (loss)	\$ 6,732	\$15,119	\$ 11,264	\$ 11,144	\$ (37,527)	\$ 6,732			

	Predecessor										
	For the Six Months Ended July 1, 2009										
			Guaranto	r	Non Guarantor						
	PTGI	IHC	Subsidiari		Subsidiaries	Eliminations	Consolidated				
NET REVENUE	\$ —	\$ —	\$ 65,3	61	\$ 325,855	\$ —	\$ 391,216				
OPERATING EXPENSES											
Cost of revenue (exclusive of depreciation included below)	_	_	52,0	58	203,230	_	255,288				
Selling, general and administrative	4,638	23	12,5	87	78,588	_	95,836				
Depreciation and amortization		_	1,3	17	11,029	_	12,346				
(Gain) loss on sale or disposal of assets			(1	77)	134		(43)				
Total operating expenses	4,638	23	65,7	85	292,981	_	363,427				
INCOME (LOSS) FROM OPERATIONS	(4,638)	(23)	(4:	24)	32,874		27,789				
INTEREST EXPENSE	(794)	(3,331)	(7,8		(2,143)	_	(14,135)				
ACCRETION (AMORTIZATION) ON DEBT PREMIUM/DISCOUNT, net	(129)	318		_ ´	`_ `	_	189				
INTEREST AND OTHER INCOME	`— ´	_		8	388	_	396				
FOREIGN CURRENCY TRANSACTION GAIN (LOSS)	2,632	8,349	(7)	05)	10,845	_	21,121				
INTERCOMPANY INTEREST	(4,169)	14,549	(8,7	64)	(1,616)	_	_				
MANAGEMENT FEE	_	_	4,1	52	(4,152)	_	_				
ROYALTY FEE		5,277		_	(5,277)						
INCOME (LOSS) BEFORE REORGANIZATION ITEMS, INCOME TAXES AND EQUITY IN											
NET INCOME (LOSS) OF SUBSIDIARIES	(7,098)	25,139	(13,6	00)	30,919	_	35,360				
REORGANIZATION ITEMS—NET	(34,650)	79,592	286,2	79	108,873	_	440,094				
INCOME (LOSS) BEFORE INCOME TAX AND EQUITY IN NET INCOME (LOSS) OF											
SUBSIDIARÍES	(41,748)	104,731	272,6	79	139,792	_	475,454				
INCOME TAX EXPENSE	` — ′	(380)	(53)	(3,474)	_	(3,907)				
INCOME (LOSS) BEFORE EQUITY IN NET INCOME OF SUBSIDIARIES	(41,748)	104,351	272,6	26	136,318		471,547				
EQUITY IN NET INCOME OF SUBSIDIARIES	512,902	_	321,7		_	(834,645)	_				
INCOME FROM CONTINUING OPERATIONS	471,154	104,351	594,3	69	136,318	(834,645)	471,547				
LOSS FROM DISCONTINUED OPERATIONS,	1,10 .	10 1,001	55 .,5	00	150,515	(65 1,6 15)	17 1,5 17				
net of tax	_	_	_	_	(676)	_	(676)				
GAIN FROM SALE OF DISCONTINUED OPERATIONS, net of tax	_	_	_	_	251	_	251				
NET INCOME.	471,154	104,351	594,3	69	135,893	(834,645)	471,122				
Less: Net loss attributable to the noncontrolling interest			_		32	_	32				
NET INCOME ATTRIBUTABLE TO PRIMUS TELECOMMUNICATIONS GROUP,			•								
INCORPORATED	\$471,154	\$104,351	\$ 594,3	69	\$ 135,925	\$ (834,645)	\$ 471,154				
AMOUNTS ATTRIBUTABLE TO COMMON SHAREHOLDERS OF PRIMUS	ψ 17 1,10 1	ψ10 i,001	ψ 55 1,5	00	100,020	\$ (65.1,6.15)	ψ 17 1,13 T				
TELECOMMUNICATIONS GROUP, INCORPORATED											
Income from continuing operations, net of tax	\$471,154	\$104,351	\$ 594.3	60	\$ 136,350	\$ (834,645)	\$ 471,579				
Loss from discontinued operations	Ψ4/1,134	Ψ104,551	φ <i>53</i> 4,5		(676)	ψ (UJ4,U4J)	(676)				
Gain from sale of discontinued operations					251		251				
Net income	\$471,154	\$104,351	\$ 594,3	_	\$ 135,925	\$ (834,645)	\$ 471,154				
Net income	\$4/1,154	\$104,351	\$ 594,5	US	3 135,925	<u>a</u> (034,045)	φ 4/1,154				

	Predecessor									
		F	or the Year Endeo		08					
	PTGI	IHC	Guarantor Subsidiaries	Non Guarantor Subsidiaries	Eliminations	Consolidated				
NET REVENUE	\$ —	\$ —	\$ 141,065	\$ 754,798	\$ —	\$ 895,863				
OPERATING EXPENSES										
Cost of revenue (exclusive of depreciation included below)	_	_	114,897	454,968	_	569,865				
Selling, general and administrative	5,097	171	33,534	221,628		260,430				
Depreciation and amortization	_	_	3,245	29,546	_	32,791				
Gain on sale or disposal of assets			(806)	(5,222)		(6,028)				
Total operating expenses	5,097	171	150,870	700,920	_	857,058				
INCOME (LOSS) FROM OPERATIONS	(5,097)	(171)	(9,805)	53,878	_	38,805				
INTEREST EXPENSE	(5,301)	(15,900)	(27,818)	(4,869)	_	(53,888)				
ACCRETION (AMORTIZATION) ON DEBT PREMIUM/DISCOUNT, net	(858)	1,441	_	_	_	583				
GAIN ON EARLY EXTINGUISHMENT OR RESTRUCTURING OF										
DEBT	12,448	1,532	22,784	108	_	36,872				
INTEREST AND OTHER INCOME (EXPENSE)	18		(1)	3,262		3,279				
FOREIGN CURRENCY TRANSACTION LOSS	(3,523)	(16,270)	(525)	(27,245)	_	(47,563)				
INTERCOMPANY INTEREST	(9,390)	20,594	(12,075)	871						
MANAGEMENT FEE	_	_	5,925	(5,925)	_	_				
ROYALTY FEE		13,684		(13,684)						
INCOME (LOSS) BEFORE INCOME TAXES AND EQUITY IN NET										
INCOME (LOSS) OF SUBSIDIARIES	(11,703)	4,910	(21,515)	6,396	_	(21,912)				
INCOME TAX BENEFIT (EXPENSE)	211	379	558	(782)	_	366				
INCOME (LOSS) BEFORE EQUITY IN NET INCOME (LOSS) OF										
SUBSIDIARIES	(11,492)	5,289	(20,957)	5,614	_	(21,546)				
EQUITY IN NET INCOME (LOSS) OF SUBSIDIARIES	(13,539)	_	3,207	_	10,332	_				
INCOME (LOSS) FROM CONTINUING OPERATIONS	(25,031)	5,289	(17,750)	5,614	10,332	(21,546)				
LOSS FROM DISCONTINUED OPERATIONS, net of tax		_	` <u> </u>	(326)	_	(326)				
NET INCOME (LOSS)	\$(25,031)	\$ 5,289	\$ (17,750)	\$ 5,288	\$ 10,332	\$ (21,872)				
Less: Net income attributable to the noncontrolling interest		_		(3,159)	_	(3,159)				
NET INCOME (LOSS) ATTRIBUTABLE TO PRIMUS										
TELECOMMUNICATIONS GROUP, INCORPORATED	\$(25,031)	\$ 5,289	\$ (17,750)	\$ 2,129	\$ 10,332	\$ (25,031)				
AMOUNTS ATTRIBUTABLE TO COMMON SHAREHOLDERS OF PRIMUS TELECOMMUNICATIONS GROUP, INCORPORATED										
Income (loss) from continuing operations, net of tax	\$(25,031)	\$ 5,289	\$ (17,750)	\$ 2,455	\$ 10,332	\$ (24,705)				
Loss from discontinued operations				(326)		(326)				
Net income (loss)	\$(25,031)	\$ 5,289	\$ (17,750)	\$ 2,129	\$ 10,332	\$ (25,031)				
The mediae (1000)	Ψ(20,001)	Ψ 0,200	Ψ (±7,750)	Ψ 2,±23	Ψ 10,002	Ψ (25,051)				

For the Year Ended December 31, 2007 Non	lidated
	lidated
	lidated
Guarantor Guarantor PTGI IHC Subsidiaries Subsidiaries Eliminations Consc	
	896.029
OPERATING EXPENSES	,
	551.303
	281,010
Depreciation and amortization	30,529
Loss on sale or disposal of assets — — 1,463 —	1,463
<u> </u>	864,305
INCOME (LOSS) FROM OPERATIONS (4,843) (111) (4,783) 41,461 —	31,724
INTEREST EXPENSE (10,249) (12,470) (31,831) (6,797) —	(61,347)
ACCRETION (AMORTIZATION) ON DEBT PREMIUM/DISCOUNT, net (1,524) 1,075 — — —	(449)
LOSS ON EARLY EXTINGUISHMENT OR RESTRUCTURING OF DEBT (2,000) (5,144) (108) (400) —	(7,652)
EOS ON LINE EXTRAORDINATION OF RESTRICT OF	5,665
FOREIGN CURRENCY TRANSACTION GAIN 4,767 8,049 777 19,106 —	32,699
INTERCOMPANY INTEREST (1,931) 15,084 (11,734) (1,419) —	_
MANAGEMENT FEE — 6,292 (6,292) —	_
ROYALTY FEE — 14,512 (578) (13,934) —	_
INCOME (LOSS) BEFORE INCOME TAXES AND EQUITY IN NET INCOME OF	
SUBSIDIARIES (15.231) 20.995 (41.895) 36.771 —	640
INCOME TAX BENEFIT (EXPENSE) (355) (1,552) (230) 11,369 —	9,232
INCOME (LOSS) BEFORE EQUITY IN NET INCOME OF SUBSIDIARIES (15,586) 19,443 (42,125) 48,140 —	9,872
EQUITY IN NET INCOME OF SUBSIDIARIES 31,322 — 78,913 — (110,235)	<u> </u>
INCOME FROM CONTINUING OPERATIONS 15,736 19,443 36,788 48,140 (110,235)	9,872
LOSS FROM DISCONTINUED OPERATIONS, net of tax — — — (268) —	(268)
GAIN ON SALE OF DISCONTINUED OPERATIONS, net of tax — — — 6,132 —	6,132
NET INCOME \$ 15,736 \$ 19,443 \$ 36,788 \$ 54,004 \$ (110,235) \$	15,736
Less: Net income attributable to the noncontrolling interest	
NET INCOME ATTRIBUTABLE TO PRIMUS TELECOMMUNICATIONS GROUP.	
INCORPORATED \$ 15,736 \$ 19,443 \$ 36,788 \$ 54,004 \$ (110,235) \$	15,736
AMOUNTS ATTRIBUTABLE TO COMMON SHAREHOLDERS OF PRIMUS	
TELECOMMUNICATIONS GROUP, INCORPORATED	
Income from continuing operations, net of tax \$ 15,736 \$ 19,443 \$ 36,788 \$ 48,140 \$ (110,235) \$	9,872
Loss from discontinued operations — — — (268) —	(268)
Gain from sale of discontinued operations	6,132
Net income \$ 15,736 \$ 19,443 \$ 36,788 \$ 54,004 \$ (110,235) \$	15,736

PRIMUS TELECOMMUNICATIONS GROUP, INCORPORATED CONSOLIDATING CONDENSED BALANCE SHEET

(in thousands)

	Successor									
			Dece	mber 31, 2009						
	PTGI	IHC	Guarantor Subsidiaries	Non Guarantor Subsidiaries	Eliminations	Consolidated				
ASSETS										
CURRENT ASSETS:										
Cash and cash equivalents	\$ 6,736	\$ —	\$ 1,672	\$ 34,130	\$ —	\$ 42,538				
Accounts receivable	_	_	9,831	79,511	_	89,342				
Prepaid expenses and other current assets	324		5,666	9,157		15,147				
Total current assets	7,060	_	17,169	122,798		147,027				
INTERCOMPANY RECEIVABLES	_	227,973	557,151	55,390	(840,514)	_				
INVESTMENTS IN SUBSIDIARIES	473,703	_	80,922	_	(554,625)	_				
RESTRICTED CASH	_	_	253	10,185	_	10,438				
PROPERTY AND EQUIPMENT—Net	_		10,356	137,250	_	147,606				
GOODWILL	_	29,642	318	34,260	_	64,220				
OTHER INTANGIBLE ASSETS—Net			83,497	95,310	_	178,807				
OTHER ASSETS			4,615	6,201	<u> </u>	10,816				
TOTAL ASSETS	\$480,763	\$257,615	\$ 754,281	\$ 461,394	\$(1,395,139)	\$ 558,914				
LIABILITIES AND STOCKHOLDERS' EQUITY (DEFICIT)										
CURRENT LIABILITIES:										
Accounts payable	\$ 57	\$ —	\$ 3,784	\$ 41,978	\$ —	\$ 45,819				
Accrued interconnection costs	_	_	10,427	27,134	_	37,561				
Deferred revenue	_	_	1,860	12,022	_	13,882				
Accrued expenses and other current liabilities	1,251	_	7,827	40,626	_	49,704				
Accrued income taxes	_	2,622	78	7,929	_	10,629				
Accrued interest	_	1,515	307	163	_	1,985				
Current portion of long-term obligations			62	4,212		4,274				
Total current liabilities	1,308	4,137	24,345	134,064	_	163,854				
INTERCOMPANY PAYABLES	377,754	_	171,457	291,303	(840,514)	_				
LONG-TERM OBLIGATIONS		123,472	83,874	45,896	_	253,242				
DEFFERED TAX LIABILITY	_	29,642	_	6,410	_	36,052				
OTHER LIABILITIES	5,362		495			5,857				
Total liabilities	384,424	157,251	280,171	477,673	(840,514)	459,005				
COMMITMENTS AND CONTINGENCIES										
STOCKHOLDERS' EQUITY (DEFICIT):										
Primus Telecommunications Group, Incorporated Stockholders' Equity (Deficit):										
Common stock	10	_	_	_	_	10				
Additional paid-in capital	85,533	85,245	458,783	(35,161)	(508,867)	85,533				
Accumulated deficit	6,732	15,119	11,263	11,145	(37,527)	6,732				
Accumulated other comprehensive loss	4,064		4,064	4,167	(8,231)	4,064				
Total Primus Telecommunications Group, Incorporated				.,	(-,)					
stockholders' equity (deficit)	96,339	100,364	474,110	(19,849)	(554,625)	96,339				
Noncontrolling interest				3,570	(23.,023)	3,570				
Total stockholders' equity (deficit)	96,339	100,364	474,110	(16,279)	(554,625)	99,909				
TOTAL LIABILITIES AND STOCKHOLDERS' EQUITY (DEFICIT)	\$480,763	\$257,615	\$ 754,281	\$ 461,394	<u>\$(1,395,139)</u>	\$ 558,914				

PRIMUS TELECOMMUNICATIONS GROUP, INCORPORATED

CONSOLIDATING CONDENSED BALANCE SHEET (in thousands)

	Predecessor December 31, 2008											
		PTGI]	ІНС				Guarantor		minations	Co	nsolidated
ASSETS												
CURRENT ASSETS:												
Cash and cash equivalents	\$	152	\$	_	\$	3,551	\$ 3	3,297	\$	_	\$	37,000
Accounts receivable		_		_		14,224	8	5,259				99,483
Prepaid expenses and other current assets		288		_		1,705	1	3,853		_		15,846
Total current assets		440				19,480	13	2,409				152,329
INTERCOMPANY RECEIVABLES		93,373	28	34,190		641,341	9	5,409	(1	,114,313)		_
INVESTMENTS IN SUBSIDIARIES		2,636		_		(132,306)		_		129,670		_
RESTRICTED CASH		_		_		314		7,819		_		8,133
PROPERTY AND EQUIPMENT—Net		_		_		14,041	9	8,111		_		112,152
GOODWILL		_		_		_	3	2,688		_		32,688
OTHER INTANGIBLE ASSETS—Net		_		_		_		746		_		746
OTHER ASSETS		393		230		5,326	1	8,447		_		24,396
TOTAL ASSETS	\$	96,842	\$28	34,420	\$	548,196	\$ 38	5,629	\$	(984,643)	\$	330,444
LIABILITIES AND STOCKHOLDERS' EQUITY (DEFICIT)												
CURRENT LIABILITIES:												
Accounts payable	\$	1,558	\$	_	\$	4,775	\$ 5	2,338	\$	_	\$	58,671
Accrued interconnection costs		_		_		16,462	2	4,960		_		41,422
Deferred revenue		_		_		1,225	1	2,078		_		13,303
Accrued expenses and other current liabilities		168		_		6,432	3	5,840		_		42,440
Accrued income taxes		70		3,243		1,220	1	3,680		_		18,213
Accrued interest		1,067		1,321		7,714		146				10,248
Current portion of long-term obligations		56,482	19	98,961		307,463		1,891				564,797
Total current liabilities		59,345	20	03,525		345,291	14	0,933		_		749,094
INTERCOMPANY PAYABLES		499,036		_		200,132	41	5,145	(1	,114,313)		_
LONG-TERM OBLIGATIONS						137	3	9,903				40,040
OTHER LIABILITIES		_		_		_		35		_		35
Total liabilities		558,381	20	03,525		545,560	59	6,016	(1	,114,313)		789,169
COMMITMENTS AND CONTINGENCIES												
STOCKHOLDERS' EQUITY (DEFICIT):												
Primus Telecommunications Group, Incorporated												
Stockholders' Equity (Deficit):												
Common stock		1,427		_		_		_		_		1,427
Additional paid-in capital		718,956		_		1,161,930	23	2,359	(1	,394,289)		718,956
Retained earnings (accumulated deficit)	(1,099,809)	8	30,895	(1,077,982)	(36	8,208)		,365,295	(1	,099,809)
Accumulated other comprehensive loss	ì	(82,113)		_	Ì	(81,312)		7,352)		158,664	ì	(82,113)
Total Primus Telecommunications Group,				-							_	
Incorporated stockholders' equity (deficit)		(461,539)	8	30,895		2,636	(21	3,201)		129,670		(461,539)
Noncontrolling interest				_				2,814				2,814
Total stockholders' equity (deficit)		(461,539)	- 8	30,895		2,636		0,387)		129,670		(458,725)
TOTAL LIABILITIES AND STOCKHOLDERS' EQUITY	_	(==,===)		,,,,,,,	_	_,				,		()2, 23
(DEFICIT)	\$	96,842	\$28	34,420	\$	548,196	\$ 38	5,629	\$	(984,643)	\$	330,444

	Successor									
]	For the Six Month		er 31, 2009					
		Non								
	PTGI	IHC	Guarantor Subsidiaries	Guarantor Subsidiaries	Eliminations	Consolidated				
CASH FLOWS FROM OPERATING ACTIVITIES:										
Net income	\$ 6,732	\$15,119	\$ 11,264	\$ 11,477	\$ (37,527)	\$ 7,065				
Adjustments to reconcile net income to net cash provided by (used in) operating activities:										
Reorganization items, net	515	_	(95)	142	_	562				
Provision for doubtful accounts receivable	_	_	678	4,336	_	5,014				
Stock compensation expense	_	_	1,151	_	_	1,151				
Depreciation and amortization	_	_	2,334	37,196	_	39,530				
Gain on sale or disposal of assets		_	11	280	_	291				
Accretion (amortization) of debt premium/discount, net	—	_	3	_		3				
Equity in net income of subsidiary	(10,856)		(26,671)	_	37,527					
Change in fair value of Contingent Value Rights	2,804	_	_		_	2,804				
Deferred income taxes				(3,891)		(3,891)				
Loss on early extinguishment or restructuring of debt	_		817	1,372	_	2,189				
Unrealized foreign currency transaction gain on intercompany and foreign debt Changes in assets and liabilities, net of acquisitions:		(9,399)	(6)	(11,071)	_	(20,476)				
(Increase) Decrease in accounts receivable	_	_	(851)	5,572	_	4,721				
(Increase) decrease in prepaid expenses and other current assets	(219)	_	(914)	3.013	_	1,880				
Decrease in other assets	_	_	1,016	785	_	1,801				
(Increase) decrease in intercompany balance	_	535	173,909	(174,444)	_					
Decrease in accounts payable	(90)	_	(490)	(7,100)	_	(7,680)				
Increase (Decrease) in accrued interconnection costs		_	(4,267)	1,789	_	(2,478)				
Increase (decrease), net, in deferred revenue, accrued expenses, other current liabilities and			() -)	,		() -/				
other liabilities	(702)	_	2,793	(6,012)	_	(3,921)				
Decrease in accrued income taxes	(73)	(1,320)	(494)	(9,193)	_	(11,080)				
Increase in accrued interest		1,515	288	162	_	1,965				
Net cash provided by (used in) operating activities before reorganization items	(1,889)	6,450	160,476	(145,587)		19,450				
Cash effect of reorganization items	(7,615)	_	_		_	(7,615)				
Net cash provided by (used in) operating activities	(9,504)	6,450	160,476	(145,587)		11,835				
CASH FLOWS FROM INVESTING ACTIVITIES:										
Purchase of property and equipment			(207)	(9,189)		(9,396)				
Sale of property and equipment and intangible assets			(207)	12		12				
Cash from disposition of business, net of cash disposed	_	_	_	(110)	_	(110)				
Increase in restricted cash	_	_	(15)	(4)	_	(19)				
Proceeds from intercompany balance	16,209	_	8,057		(24,266)	(13) —				
Net cash provided by (used in) investing activities	16,209		7,835	(9,291)	(24,266)	(9,513)				
CASH FLOWS FROM FINANCING ACTIVITIES:										
Proceeds from issuance of long-term obligations			83,733	44,330	_	128.063				
Deferred financing costs			(2,362)	(2,207)		(4,569)				
Principal payments on other long-term obligations	_	_	(2,362)	(32,108)	_	(127,871)				
Principal payments on other long-term congations Proceeds from (payments on) intercompany balance	_	(6,450)	(163,419)	145,603	24.266	(127,671)				
Net cash provided by (used in) financing activities		(6,450)	(177,811)	155,618	24,266	(4,377)				
EFFECTS OF EXCHANGE RATE CHANGES ON CASH AND CASH EQUIVALENTS			462	2,670		3,132				
NET CHANGE IN CASH AND CASH EQUIVALENTS	6,705		(9,037)	3,409		1,077				
CASH AND CASH EQUIVALENTS, BEGINNING OF PERIOD	31	_	10,709	30,721	_	41,461				
CASH AND CASH EQUIVALENTS, END OF PERIOD	\$ 6,736	<u>s</u> —	\$ 1,672	\$ 34,130	s —	\$ 42,538				
	+ 3,.33			,	*************************************					

	Predecessor											
	For the Six Months Ended July 1, 2009											
	PTGI	IHC	Guarantor Subsidiaries	Non Guarantor Subsidiaries	Eliminations	Consolidated						
CASH FLOWS FROM OPERATING ACTIVITIES:	FIGI	<u> </u>	Subsidiaries	Subsidiaries	Ellilliadolis	Consolidated						
Net income	\$ 471,154	\$104,351	\$ 594,369	\$ 135,893	\$ (834,645)	\$ 471,122						
Adjustments to reconcile net income to net cash provided by (used in) operating	Ψ 1, 1,10 1	ψ10 1,001	Ψ 00 1,000	4 100,000	(65 1,6 15)	0 171,122						
activities:												
Reorganization items, net	30,186	(79,592)	(287,970)	(102,718)	_	(440,094)						
Provision for doubtful accounts receivable	_		937	4,203	_	5,140						
Stock compensation expense	_	_	27	_	_	27						
Depreciation and amortization	_	_	1,317	11,029	_	12,346						
Gain on sale or disposal of assets	_	_	(177)	(117)	_	(294)						
Accretion (amortization) of debt premium/discount, net	129	(318)		_	_	(189)						
Equity in net income of subsidiary	(512,902)	_	(321,743)	_	834,645							
Deferred income taxes	_	_	141	(141)	_	_						
Unrealized foreign currency transaction gain (loss) on intercompany and foreign	(0.606)	(0.660)	==0	(10.150)		(20 =02)						
debt	(2,636)	(8,668)	778	(10,176)	_	(20,702)						
Changes in assets and liabilities, net of acquisitions: Decrease in accounts receivable			3.628	4.170		7,798						
(Increase) decrease in prepaid expenses and other current assets	183		3,626	4,170 (49)		7,798 461						
Decrease in other assets	52	17	1,036	1,349	_	2,454						
(Increase) decrease in intercompany balance		(6,885)	15,765	(8,880)		2,434						
Decrease in accounts payable	(1,411)	(0,003)	(500)	(10,883)	_	(12,794)						
Decrease in accounts payable Decrease in accrued interconnection costs	(1,411)	_	(1,768)	(3,593)	_	(5,361)						
Increase (decrease), net, in deferred revenue, accrued expenses, other current			(1,700)	(5,555)		(5,501)						
liabilities and other liabilities	8,885	_	(1,910)	(5,662)	_	1,313						
Increase (decrease) in accrued income taxes	4	699	(649)	2,059	_	2,113						
Increase (decrease) in accrued interest	397	3,314	(5,174)	(137)	_	(1,600)						
Net cash provided by (used in) operating activities before reorganization												
items	(5,959)	12,918	(1,566)	16,347	_	21,740						
Cash effect of reorganization items	(3,528)	_	(2,384)	1,317	_	(4,595)						
Net cash provided by (used in) operating activities	(9,487)	12,918	(3,950)	17,664	_	17,145						
CASH FLOWS FROM INVESTING ACTIVITIES:	(0,101)		(5,555)									
Purchase of property and equipment	_	<u></u>	(115)	(5,545)		(5,660)						
Sale of property and equipment and intangible assets	_	_	177	2	_	179						
Cash from disposition of business, net of cash disposed	_	_		232	_	232						
Cash used for business acquisitions, net of cash acquired	_	_	_	(199)	_	(199)						
(Increase) decrease in restricted cash	_	_	61	(207)	_	(146)						
Proceeds from intercompany balance	9,366	_	7,992	`_ ′	(17,358)	`_ ′						
Net cash provided by (used in) investing activities	9,366		8,115	(5,717)	(17,358)	(5,594)						
CASH FLOWS FROM FINANCING ACTIVITIES:												
Principal payments on other long-term obligations	_	_	(517)	(7,775)	_	(8,292)						
Proceeds from (payments on) intercompany balance	_	(12,918)	3,510	(7,950)	17,358	(0,202)						
Net cash provided by (used in) financing activities		(12,918)	2,993	(15,725)	17,358	(8,292)						
EFFECTS OF EXCHANGE RATE CHANGES ON CASH AND CASH		(12,310)	2,333	(13,723)	17,330	(0,232)						
EQUIVALENTS	_	_	_	1,202	_	1,202						
NET CHANGE IN CASH AND CASH EQUIVALENTS	(121)		7.158	(2,576)		4,461						
CASH AND CASH EQUIVALENTS, BEGINNING OF PERIOD	152	_	3,551	(2,5/6)	_	37,000						
,	\$ 31	<u> </u>	\$ 10,709		<u> </u>	\$ 41.461						
CASH AND CASH EQUIVALENTS, END OF PERIOD	\$ 31	<u> </u>	\$ 10,709	\$ 30,721	a —	\$ 41,461						

PRIMUS TELECOMMUNICATIONS GROUP, INCORPORATED CONSOLIDATING CONDENSED STATEMENT OF CASH FLOWS (in thousands)

	Predecessor					
	For the Year Ended December 31, 2008					,
	PTGI	<u>IHC</u>	Guarantor Subsidiaries	Non Guarantor Subsidiaries	Eliminations	Consolidated
CASH FLOWS FROM OPERATING ACTIVITIES:	# (DE 004)	# = 000	d (4===0)	# F000		. (0.4.0MO)
Net income (loss)	\$(25,031)	\$ 5,289	\$ (17,750)	\$ 5,288	\$ 10,332	\$ (21,872)
Adjustments to reconcile net income (loss) to net cash provided by (used in) operating						
activities: Provision for doubtful accounts receivable			2,359	9,581		11,940
Stock compensation expense			2,359	9,581	_	11,940
Depreciation and amortization		_	3,245	29,553	_	32,798
Gain on sale or disposal of assets	_		(806)	(5,222)		(6,028)
Accretion (amortization) of debt premium/discount, net	858	(1,441)	(600)	(3,222)	_	(583)
Equity in net income of subsidiary	13,539	(1,441)	(3,207)		(10,332)	(303)
Deferred income taxes	13,333		591	5,247	(10,332)	5,838
Gain on early extinguishment or restructuring of debt	(12,448)	(1,532)	(22,784)	(108)	_	(36,872)
Unrealized foreign currency transaction loss on intercompany and foreign debt	3,549	17,303	698	27,038	_	48,588
Changes in assets and liabilities, net of acquisitions:	5,545	17,505	050	27,050		40,500
Increase in accounts receivable	_	_	(2,581)	(13,077)	_	(15,658)
(Increase) decrease in prepaid expenses and other current assets	20	_	(450)	10,621	_	10,191
(Increase) decrease in other assets	521	49	1,442	(135)	_	1,877
(Increase) decrease in intercompany balance	_	(12,162)	3,415	8,747	_	
Increase (decrease) in accounts payable	753		(115)	(6,568)	_	(5,930)
Increase in accrued interconnection costs	_	_	1,262	981	_	2,243
Decrease, net, in deferred revenue, accrued expenses, other current liabilities and						
other liabilities	(40)	(504)	(1,777)	(3,319)	_	(5,640)
Decrease in accrued income taxes	(236)	(1,413)	(1,058)	(7,918)	_	(10,625)
Increase (decrease) in accrued interest	(904)	(7)	(976)	137		(1,750)
Net cash provided by (used in) operating activities	(19,419)	5,582	(38,230)	60,846	_	8,779
CASH FLOWS FROM INVESTING ACTIVITIES:						
Purchase of property and equipment	_	_	(1,405)	(24,036)	_	(25,441)
Sale of property and equipment and intangible assets	_	_	806	4,950	_	5,756
Cash from disposition of business, net of cash disposed	_	_	_	1,676	_	1,676
Cash used for business acquisitions, net of cash acquired	_	_	_	(583)	_	(583)
Increase in restricted cash	_	_	_	(102)	_	(102)
Proceeds from intercompany balance	30,689	_	20,614	`_ `	(51,303)	`_ `
Net cash provided by (used in) investing activities	30,689		20,015	(18,095)	(51,303)	(18,694)
CASH FLOWS FROM FINANCING ACTIVITIES:						
Purchase of the Company's debt securities	(11,217)	(317)	_	_	_	(11,534)
Principal payments on other long-term obligations	(1,200)	(4,676)	(6,343)	(4,326)	_	(16,545)
Proceeds from (payments on) intercompany balance	(-,)	(589)	27,439	(78,153)	51,303	(==,==,=
Net cash provided by (used in) financing activities	(12,417)	(5,582)	21,096	(82,479)	51,303	(28,079)
EFFECTS OF EXCHANGE RATE CHANGES ON CASH AND CASH EQUIVALENTS		(3,302)		(6,288)		(6,288)
·			2 001			-
NET CHANGE IN CASH AND CASH EQUIVALENTS	(1,147)		2,881	(46,016)	_	(44,282)
CASH AND CASH EQUIVALENTS, BEGINNING OF PERIOD	1,299		670	79,313		81,282
CASH AND CASH EQUIVALENTS, END OF PERIOD	\$ 152	<u> </u>	\$ 3,551	\$ 33,297	<u> </u>	\$ 37,000

PRIMUS TELECOMMUNICATIONS GROUP, INCORPORATED CONSOLIDATING CONDENSED STATEMENT OF CASH FLOWS (in thousands)

	Predecessor					
	For the Year Ended December 31, 2007					
	PTGI	IHC	Guarantor Subsidiaries	Non Guarantor Subsidiaries	Eliminations	Consolidated
CASH FLOWS FROM OPERATING ACTIVITIES:	A 45 500		å 25 2 00		d (440.00E)	A 45 500
Net income	\$ 15,736	\$ 19,443	\$ 36,788	\$ 54,004	\$ (110,235)	\$ 15,736
Adjustments to reconcile net income to net cash provided by (used in) operating activities:			4 =00	0.004		40.400
Provision for doubtful accounts receivable Stock compensation expense		_	1,532 246	8,961		10,493
Depreciation and amortization	_	_	3,700	26.894		246 30,594
Gain on sale or disposal of assets	_	_	3,700		_	
Accretion (amortization) of debt premium/discount, net	1,524	(1,075)		(4,668)	_	(4,668) 449
Equity in net income of subsidiary	(31,322)	(1,0/5)	(78,913)		110,235	44 9
Deferred income taxes	(31,322)	_	(76,913)	(12,510)	110,233	(12,463)
Loss on early extinguishment or restructuring of debt	2,000	5,144	108	400		7,652
Unrealized foreign currency transaction gain on intercompany and foreign debt	(8,372)	(3,718)	(5,721)	(17,051)		(34,862)
Changes in assets and liabilities, net of acquisitions:	(0,372)	(3,710)	(3,721)	(17,031)		(34,002)
Decrease in accounts receivable	_	_	1,452	3,823	_	5,275
(Increase) decrease in prepaid expenses and other current assets	480	_	(99)	(1,937)	_	(1,556)
Decrease in other assets	899	50	790	570	_	2,309
(Increase) decrease in intercompany balance	_	(22,228)	(1,502)	23,730	_	
Increase (decrease) in accounts payable	(33)	(22,220)	649	(3,268)	_	(2,652)
Increase (decrease) in accrued interconnection costs		_	(8,625)	2,381	_	(6,244)
Increase (decrease), net, in deferred revenue, accrued expenses, other current				,		
liabilities, and other liabilities	(934)	(94)	(2,266)	1,265	_	(2,029)
Increase (decrease) in accrued income taxes	(1,154)	1,736	2,247	1,523	_	4,352
Increase (decrease) in accrued interest	(1,781)	1,328	(65)	(647)		(1,165)
Net cash provided by (used in) operating activities	(22,957)	586	(49,632)	83,470	_	11,467
CASH FLOWS FROM INVESTING ACTIVITIES:						
Purchase of property and equipment	_	_	(1,248)	(43,497)	_	(44,745)
Cash from disposition of business, net of cash disposed	_	_	`_ ´	6,140	_	6,140
Cash used for business acquisitions, net of cash acquired	_	_	(200)	_	_	(200)
(Increase) decrease in restricted cash	_	_	541	(1,209)	_	(668)
Proceeds from intercompany balance	66,371	_	50,868	· - ·	(117,239)	<u> </u>
Net cash provided by (used in) investing activities	66,371		49,961	(38,566)	(117,239)	(39,473)
CASH FLOWS FROM FINANCING ACTIVITIES:						
Proceeds from issuance of long-term obligations	_	101,405	_	7,870	_	109,275
Deferred financing costs	_	-	_	(6,570)	_	(6,570)
Principal payments on other long-term obligations	(65,049)	_	(3,858)	(11,508)	_	(80,415)
Proceeds from sale of common stock	19,170	_	(5,550)	(11,500)	_	19,170
Proceeds from (payments on) intercompany balance		(101,991)	4,234	(19,482)	117,239	
Net cash provided by (used in) financing activities	(45,879)	(586)	376	(29,690)	117,239	41,460
EFFECTS OF EXCHANGE RATE CHANGES ON CASH AND CASH EQUIVALENTS	(43,673)	(566)		3,511		3,511
· · · · · · · · · · · · · · · · · · ·						
NET CHANGE IN CASH AND CASH EQUIVALENTS	(2,465)	_	705	18,725		16,965
CASH AND CASH EQUIVALENTS, BEGINNING OF PERIOD	3,764		(35)	60,588		64,317
CASH AND CASH EQUIVALENTS, END OF PERIOD	\$ 1,299	<u>\$ —</u>	\$ 670	\$ 79,313	<u> </u>	\$ 81,282

Table of Contents

23. SUBSEQUENT EVENTS

In January 2010, the Company paid \$2.7 million (3.1 million AUD) and completed the full repayment of the Australian leased fiber capacity.

In March 2010, the Company issued 143,157 shares of common stock in satisfaction of 200,000 vested RSUs per the Emergence RSU Agreement, net of shares withheld for tax obligations.

On March 29, 2010, the Company, Primus Telecommunications, Inc. and Mr. John F. DePodesta, Executive Vice President and Director, entered into a Termination Agreement which provides for the termination, as of March 31, 2010, of all director, officer and employment positions of Mr. DePodesta held with the Company and any of its affiliates. In connection with the termination the Company will make payments to Mr. DePodesta of \$1.6 million in October 2010 and 24 consecutive monthly payments of \$10,125 commencing in October 2010.

Table of Contents

SCHEDULE II

PRIMUS TELECOMMUNICATIONS GROUP, INCORPORATED VALUATION AND QUALIFYING ACCOUNTS

Activity in the Company's allowance accounts for the years ended December 31, 2009, 2008 and 2007 was as follows (in thousands):

		Ι	Doubtful Accounts Ro	eceivable			
<u>Period</u>	salance at ning of Period		harged to and Expenses Predecessor:		<u>uctions</u>		alance at l of Period
2007	\$ 17,296	\$	10,493	\$(1	5,750)	\$	12,039
2008	\$ 12,039	\$	11,940	\$(1	4,269)	\$	9,710
Six months ended July 1, 2009	\$ 9,710	\$	5,140	\$ (5,755)	\$	9,095
			Successor:				
Six months ended December 31, 2009	\$ 9,095	\$	5,015	\$ (5,947)	\$	8,163
	Deferred Tax Asset Valuation						
Period	Salance at ning of Period		harged to and Expenses	Ded	uctions		alance at l of Period
_ 	 <u>g</u>		Predecessor:				
2007	\$ 337,696	\$	(98,016)	\$	_	\$	239,680
2008	\$ 239,680	\$	(11,882)	\$	_	\$	227,798
Six months ended July 1, 2009	\$ 227,798	\$	(75,236)				152,562
ÿ ·							
			Successor:				
							118,520

COLLATERAL AGREEMENT

made by

PRIMUS TELECOMMUNICATIONS HOLDING, INC.

and certain of its Affiliates

in favor of

U.S. BANK NATIONAL ASSOCIATION,

as Collateral Trustee

Dated as of December 22, 2009

TABLE OF CONTENTS

SECTION 1	DEFINED TERMS	Page 1
1.1	Definitions	1
1.2	Other Definitional Provisions	6
SECTION 2.	GRANT OF SECURITY INTEREST	6
2.1	Grant of Security	6
SECTION 3.	REPRESENTATIONS AND WARRANTIES	7
3.1	Title; No Other Liens	7
3.2	Perfected First Priority Liens	8
3.3	Jurisdiction of Organization; Chief Executive Office	8
3.4	Inventory and Equipment	8
3.5	Farm Products	8
3.6	Investment Property	8
3.7	Receivables	8
3.8	Intellectual Property	9
3.9	Vehicles	9
3.10	Deposit Accounts and Securities Accounts	9
SECTION 4.	COVENANTS	9
4.1	Delivery of Instruments and Chattel Paper	9
4.2	Maintenance of Insurance	9
4.3	Payment of Obligations.	10
4.4	Maintenance of Perfected Security Interest; Further Documentation.	10
4.5	Changes in Name, etc.	10
4.6	Notices	11
4.7	Investment Property	11
4.8	Receivables	12
4.9	Intellectual Property	12
4.10	Commercial Tort Claims	13
SECTION 5.	REMEDIAL PROVISIONS	14
5.1	Certain Matters Relating to Receivables	14
5.2	Communications with Obligors; Grantors Remain Liable	14
5.3	Pledged Stock	15
5.4	Proceeds to be Turned Over To Collateral Trustee	16
5.5	Application of Proceeds	16
5.6	Code and Other Remedies	16
5.7	Registration Rights	17
5.8	Deficiency	18
5.9	ULC Limitation	18
	THE COLLATERAL TRUSTEE	19
6.1	Collateral Trustee's Appointment as Attorney-in-Fact, etc	19
6.2	Duty of Collateral Trustee	20
6.3	Execution of Financing Statements	21
6.4	Authority of Collateral Trustee	21
6.5	Applicable Secured Debt Representative	21

		Page
SECTION 7. N	MISCELLANEOUS	21
7.1	Amendments in Writing	21
7.2	Notices	21
7.3	No Waiver by Course of Conduct; Cumulative Remedies	21
7.4	Enforcement Expenses; Indemnification	22
7.5	Successors and Assigns	22
7.6	Set-Off	22
7.7	Counterparts	22
7.8	Severability	22
7.9	Section Headings	23
7.10	Integration	23
7.11	GOVERNING LAW	23
7.12	Submission To Jurisdiction; Waivers	23
7.13	Acknowledgements	23
7.14	Additional Grantors	24
7.15	Releases	24
7.16	Approvals	24
7.17	WAIVER OF JURY TRIAL	25
7.18	Collateral Trust Agreement	25
7.19	Force Majeure.	25
7.20	U.S.A. Patriot Act	25

Schedules [intentionally omitted] Schedule 1 Schedule 2 Description of Pledged Securities Schedule 3 Filings and Other Actions Required to Perfect Security Interest Schedule 4 Jurisdiction of Organization, Identification Number and Location of Chief Executive Office Locations of Inventory and Equipment Schedule 5 Schedule 6 Intellectual Property [intentionally omitted] Schedule 7 Schedule 8 Deposit Accounts and Securities Accounts

Schedule 9

Annexes

Annex I Assumption Agreement

Receivables

COLLATERAL AGREEMENT, dated as of December 22, 2009, made by each of the signatories hereto (together with any other entity that may become a party hereto as provided herein, the "Grantors"), in favor of U.S. Bank National Association, as U.S. Collateral Trustee (in such capacity and together with any successor, the "Collateral Trustee") for the benefit of the holders (the "Holders") from time to time of the 13% Senior Secured U.S. Notes due 2016 (the "U.S. Notes") issued by Primus Telecommunications Holding, Inc. (the "Company") pursuant to the Indenture, dated as of December 22, 2009 by and among the Company, Primus Telecommunications Canada, Inc., the guarantors party thereto, the Bank of New York Mellon, as trustee (in such capacity and together with any successor, the "Trustee"), the Collateral Trustee and Computershare Trust Company of Canada, as Canadian Collateral Trustee (as defined therein) (as amended, supplemented or otherwise modified from time to time, the "Indenture"), and each of the other Secured Parties (as defined).

<u>WITNESSETH</u>:

WHEREAS, pursuant to the terms of the Indenture, the Company has issued to the Holders the U.S. Notes upon the terms and subject to the conditions set forth therein, and the other Grantors have guaranteed the obligations of the Company thereunder;

WHEREAS, the Collateral Trustee has agreed to act as collateral trustee for the Secured Parties for the purposes of holding any and all security for the payment and performance of the obligations of the Company and the other Grantors under the Indenture and the other Indenture Documents;

WHEREAS, the Company is a member of an affiliated group of companies that includes each other Grantor;

WHEREAS, the proceeds of the extensions of credit under the U.S. Notes will be used in part to enable the Company to make valuable transfers to one or more of the other Grantors in connection with the operation of their respective businesses;

WHEREAS, the Company and the other Grantors are engaged in related businesses, and each Grantor will derive substantial direct and indirect benefit from the extensions of credit under the U.S. Notes; and

WHEREAS, it is a condition precedent to the obligation of the Holders to purchase the U.S. Notes that the Grantors shall have executed and delivered this Agreement to the Collateral Trustee;

NOW, THEREFORE, in consideration of the premises and to induce the Holders to purchase the U.S. Notes, each Grantor hereby agrees with the Collateral Trustee, for the benefit of the Secured Parties, as follows:

SECTION 1. DEFINED TERMS

- 1.1 <u>Definitions</u>. (a) Unless otherwise defined herein, terms defined in the Indenture and used herein shall have the meanings given to them in the Indenture and the following terms are used herein as defined in the New York UCC: Account, Certificated Security, Chattel Paper, Commercial Tort Claims, Document, Equipment, Farm Products, Fixtures, General Intangibles, Goods, Instruments, Inventory, Letter-of-Credit Rights and Supporting Obligations.
 - (b) The following terms shall have the following meanings:
 - "Additional Notes": any additional notes issued and outstanding under the Indenture (or under any supplemental indenture).

- "Agreement": this Collateral Agreement, as the same may be amended, supplemented or otherwise modified from time to time.
- "Collateral": as defined in Section 3.
- "Collateral Account": any collateral account established by the Collateral Trustee as provided in Section 5.1 or 5.4.
- "Collateral Documents": this Agreement, the Collateral Trust Agreement and each of the security agreements and other instruments and documents executed and delivered pursuant to the foregoing or this Agreement.

"Collateral Trust Agreement": a collateral trust agreement in substantially the form of Exhibit J to the Indenture to be entered into by and among the Company, the Guarantors from time to time party thereto, the Priority Lien Representatives (as defined therein) from time to time party thereto, the Junior Lien Representatives (as defined therein) from time to time party thereto, the Trustee and the Collateral Trustee, as amended, supplemented, amended and restated or otherwise modified and in effect from time to time.

"Company Obligations": the collective reference to the unpaid principal of and interest on the U.S. Notes and Additional Notes, to the extent such notes are outstanding, and all other obligations and liabilities of the Company to the Collateral Trustee or any Secured Party, whether direct or indirect, absolute or contingent, due or to become due, or now existing or hereafter incurred, which may arise under, out of, or in connection with, this Agreement, the U.S. Notes, the Indenture, the other Indenture Documents, the other Collateral Documents, or any other document made, delivered or given in connection with any of the foregoing, and in each case whether on account of principal, interest, reimbursement obligations, fees, indemnities, cost, expense or otherwise (including, without limitation, all fees and disbursements of counsel to the Collateral Trustee or to any other Secured Party that are required to be paid by the Company pursuant to the terms of any of the foregoing agreements).

"Copyrights": (i) all copyrights arising under the laws of the United States, any other country or any political subdivision thereof, whether registered or unregistered and whether published or unpublished (including, without limitation, those listed in Schedule 6), all registrations and recordings thereof, and all applications in connection therewith, including, without limitation, all registrations, recordings and applications in the United States Copyright Office, and (ii) the right to obtain all renewals thereof.

"Copyright Licenses": any written agreement naming any Grantor as licensor or licensee (including, without limitation, those listed in Schedule 6), granting any right under any Copyright, including, without limitation, the grant of rights to manufacture, distribute, exploit and sell materials derived from any Copyright.

"Deposit Account": as defined in the Uniform Commercial Code of any applicable jurisdiction and, in any event, including, without limitation, any demand, time, savings, passbook or similar account maintained with a bank.

"<u>Domestic Perfection Action</u>": any of (a) filing of financing statements under the Uniform Commercial Code in effect in any jurisdiction in the United States, (b) filings with the

United States Copyright Office or the United States Patent and Trademark Office, (c) execution and delivery of deposit account control agreements in respect of any deposit account maintained in the United States other than (i) deposit accounts holding cash in an aggregate amount not exceeding \$500,000 and (ii) any deposit account used solely as a payroll or benefits account; provided that in no event shall the aggregate amount in all such deposit accounts used as (x) payroll accounts exceed \$1,500,000 at any time and (y) benefits accounts exceed \$500,000 at any time, (d) Account No. 899660 maintained at Bank of America, N.A., which at no time shall hold cash and investment property exceeding \$5,000,000 in the aggregate, and (e) possession in the United States of stock certificates, promissory notes, and other instruments and similar collateral (other than any promissory note for which an affidavit of loss has been delivered to the Collateral Trustee, provided that any replacement promissory note shall be delivered to the Collateral Trustee promptly upon any Grantor obtaining possession of such replacement promissory note).

"Excluded Assets": the collective reference to (i) any contract, General Intangible, Copyright License, Patent License or Trademark License ("Intangible Assets") or any other property, in each case to the extent the grant by the relevant Grantor of a security interest pursuant to this Agreement in such Grantor's right, title and interest in such Intangible Asset or other property (A) is prohibited by legally enforceable provisions of such contract, agreement, instrument or indenture, or by any applicable law, (B) would give any other party to such contract, agreement, instrument or indenture a legally enforceable right to terminate its obligations thereunder or (C) is permitted only with the consent of another party, if the requirement to obtain such consent is legally enforceable and such consent has not been obtained; provided, that in any event any Receivable or any money or other amounts due or to become due under any such contract, agreement, instrument or indenture shall not be Excluded Assets to the extent that any of the foregoing is (or if it contained a provision limiting the transferability or pledge thereof would be) subject to Section 9-406 of the New York UCC and (ii) Foreign Subsidiary Voting Stock excluded from the definition of "Pledged Stock" set forth in this Section 1.1.

"Foreign Subsidiary": any Subsidiary that is not a Domestic Subsidiary.

"Foreign Subsidiary Holding Company": any Restricted Subsidiary of Parent which was formed under the laws of the United States or any state of the United States or the District of Columbia substantially all of the assets of which consist of the Capital Stock and intercompany obligations of (1) one or more Restricted Subsidiaries that were not formed under the laws of the United States or any state of the United States or the District of Columbia and (2) other Foreign Subsidiary Holding Companies.

"<u>Foreign Subsidiary Voting Stock</u>": the voting Capital Stock of any Foreign Subsidiary or Foreign Subsidiary Holding Company, in each case, owned directly by the Company or any other Grantor. For avoidance of doubt, Foreign Subsidiary Voting Stock shall not include any voting Capital Stock of any Foreign Subsidiary or Foreign Subsidiary Holding Company owned by another Foreign Subsidiary or Foreign Subsidiary Holding Company.

"Guarantor Obligations": with respect to any Guarantor, the collective reference to all obligations and liabilities of such Guarantor which may arise under or in connection with the Indenture, the other Indenture Documents or any Collateral Document to which such Guarantor is a party, in each case whether on account of guarantee obligations, reimbursement obligations, fees, indemnities, costs, expenses or otherwise (including, without limitation, all fees and disbursements of counsel to the Collateral Trustee or to any Secured Party that are required to be paid by such Guarantor pursuant to the terms of this Agreement, the Indenture or any other Collateral Document).

"Guarantors": the collective reference to each Grantor other than the Company.

"<u>Indenture Documents</u>": the Indenture, the U.S. Notes and the other documentation from time to time executed and delivered in connection therewith, including any supplemental indentures.

"Intellectual Property": the collective reference to all rights, priorities and privileges relating to intellectual property, whether arising under United States, multinational or foreign laws or otherwise, including, without limitation, the Copyrights, the Copyright Licenses, the Patents, the Patent Licenses, the Trademarks and the Trademark Licenses, and all rights to sue at law or in equity for any infringement or other impairment thereof, including the right to receive all proceeds and damages therefrom.

"Intercompany Note": any promissory note evidencing loans made by any Grantor to any other Grantor.

"Intercreditor Agreement": the Intercreditor Agreement, dated as of February 26, 2007 and amended by that certain first amendment as of July 1, 2009, by and among the Grantors party thereto, the Collateral Trustee (as successor First Lien Collateral Agent (as defined therein)), and U.S. Bank National Association as Second Lien Collateral Agent (as defined therein).

"Investment Property": the collective reference to (i) all "investment property" as such term is defined in Section 9-102(a)(49) of the New York UCC (other than any Foreign Subsidiary Voting Stock excluded from the definition of "Pledged Stock" in this Section 1.1) and (ii) whether or not constituting "investment property" as so defined, all Pledged Notes and all Pledged Stock.

"Issuers": the collective reference to each issuer of any Investment Property.

"New York UCC": the Uniform Commercial Code as from time to time in effect in the State of New York.

"Obligations": (i) in the case of the Company, the Company Obligations, and (ii) in the case of each Guarantor, its Guarantor Obligations and (iii) in the case of PTII, the PTII Obligations.

"Parent": Primus Telecommunications Group, Incorporated, a Delaware corporation.

"<u>Patents</u>": (i) all letters patent of the United States, any other country or any political subdivision thereof, all reissues and extensions thereof and all goodwill associated therewith, including, without limitation, any of the foregoing referred to in <u>Schedule 6</u>, (ii) all applications for letters patent of the United States or any other country and all divisions, continuations and continuations-in-part thereof, including, without limitation, any of the foregoing referred to in Schedule 6, and (iii) all rights to obtain any reissues or extensions of the foregoing.

"<u>Patent License</u>": all agreements, whether written or oral, providing for the grant by or to any Grantor of any right to manufacture, use or sell any invention covered in whole or in part by a Patent, including, without limitation, any of the foregoing referred to in <u>Schedule 6</u>.

"<u>Pledged Notes</u>": all promissory notes, if any, listed on <u>Schedule 2</u> and all other promissory notes issued to or held by any Grantor (other than promissory notes issued in connection with extensions of trade credit by any Grantor in the ordinary course of business).

"Pledged Securities": the collective reference to the Pledged Notes and the Pledged Stock.

"Pledged ULC Shares": the Pledged Stock which are shares in the capital stock of a ULC.

"Pledged Stock": the shares of Capital Stock listed on Schedule 2, together with any other shares, stock certificates, options or rights of any nature whatsoever in respect of the Capital Stock of any Person that may be issued or granted to, or held directly by, any Grantor while this Agreement is in effect; provided that in no event shall more than 65% of the total outstanding voting Capital Stock of any Foreign Subsidiary or Foreign Subsidiary Holding Company be required to be pledged hereunder.

"Proceeds": all "proceeds" as such term is defined in Section 9-102(a)(64) of the Uniform Commercial Code in effect in the State of New York on the date hereof and, in any event, including, without limitation, all dividends or other income from the Investment Property, collections thereon or distributions or payments with respect thereto.

"PTII": Primus Telecommunications International, Inc.

"PTII Obligations": with respect to PTII, the collective reference to all obligations and liabilities of PTII and the Company which may arise under or in connection with the Indenture or any Collateral Document to which PTII or the Company is a party, in each case whether on account of (a) guarantee obligations, reimbursement obligations, fees, indemnities, costs, expenses or otherwise (including, without limitation, all fees and disbursements of counsel to the Collateral Trustee or to any Secured Party that are required to be paid by PTII or the Company pursuant to the terms of this Agreement, the Indenture or any other Collateral Document) or (b) the unpaid principal of and interest on the Notes and all other obligations and liabilities of the Company to the Collateral Trustee or any Secured Party, whether direct or indirect, absolute or contingent, due or to become due, or not existing or hereafter incurred, which may arise under, out of, or in connection with, this Agreement, the Notes, the Indenture, the other Collateral Documents, or any other document, made delivered or given in connection with any of the foregoing, and in each case whether on account of principal, interest, reimbursement obligations, fees, indemnities, cost, expense or otherwise (including, without limitation, all fees and disbursements of counsel to the Collateral Trustee or to any other Secured Party that are required to be paid by the Company pursuant to the terms of the foregoing agreements).

"Receivable": any right to payment for goods sold, leased, licensed, assigned or otherwise disposed of, or for services rendered, whether or not such right is evidenced by an Instrument or Chattel Paper and whether or not it has been earned by performance (including, without limitation, any Account).

"Secured Parties": the collective reference to the Collateral Trustee and the Trustee and the Holders.

"Securities Account": as defined in the Uniform Commercial Code of any applicable jurisdiction.

"Securities Act of 1933, as amended.

"Supplemental Security Documents": as defined in the Collateral Trust Agreement.

"Trademarks": (i) all trademarks, trade names, corporate names, company names, business names, fictitious business names, trade styles, service marks, logos and other source or business identifiers, and all goodwill associated therewith, now existing or hereafter adopted or acquired, all registrations and recordings thereof, and all applications in connection therewith, whether in the United States Patent and Trademark Office or in any similar office or agency of the United States, any State thereof or any other country or any political subdivision thereof, or otherwise, and all common-law rights related thereto, including, without limitation, any of the foregoing referred to in Schedule 6, and (ii) the right to obtain all renewals thereof.

"<u>Trademark License</u>": any agreement, whether written or oral, providing for the grant by or to any Grantor of any right to use any Trademark, including, without limitation, any of the foregoing referred to in <u>Schedule 6</u>.

"<u>ULC</u>": any unlimited liability company or unlimited liability corporation existing under the laws of any province or territory of Canada and any successor or any such unlimited liability company or unlimited liability corporation.

"Vehicles": all cars, trucks, trailers, construction and earth moving equipment and other vehicles covered by a certificate of title law of any state and all tires and other appurtenances to any of the foregoing.

- 1.2 Other Definitional Provisions. (a) The words "hereof," "herein", "hereto" and "hereunder" and words of similar import when used in this Agreement shall refer to this Agreement as a whole and not to any particular provision of this Agreement, and Section and Schedule references are to this Agreement unless otherwise specified.
 - (b) The meanings given to terms defined herein shall be equally applicable to both the singular and plural forms of such terms.
 - (c) Where the context requires, terms relating to the Collateral or any part thereof, when used in relation to a Grantor, shall refer to such Grantor's Collateral or the relevant part thereof.

SECTION 2. GRANT OF SECURITY INTEREST

2.1 Grant of Security.

Each Grantor hereby assigns and transfers to the Collateral Trustee, and hereby grants to the Collateral Trustee, for the ratable benefit of the Secured Parties, a security interest in, all of the following property now owned or at any time hereafter acquired by such Grantor or in which such Grantor now has or at any time in the future may acquire any right, title or interest (collectively, the "Collateral"), as collateral security for the prompt and complete payment and performance when due (whether at the stated maturity, by acceleration or otherwise) of such Grantor's Obligations:

- (a) all Accounts;
- (b) all Chattel Paper;
- (c) all Deposit Accounts;
- (d) all Documents;

- (e) all Equipment;
- (f) all General Intangibles;
- (g) all Instruments;
- (h) all Intellectual Property;
- (i) all Inventory;
- (j) all Investment Property;
- (k) all Vehicles;
- (l) all Letter-of-Credit Rights;
- (m) all Commercial Tort Claims to the extent they have been notified to the Collateral Trustee pursuant to Section 4.10;
- (n) all Goods and other property not otherwise described above;
- (o) all books and records pertaining to the Collateral; and
- (p) to the extent not otherwise included, all Proceeds and products of any and all of the foregoing, all Supporting Obligations in respect of any of the foregoing and all collateral security and guarantees given by any Person with respect to any of the foregoing;

provided, that the Collateral shall not include any Excluded Assets.

SECTION 3. REPRESENTATIONS AND WARRANTIES

To induce each of the Holders to purchase the U.S. Notes, each Grantor hereby represents and warrants to the Collateral Trustee and each Secured Party that:

3.1 <u>Title</u>; No Other Liens. Except for the security interest granted to the Collateral Trustee for the ratable benefit of the Secured Parties pursuant to this Agreement and the other Liens permitted to exist on the Collateral by the Indenture, such Grantor owns its right, title and interest in and to each item of the Collateral free and clear of any and all Liens or claims of others. No financing statement or other public notice with respect to all or any part of the Collateral is on file or of record in any public office, except (a) such as have been filed in favor of the Collateral Trustee, for the ratable benefit of the Secured Parties, pursuant to this Agreement, or (b) or as are permitted by the Indenture. For the avoidance of doubt, it is understood and agreed that any Grantor may, as part of its business, grant licenses to third parties to use Intellectual Property owned or developed by a Grantor. For purposes of this Agreement and the other Loan Documents, such licensing activity shall not constitute a "Lien" on such Intellectual Property. Each of the Collateral Trustee and each Secured Party understands that any such licenses may be exclusive to the applicable licensees, and such exclusivity provisions may limit the ability of the Collateral Trustee to utilize, sell, lease or transfer the related Intellectual Property or otherwise realize value from such Intellectual Property pursuant hereto.

- 3.2 <u>Perfected First Priority Liens</u>. Subject to Section 7.18 and except as otherwise contemplated by the terms of the Indenture, the security interests granted by such Grantor pursuant to this Agreement (a) upon completion of the filings and other actions specified on <u>Schedule 3</u> (which, in the case of all filings and other documents referred to on said Schedule except with respect to (i) Vehicles and (ii) other Collateral in respect of which a security interest cannot be perfected by Domestic Perfection Actions, have been delivered to the Collateral Trustee in completed and duly executed form), will constitute valid perfected security interests in all of the Collateral with respect to which perfection may be accomplished by Domestic Perfection Actions in favor of the Collateral Trustee, for the ratable benefit of the Secured Parties, as collateral security for such Grantor's Obligations, enforceable in accordance with the terms hereof and (b) are prior to all other Liens on the Collateral except for Permitted Priority Liens.
- 3.3 <u>Jurisdiction of Organization; Chief Executive Office</u>. On the date hereof, such Grantor's jurisdiction of organization, identification number from the jurisdiction of organization (if any), and the location of such Grantor's chief executive office or sole place of business or principal residence, as the case may be, are specified on <u>Schedule 4</u>. Such Grantor has furnished to the Collateral Trustee a certified charter, certificate of incorporation or other organization document and long-form good standing certificate as of a date which is recent to the date hereof.
- 3.4 <u>Inventory and Equipment</u>. On the date hereof, the material portion of Inventory and the Equipment (other than mobile goods) are kept at the locations listed on <u>Schedule 5</u>.
 - 3.5 Farm Products. None of the Collateral constitutes, or is the Proceeds of, Farm Products.
- 3.6 <u>Investment Property</u>. (a) The shares of Pledged Stock pledged by such Grantor hereunder constitute all the issued and outstanding shares of all classes of the Capital Stock of each Issuer owned by such Grantor or, in the case of Foreign Subsidiary Voting Stock, if less, 65% of the outstanding Foreign Subsidiary Voting Stock of each relevant Issuer.
 - (b) To the knowledge of such Grantor, all the shares of the Pledged Stock have been duly and validly issued and are fully paid and nonassessable.
- (c) To the knowledge of such Grantor, each of the Pledged Notes constitutes the legal, valid and binding obligation of the obligor with respect thereto, enforceable in accordance with its terms, subject to the effects of bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium and other similar laws relating to or affecting creditors' rights generally, general equitable principles (whether considered in a proceeding in equity or at law) and an implied covenant of good faith and fair dealing.
- (d) Such Grantor is the record and beneficial owner of, and has good and marketable title to, the Investment Property pledged by it hereunder, free of any and all Liens or options in favor of, or claims of, any other Person, except the security interest created by this Agreement or as permitted by the Indenture.
- 3.7 <u>Receivables</u>. (a) <u>Schedule 9</u> is a list of the Receivables evidenced by any Instrument or Chattel Paper which has been delivered to the Collateral Trustee.
- (b) None of the obligors on any Receivable is a Governmental Authority, except for Receivables the face amount of which in the aggregate constitute not more than 5% of the face amount of all Receivables.

- 3.8 <u>Intellectual Property</u>. (a) <u>Schedule 6</u> lists all material Intellectual Property which is the subject of an application or registration owned by such Grantor in its own name on the date hereof.
- (b) On the date hereof, to its knowledge, all material Intellectual Property of such Grantor described on <u>Schedule 6</u> is valid, subsisting, unexpired and enforceable, has not been abandoned and to the Grantor's knowledge does not infringe upon the intellectual property rights of any other Person.
- (c) Except (i) for licensing or franchise agreements between any two or more of Parent and its Subsidiaries, (ii) the licensing of one provisional patent to various companies to enable them to provide void services, (iii) the licensing of Trademarks to independent sales agents, and (iv) as set forth in <u>Schedule 6</u>, on the date hereof, none of the Intellectual Property is the subject of any licensing or franchise agreement pursuant to which such Grantor is the licensor or franchisor.
- (d) No holding, decision or judgment has been rendered by any Governmental Authority which would limit, cancel or question the validity of, or such Grantor's rights in, any Intellectual Property in any respect that could reasonably be expected to have a Material Adverse Effect.
- (e) No action or proceeding is pending, or, to the knowledge of such Grantor, threatened, on the date hereof (i) seeking to limit, cancel or question the validity of any material Intellectual Property or such Grantor's ownership interest therein, or (ii) which, if adversely determined, would have a material adverse effect on the value of any material Intellectual Property.
 - 3.9 Vehicles. On the date hereof, the aggregate book value of all Vehicles owned by all Grantors is less than \$500,000.
- 3.10 <u>Deposit Accounts and Securities Accounts</u>. <u>Schedule 8</u> lists each Deposit Account and each Securities Account maintained by such Grantor on the date hereof.

SECTION 4. COVENANTS

Each Grantor covenants and agrees with the Collateral Trustee and the Secured Parties that, from and after the date of this Agreement until the Obligations shall have been paid in full:

- 4.1 <u>Delivery of Instruments and Chattel Paper</u>. If any amount payable under or in connection with any of the Collateral shall be or become evidenced by any Instrument, Certificated Security or Chattel Paper shall, not later than the end of the fiscal quarter in which any such amount becomes so evidenced, be delivered to the Collateral Trustee (subject to the Collateral Trust Agreement), duly indorsed in a manner satisfactory to the Collateral Trustee, to be held as Collateral pursuant to this Agreement; <u>provided</u>, that the Grantors shall not be obligated to deliver to the Collateral Trustee any Instruments, Certificated Securities or Chattel Paper held by any Grantor at any time to the extent that the aggregate face amount of all such Instruments and Chattel Paper held by all Grantors at such time does not exceed \$3,000,000 or such Instruments and Chattel Paper reflect an agreement between any two or more of Parent and its Subsidiaries.
- 4.2 <u>Maintenance of Insurance</u>. (a) Such Grantor will maintain, with financially sound and reputable companies, insurance policies insuring the Inventory, Equipment and Vehicles against loss by fire, explosion, theft and such other casualties as shall be customary for similar businesses in similar locations.

- (b) All such insurance shall (i) provide that no cancellation, material reduction in amount or material change in coverage thereof shall be effective until the applicable insurer has complied with applicable state law (ii) name the Collateral Trustee as additional insured party or loss payee, and (iii) provide that the applicable insurer shall endeavor to provide the Collateral Trustee with at least 30 day prior notice of cancellation thereof.
- (c) The Company shall deliver to the Collateral Trustee such additional information concerning such insurance as the Collateral Trustee may from time to time reasonably request or as otherwise required by the Indenture Documents, <u>provided</u>, <u>however</u>, that the Collateral Trustee shall have no obligation to monitor the insurance or request such further information.
- 4.3 <u>Payment of Obligations</u>. Such Grantor will pay and discharge or otherwise satisfy at or before maturity or before they become delinquent, as the case may be, all taxes, assessments and governmental charges or levies imposed upon the Collateral or in respect of income or profits therefrom, as well as all claims of any kind (including, without limitation, claims for labor, materials and supplies) against or with respect to the Collateral, except that no such tax, assessment, charge, levy or claim need be paid if the amount or validity thereof is currently being contested in good faith by appropriate proceedings, reserves in conformity with GAAP with respect thereto have been provided on the books of such Grantor and such proceedings could not reasonably be expected to result in the sale, forfeiture or loss of any material portion of the Collateral or such Grantor's interest in such material portion of the Collateral.
- 4.4 <u>Maintenance of Perfected Security Interest; Further Documentation</u>. (a) Such Grantor shall maintain the security interest created by this Agreement as a perfected security interest in the manner, and having at least the priority described in Section 3.2 and shall use its commercially reasonable efforts to defend such security interest against the claims and demands of all Persons whomsoever.
- (b) Such Grantor will furnish to the Collateral Trustee from time to time statements and schedules further identifying and describing the assets and property of such Grantor and such other reports in connection with the Collateral as the Collateral Trustee may reasonably request, all in reasonable detail, provided, however, that the Collateral Trustee shall have no obligation to monitor such assets and property or requires such statements and schedules.
- (c) At any time and from time to time, and at the sole expense of such Grantor, such Grantor will promptly and duly execute and deliver, and have recorded, such further instruments and documents and take such further actions as may be necessary or as the Collateral Trustee may reasonably request for the purpose of obtaining or preserving the full benefits of this Agreement and any Supplemental Security Agreements and of the rights and powers herein granted, including, without limitation, (i) the filing of any financing or continuation statements under the Uniform Commercial Code (or other similar laws) in effect in any jurisdiction in the United States with respect to the security interests created hereby and (ii) in the case of Investment Property, Deposit Accounts and Letter-of-Credit Rights, taking, to the extent required by the Indenture, any actions necessary to enable the Collateral Trustee to obtain "control" (within the meaning of the applicable Uniform Commercial Code) with respect thereto, subject to the terms of the Collateral Trust Agreement to the extent then in effect.
- 4.5 <u>Changes in Name, etc.</u> Such Grantor will not, except (i) during the period beginning on the date hereof and entity 15 days thereafter, upon prior notice and (ii) upon 15 days' prior written notice, to the Collateral Trustee and, in each case, upon filing, execution or delivery of all additional financing statements and other documents necessary to maintain the validity, perfection and priority of the security interests provided for herein:
 - (i) change its jurisdiction of organization or the location of its chief executive office or sole place of business or principal residence from that referred to in Section 3.3; or

- (ii) change its name.
- 4.6 Notices. Such Grantor will advise the Collateral Trustee:
- (a) No less often than on a quarterly basis, in reasonable detail, of any Lien (other than security interests created hereby or Liens permitted under the Indenture) on any of the Collateral which would adversely affect the ability of the Collateral Trustee to exercise any of its remedies hereunder; and
- (b) Promptly, in reasonable detail, of the occurrence of any other event which could reasonably be expected to have a material adverse effect on the aggregate value of the Collateral or on the security interests created hereby.
- 4.7 Investment Property. (a) If such Grantor shall become entitled to receive or shall receive any certificate (including, without limitation, any certificate representing a dividend or a distribution in connection with any reclassification, increase or reduction of capital or any certificate issued in connection with any reorganization), option or rights in respect of the Capital Stock of any Issuer of Pledged Stock, whether in addition to, in substitution of, as a conversion of, or in exchange for, any shares of the Pledged Stock, or otherwise in respect thereof, such Grantor shall, to the extent applicable and within the time period required under Section 4.1, accept the same as the agent of the Collateral Trustee and the Secured Parties, hold the same in trust for the Collateral Trustee and the Secured Parties and deliver the same promptly to the Collateral Trustee in the exact form received, duly indorsed by such Grantor to the Collateral Trustee, if required, together with an undated stock power covering such certificate duly executed in blank by such Grantor and with, if the Collateral Trustee so requests, signature guaranteed, to be held by the Collateral Trustee, subject to the terms hereof, as additional collateral security for the Obligations. Any sums paid upon or in respect of the Investment Property pledged hereunder upon the liquidation or dissolution of any Issuer shall be paid over to the Collateral Trustee to be held by it hereunder as additional collateral security for the Obligations, and in case any distribution of capital shall be made on or in respect of such Investment Property, or any property shall be distributed upon or with respect to such Investment Property pursuant to the recapitalization or reclassification of the capital of any Issuer or pursuant to the reorganization thereof, the property so distributed shall, unless otherwise subject to a perfected security interest in favor of the Collateral Trustee, be delivered to the Collateral Trustee promptly to be held by it hereunder as additional collateral security for the Obligations. If any sums of money or property so paid or distributed in respect of such Investment Property shall be received by such Grantor, subject to the terms of the Collateral Trust Agreement, such Grantor shall, until such money or property is paid or delivered to the Collateral Trustee, hold such money or property in trust for the Secured Parties, segregated from other funds of such Grantor, as additional collateral security for the Obligations. Notwithstanding any of the foregoing, the Grantors shall not be required to hold in trust for the Collateral Trustee, pay over to the Collateral Trustee or deliver to the Collateral Trustee as Collateral any proceeds of any liquidation or dissolution of any Issuer, or any distribution of capital or property in respect of any Investment Property, to the extent that (i) such liquidation, dissolution or distribution would be permitted by the Indenture and (ii) the proceeds thereof are applied toward prepayment of the U.S. Notes to the extent required by Indenture.
- (b) Without the prior written consent of the Collateral Trustee, such Grantor will not (i) except as otherwise permitted by the Indenture, vote to enable, or take any other action to permit, any Issuer to issue any stock or other equity securities of any nature or to issue any other securities convertible into or granting the right to purchase or exchange for any stock or other equity securities of any nature of

any Issuer, unless such securities are delivered to the Collateral Trustee, concurrently with the issuance thereof, to be held by the Collateral Trustee as Collateral, (ii) sell, assign, transfer, exchange, or otherwise dispose of, or grant any option with respect to, the Investment Property pledged hereunder or Proceeds thereof (except pursuant to a transaction permitted by the Indenture), (iii) create, incur or permit to exist any Lien or option in favor of, or any claim of any Person with respect to, any of such Investment Property or Proceeds thereof, or any interest therein, except for the security interests created by this Agreement and Permitted Liens or (iv) except as otherwise permitted by the Indenture, enter into any agreement or undertaking restricting the right or ability of such Grantor or the Collateral Trustee to sell, assign or transfer any of the Pledged Securities or Proceeds thereof.

- (c) In the case of each Grantor which is an Issuer, such Issuer agrees that (i) it will be bound by the terms of this Agreement relating to the Pledged Securities issued by it and will comply with such terms insofar as such terms are applicable to it, (ii) it will notify the Collateral Trustee promptly in writing upon the receipt of any collateral in respect of the Investment Property described in Section 4.7(a) and (iii) the terms of Sections 5.3(c) and 5.7 shall apply to it, mutatis mutandis, with respect to all actions that may be required of it pursuant to Section 5.3(c) or 5.7 with respect to the Pledged Securities issued by it.
- (d) Each Issuer that is a partnership or a limited liability company (i) confirms that none of the terms of any equity interest issued by it provides that such equity interest is a "security" within the meaning of Sections 8-102 and 8-103 of the New York UCC (a "Security"), (ii) agrees that it will take no action to cause or permit any such equity interest to become a Security, (iii) agrees that it will not issue any certificate representing any such equity interest and (iv) agrees that if, notwithstanding the foregoing, any such equity interest shall be or become a Security, such Issuer will (and the Grantor that holds such equity interest hereby instructs such Issuer to) comply with instructions originated by the Collateral Trustee without further consent by such Grantor.
- 4.8 <u>Receivables</u>. (a) Other than in the ordinary course of business consistent with its past practice, or unless the applicable Grantor concludes that any of the following is in the best interest of such Grantor, such Grantor will not (i) grant any extension of the time of payment of any Receivable, (ii) compromise or settle any Receivable for less than the full amount thereof, (iii) release, wholly or partially, any Person liable for the payment of any Receivable, (iv) allow any credit or discount whatsoever on any Receivable or (v) amend, supplement or modify any Receivable in any manner that could adversely affect the value thereof.
- (b) Such Grantor will deliver to the Collateral Trustee no less often than on a quarterly basis a copy of each material demand, notice or document received by it that questions or calls into doubt the validity or enforceability of more than 5% of the aggregate amount of the then outstanding Receivables.
- 4.9 Intellectual Property. (a) Such Grantor (either itself or through licensees) will (i) to such extent as such Grantor determines it is desirable to do so, continue to use each material Trademark on each and every material trademark class of goods applicable to its current line as reflected in its current catalogs, brochures and price lists in order to maintain such Trademark in full force free from any claim of abandonment for non-use, (ii) maintain the quality of products and services offered under such Trademark, (iii) use such Trademark with the appropriate notice of registration and all other notices and legends required by applicable Requirements of Law, and (iv) not (and not permit any licensee or sublicensee thereof to) do any act or knowingly omit to do any act whereby such Trademark may become invalidated or materially impaired in any way.

- (b) Such Grantor (either itself or through licensees) will not do any act, or omit to do any act, whereby any material Patent may become forfeited, abandoned or dedicated to the public.
- (c) Such Grantor (either itself or through licensees) will not (and will not permit any licensee or sublicensee thereof to) do any act or knowingly omit to do any act whereby any material portion of any material Copyrights may become invalidated or otherwise impaired. Such Grantor will not (either itself or through licensees) do any act whereby any material portion of any material Copyrights may fall into the public domain.
- (d) Such Grantor (either itself or through licensees) will not do any act that knowingly uses any material Intellectual Property to infringe upon the intellectual property rights of any other Person.
- (e) Such Grantor will promptly notify the Collateral Trustee in writing if it knows, or has reason to know, that any application or registration relating to any material Intellectual Property may become forfeited, abandoned or dedicated to the public, or of any adverse determination or development (including, without limitation, the institution of, or any such determination or development in, any proceeding in the United States Patent and Trademark Office, or the United States Copyright Office or any court in the United States regarding such Grantor's ownership of, or the validity of, any material Intellectual Property or such Grantor's right to register the same or to own and maintain the same.
- (f) Whenever such Grantor, either by itself or through any agent, employee, licensee or designee, shall file an application for the registration of any material Intellectual Property with the United States Patent and Trademark Office or the United States Copyright Office, such Grantor shall report such filing to the Collateral Trustee no less often than on a quarterly basis. Such Grantor shall execute and deliver, and have recorded, any and all agreements, instruments, documents, and papers as are required to evidence the Collateral Trustee's, and the Secured Parties' security interest in any material Copyright, Patent or Trademark and the goodwill and general intangibles of such Grantor relating thereto or represented thereby.
- (g) Such Grantor will take all reasonable and necessary steps, including, without limitation, in any proceeding before the United States Patent and Trademark Office or the United States Copyright Office, that such Grantor determines may be desirable to maintain and pursue each application relating to any material Intellectual Property (and to obtain the relevant registration) and to maintain each registration of the material Intellectual Property, including, without limitation, filing of applications for renewal, affidavits of use and affidavits of incontestability.

In the event that any material Intellectual Property is infringed, misappropriated or diluted by a third party, such Grantor shall (i) take such actions as such Grantor shall reasonably deem appropriate under the circumstances to protect such Intellectual Property and (ii) if such Intellectual Property is of material economic value, notify the Collateral Trustee after it learns thereof no less often than on a quarterly basis and, if such Grantor determines in its reasonable judgment that legal action is appropriate, take legal action to contest the infringement, misappropriation or dilution, to seek injunctive relief where appropriate and to recover any and all damages for such infringement, misappropriation or dilution.

4.10 <u>Commercial Tort Claims</u>. If any Grantor shall at any time commence a suit, action or proceeding with respect to any Commercial Tort Claim held by it with a value which such Grantor reasonably believes to be of \$3,000,000 or more, such Grantor shall promptly notify the

Collateral Trustee thereof no less often than on a quarterly basis in a writing signed by such Grantor and describing the details thereof and shall grant to the Collateral Trustee for the benefit of the Secured Parties in such writing a security interest therein and in the proceeds thereof, all upon the terms of this Agreement, with such writing to be in form and substance reasonably satisfactory to the Collateral Trustee.

SECTION 5. REMEDIAL PROVISIONS

- 5.1 <u>Certain Matters Relating to Receivables</u>. (a) The Collateral Trustee shall have the right, at any time after the occurrence and during the continuance of an Event of Default, to make test verifications of the Receivables in any manner and through any medium that it reasonably considers advisable, and each Grantor shall furnish all such assistance and information as the Collateral Trustee may reasonably require in connection with such test verifications. At any time and from time to time after the occurrence and during the continuance of an Event of Default, upon the Collateral Trustee's request and at the expense of the relevant Grantor, such Grantor shall cause independent public accountants or others satisfactory to the Collateral Trustee to furnish to the Collateral Trustee reports showing reconciliations, aging and test verifications of, and trial balances for, the Receivables.
- (b) The Collateral Trustee hereby authorizes each Grantor to collect such Grantor's Receivables, subject to the Collateral Trustee's direction and control after the occurrence and during the continuance of an Event of Default, and the Collateral Trustee may curtail or terminate said authority at any time after the occurrence and during the continuance of an Event of Default. If required by the Collateral Trustee at any time after the occurrence and during the continuance of an Event of Default, any payments of Receivables, when collected by any Grantor, (i) shall be forthwith (and, in any event, within two Business Days) deposited by such Grantor in the exact form received, duly indorsed by such Grantor to the Collateral Trustee if required, in a Collateral Account maintained under the sole dominion and control of the Collateral Trustee, subject to withdrawal by the Collateral Trustee for the account of the Secured Parties only as provided in Section 5.5, and (ii) until so turned over, shall be held by such Grantor in trust for the Collateral Trustee and the Secured Parties, segregated from other funds of such Grantor. Each such deposit of Proceeds of Receivables shall be accompanied by a report identifying in reasonable detail the nature and source of the payments included in the deposit.
- (c) At any time after the occurrence and during the continuance of an Event of Default, at the Collateral Trustee's request, each Grantor shall deliver to the Collateral Trustee all original and other documents evidencing, and relating to, the agreements and transactions which gave rise to the Receivables, including, without limitation, all original orders, invoices and shipping receipts.
- (d) At any time after the occurrence and during the continuance of an Event of Default, at the Collateral Trustee's request, each Grantor will cooperate with the Collateral Trustee to establish a system of lockbox accounts, under the sole dominion and control of the Collateral Trustee, into which all Receivables shall be paid and from which all collected funds will be transferred to a Collateral Account.
- 5.2 <u>Communications with Obligors; Grantors Remain Liable</u>. (a) The Collateral Trustee in its own name or in the name of others may at any time after the occurrence and during the continuance of an Event of Default communicate with obligors under the Receivables to verify with them to the Collateral Trustee's satisfaction the existence, amount and terms of any Receivables.
- (b) Upon the request of the Collateral Trustee at any time after the occurrence and during the continuance of an Event of Default, each Grantor shall promptly notify obligors on the

Receivables in writing that the Receivables have been assigned to the Collateral Trustee for the ratable benefit of the Secured Parties and that payments in respect thereof shall be made directly to the Collateral Trustee.

- (c) Anything herein to the contrary notwithstanding, each Grantor shall remain liable under each of the Receivables (or any agreement giving rise thereto) to observe and perform all the conditions and obligations to be observed and performed by it thereunder, all in accordance with the terms of any agreement giving rise thereto (provided that no Grantor shall be obligated in any manner to observe and perform any such conditions and obligations if the determination to not so observe and perform is made in good faith in the exercise of Grantor's reasonable business judgment). Neither the Collateral Trustee nor any Secured Party shall have any obligation or liability under any Receivable (or any agreement giving rise thereto) by reason of or arising out of this Agreement or the receipt by the Collateral Trustee or any Secured Party of any payment relating thereto, nor shall the Collateral Trustee or any Secured Party be obligated in any manner to perform any of the obligations of any Grantor under or pursuant to any Receivable (or any agreement giving rise thereto), to make any payment, to make any inquiry as to the nature or the sufficiency of any payment received by it or as to the sufficiency of any performance by any party thereunder, to present or file any claim, to take any action to enforce any performance or to collect the payment of any amounts which may have been assigned to it or to which it may be entitled at any time or times.
- 5.3 <u>Pledged Stock</u>. (a) Unless an Event of Default shall have occurred and be continuing and the Collateral Trustee shall have given notice to the relevant Grantor of the Collateral Trustee's intent to exercise its corresponding rights pursuant to Section 5.3(b), each Grantor shall be permitted to receive all cash dividends paid in respect of the Pledged Stock and all payments made in respect of the Pledged Notes, in each case paid in the normal course of business of the relevant Issuer, and to exercise all voting, corporate and other rights with respect to the Pledged Securities; <u>provided</u>, <u>however</u>, that no vote shall be cast or corporate right exercised or other action taken which could reasonably be expected to impair a material portion of Collateral (except in connection with a transaction permitted by the Indenture).
- (b) If an Event of Default shall occur and be continuing and the Collateral Trustee shall give notice of its intent to exercise such rights to the relevant Grantor or Grantors, (i) the Collateral Trustee shall have the right to receive any and all cash dividends, payments or other Proceeds paid in respect of the Pledged Securities and make application thereof to the Obligations in the order set forth in Section 5.5, and (ii) any or all of the Pledged Securities shall be registered in the name of the Collateral Trustee or its nominee, and the Collateral Trustee or its nominee may thereafter exercise (x) all voting, corporate and other rights pertaining to such Pledged Securities at any meeting of shareholders of the relevant Issuer or Issuers or otherwise and (y) any and all rights of conversion, exchange and subscription and any other rights, privileges or options pertaining to such Pledged Securities as if it were the absolute owner thereof (including, without limitation, the right to exchange at its discretion any and all of the Pledged Securities upon the merger, consolidation, reorganization, recapitalization or other fundamental change in the corporate structure of any Issuer, or upon the exercise by any Grantor or the Collateral Trustee of any right, privilege or option pertaining to such Pledged Securities, and in connection therewith, the right to deposit and deliver any and all of the Pledged Securities with any committee, depositary, transfer agent, registrar or other designated agency upon such terms and conditions as the Collateral Trustee may determine), all without liability except to account for property actually received by it, but the Collateral Trustee shall have no duty to any Grantor to exercise any such right, privilege or option and shall not be responsible for any failure to do so or delay in so doing.
- (c) Each Grantor hereby authorizes and instructs each Issuer of any Pledged Securities pledged by such Grantor hereunder to (i) comply with any instruction received by it from the

Collateral Trustee in writing that (x) states that an Event of Default has occurred and is continuing and (y) is otherwise in accordance with the terms of this Agreement, without any other or further instructions from such Grantor, and each Grantor agrees that each Issuer shall be fully protected in so complying, and (ii) to the extent otherwise required hereby, pay any dividends or other payments with respect to the Pledged Securities directly to the Collateral Trustee.

- 5.4 Proceeds to be Turned Over To Collateral Trustee. In addition to the rights of the Collateral Trustee and the Secured Parties specified in Section 5.1 with respect to payments of Receivables, if an Event of Default shall occur and be continuing all Proceeds received by any Grantor consisting of cash, checks and Instruments shall be held by such Grantor in trust for the Collateral Trustee and the Secured Parties, segregated from other funds of such Grantor, and shall, forthwith upon receipt by such Grantor, be turned over to the Collateral Trustee in the exact form received by such Grantor (duly indorsed by such Grantor to the Collateral Trustee, if required). All Proceeds received by the Collateral Trustee hereunder shall be held by the Collateral Trustee in a Collateral Account maintained under its sole dominion and control. All Proceeds while held by the Collateral Trustee in a Collateral Account (or by such Grantor in trust for the Collateral Trustee and the Secured Parties) shall continue to be held as collateral security for all the Obligations and shall not constitute payment thereof until applied as provided in Section 5.5.
- 5.5 <u>Application of Proceeds</u>. If an Event of Default shall have occurred and be continuing, at any time at the Collateral Trustee's election, the Collateral Trustee may apply all or any part of Proceeds constituting Collateral, whether or not held in any Collateral Account in payment of the Obligations in accordance with the provisions of the Indenture or, if any Collateral Trust Agreement is then in effect, in accordance with the provisions of such Collateral Trust Agreement. Any balance of such proceeds remaining after the Obligations shall have been paid in full shall be paid over to the Grantors or to whomsoever may be lawfully entitled to receive the same.
- 5.6 Code and Other Remedies. If an Event of Default shall occur and be continuing, the Collateral Trustee, on behalf of the Secured Parties, may exercise, in addition to all other rights and remedies granted to it in this Agreement and in any other instrument or agreement securing, evidencing or relating to the Obligations, all rights and remedies of a secured party under the New York UCC or any other applicable law or in equity. Without limiting the generality of the foregoing, the Collateral Trustee, without demand of performance or other demand, presentment, protest, advertisement or notice of any kind (except any notice required by law referred to below) to or upon any Grantor or any other Person (all and each of which demands, defenses, advertisements and notices are hereby waived), may in such circumstances forthwith collect, receive, appropriate and realize upon the Collateral, or any part thereof, and/or may forthwith sell, lease, assign, give option or options to purchase, or otherwise dispose of and deliver the Collateral or any part thereof (or contract to do any of the foregoing), in one or more parcels at public or private sale or sales, at any exchange, broker's board or office of the Collateral Trustee or any Secured Party or elsewhere upon such terms and conditions as it may deem advisable and at such prices as it may deem best, for cash or on credit or for future delivery without assumption of any credit risk. The Collateral Trustee or any Secured Party shall have the right upon any such public sale or sales, and, to the extent permitted by law, upon any such private sale or sales, to purchase the whole or any part of the Collateral so sold, free of any right or equity of redemption in any Grantor, which right or equity is hereby waived and released. Each Grantor further agrees, at the Collateral Trustee's request, to assemble the Collateral and make it available to the Collateral Trustee at places which the Collateral Trustee shall reasonably select, whether at such Grantor's premises or elsewhere. The Collateral Trustee shall apply the net proceeds of any action taken by it pursuant to this Section 5.6 with respect to any Grantor's Collateral, after deducting all reasonable costs and expenses of every kind incurred in connection therewith or incidental to the care or safekeeping of any of the Collateral of such Grantor or in any way relating to the Collateral of such Grantor or the rights of the Collateral Trustee and the Secured Parties

hereunder with respect thereto, including, without limitation, reasonable attorneys' fees and disbursements, to the payment in whole or in part of the Obligations of such Grantor, in the order specified in Section 5.5, and only after such application and after the payment by the Collateral Trustee of any other amount required by any provision of law, including, without limitation, Section 9-615(a)(3) of the New York UCC, need the Collateral Trustee account for the surplus, if any, to any Grantor. To the extent permitted by applicable law, each Grantor waives all claims, damages and demands it may acquire against the Collateral Trustee or any Secured Party arising out of the exercise by them of any rights hereunder. If any notice of a proposed sale or other disposition of Collateral shall be required by law, such notice shall be deemed reasonable and proper if given at least 10 days before such sale or other disposition.

- 5.7 Registration Rights. (a) If the Collateral Trustee shall determine to exercise its right to sell any or all of the Pledged Stock pursuant to Section 5.6, and if in the opinion of the Collateral Trustee it is necessary or advisable to have the Pledged Stock, or that portion thereof to be sold, registered under the provisions of the Securities Act, the relevant Grantor will cause the Issuer thereof to (i) execute and deliver, and cause the directors and officers of such Issuer to execute and deliver, all such instruments and documents, and do or cause to be done all such other acts as may be, in the opinion of the Collateral Trustee, necessary or advisable to register the Pledged Stock, or that portion thereof to be sold, under the provisions of the Securities Act, (ii) use its best efforts to cause the registration statement relating thereto to become effective and to remain effective for a period of one year from the date of the first public offering of the Pledged Stock, or that portion thereof to be sold, and (iii) make all amendments thereto and/or to the related prospectus which, in the opinion of the Collateral Trustee, are necessary or advisable, all in conformity with the requirements of the Securities Act and the rules and regulations of the Securities and Exchange Commission applicable thereto. Each Grantor agrees to cause such Issuer to comply with the provisions of the securities or "Blue Sky" laws of any and all jurisdictions which the Collateral Trustee shall designate and to make available to its security holders, as soon as practicable, an earnings statement (which need not be audited) which will satisfy the provisions of Section 11(a) of the Securities Act.
- (b) Each Grantor recognizes that the Collateral Trustee may be unable to effect a public sale of any or all the Pledged Stock, by reason of certain prohibitions contained in the Securities Act and applicable state securities laws or otherwise, and may be compelled to resort to one or more private sales thereof to a restricted group of purchasers which will be obliged to agree, among other things, to acquire such securities for their own account for investment and not with a view to the distribution or resale thereof. Each Grantor acknowledges and agrees that any such private sale may result in prices and other terms less favorable than if such sale were a public sale and, notwithstanding such circumstances, agrees that any such private sale shall be deemed to have been made in a commercially reasonable manner. The Collateral Trustee shall be under no obligation to delay a sale of any of the Pledged Stock for the period of time necessary to permit the Issuer thereof to register such securities for public sale under the Securities Act, or under applicable state securities laws, even if such Issuer would agree to do so.
- (c) Each Grantor agrees to use its best efforts to do or cause to be done all such other acts as may be necessary to make such sale or sales of all or any portion of the Pledged Stock pursuant to this Section 5.7 valid and binding and in compliance with any and all other applicable Requirements of Law. Each Grantor further agrees that a breach of any of the covenants contained in this Section 5.7 will cause irreparable injury to the Collateral Trustee and the Secured Parties, that the Collateral Trustee and the Secured Parties have no adequate remedy at law in respect of such breach and, as a consequence, that each and every covenant contained in this Section 5.7 shall be specifically enforceable against such Grantor, and such Grantor hereby waives and agrees not to assert any defenses against an action for specific performance of such covenants except for a defense that no Event of Default has occurred under the Indenture.

5.8 <u>Deficiency</u>. Each Grantor shall remain liable for any deficiency if the proceeds of any sale or other disposition of the Collateral are insufficient to pay its Obligations and the fees and disbursements of any attorneys employed by the Collateral Trustee or any Secured Party to collect such deficiency.

5.9 <u>ULC Limitation</u>. Notwithstanding any provisions to the contrary contained in this Agreement, the Indenture or any other document or agreement among all or some of the parties hereto, each Grantor is as of the date of this Agreement the sole registered and beneficial owner of all Pledged ULC Shares more particularly described in Schedule 2 to this Agreement and will remain so until such time as such Pledged ULC Shares are fully and effectively transferred into the name of the Secured Party or any other person on the books and records of such ULC. Nothing in this Agreement, the Indenture or any other document or agreement delivered among all or some of the parties hereto is intended to or shall constitute the Secured Party or any person other than a Grantor to be a member or shareholder of any ULC until such time as written notice is given to the applicable Grantor and all further steps are taken so as to register the Secured Party or other person as holder of the Pledged ULC Shares. The granting of the pledge and security interest pursuant to Section 2.1 does not make the Secured Party a successor to any Grantor as a member or shareholder of any ULC, and neither the Secured Party nor any of its respective successors or assigns hereunder shall be deemed to become a member or shareholder of any ULC by accepting this Agreement or exercising any right granted herein unless and until such time, if any, when the Secured Party or any successor or assign expressly becomes a registered member or shareholder of any ULC. Each Grantor shall be entitled to receive and retain for its own account any dividends or other distributions if any, in respect of the Collateral, and shall have the right to vote such Pledged ULC Shares and to control the direction, management and policies of the ULC issuing such Pledged ULC Shares to the same extent as such Grantor would if such Pledged ULC Shares were not pledged to the Secured Party or to any other person pursuant hereto. To the extent any provision hereof would have the effect of constituting the Secured Party to be a member or shareholder of any ULC prior to such time, such provision shall be severed herefrom and ineffective with respect to the relevant Pledged ULC Shares without otherwise invalidating or rendering unenforceable this Agreement or invalidating or rendering unenforceable such provision insofar as it relates to Collateral other than Pledged ULC Shares. Notwithstanding anything herein to the contrary (except to the extent, if any, that the Secured Party or any of its successors or assigns hereafter expressly becomes a registered member or shareholder of any ULC), neither the Secured Party nor any of its respective successors or assigns shall be deemed to have assumed or otherwise become liable for any debts or obligations of any ULC. Except upon the exercise by the Secured Party or other persons of rights to sell or otherwise dispose of Pledged ULC Shares or other remedies following the occurrence and during the continuance of an Event of Default, each Grantor shall not cause or permit, or enable any ULC in which it holds Pledged ULC Shares to cause or permit, the Secured Party to: (a) be registered as member or shareholder of such ULC; (b) have any notation entered in its favour in the share register of such ULC; (c) be held out as member or shareholder of such ULC; (d) receive, directly or indirectly, any dividends, property or other distributions from such ULC by reason of the Secured Party or other person holding a security interest in the Pledged ULC Shares; or (e) act as a member or shareholder of such ULC, or exercise any rights of a member or shareholder of such ULC, including the right to attend a meeting of such ULC or vote the shares of such ULC.

SECTION 6. THE COLLATERAL TRUSTEE

- 6.1 <u>Collateral Trustee's Appointment as Attorney-in-Fact, etc.</u> (a) Each Grantor hereby irrevocably constitutes and appoints the Collateral Trustee and any officer or agent thereof, with full power of substitution, as its true and lawful attorney-in-fact with full irrevocable power and authority in the place and stead of such Grantor and in the name of such Grantor or in its own name, for the purpose of carrying out the terms of this Agreement, to take any and all appropriate action and to execute any and all documents and instruments which may be necessary or desirable to accomplish the purposes of this Agreement, and, without limiting the generality of the foregoing, each Grantor hereby gives the Collateral Trustee the power and right, but not the obligation, on behalf of such Grantor, without notice to or assent by such Grantor, to do any or all of the following:
 - (i) in the name of such Grantor or its own name, or otherwise, take possession of and indorse and collect any checks, drafts, notes, acceptances or other instruments for the payment of moneys due under any Receivable or with respect to any other Collateral and file any claim or take any other action or proceeding in any court of law or equity or otherwise deemed appropriate by the Collateral Trustee for the purpose of collecting any and all such moneys due under any Receivable or with respect to any other Collateral when payable;
 - (ii) in the case of any Intellectual Property, execute and deliver, and have recorded, any and all agreements, instruments, documents and papers as the Collateral Trustee may request to evidence the Collateral Trustee's and the Secured Parties' security interest in such Intellectual Property and the goodwill and general intangibles of such Grantor relating thereto or represented thereby;
 - (iii) pay or discharge taxes and Liens levied or placed on or threatened against the Collateral, effect any repairs or any insurance called for by the terms of this Agreement and pay all or any part of the premiums therefor and the costs thereof;
 - (iv) execute, in connection with any sale provided for in Section 5.6 or 5.7, any indorsements, assignments or other instruments of conveyance or transfer with respect to the Collateral;
 - (v) (1) direct any party liable for any payment under any of the Collateral to make payment of any and all moneys due or to become due thereunder directly to the Collateral Trustee or as the Collateral Trustee shall direct; (2) ask or demand for, collect, and receive payment of and receipt for, any and all moneys, claims and other amounts due or to become due at any time in respect of or arising out of any Collateral; (3) sign and indorse any invoices, freight or express bills, bills of lading, storage or warehouse receipts, drafts against debtors, assignments, verifications, notices and other documents in connection with any of the Collateral; (4) commence and prosecute any suits, actions or proceedings at law or in equity in any court of competent jurisdiction to collect the Collateral or any portion thereof and to enforce any other right in respect of any Collateral; (5) defend any suit, action or proceeding brought against such Grantor with respect to any Collateral; (6) settle, compromise or adjust any such suit, action or proceeding and, in connection therewith, give such discharges or releases as the Collateral Trustee may deem appropriate; (7) assign any Copyright, Patent or Trademark (along with the goodwill of the business to which any such Copyright, Patent or Trademark pertains), throughout the world for such term or terms, on such conditions, and in such manner, as the Collateral Trustee shall in its sole discretion determine; and (8) generally, sell, transfer, pledge and make any agreement with respect to or otherwise deal with any of the Collateral as fully and completely as though the Collateral Trustee were the absolute owner thereof for all purposes, and do, at the Collateral Trustee's option and such Grantor's expense, at any time, or from time to time, all acts

and things which the Collateral Trustee deems necessary to protect, preserve or realize upon the Collateral and the Collateral Trustee's and the Secured Parties' security interests therein and to effect the intent of this Agreement, all as fully and effectively as such Grantor might do; and

(vi) license or sublicense whether on an exclusive or non-exclusive basis, any Intellectual Property for such term and on such conditions and in such manner as the Collateral Trustee shall in its sole judgment determine and, in connection therewith, such Grantor hereby grants to the Collateral Trustee for the benefit of the Secured Parties a royalty-free, world-wide irrevocable license of its Intellectual Property.

Anything in this Section 6.1 (a) to the contrary notwithstanding, the Collateral Trustee agrees that it will not exercise any rights under the power of attorney provided for in this Section 6.1(a) unless an Event of Default shall have occurred and be continuing.

- (b) If any Grantor fails to perform or comply with any of its agreements contained herein, the Collateral Trustee, at its option, but without any obligation so to do, may perform or comply, or otherwise cause performance or compliance, with such agreement.
- (c) The expenses of the Collateral Trustee incurred in connection with actions undertaken as provided in this Section 6.1, together with interest thereon at a rate per annum equal to the rate per annum at which interest would then be payable under the Indenture, from the date of payment by the Collateral Trustee to the date reimbursed by the relevant Grantor, shall be payable by such Grantor to the Collateral Trustee promptly on demand.
- (d) Each Grantor hereby ratifies all that said attorneys shall lawfully do or cause to be done by virtue hereof. All powers, authorizations and agencies contained in this Agreement are coupled with an interest and are irrevocable until this Agreement is terminated and the security interests created hereby are released.
- 6.2 <u>Duty of Collateral Trustee</u>. The Collateral Trustee's sole duty with respect to the custody, safekeeping and physical preservation of the Collateral in its possession, under Section 9-207 of the New York UCC or otherwise, shall be to deal with it in the same manner as the Collateral Trustee deals with similar property for its own account. Neither the Collateral Trustee, any Secured Party nor any of their respective officers, directors, employees or agents shall be liable for failure to demand, collect or realize upon any of the Collateral or for any delay in doing so or shall be under any obligation to sell or otherwise dispose of any Collateral upon the request of any Grantor or any other Person or to take any other action whatsoever with regard to the Collateral or any part thereof. The powers conferred on the Collateral Trustee and the Secured Parties hereunder are solely to protect the Collateral Trustee's and the Secured Parties' interests in the Collateral and shall not impose any duty upon the Collateral Trustee or any Secured Party to exercise any such powers. The Collateral Trustee and the Secured Parties shall be accountable only for amounts that they actually receive as a result of the exercise of such powers, and neither they nor any of their officers, directors, employees or agents shall be responsible to any Grantor for any act or failure to act hereunder, except for their own gross negligence or willful misconduct. The Collateral Trustee will not be required to advance or expend any funds or otherwise incur any financial liability in the performance of its duties or the exercise of its powers or rights hereunder (including, without limitation, taking any actions under or in respect of control agreements) unless it has been provided with pre-funding, security or indemnity reasonably satisfactory to it (and, for the avoidance of doubt, from parties reasonably satisfactory to it) against any and all cost, loss, liability or expense which may be incurred by it by reason of taking or continuing t

- 6.3 Execution of Financing Statements. Pursuant to any applicable law, each Grantor authorizes the Collateral Trustee to file or record financing statements and other filing or recording documents or instruments with respect to the Collateral without the signature of such Grantor in such form and in such offices as the Collateral Trustee determines appropriate to perfect the security interests of the Collateral Trustee under this Agreement. Each Grantor authorizes the Collateral Trustee to use the collateral description "all personal property of the debtor, whether now owned or hereafter acquired" or "all assets of debtor, whether now owned or hereafter acquired" in any such financing statements. Each Grantor hereby ratifies and authorizes the filing by the Collateral Trustee of any financing statement with respect to the Collateral made prior to the date hereof.
- 6.4 <u>Authority of Collateral Trustee</u>. Each Grantor acknowledges that the rights and responsibilities of the Collateral Trustee under this Agreement with respect to any action taken by the Collateral Trustee or the exercise or non-exercise by the Collateral Trustee of any option, voting right, request, judgment or other right or remedy provided for herein or resulting or arising out of this Agreement shall, as between the Collateral Trustee and the other Secured Parties, be governed by the Indenture and, to the extent then in effect, by the Collateral Trust Agreement and by such other agreements with respect thereto as may exist from time to time among them, but, as between the Collateral Trustee and the Grantors, the Collateral Trustee shall be conclusively presumed to be acting as agent for the Secured Parties with full and valid authority so to act or refrain from acting, and no Grantor shall be under any obligation, or entitlement, to make any inquiry respecting such authority. The Collateral Trustee is entitled to all the same protections hereunder that the Trustee has under the Indenture.
- 6.5 <u>Applicable Secured Debt Representative</u>. Notwithstanding anything to the contrary herein and for the avoidance of doubt, the rights, duties and obligations of the Collateral Trustee hereunder shall be subject to the terms of the Collateral Trust Agreement, to the extent in effect, and any rights, duties and obligations of the Applicable Secured Debt Representative thereunder.

SECTION 7. MISCELLANEOUS

- 7.1 <u>Amendments in Writing</u>. None of the terms or provisions of this Agreement may be waived, amended, supplemented or otherwise modified except in accordance with Section 9.01 or Section 9.02, as applicable, of the Indenture.
- 7.2 <u>Notices</u>. All notices, requests and demands to or upon the Collateral Trustee or any Grantor hereunder shall be effected in the manner provided for in Section 13.01 of the Indenture.
- 7.3 No Waiver by Course of Conduct; Cumulative Remedies. Neither the Collateral Trustee nor any Secured Party shall by any act (except by a written instrument pursuant to Section 7.1), delay, indulgence, omission or otherwise be deemed to have waived any right or remedy hereunder or to have acquiesced in any Default or Event of Default. No failure to exercise, nor any delay in exercising, on the part of the Collateral Trustee or any Secured Party, any right, power or privilege hereunder shall operate as a waiver thereof. No single or partial exercise of any right, power or privilege hereunder shall preclude any other or further exercise thereof or the exercise of any other right, power or privilege. A waiver by the Collateral Trustee or any Secured Party of any right or remedy hereunder on any one occasion shall not be construed as a bar to any right or remedy which the Collateral Trustee or such Secured Party would otherwise have on any future occasion. The rights and remedies herein provided are cumulative, may be exercised singly or concurrently and are not exclusive of any other rights or remedies provided by law.

- 7.4 Enforcement Expenses; Indemnification. (a) Each Grantor agrees to pay, or reimburse each Secured Party and the Collateral Trustee and each of their respective Affiliates and each of the directors, officers, partners, trustees, employees, attorneys and agents, (and in each case) their respective heirs, representatives, successors and assigns for, all its costs and expenses incurred in enforcing the Indemnified Liabilities (as defined in the Collateral Trust Agreement) (other than any such enforcement determined by a final, non-appealable judgment of a court to have been in bad faith) or preserving any rights under this Agreement, the Indenture and the other Collateral Documents to which such Grantor is a party, including, without limitation, the fees and disbursements of counsel (including the allocated fees and expenses of in-house counsel) to each Secured Party and of counsel to the Collateral Trustee.
- (b) Each Grantor agrees to pay, and to save the Collateral Trustee and the Secured Parties harmless from, any and all liabilities with respect to, or resulting from any delay in paying, any and all stamp, excise, sales or other taxes which may be payable or determined to be payable with respect to any of the Collateral or in connection with any of the transactions contemplated by this Agreement.
- (c) The agreements in this Section shall survive repayment of the Obligations and all other amounts payable under the Indenture, the U.S. Notes and the other Collateral Documents.
- 7.5 <u>Successors and Assigns</u>. This Agreement shall be binding upon the successors and assigns of each Grantor and shall inure to the benefit of the Collateral Trustee and the Secured Parties and their successors and assigns; <u>provided</u> that no Grantor may assign, transfer or delegate any of its rights or obligations under this Agreement without the prior written consent of the Collateral Trustee.
- 7.6 Set-Off. Each Grantor hereby irrevocably authorizes the Collateral Trustee and each Secured Party at any time and from time to time while an Event of Default shall have occurred and be continuing, without notice to such Grantor or any other Grantor, any such notice being expressly waived by each Grantor, to set-off and appropriate and apply any and all deposits (general or special, time or demand, provisional or final), in any currency, and any other credits, indebtedness or claims, in any currency, in each case whether direct or indirect, absolute or contingent, matured or unmatured, at any time held or owing by the Collateral Trustee or such Secured Party to or for the credit or the account of such Grantor, or any part thereof in such amounts as the Collateral Trustee or such Secured Party hereunder, in any currency, whether arising hereunder, under the Indenture, the U.S. Notes or any other Collateral Document, as the Collateral Trustee or such Secured Party may elect, whether or not the Collateral Trustee or any Secured Party has made any demand for payment and although such obligations, liabilities and claims may be contingent or unmatured. The Collateral Trustee and each Secured Party shall notify such Grantor promptly of any such set-off and the application made by the Collateral Trustee or such Secured Party of the proceeds thereof, provided that the failure to give such notice shall not affect the validity of such set-off and application. The rights of the Collateral Trustee and each Secured Party under this Section are in addition to other rights and remedies (including, without limitation, other rights of set-off) which the Collateral Trustee or such Secured Party may have.
- 7.7 <u>Counterparts</u>. This Agreement may be executed by one or more of the parties to this Agreement on any number of separate counterparts (including by telecopy), and all of said counterparts taken together shall be deemed to constitute one and the same instrument.
- 7.8 <u>Severability</u>. Any provision of this Agreement which is prohibited or unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective to the extent of such prohibition or unenforceability without invalidating the remaining provisions hereof, and any such prohibition or unenforceability in any jurisdiction shall not invalidate or render unenforceable such provision in any other jurisdiction.

- 7.9 <u>Section Headings</u>. The Section headings used in this Agreement are for convenience of reference only and are not to affect the construction hereof or be taken into consideration in the interpretation hereof.
- 7.10 <u>Integration</u>. This Agreement, the Indenture, the U.S. Notes and the other Collateral Documents represent the agreement of the Grantors, the Collateral Trustee and the Secured Parties with respect to the subject matter hereof and thereof, and there are no promises, undertakings, representations or warranties by the Collateral Trustee or any Secured Party relative to subject matter hereof and thereof not expressly set forth or referred to herein or in the Indenture, the U.S. Notes and the other Collateral Documents.

7.11 <u>GOVERNING LAW</u>. THIS AGREEMENT SHALL BE GOVERNED BY, AND CONSTRUED AND INTERPRETED IN ACCORDANCE WITH, THE LAW OF THE STATE OF NEW YORK WITHOUT REGARD TO CONFLICTS OF LAW PRINCIPLES THEREOF.

- 7.12 <u>Submission To Jurisdiction</u>; <u>Waivers</u>. Each Grantor hereby irrevocably and unconditionally:
- (a) submits for itself and its property in any legal action or proceeding relating to this Agreement and the other Collateral Documents to which it is a party, or for recognition and enforcement of any judgment in respect thereof, to the non-exclusive general jurisdiction of the courts of the State of New York, the courts of the United States of America for the Southern District of New York, and appellate courts from any thereof;
- (b) consents that any such action or proceeding may be brought in such courts and waives any objection that it may now or hereafter have to the venue of any such action or proceeding in any such court or that such action or proceeding was brought in an inconvenient court and agrees not to plead or claim the same;
- (c) agrees that service of process in any such action or proceeding may be effected by mailing a copy thereof by registered or certified mail (or any substantially similar form of mail), postage prepaid, to such Grantor at its address referred to in Section 7.2 or at such other address of which the Collateral Trustee shall have been notified pursuant thereto;
- (d) agrees that nothing herein shall affect the right to effect service of process in any other manner permitted by law or shall limit the right to sue in any other jurisdiction; and
- (e) waives, to the maximum extent not prohibited by law, any right it may have to claim or recover in any legal action or proceeding referred to in this Section any special, exemplary, punitive or consequential damages.
 - 7.13 Acknowledgements. Each Grantor hereby acknowledges that:
- (a) it has been advised by counsel in the negotiation, execution and delivery of this Agreement and the other Collateral Documents to which it is a party;
- (b) neither the Collateral Trustee nor any Secured Party has any fiduciary relationship with or duty to any Grantor arising out of or in connection with this Agreement or any of the

other Collateral Documents, and the relationship between the Grantors, on the one hand, and the Collateral Trustee and Secured Parties, on the other hand, in connection herewith or therewith is solely that of debtor and creditor; and

- (c) no joint venture is created hereby or by the other Collateral Documents or otherwise exists by virtue of the transactions contemplated hereby among the Secured Parties or among the Grantors and the Secured Parties.
- 7.14 <u>Additional Grantors</u>. Each Affiliate of the Company that is required to become a party to this Agreement pursuant to the Indenture shall become a Grantor for all purposes of this Agreement upon execution and delivery by such Affiliate of the Company of an Assumption Agreement in the form of Annex I hereto.
- 7.15 Releases. (a) The Collateral shall be released from the Liens created hereby at the times, and to the extent, provided under Section 12.04 of the Indenture and (in the case of a complete release of the Collateral) this and all obligations (other than those expressly stated to survive such termination) of the Collateral Trustee and each Grantor hereunder shall terminate, all without delivery of any instrument or performance of any act by any party, and all rights to the Collateral shall revert to the Grantors. At the request and sole expense of any Grantor following any such termination, the Collateral Trustee shall deliver to such Grantor, on a date to be reasonably specified by such Grantor, any Collateral held by the Collateral Trustee hereunder, and execute and deliver to such Grantor such documents as such Grantor shall reasonably request to evidence such termination.
- 7.16 Approvals. Any provision contained herein to the contrary notwithstanding, no action shall be taken hereunder by the Collateral Trustee with respect to the Collateral unless and until all applicable requirements, if any, under applicable federal and state laws and the respective rules and regulations thereunder and thereof, as well as any other laws, rules and regulations of any Governmental Authority applicable to or having jurisdiction over any Grantor, have in the reasonable judgment of the Collateral Trustee been fully satisfied to the extent necessary to take such action and there have been obtained such consents, approvals and authorizations as may be required to be obtained from applicable federal, state and local regulatory authorities and municipalities and any other Governmental Authority under the terms of any franchise, license or similar operating right held by such Grantor in order to take such action. It is the intention of the parties hereto that the grant to the Collateral Trustee of security interests in the Collateral, and all rights and remedies by the Collateral Trustee with respect to the Collateral, shall in all relevant aspects be subject to and governed by said statutes, rules and regulations and that nothing in this Agreement shall be construed to diminish the control exercised by any Grantor, except in accordance with the provisions of any statutory requirements and rules and regulations. By its acceptance of this Agreement, the Collateral Trustee agrees that it will not take any action pursuant to this Agreement which constitutes or results in any assignment of a license or franchise or any change of control over the communications properties owned and operated by any Grantor, if such assignment of license or franchise or change of control would, under then existing law or under any franchise, require the prior approval of Governmental Authority, without first obtaining such approval. Upon the exercise by the Collateral Trustee of any power, right, privilege or remedy pursuant to this Agreement which requires any consent, approval, recording, qualification or authorization of any Governmental Authority, each Grantor will execute and deliver, or will cause the execution and delivery of, all applications, certificates, instruments and other documents and papers that the Collateral Trustee may reasonably require in order for such governmental consent, approval, recording, qualification or authorization to be obtained. Each Grantor agrees to use its best efforts to cause such governmental consents, approvals, recordings, qualifications and authorizations to be forthcoming.

7.17 <u>WAIVER OF JURY TRIAL</u>. EACH GRANTOR AND, BY ACCEPTANCE OF THE BENEFITS HEREOF, EACH AGENT AND EACH SECURED PARTY, HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES TRIAL BY JURY IN ANY LEGAL ACTION OR PROCEEDING RELATING TO THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT AND FOR ANY COUNTERCLAIM THEREIN.

7.18 <u>Collateral Trust Agreement</u>. Notwithstanding anything herein to the contrary, the lien and security interest granted to the Collateral Trustee pursuant to this Agreement and the exercise of any right or remedy by such Collateral Trustee hereunder are subject to the provisions of any Collateral Trust Agreement then in effect. In the event of any conflict between the terms of any Collateral Trust Agreement then in effect and this Agreement, the terms of such Collateral Trust Agreement will govern.

7.19 <u>Force Majeure</u>. In no event shall the Collateral Trustee be responsible or liable for any failure or delay in the performance of its obligations hereunder arising out of or caused by, directly or indirectly, forces beyond its control, including, without limitation, strikes, work stoppages, accidents, acts of war or terrorism, civil or military disturbances, nuclear or natural catastrophes or acts of God, and interruptions, loss or malfunctions of utilities, communications or computer (software and hardware) services; it being understood that the Collateral Trustee shall use reasonable efforts which are consistent with accepted practices in the banking industry to resume performance as soon as practicable under the circumstances.

7.20 <u>U.S.A. Patriot Act</u>. The parties hereto acknowledge that in accordance with Section 326 of the U.S.A. Patriot Act, the Collateral Trustee, like all financial institutions and in order to help fight the funding of terrorism and money laundering, is required to obtain, verify, and record information that identifies each person or legal entity that establishes a relationship or opens an account with the Collateral Trustee. The parties to this Agreement agree that they will provide the Collateral Trustee with such information as it may request in order for the Collateral Trustee to satisfy the requirements of the U.S.A. Patriot Act.

[Signature page follows.]

written.	
	<u>COMPANY</u> :
	PRIMUS TELECOMMUNICATIONS HOLDING, INC.
	By:
	Name:
	Title:
	GUARANTORS :
	PRIMUS TELECOMMUNICATIONS GROUP, INCORPORATED
	By:
	Name:
	Title:
	PRIMUS TELECOMMUNICATIONS IHC, INC.
	PRIMUS TELECOMMUNICATIONS INTERNATIONAL, INC.
	LEAST COST ROUTING, INC.
	TRESCOM U.S.A., INC.
	iPRIMUS.COM, INC.
	Ву:
	Name:

Title:

IN WITNESS WHEREOF, each of the undersigned has caused this Collateral Agreement to be duly executed and delivered as of the date first above

PRIMUS TELECOMMUNICATIONS, INC.
iPRIMUS USA, INC.
By:
Name:
Title:
LINGO HOLDINGS, INC.
LINGO, INC.
LINGO NETWORK SERVICES, INC.
Ву:
Name:
Title:
COLLATERAL TRUSTEE:
U.S. BANK NATIONAL ASSOCIATION, as Collateral Trustee
By:
Name:
Title:

ASSUMPTION AGREEMENT

This ASSUMPTION AGREEMENT, dated as of [mm/dd/yy], is delivered by [NAME OF GRANTOR,] a [STATE OF INCORPORATION] [corporation/limited liability company/limited partnership] (the "Additional Grantor") pursuant to the Collateral Agreement, dated as of December [__], 2009 (as it may from time to time be amended, restated, modified or supplemented, the "Collateral Agreement"; all capitalized terms not defined herein shall have the meaning ascribed to them in the Collateral Agreement), made by Primus Telecommunications Holding, Inc. and the other Grantors named therein in favor of U.S. Bank National Association, as Collateral Trustee.

By executing and delivering this Assumption Agreement, the Additional Grantor, as provided in Section 7.14 of the Collateral Agreement, hereby becomes a party to the Collateral Agreement as a Grantor thereunder with the same force and effect as if originally named therein as a Grantor and, without limiting the generality of the foregoing, hereby expressly assumes all obligations and liabilities of a Grantor thereunder. The Additional Grantor confirms the grant to the Collateral Trustee set forth in the Collateral Agreement of, and does hereby assign and transfer to the Collateral Trustee, and hereby grants to the Collateral Trustee, for the ratable benefit of the Secured Parties, a security interest in, all of the Collateral now owned or at any time hereafter acquired by the Additional Grantor or in which the Additional Grantor now or at any time in the future may acquire any right, title or interest, as collateral security for the prompt and complete payment and performance when due (whether at the stated maturity, by acceleration or otherwise) of such Additional Grantor's Obligations.

Additional Grantor represents and warrants that the attached Supplements to Schedules, Annex I-A to the Assumption Agreement, accurately and completely set forth all additional information required pursuant to the Collateral Agreement and hereby agrees that such Supplements to Schedules shall constitute part of the Schedules to the Collateral Agreement. The Additional Grantor hereby represents and warrants, to the extent applicable, that each of the representations and warranties of the Additional Grantor contained in Section 3 of the Collateral Agreement is true and correct on and as of the date hereof (after giving effect to this Assumption Agreement) as if made on and as of such date (except to the extent such representation and warranty expressly relates to an earlier date).

GOVERNING LAW. THIS AGREEMENT SHALL BE GOVERNED BY, AND CONSTRUED AND INTERPRETED IN ACCORDANCE WITH, THE LAW OF THE STATE OF NEW YORK WITHOUT REGARD TO CONFLICTS OF LAW PRINCIPLES THEREOF.

IN WITNESS WHEREOF, Additional Grantor has caused this Assumption Agreement to be duly executed and delivered by its duly authorized officer as of [mm/dd/yy].

[NAME OF ADDITIONAL GRANTOR]	
By:	
Name:	
Title:	

Supplement to Schedule 2

DESCRIPTION OF PLEDGED SECURITIES

Supplement to Schedule 3

FILINGS AND OTHER ACTIONS REQUIRED TO PERFECT SECURITY INTEREST

Supplement to Schedule 4

JURISDICTION OF ORGANIZATION, IDENTIFICATION NUMBER AND LOCATION OF CHIEF EXECUTIVE OFFICE

Supplement to Schedule 5

LOCATION OF INVENTORY AND EQUIPMENT

Supplement to Schedule 6

INTELLECTUAL PROPERTY

Supplement to Schedule 8

DEPOSIT ACCOUNTS AND SECURITIES ACCOUNTS

Supplement to Schedule 9

RECEIVABLES

2

CANADIAN COLLATERAL AGREEMENT

made by

PRIMUS TELECOMMUNICATIONS CANADA INC.

and certain of its Affiliates

in favor of

COMPUTERSHARE TRUST COMPANY OF CANADA,

as Canadian Collateral Trustee

Dated as of December 22, 2009

TABLE OF CONTENTS

SECTION 1. DEFINED TERMS		<u>Page</u> 2
1.1	Defined Texas Definitions	2
1.1	Other Definitional Provisions	6
SECTION 2.	6	
2.1	Grant of Security	6
2.2	Exception to Last Day	7
2.3	Attachment	7
SECTION 3.	REPRESENTATIONS AND WARRANTIES	8
3.1	Title; No Other Liens	8
3.2	Perfected First Priority Liens	8
3.3	Jurisdiction of Organization; Chief Executive Office	8
3.4	Inventory and Equipment	8
3.5	[Intentionally Deleted]	9
3.6	Investment Property	9
3.7	Receivables	9
3.8	Intellectual Property	9
3.9	Vehicles	10
3.10	Deposit Accounts and Securities Accounts	10
3.11	[Intentionally Deleted]	10
SECTION 4. COVENANTS		10
4.1	Delivery of Instruments and Chattel Paper	10
4.2	Maintenance of Insurance	10
4.3	Payment of Obligations.	11
4.4	Maintenance of Perfected Security Interest; Further Documentation.	11
4.5	Changes in Name, etc.	11
4.6	Notices	12
4.7	Investment Property	12
4.8	Receivables	13
4.9	Intellectual Property	14
4.10	[Intentionally Deleted]	15
SECTION 5.	REMEDIAL PROVISIONS	15
5.1	Certain Matters Relating to Receivables	15
5.2	Communications with Obligors; Grantors Remain Liable	16
5.3	Pledged Stock	17
5.4	Proceeds to be Turned Over To Canadian Collateral Trustee	18
5.5	Application of Proceeds	18
5.6	PPSA and Other Remedies	18
5.7	Registration Rights	19

19

		Page
5.8	Deficiency	20
5.9	Appointment of Receiver	20
5.10	ULC Limitation	21
SECTION 6.	22	
6.1	Canadian Collateral Trustee's Appointment as Attorney-in-Fact, etc	22
6.2	Duty of Canadian Collateral Trustee	23
6.3	Execution of Financing Statements	24
6.4	Authority of Canadian Collateral Trustee	24
6.5	Applicable Secured Debt Representative	25
SECTION 7.	MISCELLANEOUS	25
7.1	Amendments in Writing	25
7.2	Notices	25
7.3	No Waiver by Course of Conduct; Cumulative Remedies	25
7.4	Enforcement Expenses; Indemnification	25
7.5	Successors and Assigns	26
7.6	Set-Off	26
7.7	Counterparts	26
7.8	Severability	26
7.9	Section Headings	26
7.10	Integration	27
7.11	GOVERNING LAW	27
7.12	Submission To Jurisdiction; Waivers	27
7.13	Acknowledgements	27
7.14	Additional Grantors	28
7.15	Releases	28
7.16	Approvals	28
7.17	WAIVER OF JURY TRIAL	29
7.18	Collateral Trust Agreement	29
7.19	Force Majeure	29
7.20	U.S.A. Patriot Act	29
7.21	Permitted Liens	29
7.22	Amalgamation	30

Schedules [intentionally omitted] Schedule 1 Schedule 2 Description of Pledged Securities Schedule 3 Filings and Other Actions Required to Perfect Security Interest Schedule 4 Jurisdiction of Organization, Identification Number and Location of Chief Executive Office Schedule 5 Locations of Inventory and Equipment Intellectual Property
[intentionally omitted]
Deposit Accounts and Securities Accounts Schedule 6 Schedule 7 Schedule 8 Receivables Schedule 9 Annexes Annex I Assumption Agreement

CANADIAN COLLATERAL AGREEMENT, dated as of December 22, 2009, made by each of the signatories hereto (together with any other entity that may become a party hereto as provided herein, the "Grantors"), in favor of Computershare Trust Company of Canada, as Canadian Collateral Trustee (in such capacity and together with any successor, the "Canadian Collateral Trustee") for the benefit of the holders (the "Holders") from time to time of the 13% Senior Secured Canadian Notes due 2016 (the "Canadian Notes") issued by Primus Telecommunications Canada Inc. (the "Company") pursuant to the Indenture, dated as of December 22, 2009 by and among the Company, Primus Telecommunications Holding, Inc., Primus Telecommunications Group, Incorporated, the guarantors party thereto, The Bank of New York Mellon as trustee (in such capacity and together with any successor, the "Trustee") the Canadian Collateral Trustee, the US Collateral Trustee (as defined therein) (as amended, supplemented or otherwise modified from time to time, the "Indenture"), and each of the other Secured Parties (as defined).

WITNESSETH:

WHEREAS, pursuant to the terms of the Indenture, the Company has issued to the Holders the Canadian Notes upon the terms and subject to the conditions set forth therein, and the other Grantors have guaranteed the obligations of the Company thereunder;

WHEREAS, the Canadian Collateral Trustee has agreed to act as Canadian Collateral Trustee for the Secured Parties for the purposes of holding any and all security for the payment and performance of the obligations of the Company and the other Grantors under the Indenture and the other Indenture Documents;

WHEREAS, the Company is a member of an affiliated group of companies that includes each other Grantor;

WHEREAS, the proceeds of the extensions of credit under the Canadian Notes will be used in part to enable the Company to make valuable transfers to one or more of the other Grantors in connection with the operation of their respective businesses;

WHEREAS, the Company and the other Grantors are engaged in related businesses, and each Grantor will derive substantial direct and indirect benefit from the extensions of credit under the Canadian Notes; and

WHEREAS, it is a condition precedent to the obligation of the Holders to purchase the Canadian Notes that the Grantors shall have executed and delivered this Agreement to the Canadian Collateral Trustee;

NOW, THEREFORE, in consideration of the premises and to induce the Holders to purchase the Canadian Notes, each Grantor hereby agrees with the Canadian Collateral Trustee, for the benefit of the Secured Parties, as follows:

SECTION 1. DEFINED TERMS

- 1.1 <u>Definitions</u>. (a) Unless otherwise defined herein, terms defined in the Indenture and used herein shall have the meanings given to them in the Indenture and the following terms are used herein as defined in the PPSA: Account, Certificated Security, Chattel Paper, Documents of Title, Equipment, Goods, Instruments, Intangibles, Inventory, motor vehicle, Serial Numbered Goods, Securities Account, Security and Security Entitlement.
 - (b) The following terms shall have the following meanings:
 - "Additional Notes": any additional notes issued and outstanding under the Indenture (or under any supplemental indenture).
 - "Agreement": this Canadian Collateral Agreement, as the same may be amended, supplemented or otherwise modified from time to time.
 - "CIPO": means the Canadian Intellectual Copyright Office.
 - "Collateral": as defined in Section 3.
 - "Collateral Account": any collateral account established by the Canadian Collateral Trustee as provided in Section 5.1 or 5.4.
 - "Collateral Documents": this Agreement, the Collateral Trust Agreement and each of the security agreements and other instruments and documents executed and delivered pursuant to the foregoing or this Agreement.
 - "Collateral Trust Agreement": a collateral trust agreement dated the date hereof among the Company, the Guarantors from time to time party thereto, the Priority Lien Representatives (as defined therein) from time to time party thereto, the Parity Lien Representatives (as defined therein) from time to time party thereto, the Trustee and the Canadian Collateral Trustee, as amended, supplemented, amended and restated or otherwise modified and in effect from time to time.

"Company Obligations": the collective reference to the unpaid principal of and interest on the Canadian Notes and Additional Notes, to the extent such notes are outstanding, and all other obligations and liabilities of the Company to the Canadian Collateral Trustee or any Secured Party, whether direct or indirect, absolute or contingent, due or to become due, or now existing or hereafter incurred, which may arise under, out of, or in connection with, this Agreement, the Canadian Notes, the Indenture, the other Indenture Documents, the other Collateral Documents, or any other document made, delivered or given in connection with any of the foregoing, and in each case whether on account of principal, interest, reimbursement obligations, fees, indemnities, cost, expense or otherwise (including, without limitation, all fees and disbursements of counsel to the Canadian Collateral Trustee or to any other Secured Party that are required to be paid by the Company pursuant to the terms of any of the foregoing agreements).

"Copyrights": (i) all copyrights arising under the laws of Canada, any other country or any political subdivision thereof, whether registered or unregistered and whether published or unpublished (including, without limitation, those listed in Schedule 6), all registrations and recordings thereof, and all applications in connection therewith, including, without limitation, all registrations, recordings and applications in the CIPO, and (ii) the right to obtain all renewals thereof.

"Copyright Licenses": any written agreement naming any Grantor as licensor or licensee (including, without limitation, those listed in Schedule 6), granting any right under any Copyright, including, without limitation, the grant of rights to manufacture, distribute, exploit and sell materials derived from any Copyright.

"Deposit Account": any demand, time, savings, passbook or similar account maintained with a bank or other deposit taking institution.

"<u>Designs</u>" means all of the following now owned or hereafter acquired by any Grantor: (a) all industrial designs and intangibles of like nature (whether registered or unregistered), now owned or existing or hereafter adopted or acquired, all registrations and recordings thereof, and all applications in connection therewith, including all registrations, recordings and applications in the CIPO or in any similar office or agency in any other country or any political subdivision thereof, and (b) all reissues, extensions or renewals thereof.

"Design Licence" means rights under any written agreement now owned or hereafter acquired by any Grantor granting any right to use any Design.

"Domestic Perfection Action": any of (a) filing of financing statements under the PPSA, (b) filings with CIPO Office, and (c) possession in Ontario of stock certificates, promissory notes, and other instruments and similar collateral. (other than any instruments for which an affidavit of loss has been delivered to the Canadian Collateral Trustee, provided that any replacement instruments shall be delivered to the Canadian Collateral Trustee promptly upon any Grantor obtaining possession of such replacement instrument).

"Excluded Assets": the collective reference to any contract, General Intangible, Copyright License, Patent License or Trademark License ("Intangible Assets"), in each case to the extent the grant by the relevant Grantor of a security interest pursuant to this Agreement in such Grantor's right, title and interest in such Intangible Asset or other property (A) is prohibited by legally enforceable provisions of such contract, agreement, instrument or indenture, or by any applicable law, (B) would give any other party to such contract, agreement, instrument or indenture a legally enforceable right to terminate its obligations thereunder or (C) is permitted only with the consent of another party, if the requirement to obtain such consent is legally enforceable and such consent has not been obtained; provided, that Excluded Assets shall not include any proceeds, products, substitutions or replacements of Excluded Assets (unless such proceeds, products, substitutions or replacements would otherwise constitute Excluded Assets).

"Guarantor Obligations": with respect to any Guarantor, the collective reference to all obligations and liabilities of such Guarantor which may arise under or in connection with the Indenture, the other Indenture Documents or any Collateral Document to which such Guarantor is a party, in each case whether on account of guarantee obligations, reimbursement obligations, fees, indemnities, costs, expenses or otherwise (including, without limitation, all fees and disbursements of counsel to the Canadian Collateral Trustee or to any Secured Party that are required to be paid by such Guarantor pursuant to the terms of this Agreement, the Indenture or any other Collateral Document).

"Guarantors": the collective reference to each Grantor other than the Company.

"Indenture Documents": the Indenture, the Canadian Notes and the other documentation from time to time executed and delivered in connection therewith, including any supplemental indentures.

"Intellectual Property": the collective reference to all rights, priorities and privileges relating to intellectual property, whether arising under Canada, multinational or foreign laws or otherwise, including, without limitation, the Copyrights, the Copyright Licenses, the Designs, the Patents, the Designs Licenses, the Patent Licenses, the Trademarks and the Trademark Licenses, and all rights to sue at law or in equity for any infringement or other impairment thereof, including the right to receive all proceeds and damages therefrom.

"Intercompany Note": any promissory note evidencing loans made by any Grantor to any other Grantor.

"Intercreditor Agreement": the Intercreditor Agreement, dated as of February 26, 2007 and amended by that certain first amendment as of July 1, 2009, by and among the Grantors party thereto, the Canadian Collateral Trustee (as successor First Lien Collateral Agent (as defined therein)), and U.S. Bank National Association as Second Lien Collateral Agent (as defined therein).

"Investment Property": means a security, whether certificated or uncertificated, security entitlement, securities account, commodity contract or commodity account and shall include, for greater certainty, all Pledged Notes and all Pledged Stock.

"Issuers": the collective reference to each issuer of any Investment Property.

"Obligations": (i) in the case of the Company, the Company Obligations, and (ii) in the case of each Guarantor, its Guarantor Obligations.

"Parent": Primus Telecommunications Group, Incorporated, a Delaware corporation.

"Patents": (i) all letters patent of Canada, any other country or any political subdivision thereof, all reissues and extensions thereof and all goodwill associated therewith, including, without limitation, any of the foregoing referred to in Schedule 6,

(ii) all applications for letters patent of Canada or any other country and all divisions, continuations and continuations-in-part thereof, including, without limitation, any of the foregoing referred to in <u>Schedule 6</u>, and (iii) all rights to obtain any reissues or extensions of the foregoing.

"<u>Patent License</u>": all agreements, whether written or oral, providing for the grant by or to any Grantor of any right to manufacture, use or sell any invention covered in whole or in part by a Patent, including, without limitation, any of the foregoing referred to in <u>Schedule 6</u>.

"<u>Pledged Notes</u>": all promissory notes, if any, listed on <u>Schedule 2</u> and all other promissory notes issued to or held by any Grantor (other than promissory notes issued in connection with extensions of trade credit by any Grantor in the ordinary course of business).

"Pledged Securities": the collective reference to the Pledged Notes and the Pledged Stock.

"<u>Pledged Stock</u>": the shares of Capital Stock listed on <u>Schedule 2</u>, together with any other shares, stock certificates, options or rights of any nature whatsoever in respect of the Capital Stock of any Person that may be issued or granted to, or held directly by, any Grantor while this Agreement is in effect.

"Pledged ULC Shares" shall mean the Pledged Stock which are shares in the capital stock of a ULC.

"<u>ULC</u>" shall mean any unlimited liability company or unlimited liability corporation existing under the laws of any province or territory of Canada and any successor to any such unlimited liability company or unlimited liability corporation.

"PPSA" means the *Personal Property Security Act* (Ontario); provided, that if the attachment, perfection or priority of the Secured Party's security interests in any Collateral are governed by the personal property security laws of any jurisdiction other than Ontario, PPSA shall mean those personal property laws in such other jurisdiction in Canada for the purpose of the provisions hereof relating to such attachment, perfection or priority and for the definitions related to such provisions.

"Proceeds": all "proceeds" as such term is defined in Section 9-102(a)(64) of the Uniform Commercial Code in effect in the State of New York on the date hereof and, in any event, including, without limitation, all dividends or other income from the Investment Property, collections thereon or distributions or payments with respect thereto.

"Receivable": any right to payment for goods sold, leased, licensed, assigned or otherwise disposed of, or for services rendered, whether or not such right is evidenced by an Instrument or Chattel Paper and whether or not it has been earned by performance (including, without limitation, any Account).

- "Secured Parties": the collective reference to the Canadian Collateral Trustee and the Trustee and the Holders.
- "Securities Act": the Securities Act (Ontario), as amended.
- "Supplemental Security Documents": as defined in the Collateral Trust Agreement.
- "Trademarks": (i) all trademarks, trade names, corporate names, company names, business names, fictitious business names, trade styles, service marks, logos and other source or business identifiers, and all goodwill associated therewith, now existing or hereafter adopted or acquired, all registrations and recordings thereof, and all applications in connection therewith, whether in CIPO or in any similar office or agency of Canada, any province or territory thereof or any other country or any political subdivision thereof, or otherwise, and all common-law rights related thereto, including, without limitation, any of the foregoing referred to in Schedule 6, and (ii) the right to obtain all renewals thereof.
- "<u>Trademark License</u>": any agreement, whether written or oral, providing for the grant by or to any Grantor of any right to use any Trademark, including, without limitation, any of the foregoing referred to in <u>Schedule 6</u>.
- 1.2 Other Definitional Provisions. (a) The words "hereof," "herein", "hereto" and "hereunder" and words of similar import when used in this Agreement shall refer to this Agreement as a whole and not to any particular provision of this Agreement, and Section and Schedule references are to this Agreement unless otherwise specified.
 - (b) The meanings given to terms defined herein shall be equally applicable to both the singular and plural forms of such terms.
- (c) Where the context requires, terms relating to the Collateral or any part thereof, when used in relation to a Grantor, shall refer to such Grantor's Collateral or the relevant part thereof.

SECTION 2. GRANT OF SECURITY INTEREST

- 2.1 <u>Grant of Security</u>. Each Grantor hereby assigns and transfers to the Canadian Collateral Trustee, and hereby grants to the Canadian Collateral Trustee, for the ratable benefit of the Secured Parties, a security interest in, all of its present and after acquired personal property including without limitation, all of the following property now owned or at any time hereafter acquired by such Grantor or in which such Grantor now has or at any time in the future may acquire any right, title or interest (collectively, the "<u>Collateral</u>"), as collateral security for the prompt and complete payment and performance when due (whether at the stated maturity, by acceleration or otherwise) of the Obligations:
 - (a) all Accounts;

- (b) all Chattel Paper;
- (c) all Deposit Accounts;
- (d) all Documents of Title;
- (e) all Equipment;
- (f) all Intangibles;
- (g) all Instruments;
- (h) all Intellectual Property;
- (i) all Inventory;
- (j) all Investment Property;
- (k) all Motor Vehicles;
- (l) all Goods and other property not otherwise described above;
- (m) all books and records pertaining to the Collateral; and
- (n) to the extent not otherwise included, all Proceeds and products of any and all of the foregoing and all collateral security and guarantees given by any Person with respect to any of the foregoing;

<u>provided</u>, that the Collateral shall not include any Excluded Assets and, in the case of Intellectual property Section 2.1 shall operate to create a security interest therein but shall not constitute an assignment or transfer thereof.

- 2.2 Exception to Last Day. The security interest granted hereby shall not extend or apply to, and Collateral shall not include, the last day of the term of any lease or agreement therefor, but upon enforcement of the security interest granted hereby, each Grantor shall stand possessed of such last day in trust or assign the same to any person acquiring such term.
- 2.3 <u>Attachment</u>. Each Grantor acknowledges that (i) value has been given, (ii) it has rights in the Collateral, (iii) it has not agreed to postpone the time for attachment of the Lien granted hereunder, and (iv) it has received a copy of this Agreement.

SECTION 3. REPRESENTATIONS AND WARRANTIES

To induce each of the Holders to purchase the Canadian Notes, each Grantor hereby represents and warrants to the Canadian Collateral Trustee and each Secured Party that:

- 3.1 <u>Title</u>; No Other Liens. Except for the security interest granted to the Canadian Collateral Trustee for the ratable benefit of the Secured Parties pursuant to this Agreement and the other Liens permitted to exist on the Collateral by the Indenture, such Grantor owns its right, title and interest in and to each item of the Collateral free and clear of any and all Liens or claims of others. No financing statement or other public notice with respect to all or any part of the Collateral is on file or of record in any public office, except (a) such as have been filed in favor of the Canadian Collateral Trustee, for the ratable benefit of the Secured Parties, pursuant to this Agreement, or (b) or as are permitted by the Indenture. For the avoidance of doubt, it is understood and agreed that any Grantor may, as part of its business, grant licenses to third parties to use Intellectual Property owned or developed by a Grantor. For purposes of this Agreement and the other Loan Documents, such licensing activity shall not constitute a "Lien" on such Intellectual Property. Each of the Canadian Collateral Trustee and each Secured Party understands that any such licenses may be exclusive to the applicable licensees, and such exclusivity provisions may limit the ability of the Canadian Collateral Trustee to utilize, sell, lease or transfer the related Intellectual Property or otherwise realize value from such Intellectual Property pursuant hereto.
- 3.2 <u>Perfected First Priority Liens</u>. Subject to Section 7.18 and except as otherwise contemplated by the terms of the Indenture, the security interests granted by such Grantor pursuant to this Agreement (a) upon completion of the filings and other actions specified on <u>Schedule 3</u> (which, in the case of all filings and other documents referred to on said Schedule) except with respect to Collateral in respect of which a security interest cannot be perfected by Domestic Perfection Actions, have been delivered to the Canadian Collateral Trustee in completed and duly executed form, will constitute valid perfected security interests in all of the Collateral with respect to which perfection may be accomplished by Domestic Perfection Actions in favor of the Canadian Collateral Trustee, for the ratable benefit of the Secured Parties, as collateral security for such Grantor's Obligations, enforceable in accordance with the terms hereof and (b) are prior to all other Liens on the Collateral except for Permitted Priority Liens.
- 3.3 <u>Jurisdiction of Organization; Chief Executive Office</u>. On the date hereof, such Grantor's jurisdiction of organization, identification number from the jurisdiction of organization (if any), and the location of such Grantor's chief executive office or sole place of business or principal residence, as the case may be, are specified on <u>Schedule 4</u>. Such Grantor has furnished to the Canadian Collateral Trustee a certified charter, certificate of incorporation or other organization document and good standing certificate as of a date which is recent to the date hereof.
- 3.4 <u>Inventory and Equipment</u>. On the date hereof, the material portion of Inventory and the Equipment (other than mobile goods) are kept at the locations listed on <u>Schedule 5</u>.

3.5 [Intentionally Deleted].

- 3.6 <u>Investment Property</u>. (a) The shares of Pledged Stock pledged by such Grantor hereunder constitute all the issued and outstanding shares of all classes of the Capital Stock of each Issuer owned by such Grantor.
 - (b) To the knowledge of such Grantor, all the shares of the Pledged Stock have been duly and validly issued and are fully paid and nonassessable.
- (c) To the knowledge of such Grantor, each of the Pledged Notes constitutes the legal, valid and binding obligation of the obligor with respect thereto, enforceable in accordance with its terms, subject to the effects of bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium and other similar laws relating to or affecting creditors' rights generally, general equitable principles (whether considered in a proceeding in equity or at law) and an implied covenant of good faith and fair dealing.
- (d) Such Grantor is the record and beneficial owner of, and has good and marketable title to, the Investment Property pledged by it hereunder, free of any and all Liens or options in favor of, or claims of, any other Person, except the security interest created by this Agreement or as permitted by the Indenture.
- 3.7 <u>Receivables</u>. (a) <u>Schedule 9</u> is a list of the Receivables evidenced by any Instrument or Chattel Paper which has been delivered to the Canadian Collateral Trustee.
- (b) None of the obligors on any Receivable is a Governmental Authority, except for Receivables the face amount of which in the aggregate constitute not more than 5% of the face amount of all Receivables.
- 3.8 <u>Intellectual Property</u>. (a) <u>Schedule 6</u> lists all material Intellectual Property which is the subject of an application or registration owned by such Grantor in its own name on the date hereof.
- (b) On the date hereof, to its knowledge, all material Intellectual Property of such Grantor described on <u>Schedule 6</u> is valid, subsisting, unexpired and enforceable, has not been abandoned and to the Grantor's knowledge does not infringe upon the intellectual property rights of any other Person.
- (c) Except (i) for licensing or franchise agreements between any two or more of Parent and its Subsidiaries, (ii) the licensing of one provisional patent to various companies to enable them to provide void services, (iii) the licensing of Trademarks to independent sales agents, and (iv) as set forth in <u>Schedule 6</u>, on the date hereof, none of the Intellectual Property is the subject of any licensing or franchise agreement pursuant to which such Grantor is the licensor or franchisor.
- (d) No holding, decision or judgment has been rendered by any Governmental Authority which would limit, cancel or question the validity of, or such Grantor's rights in, any Intellectual Property in any respect that could reasonably be expected to have a Material Adverse Effect.

- (e) No action or proceeding is pending, or, to the knowledge of such Grantor, threatened, on the date hereof (i) seeking to limit, cancel or question the validity of any material Intellectual Property or such Grantor's ownership interest therein, or (ii) which, if adversely determined, would have a material adverse effect on the value of any material Intellectual Property.
 - 3.9 Vehicles. On the date hereof, the aggregate book value of all Vehicles owned by all Grantors is less than \$500,000.
- 3.10 <u>Deposit Accounts and Securities Accounts</u>. <u>Schedule 8</u> lists each Deposit Account and each Securities Account maintained by such Grantor on the date hereof.
 - 3.11 [Intentionally Deleted].

SECTION 4. COVENANTS

Each Grantor covenants and agrees with the Canadian Collateral Trustee and the Secured Parties that, from and after the date of this Agreement until the Obligations shall have been paid in full:

- 4.1 <u>Delivery of Instruments and Chattel Paper</u>. If any amount payable under or in connection with any of the Collateral shall be or become evidenced by any Instrument, Certificated Security or Chattel Paper shall, not later than the end of the fiscal quarter in which any such amount becomes so evidenced, be delivered to the Canadian Collateral Trustee (subject to the Collateral Trust Agreement), duly indorsed in a manner satisfactory to the Canadian Collateral Trustee, to be held as Collateral pursuant to this Agreement; <u>provided</u>, that the Grantors shall not be obligated to deliver to the Canadian Collateral Trustee any Instruments, Certificated Securities or Chattel Paper held by any Grantor at any time to the extent that the aggregate face amount of all such Instruments and Chattel Paper held by all Grantors at such time does not exceed \$1,000,000 or such Instruments and Chattel Paper reflect an agreement between any two or more of Parent and its Subsidiaries.
- 4.2 <u>Maintenance of Insurance</u>. (a) Such Grantor will maintain, with financially sound and reputable companies, insurance policies insuring the Inventory, Equipment and Vehicles against loss by fire, explosion, theft and such other casualties as shall be customary for similar businesses in similar locations.
- (b) All such insurance shall (i) provide that no cancellation, material reduction in amount or material change in coverage thereof shall be effective until the applicable insurer has complied with applicable law (ii) name the Canadian Collateral Trustee as additional insured party or loss payee, and (iii) provide that the applicable insurer shall endeavor to provide the Canadian Collateral Trustee with at least 30 day prior notice of cancellation thereof.

- (c) The Company shall deliver to the Canadian Collateral Trustee such additional information concerning such insurance as the Canadian Collateral Trustee may from time to time reasonably request or as otherwise required by the Indenture Documents, <u>provided</u>, <u>however</u>, that the Canadian Collateral Trustee shall have no obligation to monitor the insurance or request such further information.
- 4.3 <u>Payment of Obligations</u>. Such Grantor will pay and discharge or otherwise satisfy at or before maturity or before they become delinquent, as the case may be, all taxes, assessments and governmental charges or levies imposed upon the Collateral or in respect of income or profits therefrom, as well as all claims of any kind (including, without limitation, claims for labor, materials and supplies) against or with respect to the Collateral, except that no such tax, assessment, charge, levy or claim need be paid if the amount or validity thereof is currently being contested in good faith by appropriate proceedings, reserves in conformity with GAAP with respect thereto have been provided on the books of such Grantor and such proceedings could not reasonably be expected to result in the sale, forfeiture or loss of any material portion of the Collateral or such Grantor's interest in such material portion of the Collateral.
- 4.4 <u>Maintenance of Perfected Security Interest; Further Documentation</u>. (a) Such Grantor shall maintain the security interest created by this Agreement as a perfected security interest in the manner, and having at least the priority described in Section 3.2 and shall use its commercially reasonable efforts to defend such security interest against the claims and demands of all Persons whomsoever.
- (b) Such Grantor will furnish to the Canadian Collateral Trustee from time to time statements and schedules further identifying and describing the assets and property of such Grantor and such other reports in connection with the Collateral as the Canadian Collateral Trustee may reasonably request, all in reasonable detail, <u>provided</u>, <u>however</u>, that the Canadian Collateral Trustee shall have no obligation to monitor such assets and property or requires such statements and schedules.
- (c) At any time and from time to time, and at the sole expense of such Grantor, such Grantor will promptly and duly execute and deliver, and have recorded, such further instruments and documents and take such further actions as may be necessary or as the Canadian Collateral Trustee may reasonably request for the purpose of obtaining or preserving the full benefits of this Agreement and any Supplemental Security Agreements and of the rights and powers herein granted, including, without limitation, (i) the filing of any financing or continuation statements under the PPSA (or other similar foreign laws, as applicable) with respect to the security interests created hereby, and (ii) in the case of Investment Property, taking, to the extent required by the Indenture, any actions necessary to enable the Canadian Collateral Trustee to obtain "control" (within the meaning of the applicable PPSA) with respect thereto, subject to the terms of the Collateral Trust Agreement to the extent then in effect.
- 4.5 <u>Changes in Name, etc.</u> Such Grantor will not, except (i) during the period beginning on the date hereof and ending 15 days thereafter, upon prior notice and (ii) thereafter upon 15 days prior written notice, to the Canadian Collateral Trustee and, in each case, upon

filing, execution or delivery of all additional financing statements and other documents necessary to maintain the validity, perfection and priority of the security interests provided for herein:

- (i) change its jurisdiction of organization or the location of its chief executive office or sole place of business or principal residence from that referred to in Section 3.3; or
 - (ii) change its name.
 - 4.6 Notices. Such Grantor will advise the Canadian Collateral Trustee:
- (a) No less often than on a quarterly basis, in reasonable detail, of any Lien (other than security interests created hereby or Liens permitted under the Indenture) on any of the Collateral which would adversely affect the ability of the Canadian Collateral Trustee to exercise any of its remedies hereunder; and
- (b) Promptly, in reasonable detail, of the occurrence of any other event which could reasonably be expected to have a material adverse effect on the aggregate value of the Collateral or on the security interests created hereby.
- 4.7 Investment Property. (a) If such Grantor shall become entitled to receive or shall receive any certificate (including, without limitation, any certificate representing a dividend or a distribution in connection with any reclassification, increase or reduction of capital or any certificate issued in connection with any reorganization), option or rights in respect of the Capital Stock of any Issuer of Pledged Stock, whether in addition to, in substitution of, as a conversion of, or in exchange for, any shares of the Pledged Stock, or otherwise in respect thereof, such Grantor shall, to the extent applicable and within the time period required under Section 4.1, accept the same as the agent of the Canadian Collateral Trustee and the Secured Parties, hold the same in trust for the Canadian Collateral Trustee and the Secured Parties and deliver the same promptly to the Canadian Collateral Trustee in the exact form received, duly endorsed by such Grantor to the Canadian Collateral Trustee, if required, together with an undated stock power covering such certificate duly executed in blank by such Grantor and with, if the Canadian Collateral Trustee so requests, signature guaranteed, to be held by the Canadian Collateral Trustee, subject to the terms hereof, as additional collateral security for the Obligations. Any sums paid upon or in respect of the Investment Property pledged hereunder upon the liquidation or dissolution of any Issuer shall be paid over to the Canadian Collateral Trustee to be held by it hereunder as additional collateral security for the Obligations, and in case any distribution of capital shall be made on or in respect of such Investment Property, or any property shall be distributed upon or with respect to such Investment Property pursuant to the recapitalization or reclassification of the capital of any Issuer or pursuant to the reorganization thereof, the property so distributed shall, unless otherwise subject to a perfected security interest in favor of the Canadian Collateral Trustee, be delivered to the Canadian Collateral Trustee promptly to be held by it hereunder as additional collateral security for the Obligations. If any sums of money or property so paid or distributed in respect of such Investment Property shall be received by such Grantor, subject to the terms of the Collateral Trust Agreement, such Grantor shall, until such money or property is paid or delivered to the Canadian Collateral Trustee, hold such money or property in trust for the Secured Parties, segregated from other funds of such

Grantor, as additional collateral security for the Obligations. Notwithstanding any of the foregoing, the Grantors shall not be required to hold in trust for the Canadian Collateral Trustee, pay over to the Canadian Collateral Trustee or deliver to the Canadian Collateral Trustee as Collateral any proceeds of any liquidation or dissolution of any Issuer, or any distribution of capital or property in respect of any Investment Property, to the extent that (i) such liquidation, dissolution or distribution would be permitted by the Indenture and (ii) the proceeds thereof are applied toward prepayment of the Canadian Notes to the extent required by Indenture.

- (b) Without the prior written consent of the Canadian Collateral Trustee, such Grantor will not (i) except as otherwise permitted by the Indenture, vote to enable, or take any other action to permit, any Issuer to issue any stock or other equity securities of any nature or to issue any other securities convertible into or granting the right to purchase or exchange for any stock or other equity securities of any nature of any Issuer, unless such securities are delivered to the Canadian Collateral Trustee, concurrently with the issuance thereof, to be held by the Canadian Collateral Trustee as Collateral, (ii) sell, assign, transfer, exchange, or otherwise dispose of, or grant any option with respect to, the Investment Property pledged hereunder or Proceeds thereof (except pursuant to a transaction permitted by the Indenture), (iii) create, incur or permit to exist any Lien or option in favor of, or any claim of any Person with respect to, any of such Investment Property or Proceeds thereof, or any interest therein, except for the security interests created by this Agreement and Permitted Liens or (iv) except as otherwise permitted by the Indenture, enter into any agreement or undertaking restricting the right or ability of such Grantor or the Canadian Collateral Trustee to sell, assign or transfer any of the Pledged Securities or Proceeds thereof.
- (c) In the case of each Grantor which is an Issuer, such Issuer agrees that (i) it will be bound by the terms of this Agreement relating to the Pledged Securities issued by it and will comply with such terms insofar as such terms are applicable to it, (ii) it will notify the Canadian Collateral Trustee promptly in writing upon its receipt of any collateral in respect of the Investment Property described in Section 4.7(a) and (iii) the terms of Sections 5.3(c) and 5.7 shall apply to it, <u>mutatis mutandis</u>, with respect to all actions that may be required of it pursuant to Section 5.3(c) or 5.7 with respect to the Pledged Securities issued by it.
- (d) Each Issuer that is a partnership or a limited liability company (i) confirms that none of the terms of any equity interest issued by it provides that such equity interest is a "security" within the meaning of the PPSA, (ii) agrees that it will take no action to cause or permit any such equity interest to become a Security, (iii) agrees that it will not issue any certificate representing any such equity interest and (iv) agrees that if, notwithstanding the foregoing, any such equity interest shall be or become a Security, such Issuer will (and the Grantor that holds such equity interest hereby instructs such Issuer to) comply with instructions originated by the Canadian Collateral Trustee without further consent by such Grantor.
- 4.8 <u>Receivables</u>. (a) Other than in the ordinary course of business consistent with its past practice, or unless the applicable Grantor concludes that any of the following is in the best interest of such Grantor, such Grantor will not (i) grant any extension of

the time of payment of any Receivable, (ii) compromise or settle any Receivable for less than the full amount thereof, (iii) release, wholly or partially, any Person liable for the payment of any Receivable, (iv) allow any credit or discount whatsoever on any Receivable or (v) amend, supplement or modify any Receivable in any manner that could adversely affect the value thereof.

- (b) Such Grantor will deliver to the Canadian Collateral Trustee no less often than on a quarterly basis a copy of each material demand, notice or document received by it that questions or calls into doubt the validity or enforceability of more than 5% of the aggregate amount of the then outstanding Receivables.
- 4.9 Intellectual Property. (a) Such Grantor (either itself or through licensees) will (i) to such extent as such Grantor determines it is desirable to do so, continue to use each material Trademark on each and every material trademark class of goods applicable to its current line as reflected in its current catalogs, brochures and price lists in order to maintain such Trademark in full force free from any claim of abandonment for non-use, (ii) maintain the quality of products and services offered under such Trademark, (iii) use such Trademark with the appropriate notice of registration and all other notices and legends required by applicable law, and (iv) not (and not permit any licensee or sublicensee thereof to) do any act or knowingly omit to do any act whereby such Trademark may become invalidated or materially impaired in any way.
- (b) Such Grantor (either itself or through licensees) will not do any act, or omit to do any act, whereby any material Patent may become forfeited, abandoned or dedicated to the public.
- (c) Such Grantor (either itself or through licensees) will not (and will not permit any licensee or sublicensee thereof to) do any act or knowingly omit to do any act whereby any material portion of any material Copyrights may become invalidated or otherwise impaired. Such Grantor will not (either itself or through licensees) do any act whereby any material portion of any material Copyrights may fall into the public domain.
- (d) Such Grantor (either itself or through licensees) will not do any act that knowingly uses any material Intellectual Property to infringe upon the intellectual property rights of any other Person.
- (e) Such Grantor will promptly notify the Canadian Collateral Trustee in writing if it knows, or has reason to know, that any application or registration relating to any material Intellectual Property may become forfeited, abandoned or dedicated to the public, or of any adverse determination or development (including, without limitation, the institution of, or any such determination or development in, any proceeding in CIPO or any court in Canada regarding such Grantor's ownership of, or the validity of, any material Intellectual Property or such Grantor's right to register the same or to own and maintain the same.
- (f) Whenever such Grantor, either by itself or through any agent, employee, licensee or designee, shall file an application for the registration of any material Intellectual Property with CIPO, such Grantor shall report such filing to the Canadian Collateral Trustee no

less often than on a quarterly basis. Such Grantor shall execute and deliver, and have recorded, any and all agreements, instruments, documents, and papers as are required to evidence the Canadian Collateral Trustee's, and the Secured Parties' security interest in any material Copyright, Patent or Trademark and the goodwill and general intangibles of such Grantor relating thereto or represented thereby.

(g) Such Grantor will take all reasonable and necessary steps, including, without limitation, in any proceeding before CIPO, that such Grantor determines may be desirable to maintain and pursue each application relating to any material Intellectual Property (and to obtain the relevant registration) and to maintain each registration of the material Intellectual Property, including, without limitation, filing of applications for renewal, affidavits of use and affidavits of incontestability.

In the event that any material Intellectual Property is infringed, misappropriated or diluted by a third party, such Grantor shall (i) take such actions as such Grantor shall reasonably deem appropriate under the circumstances to protect such Intellectual Property and (ii) if such Intellectual Property is of material economic value, notify the Canadian Collateral Trustee after it learns thereof no less often than on a quarterly basis and, if such Grantor determines in its reasonable judgment that legal action is appropriate, take legal action to contest the infringement, misappropriation or dilution, to seek injunctive relief where appropriate and to recover any and all damages for such infringement, misappropriation or dilution.

4.10 [Intentionally Deleted].

SECTION 5. REMEDIAL PROVISIONS

- 5.1 <u>Certain Matters Relating to Receivables</u>. (a) The Canadian Collateral Trustee shall have the right, at any time after the occurrence and during the continuance of an Event of Default, to make test verifications of the Receivables in any manner and through any medium that it reasonably considers advisable, and each Grantor shall furnish all such assistance and information as the Canadian Collateral Trustee may reasonably require in connection with such test verifications. At any time and from time to time after the occurrence and during the continuance of an Event of Default, upon the Canadian Collateral Trustee's request and at the expense of the relevant Grantor, such Grantor shall cause independent public accountants or others satisfactory to the Canadian Collateral Trustee to furnish to the Canadian Collateral Trustee reports showing reconciliations, aging and test verifications of, and trial balances for, the Receivables.
- (b) The Canadian Collateral Trustee hereby authorizes each Grantor to collect such Grantor's Receivables, subject to the Canadian Collateral Trustee's direction and control after the occurrence and during the continuance of an Event of Default, and the Canadian Collateral Trustee may curtail or terminate said authority at any time after the occurrence and during the continuance of an Event of Default. If required by the Canadian Collateral Trustee at any time after the occurrence and during the continuance of an Event of Default, any payments of Receivables, when collected by any Grantor, (i) shall be forthwith (and, in any event, within

two Business Days) deposited by such Grantor in the exact form received, duly indorsed by such Grantor to the Canadian Collateral Trustee if required, in a Collateral Account maintained under the sole dominion and control of the Canadian Collateral Trustee, subject to withdrawal by the Canadian Collateral Trustee for the account of the Secured Parties only as provided in Section 5.5, and (ii) until so turned over, shall be held by such Grantor in trust for the Canadian Collateral Trustee and the Secured Parties, segregated from other funds of such Grantor. Each such deposit of Proceeds of Receivables shall be accompanied by a report identifying in reasonable detail the nature and source of the payments included in the deposit.

- (c) At any time after the occurrence and during the continuance of an Event of Default, at the Canadian Collateral Trustee's request, each Grantor shall deliver to the Canadian Collateral Trustee all original and other documents evidencing, and relating to, the agreements and transactions which gave rise to the Receivables, including, without limitation, all original orders, invoices and shipping receipts.
- (d) At any time after the occurrence and during the continuance of an Event of Default, at the Canadian Collateral Trustee's request, each Grantor will cooperate with the Canadian Collateral Trustee to establish a system of lockbox accounts, under the sole dominion and control of the Canadian Collateral Trustee, into which all Receivables shall be paid and from which all collected funds will be transferred to a Collateral Account.
- 5.2 <u>Communications with Obligors; Grantors Remain Liable</u>. (a) The Canadian Collateral Trustee in its own name or in the name of others may at any time after the occurrence and during the continuance of an Event of Default communicate with obligors under the Receivables to verify with them to the Canadian Collateral Trustee's satisfaction the existence, amount and terms of any Receivables.
- (b) Upon the request of the Canadian Collateral Trustee at any time after the occurrence and during the continuance of an Event of Default, each Grantor shall promptly notify obligors on the Receivables in writing that the Receivables have been assigned to the Canadian Collateral Trustee for the ratable benefit of the Secured Parties and that payments in respect thereof shall be made directly to the Canadian Collateral Trustee.
- (c) Anything herein to the contrary notwithstanding, each Grantor shall remain liable under each of the Receivables (or any agreement giving rise thereto) to observe and perform all the conditions and obligations to be observed and performed by it thereunder, all in accordance with the terms of any agreement giving rise thereto (provided that no Grantor shall be obligated in any manner to observe and perform any such conditions and obligations if the determination to not so observe and perform is made in good faith and in the exercise of Grantor's reasonable business judgment). Neither the Canadian Collateral Trustee nor any Secured Party shall have any obligation or liability under any Receivable (or any agreement giving rise thereto) by reason of or arising out of this Agreement or the receipt by the Canadian Collateral Trustee or any Secured Party of any payment relating thereto, nor shall the Canadian Collateral Trustee or any Secured Party be obligated in any manner to perform any of the obligations of any Grantor under or pursuant to any Receivable (or any agreement giving rise thereto), to make any payment, to make any inquiry as to the nature or the sufficiency of any payment received by it or as to the sufficiency of any performance by any party thereunder, to

present or file any claim, to take any action to enforce any performance or to collect the payment of any amounts which may have been assigned to it or to which it may be entitled at any time or times.

- 5.3 <u>Pledged Stock</u>. (a) Unless an Event of Default shall have occurred and be continuing and the Canadian Collateral Trustee shall have given notice to the relevant Grantor of the Canadian Collateral Trustee's intent to exercise its corresponding rights pursuant to Section 5.3(b), each Grantor shall be permitted to receive all cash dividends paid in respect of the Pledged Stock and all payments made in respect of the Pledged Notes, in each case paid in the normal course of business of the relevant Issuer, and to exercise all voting, corporate and other rights with respect to the Pledged Securities; <u>provided</u>, <u>however</u>, that no vote shall be cast or corporate right exercised or other action taken which could reasonably be expected to impair a material portion of Collateral (except in connection with a transaction permitted by the indenture).
- (b) If an Event of Default shall occur and be continuing and the Canadian Collateral Trustee shall give notice of its intent to exercise such rights to the relevant Grantor or Grantors, (i) the Canadian Collateral Trustee shall have the right to receive any and all cash dividends, payments or other Proceeds paid in respect of the Pledged Securities and make application thereof to the Obligations in the order set forth in Section 5.5, and (ii) any or all of the Pledged Securities shall be registered in the name of the Canadian Collateral Trustee or its nominee may thereafter exercise (x) all voting, corporate and other rights pertaining to such Pledged Securities at any meeting of shareholders of the relevant Issuer or Issuers or otherwise and (y) any and all rights of conversion, exchange and subscription and any other rights, privileges or options pertaining to such Pledged Securities as if it were the absolute owner thereof (including, without limitation, the right to exchange at its discretion any and all of the Pledged Securities upon the merger, consolidation, reorganization, recapitalization or other fundamental change in the corporate structure of any Issuer, or upon the exercise by any Grantor or the Canadian Collateral Trustee of any right, privilege or option pertaining to such Pledged Securities, and in connection therewith, the right to deposit and deliver any and all of the Pledged Securities with any committee, depositary, transfer agent, registrar or other designated agency upon such terms and conditions as the Canadian Collateral Trustee may determine), all without liability except to account for property actually received by it, but the Canadian Collateral Trustee shall have no duty to any Grantor to exercise any such right, privilege or option and shall not be responsible for any failure to do so or delay in so doing.
- (c) Each Grantor hereby authorizes and instructs each Issuer of any Pledged Securities pledged by such Grantor hereunder to (i) comply with any instruction received by it from the Canadian Collateral Trustee in writing that (x) states that an Event of Default has occurred and is continuing and (y) is otherwise in accordance with the terms of this Agreement, without any other or further instructions from such Grantor, and each Grantor agrees that each Issuer shall be fully protected in so complying, and (ii) to the extent otherwise required hereby, pay any dividends or other payments with respect to the Pledged Securities directly to the Canadian Collateral Trustee.

- 5.4 <u>Proceeds to be Turned Over To Canadian Collateral Trustee</u>. In addition to the rights of the Canadian Collateral Trustee and the Secured Parties specified in Section 5.1 with respect to payments of Receivables, if an Event of Default shall occur and be continuing all Proceeds received by any Grantor consisting of cash, checks and Instruments shall be held by such Grantor in trust for the Canadian Collateral Trustee and the Secured Parties, segregated from other funds of such Grantor, and shall, forthwith upon receipt by such Grantor, be turned over to the Canadian Collateral Trustee in the exact form received by such Grantor (duly indorsed by such Grantor to the Canadian Collateral Trustee, if required). All Proceeds received by the Canadian Collateral Trustee hereunder shall be held by the Canadian Collateral Trustee in a Collateral Account (or by such Grantor in trust for the Canadian Collateral Trustee and the Secured Parties) shall continue to be held as collateral security for all the Obligations and shall not constitute payment thereof until applied as provided in Section 5.5.
- 5.5 <u>Application of Proceeds</u>. If an Event of Default shall have occurred and be continuing, at any time at the Canadian Collateral Trustee's election, the Canadian Collateral Trustee may apply all or any part of Proceeds constituting Collateral, whether or not held in any Collateral Account in payment of the Obligations in accordance with the provisions of the Indenture or, if any Collateral Trust Agreement is then in effect, in accordance with the provisions of such Collateral Trust Agreement. Any balance of such proceeds remaining after the Obligations shall have been paid in full shall be paid over to the Grantors or to whomsoever may be lawfully entitled to receive the same.
- 5.6 PPSA and Other Remedies. If an Event of Default shall occur and be continuing, the Canadian Collateral Trustee, on behalf of the Secured Parties, may exercise, in addition to all other rights and remedies granted to it in this Agreement and in any other instrument or agreement securing, evidencing or relating to the Obligations, all rights and remedies of a secured party under the PPSA or any other applicable law or in equity. Without limiting the generality of the foregoing, the Canadian Collateral Trustee, without demand of performance or other demand, presentment, protest, advertisement or notice of any kind (except any notice required by law referred to below) to or upon any Grantor or any other Person (all and each of which demands, defenses, advertisements and notices are hereby waived), may in such circumstances forthwith collect, receive, appropriate and realize upon the Collateral, or any part thereof, and/or may forthwith sell, lease, assign, give option or options to purchase, or otherwise dispose of and deliver the Collateral or any part thereof (or contract to do any of the foregoing), in one or more parcels at public or private sale or sales, at any exchange, broker's board or office of the Canadian Collateral Trustee or any Secured Party or elsewhere upon such terms and conditions as it may deem advisable and at such prices as it may deem best, for cash or on credit or for future delivery without assumption of any credit risk. The Canadian Collateral Trustee or any Secured Party shall have the right upon any such public sale or sales, and, to the extent permitted by law, upon any such private sale or sales, to purchase the whole or any part of the Collateral so sold, free of any right or equity of redemption in any Grantor, which right or equity is hereby waived and released. Each Grantor further agrees, at the Canadian Collateral Trustee's request, to assemble the Collateral and make it available to the Canadian Collateral Trustee at places which the Canadian Collateral Trustee shall reasona

premises or elsewhere. The Canadian Collateral Trustee shall apply the net proceeds of any action taken by it pursuant to this Section 5.6 or any Receiver under Section 5.9 with respect to any Grantor's Collateral, after deducting all reasonable costs and expenses of every kind incurred in connection therewith or incidental to the care or safekeeping of any of the Collateral of such Grantor or in any way relating to the Collateral of such Grantor or the rights of the Canadian Collateral Trustee and the Secured Parties hereunder with respect thereto, including, without limitation, reasonable attorneys' fees and disbursements, to the payment in whole or in part of the Obligations of such Grantor, in the order specified in Section 5.5, and only after such application and after the payment by the Canadian Collateral Trustee of any other amount required by any provision of law, need the Canadian Collateral Trustee account for the surplus, if any, to any Grantor. To the extent permitted by applicable law, each Grantor waives all claims, damages and demands it may acquire against the Canadian Collateral Trustee or any Secured Party arising out of the exercise by them of any rights hereunder. If any notice of a proposed sale or other disposition of Collateral shall be required by law, such notice shall be deemed reasonable and proper if given at least 10 days before such sale or other disposition.

5.7 Registration Rights. (a) If the Canadian Collateral Trustee shall determine to exercise its right to sell any or all of the Pledged Stock pursuant to Section 5.6, and if in the opinion of the Canadian Collateral Trustee it is necessary or advisable to have the Pledged Stock, or that portion thereof to be sold, registered under the provisions of the Securities Act of 1933 (the "US Securities Act"), the relevant Grantor will cause the Issuer thereof to (i) execute and deliver, and cause the directors and officers of such Issuer to execute and deliver, all such instruments and documents, and do or cause to be done all such other acts as may be, in the opinion of the Canadian Collateral Trustee, necessary or advisable to register the Pledged Stock, or that portion thereof to be sold, under the provisions of the US Securities Act, (ii) use its best efforts to cause the registration statement relating thereto to become effective and to remain effective for a period of one year from the date of the first public offering of the Pledged Stock, or that portion thereof to be sold, and (iii) make all amendments thereto and/or to the related prospectus which, in the opinion of the Canadian Collateral Trustee, are necessary or advisable, all in conformity with the requirements of the Securities Act and the rules and regulations of the Securities and Exchange Commission applicable thereto. Each Grantor agrees to cause such Issuer to comply with the provisions of the securities or "Blue Sky" laws of any and all jurisdictions which the Canadian Collateral Trustee shall designate and to make available to its security holders, as soon as practicable, an earnings statement (which need not be audited) which will satisfy the provisions of Section 11(a) of the US Securities Act.

(b) Each Grantor recognizes that the Canadian Collateral Trustee may be unable to effect a public sale of any or all the Pledged Stock, by reason of certain prohibitions contained in the Securities Act and applicable state securities laws or otherwise, and may be compelled to resort to one or more private sales thereof to a restricted group of purchasers which will be obliged to agree, among other things, to acquire such securities for their own account for investment and not with a view to the distribution or resale thereof. Each Grantor acknowledges and agrees that any such private sale may result in prices and other terms less favorable than if such sale were a public sale and, notwithstanding such circumstances, agrees that any such private sale shall be deemed to have been made in a commercially reasonable manner. The Canadian Collateral Trustee shall be under no obligation to delay a sale of any of the Pledged

Stock for the period of time necessary to permit the Issuer thereof to register such securities for public sale under the Securities Act, or under applicable state securities laws, even if such Issuer would agree to do so.

- (c) Each Grantor agrees to use its best efforts to do or cause to be done all such other acts as may be necessary to make such sale or sales of all or any portion of the Pledged Stock pursuant to this Section 5.7 valid and binding and in compliance with any and all other applicable law. Each Grantor further agrees that a breach of any of the covenants contained in this Section 5.7 will cause irreparable injury to the Canadian Collateral Trustee and the Secured Parties, that the Canadian Collateral Trustee and the Secured Parties have no adequate remedy at law in respect of such breach and, as a consequence, that each and every covenant contained in this Section 5.7 shall be specifically enforceable against such Grantor, and such Grantor hereby waives and agrees not to assert any defenses against an action for specific performance of such covenants except for a defense that no Event of Default has occurred under the Indenture.
- 5.8 <u>Deficiency</u>. Each Grantor shall remain liable for any deficiency if the proceeds of any sale or other disposition of the Collateral are insufficient to pay its Obligations and the fees and disbursements of any attorneys employed by the Canadian Collateral Trustee or any Secured Party to collect such deficiency.
- 5.9 Appointment of Receiver. Upon the occurrence and during the continuance of any Event of Default, the Canadian Collateral Trustee may appoint or reappoint by instrument in writing, any Person or Persons, whether an officer or officers or an employee or employees of the Canadian Collateral Trustee or not, to be an interim receiver, receiver or receivers (hereinafter called a "Receiver", which term when used herein shall include a receiver and manager) of Collateral (including any interest, income or profits therefrom) and may remove any Receiver so appointed and appoint another in his/her/its stead. Any such Receiver shall, so far as concerns responsibility for his/her/its acts, be deemed the agent of the applicable Grantor and not the Canadian Collateral Trustee or any of Holders, and neither the Canadian Collateral Trustee nor any Holder shall be in any way responsible for any misconduct, negligence or non-feasance on the part of any such Receiver or his/her/its servants, agents or employees. Subject to the provisions of the instrument appointing him/her/it and the provisions of applicable law, any such Receiver shall have power to take possession of Collateral, to preserve Collateral or its value, to carry on or concur in carrying on all or any part of the business of the applicable Grantor and to sell, lease, license or otherwise dispose of or concur in selling, leasing, licensing or otherwise disposing of Collateral. To facilitate the foregoing powers, any such Receiver may, to the exclusion of all others, including the applicable Grantor, enter upon, use and occupy all premises owned or occupied by the applicable Grantor wherein Collateral may be situate, maintain Collateral upon such premises, borrow money on a secured or unsecured basis and use Collateral directly in carrying on the applicable Grantor's business or as security for loans or advances to enable the Receiver to carry on the applicable Grantor's business or otherwise, as such Receiver shall, in its discretion, determine. Except as may be otherwise directed by the Canadian Collateral Trustee, all money received from time to time by such Receiver in carrying out his/her/its appointment shall be received in trust for and be paid over to the Canadian Collateral Trustee, Every such Receiver may, in the discretion of the Canadian Collateral Trustee, be vested with all or any of the rights and powers of the Canadian Collateral Trustee.

The Canadian Collateral Trustee may, either directly or through its Canadian Collateral Trustees or nominees, exercise any or all of the powers and rights given to a Receiver by virtue of Section 5.9.

5.10 ULC Limitation. Notwithstanding any provisions to the contrary contained in this Agreement, the Indenture or any other document or agreement among all or some of the parties hereto, each Grantor is as of the date of this Agreement the sole registered and beneficial owner of all Pledged ULC Shares more particularly described in Schedule 2 to this Agreement and will remain so until such time as such Pledged ULC Shares are fully and effectively transferred into the name of the Secured Party or any other person on the books and records of such ULC. Nothing in this Agreement, the Indenture or any other document or agreement delivered among all or some of the parties hereto is intended to or shall constitute the Secured Party or any person other than a Grantor to be a member or shareholder of any ULC until such time as written notice is given to the applicable Grantor and all further steps are taken so as to register the Secured Party or other person as holder of the Pledged ULC Shares. The granting of the pledge and security interest pursuant to Section 2.1 does not make the Secured Party a successor to any Grantor as a member or shareholder of any ULC, and neither the Secured Party nor any of its respective successors or assigns hereunder shall be deemed to become a member or shareholder of any ULC by accepting this Agreement or exercising any right granted herein unless and until such time, if any, when the Secured Party or any successor or assign expressly becomes a registered member or shareholder of any ULC. Each Pledgor shall be entitled to receive and retain for its own account any dividends or other distributions if any, in respect of the Collateral, and shall have the right to vote such Pledged ULC Shares and to control the direction, management and policies of the ULC issuing such Pledged ULC Shares to the same extent as such Grantor would if such Pledged ULC Shares were not pledged to the Secured Party or to any other person pursuant hereto. To the extent any provision hereof would have the effect of constituting the Secured Party to be a member or shareholder of any ULC prior to such time, such provision shall be severed herefrom and ineffective with respect to the relevant Pledged ULC Shares without otherwise invalidating or rendering unenforceable this Agreement or invalidating or rendering unenforceable such provision insofar as it relates to Collateral other than Pledged ULC Shares. Notwithstanding anything herein to the contrary (except to the extent, if any, that the Secured Party or any of its successors or assigns hereafter expressly becomes a registered member or shareholder of any ULC), neither the Secured Party nor any of its respective successors or assigns shall be deemed to have assumed or otherwise become liable for any debts or obligations of any ULC. Except upon the exercise by the Secured Party or other persons of rights to sell or otherwise dispose of Pledged ULC Shares or other remedies following the occurrence and during the continuance of an Event of Default, each Pledgor shall not cause or permit, or enable any ULC in which it holds Pledged ULC Shares to cause or permit, the Secured Party to: (a) be registered as member or shareholder of such ULC; (b) have any notation entered in its favour in the share register of such ULC; (c) be held out as member or shareholder of such ULC; (d) receive, directly or indirectly, any dividends, property or other distributions from such ULC by reason of the Secured Party or other person holding a security interest in the Pledged ULC Shares; or (e) act as a member or shareholder of such ULC, or exercise any rights of a member or shareholder of such ULC, including the right to attend a meeting of such ULC or vote the shares of such ULC.

SECTION 6. THE CANADIAN COLLATERAL TRUSTEE

- 6.1 <u>Canadian Collateral Trustee's Appointment as Attorney-in-Fact, etc.</u> (a) Each Grantor hereby irrevocably constitutes and appoints the Canadian Collateral Trustee and any officer or agent thereof, with full power of substitution, as its true and lawful attorney-in-fact with full irrevocable power and authority in the place and stead of such Grantor and in the name of such Grantor or in its own name, for the purpose of carrying out the terms of this Agreement, to take any and all appropriate action and to execute any and all documents and instruments which may be necessary or desirable to accomplish the purposes of this Agreement, and, without limiting the generality of the foregoing, each Grantor hereby gives the Canadian Collateral Trustee the power and right, but not the obligation, on behalf of such Grantor, without notice to or assent by such Grantor, to do any or all of the following:
 - (i) in the name of such Grantor or its own name, or otherwise, take possession of and indorse and collect any checks, drafts, notes, acceptances or other instruments for the payment of moneys due under any Receivable or with respect to any other Collateral and file any claim or take any other action or proceeding in any court of law or equity or otherwise deemed appropriate by the Canadian Collateral Trustee for the purpose of collecting any and all such moneys due under any Receivable or with respect to any other Collateral when payable;
 - (ii) in the case of any Intellectual Property, execute and deliver, and have recorded, any and all agreements, instruments, documents and papers as the Canadian Collateral Trustee may request to evidence the Canadian Collateral Trustee's and the Secured Parties' security interest in such Intellectual Property and the goodwill and general intangibles of such Grantor relating thereto or represented thereby;
 - (iii) pay or discharge taxes and Liens levied or placed on or threatened against the Collateral, effect any repairs or any insurance called for by the terms of this Agreement and pay all or any part of the premiums therefor and the costs thereof;
 - (iv) execute, in connection with any sale provided for in Section 5.6 or 5.7, any endorsements, assignments or other instruments of conveyance or transfer with respect to the Collateral;
 - (v) (1) direct any party liable for any payment under any of the Collateral to make payment of any and all moneys due or to become due thereunder directly to the Canadian Collateral Trustee or as the Canadian Collateral Trustee shall direct; (2) ask or demand for, collect, and receive payment of and receipt for, any and all moneys, claims and other amounts due or to become due at any time in respect of or arising out of any Collateral; (3) sign and indorse any invoices, freight or express bills, bills of lading, storage or warehouse receipts, drafts against debtors, assignments, verifications, notices and other documents in connection with any of the Collateral; (4) commence and prosecute any suits, actions or proceedings at law or in equity in any court of competent jurisdiction to collect the Collateral or any portion thereof and to enforce any other right in respect of any Collateral; (5) defend any suit, action or proceeding brought against such Grantor with respect to any Collateral; (6) settle, compromise or adjust any such

suit, action or proceeding and, in connection therewith, give such discharges or releases as the Canadian Collateral Trustee may deem appropriate; (7) assign any Design, Copyright, Patent or Trademark (along with the goodwill of the business to which any such Design, Copyright, Patent or Trademark pertains), throughout the world for such term or terms, on such conditions, and in such manner, as the Canadian Collateral Trustee shall in its sole discretion determine; and (8) generally, sell, transfer, pledge and make any agreement with respect to or otherwise deal with any of the Collateral as fully and completely as though the Canadian Collateral Trustee were the absolute owner thereof for all purposes, and do, at the Canadian Collateral Trustee's option and such Grantor's expense, at any time, or from time to time, all acts and things which the Canadian Collateral Trustee deems necessary to protect, preserve or realize upon the Collateral and the Canadian Collateral Trustee's and the Secured Parties' security interests therein and to effect the intent of this Agreement, all as fully and effectively as such Grantor might do; and

(vi) license or sublicense whether on an exclusive or non-exclusive basis, any Intellectual Property for such term and on such conditions and in such manner as the Canadian Collateral Trustee shall in its sole judgment determine and, in connection therewith, such Grantor hereby grants to the Canadian Collateral Trustee for the benefit of the Secured Parties a royalty-free, world-wide irrevocable license of its Intellectual Property.

Anything in this Section 6.1(a) to the contrary notwithstanding, the Canadian Collateral Trustee agrees that it will not exercise any rights under the power of attorney provided for in this Section 6.1(a) unless an Event of Default shall have occurred and be continuing.

- (b) If any Grantor fails to perform or comply with any of its agreements contained herein, the Canadian Collateral Trustee, at its option, but without any obligation so to do, may perform or comply, or otherwise cause performance or compliance, with such agreement.
- (c) The expenses of the Canadian Collateral Trustee incurred in connection with actions undertaken as provided in this Section 6.1, together with interest thereon at a rate per annum equal to the rate per annum at which interest would then be payable under the Indenture, from the date of payment by the Canadian Collateral Trustee to the date reimbursed by the relevant Grantor, shall be payable by such Grantor to the Canadian Collateral Trustee promptly on demand.
- (d) Each Grantor hereby ratifies all that said attorneys shall lawfully do or cause to be done by virtue hereof. All powers, authorizations and agencies contained in this Agreement are coupled with an interest and are irrevocable until this Agreement is terminated and the security interests created hereby are released.
- 6.2 <u>Duty of Canadian Collateral Trustee</u>. The Canadian Collateral Trustee's sole duty with respect to the custody, safekeeping and physical preservation of the Collateral in its possession, under the PPSA or otherwise, shall be to deal with it in the same manner as the Canadian Collateral Trustee deals with similar property for its own account. Neither the

Canadian Collateral Trustee, any Secured Party nor any of their respective officers, directors, employees or agents shall be liable for failure to demand, collect or realize upon any of the Collateral or for any delay in doing so or shall be under any obligation to sell or otherwise dispose of any Collateral upon the request of any Grantor or any other Person or to take any other action whatsoever with regard to the Collateral or any part thereof. The powers conferred on the Canadian Collateral Trustee and the Secured Parties hereunder are solely to protect the Canadian Collateral Trustee's and the Secured Parties' interests in the Collateral and shall not impose any duty upon the Canadian Collateral Trustee or any Secured Party to exercise any such powers. The Canadian Collateral Trustee and the Secured Parties shall be accountable only for amounts that they actually receive as a result of the exercise of such powers, and neither they nor any of their officers, directors, employees or agents shall be responsible to any Grantor for any act or failure to act hereunder, except for their own gross negligence or willful misconduct. The Canadian Collateral Trustee will not be required to advance or expend any funds or otherwise incur any financial liability in the performance of its duties or the exercise of its powers or rights hereunder (including, without limitation, taking any actions under or in respect of control agreements) unless it has been provided with pre-funding, security or indemnity reasonably satisfactory to it (and, for the avoidance of doubt, from parties reasonably satisfactory to it) against any and all cost, loss, liability or expense which may be incurred by it by reason of taking or continuing to take such action.

6.3 Execution of Financing Statements. Pursuant to any applicable law, each Grantor authorizes the Canadian Collateral Trustee to file or record financing statements, financing change statements and other filing or recording documents or instruments with respect to the Collateral without the signature of such Grantor in such form and in such offices as the Canadian Collateral Trustee determines appropriate to perfect the security interests of the Canadian Collateral Trustee under this Agreement. Each Grantor authorizes the Canadian Collateral Trustee to use the collateral description "all personal property of the debtor, whether now owned or hereafter acquired" or "all assets of debtor, whether now owned or hereafter acquired" in any such financing statements. Each Grantor hereby ratifies and authorizes the filing by the Canadian Collateral Trustee of any financing statement with respect to the Collateral made prior to the date hereof.

6.4 Authority of Canadian Collateral Trustee. Each Grantor acknowledges that the rights and responsibilities of the Canadian Collateral Trustee under this Agreement with respect to any action taken by the Canadian Collateral Trustee or the exercise or non-exercise by the Canadian Collateral Trustee of any option, voting right, request, judgment or other right or remedy provided for herein or resulting or arising out of this Agreement shall, as between the Canadian Collateral Trustee and the other Secured Parties, be governed by the Indenture and, to the extent then in effect, by the Collateral Trust Agreement and by such other agreements with respect thereto as may exist from time to time among them, but, as between the Canadian Collateral Trustee and the Grantors, the Canadian Collateral Trustee shall be conclusively presumed to be acting as agent for the Secured Parties with full and valid authority so to act or refrain from acting, and no Grantor shall be under any obligation, or entitlement, to make any inquiry respecting such authority. The Canadian Collateral Trustee is entitled to all the same protections hereunder that the Trustee has under the Indenture.

6.5 <u>Applicable Secured Debt Representative</u>. Notwithstanding anything to the contrary herein and for the avoidance of doubt, the rights, duties and obligations of the Canadian Collateral Trustee hereunder shall be subject to the terms of the Collateral Trust Agreement, to the extent in effect, and any rights, duties and obligations of the Applicable Secured Debt Representative thereunder.

SECTION 7. MISCELLANEOUS

- 7.1 <u>Amendments in Writing</u>. None of the terms or provisions of this Agreement may be waived, amended, supplemented or otherwise modified except in accordance with Section 9.01 or Section 9.02, as applicable, of the Indenture.
- 7.2 <u>Notices</u>. All notices, requests and demands to or upon the Canadian Collateral Trustee or any Grantor hereunder shall be effected in the manner provided for in Section 13.01 of the Indenture.
- 7.3 No Waiver by Course of Conduct; Cumulative Remedies. Neither the Canadian Collateral Trustee nor any Secured Party shall by any act (except by a written instrument pursuant to Section 7.1), delay, indulgence, omission or otherwise be deemed to have waived any right or remedy hereunder or to have acquiesced in any Default or Event of Default. No failure to exercise, nor any delay in exercising, on the part of the Canadian Collateral Trustee or any Secured Party, any right, power or privilege hereunder shall operate as a waiver thereof. No single or partial exercise of any right, power or privilege hereunder shall preclude any other or further exercise thereof or the exercise of any other right, power or privilege. A waiver by the Canadian Collateral Trustee or any Secured Party of any right or remedy hereunder on any one occasion shall not be construed as a bar to any right or remedy which the Canadian Collateral Trustee or such Secured Party would otherwise have on any future occasion. The rights and remedies herein provided are cumulative, may be exercised singly or concurrently and are not exclusive of any other rights or remedies provided by law.
- 7.4 Enforcement Expenses; Indemnification. (a) Each Grantor agrees to pay, or reimburse each Secured Party and the Canadian Collateral Trustee and each of their respective Affiliates and each of the directors, officers, partners, trustees, employees, attorneys and agents, (and in each case) their respective heirs, representatives, successors and assigns for, all its costs and expenses incurred in enforcing the Indemnified Liabilities (as defined in the Collateral Trust Agreement) (other than any such enforcement determined by a final, non-appealable judgment of a court to have been in bad faith) or preserving any rights under this Agreement, the Indenture and the other Collateral Documents to which such Grantor is a party, including, without limitation, the fees and disbursements of counsel (including the allocated fees and expenses of in-house counsel) to each Secured Party and of counsel to the Canadian Collateral Trustee.
- (b) Each Grantor agrees to pay, and to save the Canadian Collateral Trustee and the Secured Parties harmless from, any and all liabilities with respect to, or resulting from any delay in paying, any and all stamp, excise, sales or other taxes which may be payable or determined to be payable with respect to any of the Collateral or in connection with any of the transactions contemplated by this Agreement.

- (c) The agreements in this Section shall survive repayment of the Obligations and all other amounts payable under the Indenture, the Canadian Notes and the other Collateral Documents.
- 7.5 <u>Successors and Assigns</u>. This Agreement shall be binding upon the successors and assigns of each Grantor and shall inure to the benefit of the Canadian Collateral Trustee and the Secured Parties and their successors and assigns; <u>provided</u> that no Grantor may assign, transfer or delegate any of its rights or obligations under this Agreement without the prior written consent of the Canadian Collateral Trustee.
- 7.6 Set-Off. Each Grantor hereby irrevocably authorizes the Canadian Collateral Trustee and each Secured Party at any time and from time to time while an Event of Default shall have occurred and be continuing, without notice to such Grantor or any other Grantor, any such notice being expressly waived by each Grantor, to set-off and appropriate and apply any and all deposits (general or special, time or demand, provisional or final), in any currency, and any other credits, indebtedness or claims, in any currency, in each case whether direct or indirect, absolute or contingent, matured or unmatured, at any time held or owing by the Canadian Collateral Trustee or such Secured Party to or for the credit or the account of such Grantor, or any part thereof in such amounts as the Canadian Collateral Trustee or such Secured Party may elect, against and on account of the obligations and liabilities of such Grantor to the Canadian Collateral Trustee or such Secured Party hereunder, in any currency, whether arising hereunder, under the Indenture, the Canadian Notes or any other Collateral Document, as the Canadian Collateral Trustee or such Secured Party may elect, whether or not the Canadian Collateral Trustee or any Secured Party has made any demand for payment and although such obligations, liabilities and claims may be contingent or unmatured. The Canadian Collateral Trustee and each Secured Party shall notify such Grantor promptly of any such set-off and the application made by the Canadian Collateral Trustee or such Secured Party of the proceeds thereof, provided that the failure to give such notice shall not affect the validity of such set-off and application. The rights of the Canadian Collateral Trustee and each Secured Party may have.
- 7.7 <u>Counterparts</u>. This Agreement may be executed by one or more of the parties to this Agreement on any number of separate counterparts (including by telecopy), and all of said counterparts taken together shall be deemed to constitute one and the same instrument.
- 7.8 <u>Severability</u>. Any provision of this Agreement which is prohibited or unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective to the extent of such prohibition or unenforceability without invalidating the remaining provisions hereof, and any such prohibition or unenforceability in any jurisdiction shall not invalidate or render unenforceable such provision in any other jurisdiction.
- 7.9 <u>Section Headings</u>. The Section headings used in this Agreement are for convenience of reference only and are not to affect the construction hereof or be taken into consideration in the interpretation hereof.

7.10 <u>Integration</u>. This Agreement, the Indenture, the Canadian Notes and the other Collateral Documents represent the agreement of the Grantors, the Canadian Collateral Trustee and the Secured Parties with respect to the subject matter hereof and thereof, and there are no promises, undertakings, representations or warranties by the Canadian Collateral Trustee or any Secured Party relative to subject matter hereof and thereof not expressly set forth or referred to herein or in the Indenture, the Canadian Notes and the other Collateral Documents.

7.11 GOVERNING LAW. THIS AGREEMENT SHALL BE GOVERNED BY, AND CONSTRUED AND INTERPRETED IN ACCORDANCE WITH, THE LAW OF THE PROVINCE OF ONTARIO WITHOUT REGARD TO CONFLICTS OF LAW PRINCIPLES THEREOF.

- 7.12 Submission To Jurisdiction; Waivers. Each Grantor hereby irrevocably and unconditionally:
- (a) submits for itself and its property in any legal action or proceeding relating to this Agreement and the other Collateral Documents to which it is a party, or for recognition and enforcement of any judgment in respect thereof, to the non-exclusive general jurisdiction of the courts of the State of New York, the courts of the United States of America for the Southern District of New York and the courts of the Province of Ontario, and appellate courts from any of the foregoing;
- (b) consents that any such action or proceeding may be brought in such courts and waives any objection that it may now or hereafter have to the venue of any such action or proceeding in any such court or that such action or proceeding was brought in an inconvenient court and agrees not to plead or claim the same;
- (c) agrees that service of process in any such action or proceeding may be effected by mailing a copy thereof by registered or certified mail (or any substantially similar form of mail), postage prepaid, to such Grantor at its address referred to in Section 7.2 or at such other address of which the Canadian Collateral Trustee shall have been notified pursuant thereto;
- (d) agrees that nothing herein shall affect the right to effect service of process in any other manner permitted by law or shall limit the right to sue in any other jurisdiction; and
- (e) waives, to the maximum extent not prohibited by law, any right it may have to claim or recover in any legal action or proceeding referred to in this Section any special, exemplary, punitive or consequential damages.
 - 7.13 Acknowledgements. Each Grantor hereby acknowledges that:
- (a) it has been advised by counsel in the negotiation, execution and delivery of this Agreement and the other Collateral Documents to which it is a party;
- (b) neither the Canadian Collateral Trustee nor any Secured Party has any fiduciary relationship with or duty to any Grantor arising out of or in connection with this Agreement or any of the other Collateral Documents, and the relationship between the Grantors, on the one hand, and the Canadian Collateral Trustee and Secured Parties, on the other hand, in connection herewith or therewith is solely that of debtor and creditor; and

(c) no joint venture is created hereby or by the other Collateral Documents or otherwise exists by virtue of the transactions contemplated hereby among the Secured Parties or among the Grantors and the Secured Parties.

7.14 <u>Additional Grantors</u>. Each Affiliate of the Company that is required to become a party to this Agreement pursuant to the Indenture shall become a Grantor for all purposes of this Agreement upon execution and delivery by such Affiliate of the Company of an Assumption Agreement in the form of Annex I hereto.

7.15 Releases. (a) The Collateral shall be released from the Liens created hereby at the times, and to the extent, provided under Section 12.04 of the Indenture and (in the case of a complete release of the Collateral) this and all obligations (other than those expressly stated to survive such termination) of the Canadian Collateral Trustee and each Grantor hereunder shall terminate, all without delivery of any instrument or performance of any act by any party, and all rights to the Collateral shall revert to the Grantors. At the request and sole expense of any Grantor following any such termination, the Canadian Collateral Trustee shall deliver to such Grantor, on a date to be reasonably specified by such Grantor, any Collateral held by the Canadian Collateral Trustee hereunder, and execute and deliver to such Grantor such documents as such Grantor shall reasonably request to evidence such termination.

7.16 Approvals. Any provision contained herein to the contrary notwithstanding, no action shall be taken hereunder by the Canadian Collateral Trustee with respect to the Collateral unless and until all applicable requirements, if any, under applicable federal, provincial and territorial laws and the respective rules and regulations thereunder and thereof, as well as any other laws, rules and regulations of any Governmental Authority applicable to or having jurisdiction over any Grantor, have in the reasonable judgment of the Canadian Collateral Trustee been fully satisfied to the extent necessary to take such action and there have been obtained such consents, approvals and authorizations as may be required to be obtained from applicable federal, state and local regulatory authorities and municipalities and any other Governmental Authority under the terms of any franchise, license or similar operating right held by such Grantor in order to take such action. It is the intention of the parties hereto that the grant to the Canadian Collateral Trustee of security interests in the Collateral, and all rights and remedies by the Canadian Collateral Trustee with respect to the Collateral, shall in all relevant aspects be subject to and governed by said statutes, rules and regulations and that nothing in this Agreement shall be construed to diminish the control exercised by any Grantor, except in accordance with the provisions of any statutory requirements and rules and regulations. By its acceptance of this Agreement, the Canadian Collateral Trustee agrees that it will not take any action pursuant to this Agreement which constitutes or results in any assignment of a license or franchise or any change of control over the communications properties owned and operated by any Grantor, if such assignment of license or franchise or change of control would, under then existing law or under any franchise, require the prior approval of Governmental Authority, without first obtaining such approval, recording, qualification or auth

will execute and deliver, or will cause the execution and delivery of, all applications, certificates, instruments and other documents and papers that the Canadian Collateral Trustee may reasonably require in order for such governmental consent, approval, recording, qualification or authorization to be obtained. Each Grantor agrees to use its best efforts to cause such governmental consents, approvals, recordings, qualifications and authorizations to be forthcoming.

7.17 WAIVER OF JURY TRIAL. EACH GRANTOR AND, BY ACCEPTANCE OF THE BENEFITS HEREOF, THE CANADIAN COLLATERAL TRUSTEE AND EACH SECURED PARTY, HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES TRIAL BY JURY IN ANY LEGAL ACTION OR PROCEEDING RELATING TO THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT AND FOR ANY COUNTERCLAIM THEREIN.

7.18 <u>Collateral Trust Agreement</u>. Notwithstanding anything herein to the contrary, the lien and security interest granted to the Canadian Collateral Trustee pursuant to this Agreement and the exercise of any right or remedy by such Canadian Collateral Trustee hereunder are subject to the provisions of any Collateral Trust Agreement then in effect. In the event of any conflict between the terms of any Collateral Trust Agreement then in effect and this Agreement, the terms of such Collateral Trust Agreement will govern.

7.19 <u>Force Majeure</u>. In no event shall the Canadian Collateral Trustee be responsible or liable for any failure or delay in the performance of its obligations hereunder arising out of or caused by, directly or indirectly, forces beyond its control, including, without limitation, strikes, work stoppages, accidents, acts of war or terrorism, civil or military disturbances, nuclear or natural catastrophes or acts of God, and interruptions, loss or malfunctions of utilities, communications or computer (software and hardware) services; it being understood that the Canadian Collateral Trustee shall use reasonable efforts which are consistent with accepted practices in the banking industry to resume performance as soon as practicable under the circumstances.

7.20 <u>U.S.A.</u> Patriot Act. The parties hereto acknowledge that in accordance with Section 326 of the U.S.A. Patriot Act and other similar laws of Canada, the Canadian Collateral Trustee, like all financial institutions and in order to help fight the funding of terrorism and money laundering, is required to obtain, verify, and record information that identifies each person or legal entity that establishes a relationship or opens an account with the Canadian Collateral Trustee. The parties to this Agreement agree that they will provide the Canadian Collateral Trustee with such information as it may request in order for the Canadian Collateral Trustee to satisfy the requirements of the U.S.A. Patriot Act and other similar laws of Canada.

7.21 <u>Permitted Liens</u>. The inclusion or reference to Permitted Liens in this Agreement or in any other Canadian Secured Debt Document is not intended to subordinate and shall not subordinate, and shall not be interpreted as subordinating, the Lien and security interest created by this Agreement or any other Canadian Secured Debt Document to any Permitted Liens.

- 7.22 <u>Amalgamation</u>. Each Grantor acknowledges and agrees that, in the event it amalgamates with any other corporation or corporations, it is the intention of the parties hereto that the term "Grantor", when used herein, shall apply to each of the amalgamating corporations and to the amalgamated corporation, such that the security interest granted hereby:
- (a) shall extend to "Collateral" (as that term is herein defined) owned by each of the amalgamating corporations and the amalgamated corporation at the time of amalgamation and to any "Collateral" thereafter owned or acquired by the amalgamated corporation, and
- (b) shall secure all "Obligations" (as that term is herein defined) of each of the amalgamating corporations and the amalgamated corporation to Agent and Secured Parties at the time of amalgamation and all "Obligations" of the amalgamated corporation to Agent and Secured Parties thereafter arising. The security interest created hereby shall attach to all "Collateral" owned by each corporations amalgamating with any Grantor, and by the amalgamated company, at the time of the amalgamation, and shall attach to all "Collateral" thereafter owned or acquired by the amalgamated corporation when such becomes owned or is acquired.

[Signature page follows.]

COMPANY:

PRIMUS TELECOMMUNICATIONS CANADA INC.

By:
Name:
Title:

GUARANTORS:

3082833 NOVA SCOTIA COMPANY

By:
Name:
Title:
Title:

TELESONIC COMMUNICATIONS INC.

By: Name: Title:

IN WITNESS WHEREOF, each of the undersigned has caused this Canadian Collateral Agreement to be duly executed and delivered as of the date

ASSUMPTION AGREEMENT

This ASSUMPTION AGREEMENT, dated as of [mm/dd/yy], is delivered by [NAME OF GRANTOR,] a [JURISDICTION] [corporation/limited liability company/limited partnership] (the "Additional Grantor") pursuant to the Canadian Collateral Agreement, dated as of December 22, 2009 (as it may from time to time be amended, restated, modified or supplemented, the "Canadian Collateral Agreement"; all capitalized terms not defined herein shall have the meaning ascribed to them in the Collateral Agreement), made by Primus Telecommunications Canada Inc. and the other Grantors named therein in favor of Computershare Trust Company of Canada, as Canadian Collateral Trustee.

By executing and delivering this Assumption Agreement, the Additional Grantor, as provided in Section 7.14 of the Canadian Collateral Agreement, hereby becomes a party to the Canadian Collateral Agreement as a Grantor thereunder with the same force and effect as if originally named therein as a Grantor and, without limiting the generality of the foregoing, hereby expressly assumes all obligations and liabilities of a Grantor thereunder. The Additional Grantor confirms the grant to the Canadian Collateral Trustee set forth in the Canadian Collateral Agreement of, and does hereby assign and transfer to the Canadian Collateral Trustee, and hereby grants to the Canadian Collateral Trustee, for the ratable benefit of the Secured Parties, a security interest in, all of the Collateral now owned or at any time hereafter acquired by the Additional Grantor or in which the Additional Grantor now or at any time in the future may acquire any right, title or interest, as collateral security for the prompt and complete payment and performance when due (whether at the stated maturity, by acceleration or otherwise) of such Additional Grantor's Obligations.

Additional Grantor represents and warrants that the attached Supplements to Schedules, Annex I-A to the Assumption Agreement, accurately and completely set forth all additional information required pursuant to the Canadian Collateral Agreement and hereby agrees that such Supplements to Schedules shall constitute part of the Schedules to the Canadian Collateral Agreement. The Additional Grantor hereby represents and warrants, to the extent applicable, that each of the representations and warranties of the Additional Grantor contained in Section 3 of the Canadian Collateral Agreement is true and correct on and as of the date hereof (after giving effect to this Assumption Agreement) as if made on and as of such date (except to the extent such representation and warranty expressly relates to an earlier date).

GOVERNING LAW

THIS AGREEMENT SHALL BE GOVERNED BY, AND CONSTRUED AND INTERPRETED IN ACCORDANCE WITH, THE LAW OF THE PROVINCE OF ONTARIO WITHOUT REGARD TO CONFLICTS OF LAW PRINCIPLES THEREOF.

nm/dd/yy].		
	[NAME OF ADDIT	TONAL GRANTOR]
	Ву:	
	Name: Title:	

Supplement to Schedule 2

DESCRIPTION OF PLEDGED SECURITIES

Supplement to Schedule 3

FILINGS AND OTHER ACTIONS REQUIRED TO PERFECT SECURITY INTEREST

Supplement to Schedule 4

JURISDICTION OF ORGANIZATION, IDENTIFICATION NUMBER AND LOCATION OF CHIEF EXECUTIVE OFFICE

Supplement to Schedule 5

LOCATION OF INVENTORY AND EQUIPMENT

Supplement to Schedule 6

INTELLECTUAL PROPERTY

Supplement to Schedule 8

DEPOSIT ACCOUNTS AND SECURITIES ACCOUNTS

Supplement to Schedule 9

RECEIVABLES

Schedule 2 Description of Pledged Securities

Current Legal Entities Owned by Entity/or Issuer Primus Telecommunications Canada Inc.	Record Owners of Entity 3082833 Nova Scotia Company	<u>Certificate No.</u> C-1	No. Shares/Interest 825,104 Common Shares	Percent of Interests Owned that are Pledged 100%
Telesonic Communications Inc.	Primus Telecommunications Canada Inc.	C-1	1,414,289 Common Shares	100%
Primus Telecommunications Canada Inc.	3082833 Nova Scotia Company		Term Debenture in amount of Cdn\$237,227,000	100%

Schedule 3 Filings and Other Actions Required to Perfect Security Interests

1. Registrations against Primus Telecommunications Canada Inc. under *Personal Property Security Acts* in the following jurisdictions:

Ontario

British Columbia

Alberta

Saskatchewan

Manitoba

Nova Scotia

New Brunswick

2. Registrations against 3082833 Nova Scotia Company under *Personal Property Security Acts* in the following jurisdictions:

Ontario

British Columbia

Nova Scotia

- 3. Registration against Telesonic Communications Inc. under the *Personal Property Security Act* (Ontario).
- 4. Registration against Primus Telecommunications Canada Inc. with the Quebec Registry of Personal and Movable Real Rights.
- 5. Registration against Primus Telecommunications Canada Inc. under the Uniform Commercial Code (DC).

Schedule 4 Jurisdictions of Organization, Indentification Number and Location of Chief Executive Office

Company	Jurisdiction of Organization	Identification Number	Chief Executive Office	
Primus Telecommunications Canada Inc.	Ontario	1683879	5343 Dundas Street West,	
			Suite 400	
			Toronto	
Telesonic Communications Inc.	Ontario	1598316	5343 Dundas Street West,	
			Suite 400	
			Toronto	
3082833 Nova Scotia Company	Nova Scotia	3082833	900-1959 Upper Water Street	
1 3			Halifax, NS	

Schedule 5

Location of Inventory and Equipment Company/Relevant Subsidiary 3082833 Nova Scotia Company 900-1959 Upper Water Street Halifax, NS Primus Telecommunications Canada Inc. 5343 Dundas Street West, Suite 400 Toronto Primus Telecommunications Canada Inc. 333 Seymour Street Suite 530 and 540 Vancouver, B.C. V6B 5A6 Primus Telecommunications Canada Inc. 970 Green Valley Road London, Ontario N6N 1E3 Primus Telecommunications Canada Inc. 710 Dorval Drive Suite 700 Oakville, Ontario L6K 3V7 Primus Telecommunications Canada Inc. 31 Auriga Drive Ottawa, Ontario K2E 1C4 Primus Telecommunications Canada Inc. 135 St. Francois Street Edmunston, New Brunswick E3V 1E8 Primus Telecommunications Canada Inc. 555 West Hastings Street Suite 2050 and 2060 Vancouver, B.C. V6B 4N4 Primus Telecommunications Canada Inc. 151 Front Street West Suites 604A, 607, 610, 8th floor and roof Toronto, Ontario

M5J 2N1

Primus Telecommunications Canada Inc. 210 Dundas Street West

8th Floor Toronto, Ontario M5G 2E8

Company/Relevant Subsidiary Primus Telecommunications Canada Inc.	Address WaterPark Place 10 Bay St., Suite 200, Toronto, Ontario M5J 2N8
Primus Telecommunications Canada Inc.	2934 Baseline Road Ground Floor Ottawa, Ontario K2H 1B2
Primus Telecommunications Canada Inc.	195 Colonnade Road South Nepean, Ontario K2E 7K3
Primus Telecommunications Canada Inc.	230 – 605 1st Street SW Suite 230 Calgary, Alberta T2P 3S9
Primus Telecommunications Canada Inc.	8500 Leslie St. Thornhill, Ontario L3T 7M8
Primus Telecommunications Canada Inc.	25 Trailwood Drive Mississauga, Ontario L4Z 3L6
	(roof and floor space)
Primus Telecommunications Canada Inc.	10335 172 Street Suite 307 Edmonton, Alberta
Primus Telecommunications Canada Inc.	140 Renfrew Drive, Suite 200 Markham, Ontario
Primus Telecommunications Canada Inc.	3016-19 th Street N.E. #204 Calgary, Alberta
Primus Telecommunications Canada Inc.	Edmonton City Centre #300 Edmonton, Alberta
Primus Telecommunications Canada Inc.	160 Market St Hamilton, Ontario

Company/Relevant Subsidiary	Address
Primus Telecommunications Canada Inc.	200 Bay Street South Hamilton, Ontario
Primus Telecommunications Canada Inc.	194 Gage Ave. So. Hamilton, Ontario
Primus Telecommunications Canada Inc.	150 Sanford Ave North Hamilton, Ontario
Primus Telecommunications Canada Inc.	4800 Dufferin Street Toronto, Ontario
Primus Telecommunications Canada Inc.	6205 Airport Road Toronto, Ontario
Primus Telecommunications Canada Inc.	33 Carlson Court Toronto, Ontario
Primus Telecommunications Canada Inc.	33 Fennell Ave E. Hamilton, Ontario
Primus Telecommunications Canada Inc.	644 Main Street Hamilton, Ontario
Primus Telecommunications Canada Inc.	105 Main St East Hamilton, Ontario
Primus Telecommunications Canada Inc.	1280 Main St W Hamilton, Ontario
Primus Telecommunications Canada Inc.	300 Mohawk Road Hamilton, Ontario
Primus Telecommunications Canada Inc.	950 Fennell Ave E Hamilton, Ontario
Primus Telecommunications Canada Inc.	55 Reading Court Etobicoke, Ontario
Primus Telecommunications Canada Inc.	30 Malta Avenue Brampton, Ontario M2R 3V8
Primus Telecommunications Canada Inc.	Hydro One Tower Clairville, Ontario
Primus Telecommunications Canada Inc.	176 Rivermead Road #3 Concord, Ontario L4K 3M7
Primus Telecommunications Canada Inc.	100 Allstate Parkway Markham, Ontario L3R 6H3

Company/Relevant Subsidiary Primus Telecommunications Canada Inc.	<u>Address</u> 50 East Valhalla Drive Markham, Ontario
Primus Telecommunications Canada Inc.	68 Corporate Drive Scarborough, Ontario M1H 3H3
Primus Telecommunications Canada Inc.	10 Torresdale Avenue Toronto, Ontario M2R 3V8
Primus Telecommunications Canada Inc.	8000 Jane Street Vaughan, Ontario L4K 5B8
Primus Telecommunications Canada Inc.	63 John Street Brampton, Ontario
Primus Telecommunications Canada Inc.	299 Walker Brampton, Ontario
Primus Telecommunications Canada Inc.	86 Market St Brantford, Ontario
Primus Telecommunications Canada Inc.	120 7th Ave SW Calgary, Alberta
Primus Telecommunications Canada Inc.	10405 104 th Ave NW Edmonton, Alberta
Primus Telecommunications Canada Inc.	20 Hunter Street West Hamilton, Ontario
Primus Telecommunications Canada Inc.	628 Upper Wentworth St Hamilton, Ontario
Primus Telecommunications Canada Inc.	1212 Main Street East Hamilton, Ontario
Primus Telecommunications Canada Inc.	450 Princess St. Kingston, Ontario
Primus Telecommunications Canada Inc.	20 Water St. North Kitchener, Ontario
Primus Telecommunications Canada Inc.	391 Albert St Kitchener, Ontario
Primus Telecommunications Canada Inc.	479 Clarence Street London, Ontario

Company/Relevant Subsidiary Primus Telecommunications Canada Inc.	<u>Address</u> 735 Hyde Park Rd London, Ontario
Primus Telecommunications Canada Inc.	1775 Whitney St London, Ontario
Primus Telecommunications Canada Inc.	725 Southdale Rd East London, Ontario
Primus Telecommunications Canada Inc.	915 Av Atwater Montreal, Quebec
Primus Telecommunications Canada Inc.	87 Rue Ontario Ouest Montreal, Quebec
Primus Telecommunications Canada Inc.	6420 Rue Saint-Dominique Montreal, Quebec
Primus Telecommunications Canada Inc.	6961 Boul Saint-Michel Montreal, Quebec
Primus Telecommunications Canada Inc.	5220 Rue De Castille, Montreal-Nord, Montreal, Quebec
Primus Telecommunications Canada Inc.	960 1Re Av., Verdun Montreal, Quebec
Primus Telecommunications Canada Inc.	3555 Rue De Cadillac Montreal, Quebec
Primus Telecommunications Canada Inc.	6920 Boul Joseph-Renaud, Anjou Montreal, Quebec
Primus Telecommunications Canada Inc.	6055 Av Monkland Montreal, Quebec
Primus Telecommunications Canada Inc.	2265 Av Papineau Montreal, Quebec
Primus Telecommunications Canada Inc.	4450 Rue Dudemaine Montreal, Quebec
Primus Telecommunications Canada Inc.	6250 Chemin De La Cote-Des-Neiges Montreal, Quebec
Primus Telecommunications Canada Inc.	15 Victoria St Oshawa, Ontario
Primus Telecommunications Canada Inc.	78 O'Connor Street Ottawa, Ontario
Primus Telecommunications Canada Inc.	1501 Bank St Ottawa, Ontario

<u>Company/Relevant Subsidiary</u> Primus Telecommunications Canada Inc.	Address 225 Montreal Rd Ottawa, Ontario
Primus Telecommunications Canada Inc.	3212 Richmond Rd Ottawa, Ontario
Primus Telecommunications Canada Inc.	190 Iona St Ottawa, Ontario
Primus Telecommunications Canada Inc.	1671 Merivale Rd Ottawa, Ontario
Primus Telecommunications Canada Inc.	76 Adelaide St West Toronto, Ontario
Primus Telecommunications Canada Inc.	1 Eagle Road Toronto, Ontario
Primus Telecommunications Canada Inc.	220 Simcoe Street Toronto, Ontario
Primus Telecommunications Canada Inc.	1040 Dufferin Street Toronto, Ontario
Primus Telecommunications Canada Inc.	15 Asquith Avenue Toronto, Ontario
Primus Telecommunications Canada Inc.	72 Eglinton Av East Toronto, Ontario
Primus Telecommunications Canada Inc.	19 Main Street Toronto, Ontario
Primus Telecommunications Canada Inc.	4 Bellevue Cres Toronto, Ontario
Primus Telecommunications Canada Inc.	31 Finch Ave East Toronto, Ontario
Primus Telecommunications Canada Inc.	103 Ernest Ave Toronto, Ontario
Primus Telecommunications Canada Inc.	200 Rogers Road Toronto, Ontario
Primus Telecommunications Canada Inc.	3783 St. Clair Ave E Toronto, Ontario
Primus Telecommunications Canada Inc.	240 Acton Ave Toronto, Ontario

Company/Relevant Subsidiary Primus Telecommunications Canada Inc.	<u>Address</u> 4171 Sheppard Ave East Toronto, Ontario
Primus Telecommunications Canada Inc.	67 Orchard Park Dr Toronto, Ontario
Primus Telecommunications Canada Inc.	2664 Islington Av Toronto, Ontario
Primus Telecommunications Canada Inc.	130 Ridelle AV Toronto, Ontario
Primus Telecommunications Canada Inc.	562 Runnymede Rd Toronto, Ontario
Primus Telecommunications Canada Inc.	184 Simpson AV Toronto, Ontario
Primus Telecommunications Canada Inc.	364 Donlands Av Toronto, Ontario
Primus Telecommunications Canada Inc.	1260 Don Mills Rd Toronto, Ontario
Primus Telecommunications Canada Inc.	40 Old Burnhamthorpe Rd Toronto, Ontario
Primus Telecommunications Canada Inc.	1421 Victoria Park Toronto, Ontario
Primus Telecommunications Canada Inc.	180 Islington Ave Toronto, Ontario
Primus Telecommunications Canada Inc.	3665 Lawrence Ave E Toronto, Ontario
Primus Telecommunications Canada Inc.	1101 Alness St Toronto, Ontario
Primus Telecommunications Canada Inc.	768 Seymour St Vancouver, Ontario
Primus Telecommunications Canada Inc.	2212 W 10 th Ave Vancouver, Ontario
Primus Telecommunications Canada Inc.	380 East 10 th Ave Vancouver, Ontario
Primus Telecommunications Canada Inc.	1149 Goyeau Street Windsor, Ontario
Primus Telecommunications Canada Inc.	3501 Avondale Court Windsor, Ontario

Company/Relevant Subsidiary Primus Telecommunications Canada Inc.	<u>Address</u> 1706 Francois Rd Windsor, Ontario
Primus Telecommunications Canada Inc.	166 Portage Ave East Winnipeg, Ontario
Primus Telecommunications Canada Inc.	2491 Portage Ave Winnipeg, Ontario
Primus Telecommunications Canada Inc.	297 Sherbrook Street Winnipeg, Ontario
Primus Telecommunications Canada Inc.	640 Corydon Ave Winnipeg, Ontario
Primus Telecommunications Canada Inc.	220 Station St 2 nd Floor Belleville, Ontario
Primus Telecommunications Canada Inc.	22 Selby Rd Brampton, Ontario
Primus Telecommunications Canada Inc.	23 Harris Ave Brantford, Ontario
Primus Telecommunications Canada Inc.	141 Perth St CNR Stn Brockville, Ontario
Primus Telecommunications Canada Inc.	109B Station Rd Cornwall, Ontario
Primus Telecommunications Canada Inc.	9924 – 104 th Ave NW, 2 nd Floor Edmonton, Alberta
Primus Telecommunications Canada Inc.	94 Dundurn St South Hamilton, Ontario
Primus Telecommunications Canada Inc.	14 Brunel, Lot 11-12 Brunel District Muskoka Huntsville, Ontario
Primus Telecommunications Canada Inc.	350 Landsdowne St Kamloops, B.C.
Primus Telecommunications Canada Inc.	810 Montreal Street Kingston, Ontario
Primus Telecommunications Canada Inc.	185 Breithaupt St Kitchener, Ontario

<u>Company/Relevant Subsidiary</u> Primus Telecommunications Canada Inc.	<u>Address</u> Champlain College 5 Winder Street Lennoxville, Quebec
Primus Telecommunications Canada Inc.	205 York St London, Ontario
Primus Telecommunications Canada Inc.	740 Notre Dame Ouest, 3 rd Floor Montreal, Quebec
Primus Telecommunications Canada Inc.	60 Hudson St Suite 1904 New York, New York
Primus Telecommunications Canada Inc.	17 King St East Oshawa, Ontario
Primus Telecommunications Canada Inc.	555 Wentworth St East, Unit 5 Oshawa, Ontario
Primus Telecommunications Canada Inc.	495 Terminal Ave Ottawa, Ontario
Primus Telecommunications Canada Inc.	2020 Saskatchewan Dr, 2 nd Floor Regina, Saskatchewan
Primus Telecommunications Canada Inc.	400 Avenue A South Saskatoon, Saskatchewan
Primus Telecommunications Canada Inc.	Sault College 443 Northern Ave H Wing Bsmt Sault Ste. Marie, Ontario P6A 5L3
Primus Telecommunications Canada Inc.	51 Elgin St Sudbury, Ontario
Primus Telecommunications Canada Inc.	723 North McKellar St Thunder Bay, Ontario
Primus Telecommunications Canada Inc.	737 Ouellette Ave., London Telecom and Wincom POF Windsor, Ontario

<u>Company/Relevant Subsidiary</u> Primus Telecommunications Canada Inc. Address 392 Main St Winnipeg, Alberta Primus Telecommunications Canada Inc. 330 Bay Street, Suite 1600 Toronto, Ontario M5H 2S8 Primus Telecommunications Canada Inc. 166 Portage Ave. East Winnipeg, Manitoba R3C 3N8 2491 Portage Ave. East Primus Telecommunications Canada Inc. Winnipeg, Manitoba R3J 0N6 Telesonic Communications Inc. 140 Renfrew Dr., Suite 200 Markham, ON, L3R 6B3

Schedule 6 Intellectual Property

TRADEMARKS

Primus Telecommunications Canada Inc.

Trade-mark 800-363-FLAT	App./Reg. No. TMA515437	App./Reg. Date 1999-08-26	<u>Status</u> Registered
888-LTN-FLAT	TMA513635	1999-07-29	Registered
BPS	TMA632560	2005-02-10	Registered
BROADBAND PHONE SERVICE (BPS)	TMA678143	2006-12-05	Registered
BROADBAND VOICE SERVICE (BVS)	TMA678144	2006-12-05	Registered
BVS	TMA632466	2005-02-09	Registered
CALL25	TMA644503	2005-07-15	Registered
COMPARE AND REWARD	TMA603341	2004-02-26	Registered
DATASAFE ADVANCED	1217670	2004-05-21	Allowed
DATASAFE EXTRA SECURE	1217678	2004-05-21	Registered
DATASAFE SOLUTIONS	1217679	2004-05-21	Registered
EASYREACH COMMUNICATIONS	TMA498070	1998-07-30	Registered
ENTERPRISE DATASAFE	1217677	2004-05-21	Allowed
GLOBALSERVE	TMA468136	1997-01-02	Registered
GO FOR MORE	TMA658186	2006-02-07	Registered
GUARDIEN TÉLÉMARKETING	TMA749509	2007-08-28	Formulized
INTERNATIONAL SANS COMPROMIS	TMA707426	2008-02-14	Registered
LONDON TELECOM	TMA423877	1994-02-25	Registered
LONDON TELECOM NETWORK	TMA423876	1994-02-25	Registered
London Telecom Network LONDON TELECOM NETWORK & DESIGN	TMA465850	1996-11-06	Registered
Canada's Flat Rate Long Distance Company LONDON TELECOM NETWORK CANADA'S FLAT RATE LONG DISTANCE COMPANY & DESIGN	TMA535474	2000-10-23	Registered
MAGMA	TMA733207	2009-01-27	Registered
Magma Communications MAGMA & DESIGN	TMA733208	2009-01-27	Registered

Trade-mark MORE \$15	App./Reg. No. TMA693053	App./Reg. Date 2007-07-31	<u>Status</u> Registered	
MORE \$20	TMA693052	2007-07-31	Registered	
MORE \$25	TMA693051	2007-07-31	Registered	
MORE \$35	TMA693050	2007-07-31	Registered	
MORE ANYTIME	TMA667323	2006-07-11	Registered	
MORE EVENINGS AND WEEKENDS	TMA693480	2007-08-03	Registered	
MY TALKBROADBAND	TMA654488	2005-12-06	Registered	
OBTENEZ-EN PLUS	TMA706434	2008-02-04	Registered	
ONRAMP ONRAMP NETWORK SERVICES INC. & DESIGN	TMA613371	2004-06-22	Registered	
PARLEZ HAUTE VITESSE	TMA655193	2005-12-15	Registered	
PLUS! 15\$	TMA692862	2007-07-27	Registered	
PLUS! 20\$	TMA693049	2007-07-31	Registered	
PLUS! 25\$	TMA693048	2007-07-31	Registered	
PLUS! 35\$	TMA693047	2007-07-31	Registered	
PLUS! EN TOUT TEMPS	TMA674864	2006-10-13	Registered	
PLUS! SOIR ET LE WEEKEND	TMA694494	2007-08-22	Registered	
REVERSE CALLING REVERSE CALLING & DESIGN	TMA451154	1995-12-01	Registered	
SERVICE ELECTRONIQUE WIN-TEL	TMA642557	2005-06-20	Registered	
TALKBROADBAND	TMA635039	2005-00-20	Registered	
TELE-FRIEND	TMA436619	1994-12-02	Registered	
TELEMARKETING GUARD	TMA749510	2007-08-28	Searched	
TEELMINGETING GOING	TMA504990	1998-12-03	Registered	
The London Telecom Group THE LONDON TELECOM GROUP & DESIGN				
TOTALLINE COMMUNICATIONS INC. & DESIGN	TMA522632	2000-02-01	Registered	
TotalLine Communications Inc. TOTALLINE COMMUNICATIONS INC. design	TMA522250	2000-01-26	Registered	

Trade-mark_	App./Reg. No.	App./Reg. Date	Status
TRULY INTERNATIONAL	TMA714001	2008-05-09	Registered
TRULY UNLIMITED CANADA	TMA659832	2006-02-28	Registered
TRULY UNLIMITED NORTH AMERICA	TMA659969	2006-03-01	Registered
WIN-TEL	TMA630061	2005-01-12	Registered
WIN-TEL E-CARE	TMA691794	2007-07-11	Registered
W@RLDTALK	TMA473151	1997-03-20	Registered

3082833 Nova Scotia Company

WORLDTALK & GLOBE DESIGN

Trade-mark CALL HOME	Reg. No. TMA471805	Reg. Date 1997-02-27	<u>Status</u> Registered
CALL NORTH AMERICA	TMA430193	1994-07-08	Registered
CALL REGIONAL	TMA431725	1994-08-12	Registered
CLOSE CONNECTIONS	TMA430020	1994-07-01	Registered
DIME TIME	TMA510125	1999-03-26	Registered
DIX SOUS C'EST TOUT	TMA507619	1999-02-05	Registered
ENTRE NOUS	TMA436267	1994-11-25	Registered
FREE MONTH FOR LIFE	TMA477010	1997-05-29	Registered
IT'S YOUR CALL	TMA454966	1996-03-08	Registered
LA QUALITÉ EN INTERURBAIN À MOINDRE PRIX	TMA477212	1997-06-04	Registered
MON PAYS	TMA492607	1998-04-08	Registered
MY COUNTRY	TMA501131	1998-09-24	Registered
NOUS SOMMES TOUJOURS À L'ÉCOUTE	TMA490172	1998-02-18	Registered
ONE & ALL	TMA447160	1995-09-01	Registered
OUR WORLD IS FLAT	TMA526682	2000-04-18	Registered
QUALITY LONG DISTANCE FOR LESS	TMA453520	1996-02-02	Registered
TOUJOURS À L'ÉCOUTE	TMA490173	1998-02-18	Registered
UN POUR TOUS	TMA447842	1995-09-15	Registered
YOU TALK WE LISTEN	TMA490124	1998-02-18	Registered

PATENTS

Primus Telecommunications Canada, Inc.

Patent Call Screening System and Method

Details
Patent No. 2,597,377
Filing Date: August 15, 2007

Schedule 7 [Intentionally Omitted]

Schedule 8 Deposit Accounts and Securities Accounts

Account # 80002 00234 18	Account Name Primus Deposit Account	Bank Branch Scotiabank	Branch Address 44 King Street West Toronto, Ont. M5K 1A2	Account Currency Canadian
80002 03789 17	TCI Disbursements	Scotiabank	44 King Street West Toronto, Ont. M5K 1A2	Canadian
80002 03875 17	TCI Deposits	Scotiabank	44 King Street West Toronto, Ont. M5K 1A2	Canadian
80002 03842 16	TCI Payroll	Scotiabank	44 King Street West Toronto, Ont. M5K 1A2	Canadian
80002 03889 12	TCI Concentration	Scotiabank	44 King Street West Toronto, Ont. M5K 1A2	Canadian
80002 12524 10	Primus Payroll	Scotiabank	44 King Street West Toronto, Ont. M5K 1A2	Canadian
80002 12525 18	Primus Disbursements	Scotiabank	44 King Street West Toronto, Ont. M5K 1A2	Canadian
80002 12526 15	Primus	Scotiabank	44 King Street West Toronto, Ont. M5K 1A2	Canadian
80002 12527 12	Primus Concentration	Scotiabank	44 King Street West Toronto, Ont. M5K 1A2	Canadian
80002 11387 15	Primus Operating Account	Scotiabank	44 King Street West Toronto, Ont. M5K 1A2	Canadian

Account #	Account Name Bank Branch		Branch Address	Account Currency	
80002 13040 11	MIPPS Inc.	Scotiabank	44 King Street West Toronto, Ont. M5K 1A2	Canadian	
80002 06133 12	Globility	Scotiabank	44 King Street West Toronto, Ont. M5K 1A2	Canadian	
80002 06151 10	Globility Communications	Scotiabank	44 King Street West Toronto, Ont. M5K 1A2	Canadian	
80002 13015 19	Magma Deposit Account	Scotiabank	44 King Street West Toronto, Ont. M5K 1A2	Canadian	
80002 13089 12	Magma Disbursements	Scotiabank	44 King Street West Toronto, Ont. M5K 1A2	Canadian	
80002 13112 12	Magma Payroll Account	Scotiabank	44 King Street West Toronto, Ont. M5K 1A2	Canadian	
80002 13114 17	Magma Concentration	Scotiabank	44 King Street West Toronto, Ont. M5K 1A2	Canadian	
47696 01092 15	Project Samuels	Scotiabank	44 King Street West Toronto, Ont. M5K 1A2	Canadian	
0481 1010 446	Magma Communications	Bank of Montreal	510 Main Street Winchester, Ontario K0C 2K0	Canadian	
80002 62439 16	3082833 Nova Scotia Company	Scotiabank	44 King Street West Toronto, Ont. M5K 1A2	USD	

Account #	Account Name	Bank Branch	Branch Address	Account Currency
80002 61699 10	Primus Deposit Account	Scotiabank	44 King Street West Toronto, Ont. M5K 1A2	USD
80002 61700 13	Primus	Scotiabank	44 King Street West Toronto, Ont. M5K 1A2	USD
80002 61701 10	Primus Concentration	Scotiabank	44 King Street West Toronto, Ont. M5K 1A2	USD
80002 87432 15	Primus Disbursements	Scotiabank	44 King Street West Toronto, Ont. M5K 1A2	USD
80002 85441 15	MIPPS Inc.	Scotiabank	44 King Street West Toronto, Ont. M5K 1A2	USD
80002 85300 17	TCI US	Scotiabank	44 King Street West Toronto, Ont. M5K 1A2	USD
80002 85634 11	Magma US	Scotiabank	44 King Street West Toronto, Ont. M5K 1A2	USD

Schedule 9 Receivables

1. Promissory note in the amount of \$40,000.00 dated April 20, 2009 given by Primus Telecommunications Canada Inc. in favour of 9151-4877 Quebec Inc. (dba Teliphone Inc.).

The \$40,000.00 receivable is payable over a two year period and as of December 1, 2009 the balance outstanding is \$19,000.00

NEW DEBT NOTICE AND JOINDER AGREEMENT

Dated as of December 22, 2009

Reference is hereby made to the Intercreditor Agreement dated as of February 26, 2007 (as amended, restated, supplemented or otherwise modified from time to time, the "Intercreditor Agreement"; capitalized terms used but not otherwise defined herein having the meanings assigned to such terms in the Intercreditor Agreement) among Primus Telecommunications Holding, Inc., a Delaware corporation (the "Company"), the other Grantors party thereto, The Bank of New York Mellon (as successor to Lehman Commercial Paper Inc., in its capacity as administrative agent for the First Lien Obligations), as First Lien Collateral Agent, and U.S. Bank National Association, as Second Lien Collateral Agent.

Pursuant to Section 5.5 of the Intercreditor Agreement, the Second Lien Collateral Agent is hereby notified that (i) this New Debt Notice and Joinder Agreement (this "Agreement"), constitutes a New Debt Notice for purposes of Section 5.5 of the Intercreditor Agreement, (ii) the First Lien Documents have been Refinanced by the Senior Secured Notes Documents (as defined below), which Refinancing is permitted by the Second Lien Loan Documents and the Intercreditor Agreement, (iii) the Company has entered into New First Lien Loan Documents including that certain Indenture, dated as of December 22, 2009 (as amended, restated, supplemented or otherwise modified from time to time, the "Senior Secured Notes Indenture"), by and among the Company, Primus Telecommunications Canada Inc., The Bank of New York Mellon, as Trustee, U.S. Bank National Association, as U.S. Collateral Trustee (together with its successors in such capacity, the "U.S. Collateral Trustee") and Computershare Trust Company of Canada, as Canadian Collateral Trustee, and the 13% Senior Secured Notes due 2016 (the "Senior Secured Notes") of the Company issued under the Indenture, that certain Collateral Agreement, dated as of December 22, 2009 (as amended, restated, supplemented or otherwise modified from time to time, the "Senior Secured Notes Collateral Agreement", and, together with the Senior Secured Notes Indenture, the Senior Secured Notes and each of the other agreements, documents and instruments executed or delivered at any time by any Grantor in connection with the Senior Secured Notes, the "Senior Secured Notes Documents"), (iv) the U.S. Collateral Trustee is the New Agent for purposes of the Intercreditor Agreement, from and after the date hereof, the obligations under the Senior Secured Notes Documents shall be treated as First Lien Obligations for all purposes of the Intercreditor Agreement and the U.S. Collateral Trustee shall be the First Lien Collateral Agent for all purposes of the Intercreditor Agreement.

The U.S. Collateral Trustee hereby:

(i) represents that the holders of the obligation under the Senior Secured Notes Documents (the "New First Lien Claimholders") have authorized it to become a party to the Intercreditor Agreement on behalf of such New Claimholders and to act as the First Lien Collateral Agent on behalf of such New Claimholders;

- (ii) acknowledges on behalf of itself and the other New Claimholders that the obligations under the Senior Secured Notes Documents constitute First Lien Obligations for all purposes of the Intercreditor Agreement; and
- (iii) accepts and acknowledges the terms of the Intercreditor Agreement applicable to the First Lien Claimholders and agrees on its own behalf and on behalf of the other New Claimholders to be bound by the terms thereof applicable to holders of First Lien Obligations, with all the rights, duties and obligations of the First Lien Claimholders under the Intercreditor Agreement and to be bound by all the provisions thereof as fully as if they had been named as First Lien Claimholders on the effective date of the Intercreditor Agreement; and
- (iv) accepts and acknowledges the terms of the Intercreditor Agreement appli-cable to the First Lien Collateral Agent and agrees on its own behalf and on behalf of the other New Claimholders to be bound by the terms thereof applicable to the First Lien Collateral Agent, with all the rights, duties and obligations of the First Lien Collateral Agent under the Intercreditor Agreement and to be bound by all the provisions thereof as fully as if it had been named as the First Lien Collateral Agent on the effective date of the Intercreditor Agreement and agrees that its address for receiving notices pursuant to the Intercreditor Agreement shall be as follows:

U.S. Bank National Association 100 Wall Street 16th Floor New York, New York 10005 Attention: Corporate Trust Services Fax: (212) 361-6153 Telephone: (212) 361-6187

THIS AGREEMENT SHALL BE GOVERNED BY, AND SHALL BE CONSTRUED AND ENFORCED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF NEW YORK, WITHOUT REGARD TO CONFLICTS OF LAWS THEREOF.

[Signature Page Follows]

December 2009.	
	U.S. BANK NATIONAL ASSOCIATION,
	as U.S. Collateral Trustee
	By:
	Name:
	Title:
U.S. BANK NATIONAL ASSOCIATION, as Second Lien Collateral Agent	
By:	
Name:	
Title:	
PRIMUS TELECOMMUNICATIONS HOLDING, INC.	
Ву:	
Name:	
Title:	
PRIMUS TELECOMMUNICATIONS GROUP, INCORPORATED	
By:	
Name:	
Title:	
PRIMUS TELECOMMUNICATIONS, INC.	
By:	
Name:	

IN WITNESS WHEREOF, each of the undersigned has caused this Agreement to be duly executed by its authorized officer as of the 22nd day of

Signature Page to New Debt Notice and Joinder Agreement

Title:

PRIMUS TELECOMMUNICATIONS INTERNATIONAL, INC.
By:
Name:
Title:
TRESCOM U.S.A., INC.
By:
Name:
Title:
Tiue.
LEAST COST ROUTING, INC.
By:
Name:
Title:
iPRIMUS U.S.A., INC.
Ву:
Name:
Title:
iPRIMUS.COM, INC.
By:
Name:
Title:
PRIMUS TELECOMMUNICATIONS IHC, INC.
Ву:
Name:
Title:

Signature Page to New Debt Notice and Joinder Agreement

Computation of Ratio of Earnings to Fixed Charges (in thousands)

		Predecessor				_ 5	Succesor
		Six Months Year Ended December 31, Ended					x Months Ended
	2005	2006	2007	2008	July 1, 2009	December 31, 2009	
Earnings:							
Pre-tax income (loss) from continuing operations before adjustment for							
income (loss) of equity investees	\$(147,966)	\$(236,947)	\$ 640	\$(21,912)	\$475,454	\$	(2,595)
Add: Fixed charges	60,942	62,858	69,189	60,640	16,983		20,947
Add: Distributed income of equity investees	_	_	_	_	_		_
Less: Equity investment	_	_	_	_	_		_
Total earnings before fixed charges	\$ (87,024)	\$(174,089)	\$69,829	\$ 38,728	\$492,437	\$	18,352
Fixed charges:							
Interest expense	\$ 53,394	\$ 54,128	\$61,347	\$ 53,888	\$ 14,135	\$	17,323
Accretion (amortization)on debt discount/premium, net	_	1,732	449	(583)	(189)		3
Estimated interest component of rent expense	7,548	6,998	7,393	7,335	3,037		3,621
Total fixed charges	\$ 60,942	\$ 62,858	\$69,189	\$ 60,640	\$ 16,983	\$	20,947
Ratio of earnings to fixed charges(1)	<1	<1	1.01	<1	29.00	=	<1

¹⁾ The ratio of earnings to fixed charges is computed by dividing the sum of pre-tax income from continuing operations (before adjustment for minority interests in consolidated subsidiaries and loss from equity investees) plus fixed charges, by fixed charges. Fixed charges consist of interest charges, whether expensed or capitalized, and that portion of rental expense we believe to be representative of interest.

(Spain)

(Belgium)

(Austria)

(Sweden) (Delaware)

(India)

(Canada)

Operating Subsidiaries Primus Telecommunications, Inc. (Delaware) iPRIMUS USA, Inc. (Delaware) Primus Telecommunications IHC, Inc. (Delaware) STSJ Overseas Telephone Company, Inc. (Puerto Rico) St. Thomas & San Juan Telephone Company, Inc. (US Virgin Islands) Primus Telecommunications Canada, Inc. (Canada) Telesonic Communications, Inc. (Canada) MIPPS, Incorporated (Canada) Primus Telecom Holdings Pty., Ltd. (Australia) Primus Network (Australia) Pty., Ltd. (Australia) Primus Telecom Ptv, Ltd (Australia) Primus Telecommunications Pty., Ltd (Australia) (Australia) Primus Telecommunications (Australia) Pty., Ltd. (Australia) 0014 Pty, Ltd. Liquor Industry Service Technology Pty., Ltd. (Australia) Hotkey Internet Service Pty., Ltd (Australia) Primus Data Pty., Ltd. (Australia) Primus Online Pty., Ltd. (Australia) (United Kingdom) Telegroup UK Ltd. Primus Telecommunications Ltd. (United Kingdom) (United Kingdom) Planet Talk UK Limited Primus Telecommunications KK Japan (Japan) Global Access Pty., Ltd. (South Africa) P1 do Brasil Ltda (Brazil) Matrix Internet, S.A. (Brazil) US Matrix Telecommunications, Inc. (Florida) Primus Telecommunications Netherlands B.V. (Netherlands) (Netherlands) Primus Nederland B.V. Primus Telecommunications GmbH (Germany) Primus Telecommunications France S.A.S.U (France) Primus Telecommunications S.r.1. (Italy) (Italy)

Primus Telecommunications S.r.1.
Telegroup Italia S.r.1.
Primus Telecommunications Iberica SA
Primus Telecommunications Belgium N.V.
PTI Telecommunications GmbH
Telegroup Sweden AB

Lingo, Inc.

Primus Professional Services Private Limited (fka Primus Telecommunication India Private Limited) Globility Communications Corporation

Non-Operating Subsidiaries

Primus Telecommunications International, Inc. (Delaware) iPRIMUS.com, Inc. (Delaware) (Delaware) Lingo Holdings, Inc. Primus Telecommunications Holding, Inc. (Delaware) Least Cost Routing, Inc. (Florida) TresCom U.S.A. Inc. (Florida) STSJ Network Assets Inc. (US Virgin Islands) OTC Network Assets, Inc. (Puerto Rico) (Hong Kong) Stubbs, Ltd. PRIMUS Telecom SA de C.V. (El Salvador) Primus Telecommunications Holdings Ltd (United Kingdom) European Mobile Products and Solutions Limited (Ireland) Primus Telecommunications de Mexico SA de CV (Mexico) 3082833 Nova Scotia Company (Canada) 3620212 Canada Inc. (Canada) Primus Telecommunications Europe BV (Netherlands)

(Brazil)

(Brazil)

(Switzerland)

(Mauritius)

Delta One America Do Sul

Communicacoes do Brasil Ltda

Primus Telecommunications AG

Primus Telecom Mauritius Holdings

CONSENT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

We consent to the incorporation by reference in Registration Statement No. 333-160406 on Form S-8 of our report dated April 5, 2010, relating to the consolidated financial statements and financial statement schedule of Primus Telecommunications Group, Incorporated and subsidiaries (which report expresses an unqualified opinion, and includes an emphasis of matter paragraph relating to the Company's emergence from bankruptcy and implementation of fresh-start accounting, and includes an explanatory paragraph regarding the Company's adoption of a new accounting standard), appearing in this Annual Report on Form 10-K of Primus Telecommunications Group, Incorporated and subsidiaries for the year ended December 31, 2009.

/s/ Deloitte & Touche LLP McLean, Virginia April 5, 2010

CERTIFICATIONS

I, K. Paul Singh, certify that:

- 1. I have reviewed this report on Form 10-K of Primus Telecommunications Group, Incorporated;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15e and 15d-15e) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15f and 15d-15f) for the registrant and have:
 - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusion about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal controls.

Dated: April 5, 2010 By: /s/ K. PAUL SINGH

Name: K. Paul Singh Title: Chairman, President and Chief Executive Officer (Principal Executive Officer) and Director

I, Thomas R. Kloster, certify that:

- 1. I have reviewed this report on Form 10-K of Primus Telecommunications Group, Incorporated;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15e and 15d-15e) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15f and 15d-15f) for the registrant and have:
 - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusion about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal controls.

Dated: April 5, 2010 By: /s/ THOMAS R. KLOSTER

Name: Thomas R. Kloster

Title: Chief Financial Officer (Principal Financial Officer)

CERTIFICATION

Pursuant to Section 906 of the Public Company Accounting Reform and Investor Protection Act of 2002 (18 U.S.C. § 1350, as adopted), K. Paul Singh, the Chief Executive Officer of Primus Telecommunications Group, Incorporated (the "Company"), and Thomas R. Kloster, the Chief Financial Officer of the Company, each hereby certifies that, to the best of his knowledge:

- 1. The Company's Annual Report on Form 10-K for the year ended December 31, 2009, to which this Certification is attached as Exhibit 32 (the "Periodic Report"), fully complies with the requirements of Section 13(a) or Section 15(d) of the Securities Exchange Act of 1934, as amended; and
- 2. The information contained in the Periodic Report fairly presents, in all material respects, the financial condition of the Company at the end of the period covered by the Periodic Report and results of operations of the Company for the period covered by the Periodic Report.

Dated: April 5, 2010

/s/ K. PAUL SINGH /s/ THOMAS R. KLOSTER

K. Paul Singh
Thomas R. Kloster
Chairman, President and Chief Executive Officer (Principal Executive
Chief Financial Officer)

Officer) and Director

Chief Financial Officer (Principal Financial Officer)